



Thèse de doctorat de L'Université de Bordeaux

Ecole doctorale de Mathématiques et Informatique de Bordeaux

Spécialité : Mathématiques appliquées

Présentée par

Benjamin Mélinand

POUR OBTENIR LE GRADE DE

DOCTEUR DE L'UNIVERSITÉ DE BORDEAUX

SUJET :

Météotsunamis, résonance de Proudman et effet Coriolis pour les équations des vagues

Soutenue publiquement le 28 Juin 2016

Directeur de thèse : Rapporteurs :	M. Lannes David Mme Gallagher Isabelle Mme Sulem Catherine	DR CNRS (Université de Bordeaux) Professeur (Université Paris-Diderot) Professeur (University of Toronto)
Composition du jury :		
Président du jury :	M. Rousset Frédéric	Professeur (Université Paris-Sud)
Directeur de thèse :	M. Lannes David	DR CNRS (Université de Bordeaux)
Rapporteurs :	Mme Gallagher Isabelle	Professeur (Université Paris-Diderot)
	Mme Sulem Catherine	Professeur (University of Toronto)
Examinateurs :	M. Chiron David	Maître de conférence (Université de Nice Sophia Antipolis)
	M. Colin Mathieu	Maître de conférence (Université de Bordeaux)
	M. Sueur Franck	Professeur (Université de Bordeaux)

Météotsunamis, résonance de Proudman et effet Coriolis pour les équations des vagues

Mélinand Benjamin *

^{*}Institut de Mathématiques de Bordeaux UMR 5251, 351, cours de la Libération - F 33 405 TALENCE. Email: benjamin.melinand@math.u-bordeaux.fr

Remerciements

Mes premiers remerciements vont tout naturellement à David Lannes, mon directeur de thèse, pour sa disponibilité et sa patience. David a su me guider et me canaliser tout au long de ces trois années. Ses conseils avisés et sa clairvoyance m'ont aidé à m'épanouir scientifiquement. Je tiens tout particulièrement à le remercier pour sa décontraction et sa sérénité basquaise.

Je tiens ensuite à remercier mes deux rapportrices, Isabelle Gallagher et Catherine Sulem, pour leur relecture de mon manuscrit. Les nombreuses conférences d'Isabelle Gallagher auxquelles j'ai pu assister m'ont faites une forte impression. Je suis honoré de l'intérêt qu'elle porte à mes travaux. Mes discussions avec Catherine Sulem lors de mon voyage à Toronto en janvier 2016 m'ont profondément marqué. Par ses questions précises, Catherine m'a beaucoup apporté scientifiquement. Je lui suis très reconnaissant de se déplacer jusqu'à Bordeaux pour ma soutenance.

Je garde un bon souvenir de l'exposé de David Chiron à Pise sur l'équation de Gross-Pitaevski. Je suis très heureux qu'il fasse partie de mon jury. Je remercie Mathieu Colin d'être présent dans mon jury et pour nos nombreux échanges. Je retiendrai surtout la conférence à Vienne sur les équations quasilinéraires. Je suis très honoré que Frédéric Rousset ait accepté de faire partie de mon jury. J'ai découvert les EDP grâce à son cours de M2 à Rennes. Enfin, je remercie Franck Sueur pour sa présence dans mon jury, nos discussions mathématiques et nos confrontations footbalistiques. Franck a su m'en mettre plein la vue scientifiquement comme sportivement.

Mes remerciements vont aussi à tous les thésards de l'équipe EDP de Bordeaux. Merci à Francesco (pour sa fulgurance italo-footballistique), Stefano (pour décontraction), Marouane (pour nos discussions sur l'islam et sa passion pour le real), Stevan (pour son savoir sportif), Adrien (pour son prêt de bureau quand il n'était pas présent), Youssef, Krisztian, Manon, Alice, Andrea, Luca, Teddy, Leo, etc. Merci en particulier à Stevan pour nos nombreux échanges scientifiques et sportifs. Je voudrais aussi remercier mes anciens co-bureaux du DMA, Yannick (sauf le 14 juillet), Thibault (pour nos confrontations scientifiques) et Benoît (pour sa sagesse). Mention spéciale à Benoît pour avoir été un grand frère de thèse exceptionnel. Je me dois aussi de remercier tous les thésards que j'ai pu rencontrer en conférence (et en particulier aux journées EDP) et pendant mes études : Olivier (frère de promo et picard accompli), Antoine, Victor, Marco, Baptiste, Nicolas, Eugénie, Julie, Alexandre etc. Enfin je voudrais remercier quelques chercheurs qui m'ont aidé pendant mes études et mon doctorat : Jean-Claude Saut pour nos discussions sur la dispersion des vagues et l'équation d'Ostrovsky, François Castella pour sa pédagogie, Vincent Duchêne pour ses conseils avisés sur les water waves, Jean-François Coulombel pour nos discussions sur l'avenir de la recherche française, Thomas Alazard pour nos discussions sur les water waves et pour sa bonne humeur et Afaf Bouharguane pour m'avoir formé au numérique.

Je remercie chaleureusement l'IMB pour l'accueil et les conditions de travails. Merci à Karine, Ida, Catherine et Catherine pour leur efficacité et leur bonne humeur. Je voudrais aussi remercier l'équipe EDP pour sa bonne ambiance : David, Franck, Corentin, Chistophe, Jean-François (pour ses interrogations sur la religion catholique), Vincent, Rafik, Nicolas, Mathieu, Mario, Bernhard, El Maati, Guy, Alain, Agnès, Marius, Mouez, Denise, Stéphane, Afaf, Patrick etc.

Je remercie enfin ma famille et belle-famille (et oui, moi aussi j'utilise ce terme Nicole) pour leur présence à ma soutenance et dans ma vie. Je n'ose imaginer ce que je serais devenu sans eux. Merci à mon père de m'avoir toujours soutenu et de m'avoir fait découvrir les sciences. Merci à mes grands parents pour leur soutien sans faille et les nombreux repas que j'ai partagés en leur compagnie (souvent accompagnés de bonnes bouteilles). Merci aussi à mes oncles, mes tantes, ma marraine et mon parrain pour leur présence et leurs conseils de vie. Merci enfin à la famille de Catherine pour leurs conseils et leur accueil en leur sein.

Mes derniers remerciements vont à Catherine pour son sourire et son soutien constant. Elle a su me remobiliser dans les moments difficiles. Merci à toi Catherine pour ta présence.

Résumé

Dans ce travail nous nous intéressons aux comportement de vagues soumises à l'action d'une pression atmosphérique non constante, un fond mobile et la force de Coriolis. Une première partie est dédiée à l'étude de la résonance de Proudman. Nous proposons une approche mathématique rigoureuse pour étudier ce phénomène. Nous commençons par démontrer un résultat d'existence locale dans un cadre irrotationnel sur les équations des vagues (appelées aussi formulation de Zakharov/Craig-Sulem). Puis, nous justifions différents modèles asymptotiques pour généraliser cette résonance dans diverses situations physiques. Nous proposons en particulier une étude détaillée dans des eaux profondes dans un régime linéaire. Nous étudions aussi la propagation de vagues dans des eaux profondes dans un régime faiblement non-linéaire grâce aux équations de Saut-Xu et nous proposons un schéma numérique pour résoudre ces équations.

Dans une deuxième partie, nous étudions l'effet de la force de Coriolis sur les vagues. Nous démontrons un résultat d'existence locale sur les équations Castro-Lannes, équations qui généralisent la formulation de Zakharov/Craig-Sulem dans un cadre rotationnel. Nous justifions ensuite différents modèles asympotiques dans des eaux peu profondes en présence de la force de Coriolis. En particulier, nous proposons une généralisation des équations de Boussinesq (modèle asymptotique dans un régime faiblement linéaire) lorsque la force de Coriolis n'est pas négligeable. Ces équations nous permettent ensuite de justifier mathématiquement les ondes de Poincaré puis l'équation d'Ostrovsky qui généralise l'équation de Korteweg-De-Vries en présence de la force de Coriolis.

Mots clés : résonance de Proudman, effet Coriolis, équations des vagues, modèles asymptotiques.

Abstract

In this work, we are interested in the evolution of water waves under the influence of a non constant atmospheric pressure, a moving bottom and a Coriolis forcing. In a first part, we study the Proudman resonance. We propose a mathematical approach to understand this phenomenon. First, we prove a local wellposedness result in a irrotational framework on the water waves equations (also called the Zakharov/Craig-Sulem formulation). Then, we fully justify different asymptotic models. In particular, we carefully study the Proudman resonance in deep water in the linear regime. Finally, we study the propagation of water waves in a weakly nonlinear regime thanks to the Saut-Xu equations and we propose a numerical scheme in order to solve these equations.

In a second part, we study the influence of a Coriolis forcing on water waves. We prove a local wellposedness result on the Castro-Lannes equations, which generalize the Zakharov/Craig-Sulem formulation in the rotational framework. Then, we fully justify different asymptotic models where we take into account a Coriolis forcing. In particular, we generalize the Boussinesq equations (asymptotic model in a weakly nonlinear regime) in this setting. Thanks to these equations, we justify the Poincaré waves and then the Ostrovsky equation, which generalize the Korteweg-De-Vries equation when a Coriolis forcing is taken into account.

Key words : Proudman resonance, Coriolis effect, water waves equations, asymptotic models.

Contents

1 Introduction

1	Intr	itroduction		
	1.1	.1 Les équations d'Euler à surface libre		
		1.1.1 Formulation de Zakharov/Craig-Sulem-Sulem, cas d'un fluide irrotationel 1	2	
		1.1.1.1 Les équations	2	
		1.1.1.2 Adimensionnement	5	
		1.1.2 Formulation de Castro-Lannes, cas avec vorticité	8	
		1.1.2.1 Les équations	8	
		$1.1.2.2$ Adimensionnement $\ldots \ldots 2$	0	
	1.2	Modèles asymptotiques pour les équations des vagues	2	
		1.2.1 Régime d'eaux peu profondes	2	
		1.2.1.1 Équations de Saint-Venant	2	
		1.2.1.2 Équations de Green-Naghdi	2	
		1.2.2 Régime faiblement non linéaire	3	
		1.2.2.1 Équations linéaires des vagues	3	
		1.2.2.2 Équations de Saut-Xu	3	
		1.2.3 Régime d'ondes longues	3	
		$1.2.3.1$ Équation des ondes $\ldots \ldots 2$	3	
		$1.2.3.2$ Équation de Boussinesq $\ldots \ldots 2$	4	
		1.2.3.3 Équations de KdV et d'Ostrovsky	4	
	1.3	Résonance de Proudman	5	
		1.3.1 Mécanisme de la résonance de Proudman	5	
		1.3.2 Tsunamis météorologiques	6	
		1.3.3 Tsunamis par glissements de terrain	7	
		1.3.4 Résultats obtenus 2	8	
	1.4	La résonance de Proudman au delà du cadre classique, effets nonlinéaires et dis-		
		persion non locale	8	
	1.5	Perspectives de recherches	9	
		1.5.1 Vers un temps d'existence plus long dans le cas irrotationnel	9	
		1.5.2 Dispersion pour les équations linéaires des vagues	0	
		1.5.3 Vers un temps d'existence plus long dans le cas rotationnel	0	
2	A n	nathematical study of meteorological and landslide tsunamis : The Proud-		
	mai	nan resonance		
	2.1	Introduction	2	
		2.1.1 Presentation of the problem	2	
		2.1.2 Notations for this chapter	4	
	2.2	Local existence of the water waves equations	5	

		2.2.1 The model
		2.2.2 Notations for this section and statement of the main results
		2.2.5 Quashineanization $2.2.4$ Level evictors
		2.2.4 Local existence
		2.2.5 The Rayleigh-Taylor coefficient
	0.0	2.2.0 Hamiltonian system
	2.3	Asymptotic models
		2.5.1 A shahow water model for very small topography variation
		2.3.1.1 Linear asymptotic
		2.3.1.2 Resonance in snahow waters for very small topography variation a
		2.5.2 A shahow water model when the topography is not small
		2.3.2.1 Linear asymptotic $\dots \dots \dots$
		2.3.2.2 Amplification in shallow waters when p is large
		2.3.3 Linear asymptotic and resonance in intermediate depths
		2.3.4 Ioward nonlinear asymptotic models
		2.3.4.1 The nonlinear shallow water equation $\dots \dots \dots \dots \dots \dots$
		2.3.4.2 The Saut-Au equations
		$2.3.4.3 \text{Conclusion} \dots \dots \dots \dots \dots \dots \dots \dots \dots $
3	$\mathbf{A} \mathbf{s}$	plitting method for deep water with bathymetry 7
	3.1	Introduction
		3.1.1 Presentation of the problem
		3.1.2 Notations and assumptions
	3.2	The Saut-Xu system
	3.3	A splitting scheme
		3.3.1 The transport equation
		3.3.2 The dispersive equation
	3.4	Error estimates
		3.4.1 The local error estimate
		3.4.2 Global error estimate 8
	3.5	Numerical experiments
		3.5.1 Example 1: Convergence curve
		3.5.2 Example 2: Linear versus nonlinear
		3.5.3 Example 3: Boussinesq regime 8
		3.5.4 Example 4: Rapidly varying topographies
4	Cor	iolis effect on water waves
	4.1	Introduction
		4.1.1 Presentation of the problem
		4.1.2 Existence result
		4.1.3 Notations for this chapter
	4.2	The div-curl problem
		4.2.1 Notations for this part
		4.2.2 Existence and uniqueness
		4.2.3 The transformed div-curl problem
		4.2.4 Time derivatives and few remarks about the good unknown
	4.3	Well-posedness of the water waves equations
	1.0	4.3.1 Framework
		4.3.2 A priori estimate for the vorticity 11
		± 0.2 A priori estimate for the volution $y = 0$, $\dots $, $\dots $, $\dots $, $\dots $, 1 .

		4.3.3 Quasilinearization and a priori estimates	112
		4.3.4 Existence result	118
	4.4	From the Castro-Lannes system to the Euler equations, reconstruction of the	
		pressure	18
	4.5	The nonlinear shallow water equations	120
		4.5.1 The context	120
		4.5.2 WKB expansion with respect to μ	121
		4.5.3 Rigorous derivation	24
		4.5.4 The Proudman resonance, linear properties of the shallow water equations 1	125
5	Lon	g wave approximation for water waves under a Coriolis forcing 1	29
	5.1	Introduction	130
		5.1.1 Nondimensionalization and the Castro-Lannes formulation	132
		5.1.2 Notations \ldots \ldots 1	134
		5.1.3 Useful results \ldots 1	135
	5.2	Boussinesq-Coriolis equations when $\gamma = 0$	136
		5.2.1 Asymptotic expansion for the velocity and useful identities 1	137
		5.2.2 Equations for Q_x, Q_y and V^{\sharp}	139
		5.2.3 The Boussinesq-Coriolis equations	141
		5.2.4 Full justification of the Boussinesq-Coriolis equations	42
	5.3	Different asymptotic models in the Boussinesq regime over a flat bottom 1	47
		5.3.1 Strong rotation, the Poincaré waves	47
		5.3.2 The Proudman resonance in presence of a Coriolis forcing	150
		5.3.3 Weak rotation, the Ostrovsky equation	153
		5.3.4 Very weak rotation, the KdV equation	157
	5.4	Green-Naghdi equations for $\gamma = 0$ and $\beta = \mathcal{O}(\mu)$	159
		5.4.1 Improvements for the equations of Q and Q	160
		5.4.2 Equations for E	64
		5.4.3 The Green-Naghdi equations	165
		5.4.4 A simplified model in the case of a weak rotation and medium amplitude	100
		waves	66
		wavts	100
\mathbf{A}	Lap	lace problems for water waves 1	67
	A.1	The Laplace problems	168
		A.1.1 Formulation of the problems 1	168
		A.1.2 Regularity estimates of the solutions	174
		A.1.3 Shape derivatives of the $B^{\mathfrak{d}}$	176
	A.2	The Dirichlet-Neumann and the Neumann-Neumann operators	180
		A.2.1 Main properties	180
		A.2.2 Regularity Estimates	84
		A 2.3 Shape derivatives	87
\mathbf{B}	Use	ful estimates 1	91
	B.1	Estimate for \mathfrak{P}	192
	B.2	Product estimates	192
	B.3	Commutator estimates	193
	B.4	Estimates for \mathcal{H}_{μ}	194
<u>р</u> ,	ել։	nombio	٥٣
D 1	nnoā	rapine 1	90

Chapter 1

Introduction

Sommaire

1 Les équations d'Euler à surface libre		
1.1.1	Formulation de Zakharov/Craig-Sulem-Sulem, cas d'un fluide irrota-	
	tionel	12
1.1.2	Formulation de Castro-Lannes, cas avec vorticité	18
2 Moo	dèles asymptotiques pour les équations des vagues	22
1.2.1	Régime d'eaux peu profondes	22
1.2.2	Régime faiblement non linéaire	23
1.2.3	Régime d'ondes longues	23
Rés	onance de Proudman	25
1.3.1	Mécanisme de la résonance de Proudman	25
1.3.2	Tsunamis météorologiques	26
1.3.3	Tsunamis par glissements de terrain	27
1.3.4	Résultats obtenus	28
l Lar	ésonance de Proudman au delà du cadre classique, effets non-	
linéa	aires et dispersion non locale	28
6 Pers	spectives de recherches	29
1.5.1	Vers un temps d'existence plus long dans le cas irrotationnel \ldots .	29
1.5.2	Dispersion pour les équations linéaires des vagues	30
1.5.3	Vers un temps d'existence plus long dans le cas rotationnel $\ . \ . \ .$	30
	Les 1.1.1 1.1.2 Mod 1.2.1 1.2.2 1.2.3 Rés 1.3.1 1.3.2 1.3.3 1.3.4 La r linés 5 Pers 1.5.1 1.5.2 1.5.3	Les équations d'Euler à surface libre 1.1.1 Formulation de Zakharov/Craig-Sulem-Sulem, cas d'un fluide irrotationel 1.1.2 Formulation de Castro-Lannes, cas avec vorticité 1.1.2 Formulation de Castro-Lannes, cas avec vorticité Modèles asymptotiques pour les équations des vagues 1.2.1 Régime d'eaux peu profondes 1.2.2 Régime faiblement non linéaire 1.2.3 Régime d'ondes longues 1.2.4 Régime d'ondes longues 1.2.5 Résonance de Proudman 1.3.1 Mécanisme de la résonance de Proudman 1.3.2 Tsunamis météorologiques 1.3.3 Tsunamis par glissements de terrain 1.3.4 Résultats obtenus 1.3.4 Résultats obtenus 1.5.1 Vers un temps d'existence plus long dans le cas irrotationnel 1.5.2 Dispersion pour les équations linéaires des vagues 1.5.3 Vers un temps d'existence plus long dans le cas rotationnel

L'objectif principal de ce manuscrit est d'analyser le comportement de vagues soumises à l'action de forces extérieures autres que la force de gravité. Ce premier chapitre a pour but de replacer le contexte et le cheminement du présent document. Il est suivi de quatre autres chapitres. Le deuxième chapitre s'intéresse à la propagation de vagues induites par des tempêtes ou des glissements de terrain sous-marins. Pendant de tels événements, il peut se produire une amplification appelée résonance de Proudman ([120]). Ce chapitre a pour but d'étudier cette résonance dans différentes situations. Il est tiré de l'article [102]. Le troisième chapitre étudie la propagation de vagues dans des eaux dites profondes, à savoir lorsque la longueur des vagues est comparable à la hauteur d'eau. Il a été écrit en collaboration avec Afaf Bouharguane (IMB). Nous donnons un schéma numérique basé sur les équations de Saut-Xu ([125]). Différentes simulations numériques sont proposées pour valider notre schéma numérique. Nous étudions aussi l'effet de fonds fortement oscillants sur la propagation d'une vague. Ce chapitre est tiré de l'article [24]. Les deux derniers chapitres de ce manuscrit ont pour but d'étudier l'effet de la force de Coriolis sur les vagues. Dans le quatrième chapitre, nous proposons une analyse mathématique de ce phénomène et nous justifions les équations de Saint-Venant dans le cas d'eaux dites peu profondes. Ce chapitre est tiré de l'article [101]. Enfin, dans le dernier chapitre, nous étudions l'effet de la force de Coriolis dans le régime Boussinesq. Ce régime est valide pour des vagues ayant une grande longueur d'onde et une petite amplitude. Nous montrons que l'équation de Korteweig de Vries et l'équation d'Ostrovsky sont des modèles raisonnables dans un tel régime. Ce chapitre est tiré de l'article [103].

1.1 Les équations d'Euler à surface libre

Dans ce manuscrit, nous nous intéressons à l'évolution d'un fluide soumis à son propre poids et à la force de Coriolis. Le fluide est supposé délimité par deux frontières : le fond et la surface (voir la figure 1.1). Nous supposons que ces deux frontières peuvent être paramétrées par des graphes et qu'elles peuvent évoluer au cours du temps. Tout effet de déferlement est donc exclu. Le fond est imperméable. La surface sépare notre fluide de l'air et nous supposons qu'il n'y a pas d'échange entre notre fluide et l'extérieur (pas d'évaporation par exemple). Le fluide est supposé parfait et homogène, c'est à dire que tous les effets de viscosité et de conductivité thermique sont négligés, que notre fluide a une masse volumique constante et qu'il vérifie le principe de conservation de la masse. En outre, tous les effets de tension de surface sont négligés. Enfin, la pression à la surface et l'évolution du fond sont connues. Pour modéliser mathématiquement un tel fluide, nous avons recours aux équations d'Euler à surface libre.

Dans la suite, nous notons d la dimension horizontale, $z \in \mathbb{R}$ la variable verticale et $X \in \mathbb{R}^d$ la variable horizontale : X = (x, y) si d = 2. Pour des applications concrètes, d est égal à 1 ou 2. On note la masse volumique du fluide ρ , la surface $\zeta(t, X)$ et on paramètre le fond par -H + b(t, X) où H représente la profondeur moyenne et b la variation du fond. Le domaine fluide à l'instant t est noté Ω_t

$$\Omega_t = \{ (X, z), -H + b(t, X) < z < \zeta(t, X) \}.$$
(1.1)

La vitesse dans le fluide est notée $\mathbf{U} = \mathbf{U}(t, X, z)$, sa composante horizontale par \mathbf{V} et sa composante verticale par w. On note $\mathcal{P} = \mathcal{P}(t, X, z)$ la pression dans le fluide et P = P(t, X) la pression à la surface. Enfin, comme nous l'avons dit précédemment, notre fluide est soumis à la force de gravité $\mathbf{g} = -g\mathbf{e}_{\mathbf{z}}$ et à la rotation de la Terre de vecteur rotation $\mathbf{f} = \frac{f}{2}\mathbf{e}_{\mathbf{z}}$. Nous



Figure 1.1: Notations

pouvons à présent donner les équations d'Euler à surface libre

$$\begin{cases} \partial_t \mathbf{U} + (\mathbf{U} \cdot \nabla_{X,z}) \mathbf{U} + \mathbf{f} \times \mathbf{U} = -\frac{1}{\rho} \nabla_{X,z} \mathcal{P} - g \mathbf{e}_{\mathbf{z}} \operatorname{dans} \Omega_t, \\ \operatorname{div}_{X,z} \mathbf{U} = 0 \operatorname{dans} \Omega_t, \end{cases}$$
(1.2)

auxquelles nous rajoutons les conditions aux bords

$$\begin{cases} \partial_t \zeta - \underline{U} \cdot N = 0, \\ \partial_t b - U_b \cdot N_b = 0, \\ \mathcal{P}_{|z=\zeta} = P, \end{cases}$$
(1.3)

où $N = \begin{pmatrix} -\nabla_X \zeta \\ 1 \end{pmatrix}$ et $N_b = \begin{pmatrix} -\nabla_X b \\ 1 \end{pmatrix}$ sont les vecteurs normaux respectivement à la surface et au fond et $\underline{\mathbf{U}} = \begin{pmatrix} \mathbf{V} \\ \underline{\mathbf{W}} \end{pmatrix} = \mathbf{U}_{|z=\zeta}$ et $\mathbf{U}_b = \begin{pmatrix} \mathbf{V}_b \\ \mathbf{W}_b \end{pmatrix} = \mathbf{U}_{|z=-H+b}$ sont les traces de la vitesse respectivement à la surface et au fond.

Rappelons que la première équation du sytème (1.2) correspond à l'équation de bilan de la quantité de mouvement et découle du principe fondamental de la dynamique. La deuxième équation du sytème (1.2) correspond à la conservation de la masse et à l'incompressibilité du fluide. Enfin, les deux premières conditions aux bords traduisent l'absence d'échange de matière entre notre fluide et l'extérieur. La troisième condition aux bords traduit la continuité à la surface entre la pression dans notre fluide et la pression extérieure (nous avons négligé la tension de surface). À ce stade nous avons donc un système d'équations sur la vitesse U et la surface libre ζ lesquelles sont alors, avec la pression \mathcal{P} , les inconnues du problème. En fait, la pression \mathcal{P} n'est pas une vraie inconnue mais le multiplicateur de Lagrange associé à la contrainte d'incompressibilité.

Dans la suite, nous serons amenés à séparer deux situations. Premièrement, nous supposerons que la force de Coriolis et les effets dus à la vorticité sont négligeables. Cette hypothèse sera

utilisée dans les deuxième et troisième chapitres de ce manuscrit. Nous verrons que cela simplifie les équations d'Euler. Puis, dans les chapitres quatre et cinq, nous ne négligerons ni la vorticité ni la force de Coriolis. Il est assez courant de négliger la vorticité en océanographie lorsque l'on s'intéresse à la propagation de vagues loin des côtes. En revanche, cette hypothèse n'est pas réaliste lorsque l'on souhaite prendre en compte la force de Coriolis. En effet, si nous prenons le rotationnel de la première équation de (1.2) et que nous notons $\boldsymbol{\omega} = \operatorname{rot}_{X,z} \mathbf{U}$, nous obtenons l'équation suivante

$$\partial_t \boldsymbol{\omega} + (\mathbf{U} \cdot \nabla_{X,z}) \boldsymbol{\omega} - (\boldsymbol{\omega} \cdot \nabla_{X,z}) \mathbf{U} = f \partial_z \mathbf{U} \operatorname{dans} \Omega_t.$$

La force de Coriolis génère donc de la vorticité même si cette dernière est initialement nulle. Pour quantifier l'influence de la force de Coriolis sur les vagues, les physiciens utilisent le nombre de Rossby, noté Ro dans ce manuscrit. C'est un nombre sans dimension qui représente le rapport entre les forces d'inertie et l'effet de la rotation de la Terre :

$$\mathrm{Ro} = \frac{\mathrm{V}_0}{fL}$$

où V₀ est la vitesse typique horizontale, L la longueur typique de notre phénomène et f le paramètre de Coriolis valant $f = 2\omega \sin(\phi)$, où ω est la vitesse angulaire de notre référentiel tournant et ϕ la latitude. Pour la Terre, $\omega = 7,310^{-5}s^{-1}$. Ainsi, pour une latitude moyenne ϕ comprise entre 30 degrés et 60 degrés, le paramètre f est de l'ordre de $10^{-4}s^{-1}$. Nous voyons donc que, pour des vagues de longueurs d'onde de l'ordre du kilomètre avec une vitesse typique de $1m.s^{-1}$, les effets de la force de Coriolis sont négligeables. Cependant, pour des vagues ayant une plus grande longue d'onde, cette simplification n'est pas réaliste. En particulier, pour une vague créée par un tsunami, la longueur d'onde typique peut atteindre la dizaine voire la centaine de kilomètres et la force de Coriolis a tendance à atténuer son amplitude. Nous étudions cet effet dans la section 4.5.4 lorsque d = 2 et dans la section 5.3.2 lorsque d = 1 (voir aussi [61], [93]).

1.1.1 Formulation de Zakharov/Craig-Sulem-Sulem, cas d'un fluide irrotationel

1.1.1.1 Les équations

Dans cette partie, comme dans les chapitres 2 et 3, nous supposons que la force de Coriolis et les effets de vorticité sont négligeables. Nous avons donc

$$\operatorname{rot}_{X,z} \mathbf{U} = 0 \operatorname{dans} \Omega_t. \tag{1.4}$$

Si, en outre, Ω_t est simplement connexe, **U** peut s'écrire sous la forme d'un gradient. Un moyen simple d'assurer que le domaine fluide est simplement connexe est de supposer la condition suivante

$$\exists h_{\min} > 0, \zeta + H - b \ge h_{\min}. \tag{1.5}$$

Nous ferons cette hypothèse dans toute la suite du manuscrit. Nous avons donc l'existence d'un potentiel Φ , appelé potentiel des vitesses, tel que

$$\mathbf{U} = \nabla_{X,z} \Phi. \tag{1.6}$$

On peut alors réécrire les équations d'Euler (1.2) de la façon suivante

$$\begin{cases} \partial_t \Phi + \frac{1}{2} \left| \nabla_{X,z} \Phi \right|^2 + gz = -\frac{1}{\rho} \left(\mathcal{P} - P \right) \text{ dans } \Omega_t, \\ \Delta_{X,z} \Phi = 0 \text{ dans } \Omega_t. \end{cases}$$
(1.7)

La première équation s'appelle l'équation de Bernoulli. La deuxième équation montre que le potentiel des vitesses est une fonction harmonique dans le domaine fluide. Remarquons que nous avons implicitement supposé que le fluide est au repos à l'infini. Nos inconnues sont maintenant la surface ζ et le potentiel des vitesses Φ . Cependant, Φ est défini dans le domaine fluide Ω_t qui dépend de la surface ζ et qui varie au cours du temps. Ce la complique l'analyse mathématique. Un moyen de fixer le domaine est de travailler en coordonnées lagrangiennes. Avec cette approche, divers auteurs ont montré des résultats d'existence locale (voir par exemple [39], [90], [42], [155], [156]). Une autre possibilité lorsque d = 1 est de voir le domaine fluide comme un domaine de \mathbb{C} et les conditions d'incompressibilité et d'irrotationalité comme des conditions de Cauchy-Riemann sur le conjugué de la vitesse. Grâce à cette approche, Nalimov ([110]) a obtenu le premier résultat d'existence locale dans le cas d = 1, avec une profondeur infinie et des données initiales Sobolev assez petites. Yosihara ([152]) a ensuite étendu le résultat de Nalimov à un fond fixe fini et Wu ([150]) a obtenu un résultat d'existence locale dans le cas d'une profondeur infinie et de données initiales Sobolev quelconque. Wu a aussi traité le cas d = 2 ([151]) en utilisant l'analyse de Clifford. Nous n'allons pas nous baser sur ces approches mais plutôt utiliser l'approche de Zakharov [153]. Il remarqua que la connaissance de la trace à la surface du potentiel des vitesses suffit pour déterminer Φ . En effet, en notant $\Phi_{|z=\zeta} = \psi$ et en utilisant la deuxième équation des conditions aux bords (1.3) et la deuxième équation de (1.7), nous obtenons le problème de Laplace suivant

$$\begin{cases} \Delta_{X,z} \Phi = 0 \text{ dans } \Omega_t, \\ \Phi_{|z=\zeta} = \psi , \sqrt{1 + |\nabla_X b|^2} \partial_{\mathbf{n}} \Phi_{|z=-H+b} = \partial_t b. \end{cases}$$
(1.8)

Nous allons décomposer Φ en deux parties en séparant la contribution de la surface et du fond

$$\Phi = \Phi^S + \Phi^B.$$

Nous obtenons les deux problèmes de Laplace suivant

$$\begin{cases} \Delta_{X,z} \Phi^S = 0 \text{ dans } \Omega_t, \\ \Phi^S_{|z=\zeta} = \psi , \sqrt{1 + |\nabla_X b|^2} \partial_{\mathbf{n}} \Phi^S_{|z=-H+b} = 0, \end{cases}$$
(1.9)

 et

$$\begin{cases} \Delta_{X,z} \Phi^B = 0 \text{ dans } \Omega_t, \\ \Phi^B_{|z=\zeta} = 0, \ \sqrt{1+|\nabla_X b|^2} \partial_{\mathbf{n}} \Phi^B_{|z=-H+b} = \partial_t b. \end{cases}$$
(1.10)

Notons ici que ${\bf n}$ est le vecteur normal normalisé qui pointe vers le haut

$$\mathbf{n} = \begin{cases} \frac{1}{\sqrt{1 + |\nabla_X \zeta|^2}} N, \text{ à la surface,} \\ \frac{1}{\sqrt{1 + |\nabla_X b|^2}} N_b, \text{ au fond.} \end{cases}$$
(1.11)

Nous étudions en détail ces deux problèmes de Laplace dans l'Appendice A. En particulier, nous montrons l'existence et l'unicité de Φ . Nos inconnues sont maintenant la trace du potentiel des

vitesses ψ qui est définie sur \mathbb{R}^d et la surface ζ . Ainsi, via le procédé de Zakharov, nous avons fixé le domaine de nos inconnues. Il s'agit alors de trouver deux équations d'évolution sur ζ et ψ . On obtient une première équation grâce à la première équation de (1.3). Nous introduisons les opérateurs suivants : l'opérateur de Dirichlet-Neumann

$$G[\zeta, b]: \psi \mapsto \sqrt{1 + |\nabla_X \zeta|^2} \partial_{\mathbf{n}} \Phi^S_{|z=\zeta}, \qquad (1.12)$$

où Φ^S satisfait le système (1.9), et l'opérateur de Neumann-Neumann

$$G^{NN}[\zeta, b]: \partial_t b \mapsto \sqrt{1 + |\nabla_X \zeta|^2} \partial_\mathbf{n} \Phi^B_{|z=\zeta}, \tag{1.13}$$

où Φ^B satisfait (1.10). Nous pouvons alors reformuler la première équation de (1.3) en

$$\partial_t \zeta - G[\zeta, b](\psi) = G^{NN}[\zeta, b](\partial_t b).$$
(1.14)

Pour obtenir une deuxième équation d'évolution, nous prenons la trace de l'équation de Bernoulli à la surface. En utilisant la troisième équation de (1.3), nous pouvons alors supprimer la pression \mathcal{P} . Nous obtenons alors le système d'équations suivant, appelé équations des vagues,

$$\begin{cases} \partial_t \zeta - G[\zeta, b](\psi) = G^{NN}[\zeta, b](\partial_t b), \\ \partial_t \psi + g\zeta + \frac{1}{2} |\nabla_X \psi|^2 - \frac{1}{2} \frac{\left(G[\zeta, b](\psi) + G^{NN}[\zeta, b](\partial_t b) + \nabla_X \zeta \cdot \nabla_X \psi\right)^2}{(1 + |\nabla_X \zeta|^2)} = -\frac{P}{\rho}. \end{cases}$$
(1.15)

Le système (1.15) porte aussi le nom de formulation de Zakharov/Craig-Sulem-Sulem. En effet, en utilisant le travail de Zakharov, Craig, Sulem et Sulem ([47], [48]) ont obtenu ce sytème dans le cas d'un fond plat fixe et avec une pression constante à la surface. Le grand avantage de cette formulation est de ramener les équations d'Euler à surface libre à un système d'équations à la surface.

D'un point de vue mathématique, ces équations sont complètement non linéaires et liées à un problème de surface libre, ce qui rend leur étude délicate. Lannes ([78]) montre le premier résultat d'existence locale. Sa preuve repose sur un schéma de Nash-Moser. Puis, Iguchi ([70]) montre que l'on peut quasilinéariser et symétriser ces équations. Citons aussi les travaux d'Alazard, Burg et Zuily (voir par exemple [4], [5]) qui permettent d'améliorer la régularité minimale demandée pour la donnée initiale en utilisant le côté dispersif des équations des vagues dans le cas d'une profondeur finie (voir aussi l'article [52] de De Poyferré pour un critère d'explosion). Enfin, les travaux récents de Wang ([148], [147]) montrent que, pour d = 2, il y a existence globale des équations des vagues pour des petites données initiales dans le cas d'un fond plat (b = 0). Tous les résultats donnés précédemment font l'hypothèse d'une pression à la surface constante et d'un fond fixe. Iguchi ([71]) est le premier à étudier le cas d'un fond mobile pour modéliser la propagation d'une vague créée par une variation brusque de la bathymétrie. Dans la section 2.2, nous généralisons le résultat d'Iguchi ([71]) en ajoutant une pression non constante à la surface, en baissant la régularité minimale de la donnée initiale et en considérant des potentiels des vitesses plus généraux. Notons enfin le résultat de contrôle pour les équations des vagues [2] pour lequel Alazard, Baldi et Han-Kwan étudient le cas d'une pression non constante à la surface mais dans le cas périodique (voir aussi [1]).

Comme nous l'avons dit préédemment, Iguchi ([70]) prouve l'existence locale des équations des vagues en *quasilinéarisant* et en *symétrisant* les équations. Nous allons expliquer ce principe sur un exemple plus simple. On considère un système d'équations d'évolution quasilinéaire de la forme

$$\partial_t U + \mathcal{A}(U, \nabla_X)U = 0.$$

Si on peut trouver un opérateur matriciel $S(U, \nabla_X)$ défini positif tel que l'opérateur matriciel $S(U, \nabla_X)\mathcal{A}(U, \nabla_X)$ soit anti-symétrique, on dit que le système est symétrisable et on obtient l'existence locale en étudiant l'énergie $(S(U, \nabla_X)U, U)_{L^2}$. Maintenant, si nous avons un système complètement non linéaire

$$\partial_t U + f(U) = 0,$$

on dit que ce système peut être quasilinéarisé et symétrisé si, quand on dérive l'équation assez de fois, on obtient un système d'équations d'évolution sur U et ses dérivées qui est quasilinéaire symétrisable. Pour les équations des vagues, nous avons une difficulté supplémentaire. Comme nous avons un problème à surface libre, la bonne inconnue d'Alinhac va jouer un rôle crucial pour clore les estimations d'énergie. Cette notion a été introduite pour la première fois par Alinhac ([7]) dans le cadre d'ondes de raréfaction pour des systèmes hyperboliques quasilinéaires. Alinhac montre alors que pour pouvoir clore les estimations d'énergie pour un problème à surface libre quasilinéaire (il remarque qu'il manque une demi régularité pour clore ses estimations), il ne faut pas considérer des dérivées classiques pour nos inconnues, mais des dérivées adaptées à notre domaine mobile. Pour les équations des vagues, la notion de bonne inconnue d'Alinhac a été introduite par Alazard et Métivier ([6]). Elle avait cependant été utilisée par Lannes ([78]) et Iguchi ([70]) sans faire le lien avec les travaux d'Alinhac.

Pour finir sur cette partie, notons que nous avons réduit les équations d'Euler à surface libre à une équation à la surface. Nous sommes partis d'une solution d'Euler pour obtenir une solution des équations des vagues. On peut aussi faire le chemin inverse. Alazard, Burq et Zuily ([3]) ont montré que, à partir d'une solution des équations des vagues, on peut reconstruire la pression dans le domaine fluide et obtenir une solution des équations d'Euler.

1.1.1.2 Adimensionnement

Pour étudier la propagation de vagues, le système (1.15) s'avère en général trop compliqué. Il contient toute la dynamique du fluide. Nous allons donc simplifier des termes qui nous semblent insignifiants. Mais comment savoir lesquels ? Un moyen très simple et connu des physiciens est le principe d'adimensionnement. On adimensionne chaque variable du système par une quantité caractéristique. Dans notre cas nous avons 4 échelles caractéristiques : la longueur L_x des vagues que nous étudions, leur largeur L_y , leur amplitude a et la profondeur typique H de l'océan. Le fond doit aussi être adimensionné. Nous travaillons ici avec un fond mobile. Le but de ce manuscrit et en particulier le chapitre 2 est d'étudier la propagation de vagues créées par des glissements de terrain sous-marins. Notre fond b peut s'écrire de la forme $b(t, X) = b_0(X) + b_m(t, X)$ et nous avons deux échelles caractéristiques, l'amplitude du fond fixe a_{bott} et l'amplitude de notre glissement de terrain $a_{bott,m}$. Ensuite, nous pouvons créer 5 paramètres sans dimension

$$\varepsilon = \frac{a}{H}, \ \mu = \frac{H^2}{L_x^2}, \ \gamma = \frac{L_x}{L_y}, \ \beta = \frac{a_{\text{bott}}}{H}, \ \lambda = \frac{a_{\text{bott}}, m}{a_{\text{bott}}},$$

où ε est appelé paramètre de non-linearité, μ est appelé paramètre de faible profondeur, γ est appelé paramètre de transversalité, $\beta = \frac{a_{\text{bott}}}{H}$ est appelé paramètre de bathymétrie et où λ compare l'amplitude du fond fixe à celle du fond mobile. Nous pouvons adimensionner les

variables de nos équations

$$\begin{cases} x' = \frac{x}{L_x}, y' = \frac{y}{L_y}, z' = \frac{z}{H}, \zeta' = \frac{\zeta}{a}, b' = \frac{b}{a_{\text{bott}}}, b'_0 = \frac{b_0}{a_{\text{bott}}}, b'_m = \frac{b_m}{a_{\text{bott,m}}}, t' = \frac{\sqrt{gH}}{L}t, \\ \left(\Phi^S\right)' = \frac{H}{aL\sqrt{gH}}\Phi^S, \left(\Phi^B\right)' = \frac{L}{Ha_{\text{bott,m}}\sqrt{gH}}\Phi^B, \psi' = \frac{H}{aL\sqrt{gH}}\psi, P' = \frac{P}{a\rho g}. \end{cases}$$
(1.16)

L'adimensionnalisation de Φ^S , ψ et t provient de la théorie linéaire (voir paragraphe **1.3.2** dans [80]). Nous expliquons l'adimensionnalisation de Φ^B dans le chapitre 2. Dans la suite, nous omettons les primes pour simplifier les notations. Nous notons

$$\nabla_{X,z}^{\mu,\gamma} = (\sqrt{\mu}\nabla_X^{\gamma}, \partial_z)^t, \ \Delta_{X,z}^{\mu,\gamma} := \mu\Delta_X^{\gamma} + \partial_z^2, \ \nabla_X^{\gamma} = (\partial_x, \gamma\partial_y)^t \text{ et } \Delta_X^{\gamma} = \partial_z^2 + \gamma\partial_y^2.$$

Les équations des vagues s'adimensionnent alors de la façon suivante

$$\begin{cases} \partial_t \zeta - \frac{1}{\mu} G_\mu[\varepsilon\zeta,\beta b](\psi) = \frac{\beta\lambda}{\varepsilon} G_\mu^{NN}[\varepsilon\zeta,\beta b](\partial_t b), \\ \partial_t \psi + \zeta + \frac{\varepsilon}{2} |\nabla_X^\gamma \psi|^2 - \frac{\varepsilon}{2\mu} \frac{\left(G_\mu[\varepsilon\zeta,\beta b](\psi) + \frac{\lambda\beta\mu}{\varepsilon} G_\mu^{NN}[\varepsilon\zeta,\beta b](\partial_t b) + \mu\varepsilon\nabla_X^\gamma \zeta \cdot \nabla_X^\gamma \psi\right)^2}{(1 + \varepsilon^2 \mu |\nabla_X^\gamma \zeta|^2)} = -P, \end{cases}$$

$$(1.17)$$

où l'opérateur de Dirichlet-Neumann $G_{\mu}[\varepsilon\zeta,\beta b]$ est

$$G_{\mu}[\varepsilon\zeta,\beta b](\psi) := \sqrt{1 + \varepsilon^2 |\nabla_X^{\gamma}\zeta|^2} \partial_{\mathbf{n}} \Phi^S_{|z=\varepsilon\zeta} , \qquad (1.18)$$

avec Φ^S qui satisfait

$$\begin{cases} \Delta_{X,z}^{\mu,\gamma} \Phi^S = 0 \, \operatorname{dans} \, \Omega_t \, , \\ \Phi^S_{|z=\varepsilon\zeta} = \psi \, , \, \partial_{\mathbf{n}} \Phi^S_{|z=-1+\beta b} = 0, \end{cases}$$
(1.19)

et l'opérateur de Neumann-Neumann $G^{NN}_{\mu}[\varepsilon\zeta,\beta b]$ est

$$G^{NN}_{\mu}[\varepsilon\zeta,\beta b](\partial_t b) := \sqrt{1 + \varepsilon^2 |\nabla^{\gamma}_X \zeta|^2} \partial_{\mathbf{n}} \Phi^B_{|z=\varepsilon\zeta} , \qquad (1.20)$$

où Φ^B satisfait

$$\begin{cases} \Delta_{X,z}^{\mu,\gamma} \Phi^B = 0 \text{ dans } \Omega_t , \\ \Phi^B_{|z=\varepsilon\zeta} = 0 , \sqrt{1+\beta^2 |\nabla_X^{\gamma} b|^2} \partial_{\mathbf{n}} \Phi^B_{|z=-1+\beta b} = \partial_t b. \end{cases}$$
(1.21)

Notons que $\partial_{\mathbf{n}}$ est ici la dérivée conormale point ant vers le haut

$$\partial_{\mathbf{n}} \Phi = \mathbf{n} \cdot \begin{pmatrix} \sqrt{\mu} I_d \ 0 \\ 0 \ 1 \end{pmatrix} \nabla^{\mu,\gamma}_{X,z} \Phi_{|\partial\Omega}$$

Dans [9], Alvarez-Samaniego et Lannes montrent l'existence locale du système (1.17) dans le cas d'une pression constante à la surface et d'un fond fixe. Leur temps d'existence est de la forme $\frac{T}{\max(\varepsilon,\beta)}$ où T est indépendant de μ , ε et β . Dans [70] et [71], Iguchi obtient aussi un temps d'existence indépendant de μ dans le cas d'un fond mobile ou fixe mais avec un adimensionnement un peu différent. Mésognon-Gireau ([105]) généralise le résultat de [9] et montre que, si l'on rajoute de la tension de surface, on peut avoir un temps d'existence de la forme $\frac{T}{\varepsilon}$. Notons que le

rajout de la tension de surface est essentiel dans son travail et qu'il considère des coefficients de tension de surface qui ne sont pas aberrants physiquement. Dans la section 2.2, nous obtenons un temps d'existence de la forme $\frac{T}{\sqrt{\max(\varepsilon,\beta)}}$ (Théorème 2.2.4) pour le système (1.17) avec pression non constante à la surface et fond mobile.

Le temps d'existence que nous avons obtenu peut sembler moins intéressant que celui d'Alvarez-Samaniego et Lannes [9]. Pourtant il s'avère optimal si l'on considère des solutions fortes et si le fond est plat ($\beta = 0$) car nous avons un terme source de taille 1 et non de taille max(ε, β). Pour expliquer comment nous avons pu obtenir un tel temps d'existence et pourquoi il est optimal, prenons un modèle plus simple. Nous considérons le problème suivant défini sur \mathbb{R} :

$$\partial_t u + \varepsilon u \partial_x u + L u = f,$$

avec L un opérateur anti-symétrique qui commute avec ∂_x , f un terme source d'énergie finie et ε un petit paramètre. Nous cherchons des solutions fortes à ce problème. Nous allons donc effectuer des estimations d'énergie. Si nous considérons l'énergie $E = E(t) = |u(t, \cdot)|^2_{H^2}$, alors nous pouvons montrer que

$$\frac{d}{dt}E \leq \varepsilon C E^{\frac{3}{2}} + |f| E^{\frac{1}{2}}.$$

Si f est identiquement nulle, nous pouvons obtenir un temps d'existence de la forme $\frac{T}{\varepsilon}$ qui correspond au temps donné par Alvarez-Samaniego et Lannes. On appelle souvent ce temps d'existence, le temps hyperbolique. Si maintenant f n'est pas petit, une estimation brutale nous donne un temps d'existence de l'ordre de 1 qui n'est pas très intéressant si l'on souhaite étudier notre phénomène sur de longues durées. Pour améliorer ce temps d'existence, on considére une nouvelle énergie $F = F(\tau) = \varepsilon E(\frac{\tau}{\sqrt{\varepsilon}})$. Nous pouvons alors montrer que

$$\frac{d}{d\tau}F \le C(f)\left(F^{\frac{3}{2}} + F^{\frac{1}{2}}\right).$$

Ainsi, il existe un temps T > 0 et une constante C > 0,

$$\sup_{t \in \left[0, \frac{T}{\sqrt{\varepsilon}}\right]} \varepsilon E(t) = \sup_{\tau \in [0,T]} F(\tau) \le C,$$

et notre solution u existe sur un temps $\frac{T}{\sqrt{\varepsilon}}$. Il y a cependant une contrepartie : l'énergie n'est pas bornée uniformément par rapport à ε sur ce temps. Cela jouera un rôle dans la justification de modèles asymptotiques (voir Section 2.3). Nous allons maintenant montrer que ce temps d'existence est optimal. Supposons pour simplifier que L est identiquement nulle, que l'on part d'une donnée initiale nulle et que f ne dépend pas du temps. Nous cherchons une solution forte à l'équation de Burgers 1d avec terme source

$$\begin{cases} \partial_t u + \varepsilon u \partial_x u = f, \\ u_{|t=0} = 0. \end{cases}$$

Un moyen de résoudre de manière exacte cette équation est la méthode des caractéristiques. Nous notons $\Phi_0^t(x)$ la solution de l'équation différentielle ordinaire suivante

$$\begin{cases} \frac{d}{dt} \Phi_0^t(x) = \varepsilon u(t, \Phi_0^t(x)) \\ \Phi_0^0(x) = x. \end{cases}$$

Nous avons donc

$$u(t,\Phi_0^t(x))=\int_0^t f(\Phi_0^s(x))ds.$$

Cela caractérise la solution u. Nous voyons aussi que $\Phi_0^t(x)$ vérifie l'équation différentielle ordinaire suivante

$$\begin{cases} \displaystyle \frac{d^2}{dt^2} \Phi_0^t(x) = \varepsilon f(\Phi_0^t(x)), \\ \Phi_0^0(x) = x, \\ \left(\frac{d}{dt} \Phi_t^0(x)\right)_{|t=0} = 0. \end{cases}$$

Il est alors facile de montrer que l'on peut trouver des fonctions f telles que la solution de cette équation différentielle ordinaire existe sur $\left[0, \frac{T}{\sqrt{\varepsilon}}\right]$ (T indépendant de ε) et non pas sur $\left[0, \frac{T}{\varepsilon^{\alpha}}\right]$ pour $\alpha > \frac{1}{2}$. Au delà du temps $\left[0, \frac{T}{\sqrt{\varepsilon}}\right]$, les courbes caractéristiques se croisent et une solution forte ne peut plus continuer à exister. Le temps d'existence $\frac{T}{\sqrt{\varepsilon}}$ est donc un temps optimal dans ce contexte. Ainsi, le temps d'existence que nous trouvons au théorème 2.2.4 est un temps hyperbolique dans le cadre d'un système hyperbolique avec terme source.

1.1.2 Formulation de Castro-Lannes, cas avec vorticité

1.1.2.1 Les équations

Dans cette partie, comme dans les chapitres 4 et 5, nous prenons en compte la vorticité. Nous supposons aussi que le fond est fixe. Nous devons résoudre le système (1.2) avec les conditions aux bords (1.3). Nos inconnues sont la vitesse U et la surface ζ . Comme dans le cas précédent, la vitesse est définie dans le domaine fluide (1.1) qui évolue au cours du temps. Divers résultats d'existence locale ont été montrés en utilisant une approche Lagrangienne ([39], [90], [42], [155], [156]). Une autre solution consiste à redresser le domaine grâce à un difféomorphisme (voir par exemple [96]). Nous avons vu dans le cas irrotationnel que nous pouvons réduire les équations d'Euler à surface libre à un système d'équations à la surface. À cause de la vorticité, ce ne sera pas possible ici. Cependant nous pouvons essayer de réduire au maximum le nombre d'inconnues définies dans le domaine fluide. En adoptant ce point de vue, Castro et Lannes ([34]) ont proposé une généralisation de la formulation de Zakharov/Craig-Sulem-Sulem pour des fluides avec vorticité. Nous suivons leur approche dans ce manuscrit. En prenant la trace à la surface de la première équation de (1.2) et en utilisant la première équation de (1.3), nous obtenons l'équation

$$\partial_t \underline{\mathbf{U}} + (\underline{\mathbf{V}} \cdot \nabla_X) \,\underline{\mathbf{U}} + f\left(\begin{array}{c} \underline{\mathbf{V}}^{\perp} \\ 0 \end{array}\right) = - \begin{pmatrix} \nabla_X P \\ 0 \end{pmatrix} - \begin{pmatrix} 0 \\ g \end{pmatrix} - (\partial_z \mathcal{P})_{|z=\zeta} N, \tag{1.22}$$

où si $\mathbf{V} = \begin{pmatrix} V_1 \\ V_2 \end{pmatrix} \in \mathbb{R}^2$, nous définissons $\mathbf{V}^{\perp} = \begin{pmatrix} -V_2 \\ V_1 \end{pmatrix}$. On supprime alors le terme de pression $\partial_z \mathcal{P}_{|z=\zeta} N$ en prenant le produit vectoriel de l'équation précédente avec N. Nous introduisons la notation suivante :

$$\mathbf{U}_{\mathbb{I}} = \underline{\mathbf{V}} + \underline{\mathbf{w}} \nabla_{\mathbf{X}} \zeta.$$

Notons que

$$\underline{\mathbf{U}} \times N = \begin{pmatrix} -\mathbf{U}_{\mathbb{A}}^{\perp} \\ -\mathbf{U}_{\mathbb{A}}^{\perp} \cdot \nabla_{\mathbf{X}} \zeta \end{pmatrix}.$$

La quantité $U_{/\!\!/}$ définit donc entièrement $\underline{U} \times N^{\mu}$ et réciproquement. Ainsi en prenant le produit vectoriel de (1.22) avec N nous obtenons

$$\partial_t \mathbf{U}_{/\!/} + g \nabla_X \zeta + \frac{1}{2} \nabla_X \left| \mathbf{U}_{/\!/} \right|^2 - \frac{1}{2} \nabla_X \left[\left(1 + \left| \nabla_X \zeta \right|^2 \right) \underline{\mathbf{w}}^2 \right] + \left(\nabla_X^\perp \cdot \mathbf{U}_{/\!/} + f \right) \underline{\mathbf{V}}^\perp = -\nabla_X P. \quad (1.23)$$

Remarquons que dans le cas irrotationnel $U_{\parallel} = \nabla_X(\Phi_{|z=\zeta})$ si $U = \nabla_{X,z}\Phi$. De plus, si nous notons la vorticité $\boldsymbol{\omega} = \operatorname{rot}_{X,z}\mathbf{U}$ et $\underline{\boldsymbol{\omega}} = \boldsymbol{\omega}_{|z=\zeta}$, nous avons la formule suivante

$$\nabla^{\perp}_X \cdot \mathbf{U}_{/\!\!/} = \underline{\boldsymbol{\omega}} \cdot N.$$

La quantité $\nabla_X^{\perp} \cdot U_{\parallel}$ est donc entièrement déterminé par $\boldsymbol{\omega}$ et ζ . Fort de ce constat nous décomposons U_{\parallel} en deux parties :

$$\mathbf{U}_{/\!\!/} = \nabla_{\!X} \Delta_X^{-1} \nabla_{\!X} \cdot \mathbf{U}_{/\!\!/} + \nabla_{\!X}^{\perp} \Delta_X^{-1} \nabla_{\!X}^{\perp} \cdot \mathbf{U}_{/\!\!/}.$$
(1.24)

Ces opérateurs seront définis rigoureusement dans le chapitre 4. Nous posons

$$\psi := \Delta_X^{-1} \nabla_X \cdot \mathbf{U}_{/\!\!/}$$

et en appliquant l'opérateur $\Delta_X^{-1} \nabla_X \cdot$ à (1.23), nous avons

$$\partial_t \psi + \zeta + \frac{1}{2} \left| \mathbf{U}_{/\!\!/} \right|^2 - \frac{1}{2} \left(1 + \left| \nabla_X \zeta \right|^2 \right) \underline{\mathbf{w}}^2 + \Delta_X^{-1} \nabla_X \cdot \left[\left(\underline{\boldsymbol{\omega}} \cdot N + f \right) \underline{\mathbf{V}}^{\perp} \right] = -P.$$
(1.25)

Nous obtenons donc le système suivant appelé équations des vagues ou formulation de Castro-Lannes. C'est un système de trois équations avec les trois inconnues (ζ, ψ, ω) ,

$$\begin{cases} \partial_t \zeta - \underline{\mathbf{U}} \cdot N = 0, \\ \partial_t \psi + g\zeta + \frac{1}{2} \left| \mathbf{U}_{/\!\!/} \right|^2 - \frac{1}{2} \left(1 + \left| \nabla_X \zeta \right|^2 \right) \underline{\mathbf{w}}^2 + \Delta_X^{-1} \nabla_X \cdot \left[(\underline{\boldsymbol{\omega}} \cdot N + f) \, \underline{\mathbf{V}}^\perp \right] = -P, \quad (1.26) \\ \partial_t \boldsymbol{\omega} + (\mathbf{U} \cdot \nabla_{X,z}) \, \boldsymbol{\omega} = (\boldsymbol{\omega} \cdot \nabla_{X,z}) \, \mathbf{U} + f \partial_z \mathbf{U} , \text{ dans } \Omega_t, \end{cases}$$

où $\mathbf{U} := \mathbf{U}[\zeta, b](\psi, \boldsymbol{\omega}) = (\mathbf{V}, \mathbf{w})^t$ est l'unique solution du problème divergence-rotationnel suivant

$$\begin{cases} \operatorname{curl}_{X,z} \mathbf{U} = \boldsymbol{\omega} \operatorname{dans} \Omega_t, \\ \operatorname{div}_{X,z} \mathbf{U} = 0 \operatorname{dans} \Omega_t, \\ U_{\mathbb{I}} = \nabla_X \psi + \Delta_X^{-1} \nabla_X^{\perp} \left(\underline{\boldsymbol{\omega}} \cdot N \right), \\ U_b \cdot N_b = 0. \end{cases}$$
(1.27)

Dans la section 4.2, nous étudions le problème divergence-rotationnel (1.27). Nous montrons alors qu'il est bien défini. Dans [34], Castro et Lannes prouvent l'existence locale du système (1.26) dans le cas d'un fond plat, d'une pression constante à la surface et sans la force de Coriolis. Dans la section 4.3, nous traitons le cas général.

1.1.2.2 Adimensionnement

Nous pouvons procéder à un adimensionnement similaire au cas irrotationnel. La pression \mathcal{P} peut être décomposée en une contribution surfacique et une pression interne

$$\mathcal{P}(t, X, z) = P(t, X) + P_{\text{ref}} + \mathcal{P}(t, X, z),$$

avec $\widetilde{\mathcal{P}}_{|z=\zeta} = 0$. Nous adimensionnons les variables de la manière suivante

$$\begin{cases} x' = \frac{x}{L_x}, y' = \frac{y}{L_y}, z' = \frac{z}{H}, \zeta' = \frac{\zeta}{a}, b' = \frac{b}{a_{\text{bott}}}, t' = \frac{\sqrt{gH}}{L}t, \\ \mathbf{V}' = \sqrt{\frac{H}{g}} \frac{\mathbf{V}}{a}, w' = H\sqrt{\frac{H}{g}} \frac{w}{aL_x}, P' = \frac{P}{\rho ga} \text{ and } \widetilde{\mathcal{P}}' = \frac{\widetilde{\mathcal{P}}}{\rho gH} \end{cases}$$
(1.28)

et nous avons les paramètres sans dimension suivants

$$\varepsilon = \frac{a}{H}, \ \beta = \frac{a_{\text{bott}}}{H}, \ \mu = \frac{H^2}{L_x^2}, \ \gamma = \frac{L_x}{L_y} \text{ et Ro} = \frac{a}{fL_x} \sqrt{\frac{g}{H}}$$

Notre adimensionnement est basé sur une étude linéaire de ces équations (voir par exemple [34] et [80]). Puisque nous nous intéressons à des effets surfaciques, nous avons utilisé la quantité a comme distance caractéristique dans l'adimensionnement des vitesses horizontale et verticale. De même, pour créer le nombre de Rossby, nous avons pris la vitesse horizontale surfacique caractéristique $a\sqrt{\frac{g}{H}}$ comme vitesse horizontale typique. Cet adimensionnement diffère de celui fréquemment utilisé pour étudier les fluides géophysiques pour lequel la vitesse horizontale typique considérée est \sqrt{gH} (voir par exemple [117] ou [37] pour un tel adimensionnement). Il reste à adimensionner la vorticité $\boldsymbol{\omega} = \operatorname{curl}_{X,z} \mathbf{U}$. Nous allons nous placer dans le cas d'un écoulement faiblement cisaillé (voir par exemple [139], [121], [33]). Pour comprendre cette hypothèse, analysons de plus près les conséquences de notre adimensionnement quand nous supposons que μ est petit. En utilisant la condition d'incompressibilité, nous avons dans le domaine fluide

$$\mu \partial_{x'} \mathbf{V}'_x + \gamma \mu \partial_{y'} \mathbf{V}'_y + \partial_{z'} \mathbf{w}' = 0.$$

On introduit alors les notations suivantes

$$\nabla_{X',z'}^{\mu,\gamma} = \begin{pmatrix} \sqrt{\mu} \nabla_{X'}^{\gamma} \\ \partial_{z'} \end{pmatrix}, \operatorname{curl}_{X',z'}^{\mu,\gamma} = \nabla_{X',z'}^{\mu,\gamma} \times, \operatorname{div}_{X',z'}^{\mu,\gamma} = \nabla_{X',z'}^{\mu,\gamma} \cdot$$
(1.29)

 \mathbf{et}

$$\mathbf{U}^{\mu} = \begin{pmatrix} \sqrt{\mu} \mathbf{V}' \\ \mathbf{w}' \end{pmatrix}, \ \underline{\mathbf{U}}^{\mu} = \mathbf{U}^{\mu}_{|z'=\varepsilon\zeta'}, \ \mathbf{U}^{\mu}_{b} = \mathbf{U}^{\mu}_{|z'=-1+\beta b'}, \\ N^{\mu,\gamma} = \begin{pmatrix} -\varepsilon\sqrt{\mu}\nabla^{\gamma}_{X'}\zeta \\ 1 \end{pmatrix}, \ N^{\mu}_{b} = \begin{pmatrix} -\beta\sqrt{\mu}\nabla^{\gamma}_{X'}b \\ 1 \end{pmatrix}.$$
(1.30)

Ainsi, on peut réécrire la condition d'incompressibilité en

$$\nabla_{X',z'}^{\mu,\gamma} \cdot \mathbf{U}^{\mu} = \mu \nabla_{X'}^{\gamma} \cdot \mathbf{V} + \partial_{z'} \mathbf{w} = 0.$$

De plus, la deuxième équation de (1.3) (le fond est fixe ici) nous donne que

$$-\beta \mu \mathbf{V}_b' \cdot \nabla_{X'}^{\gamma} b' + \mathbf{w}_b' = 0.$$

Nous voyons alors que la vitesse verticale w' est d'ordre $\mathcal{O}(\mu)$ si V' est d'ordre $\mathcal{O}_{\mu}(1)$. De plus, nous avons

$$\operatorname{curl}_{X',z'}^{\mu,\gamma} \mathbf{U}^{\mu} = \frac{H^2}{aL_x} \sqrt{\frac{H}{g}} \boldsymbol{\omega}$$

Nous pouvons être tentés d'adimensionner ω de la façon suivante

$$oldsymbol{\omega}' = rac{H^2}{aL_x}\sqrt{rac{H}{g}}oldsymbol{\omega}.$$

En prenant la partie horizontale de la vorticité, noté ω_h' , nous obtenons alors que

$$\boldsymbol{\omega}_h' = \mu \partial_{z'} \mathbf{V}^{\perp} - \sqrt{\mu} (\nabla_{X'}^{\gamma})^{\perp} \mathbf{w}.$$

Nous voyons que ω'_h est d'ordre $\mathcal{O}(\mu)$ si $\partial_{z'} \mathbf{V}'$ est d'ordre $\mathcal{O}_{\mu}(1)$. L'hypothèse d'un écoulement faiblement cisaillé consiste précisement à supposer que $\partial_{z'} \mathbf{V}'$ est d'ordre $\mathcal{O}_{\mu}(1)$ (voir [139], [121]). La vorticité est alors de l'ordre de $\mathcal{O}(\mu)$. Remarquons que, dans le cas irrotationnel, cette hypothèse est automatiquement vérifiée. Nous adimensionnons alors $\boldsymbol{\omega}$ de la manière suivante

$$\boldsymbol{\omega}' = \frac{L_x}{a} \sqrt{\frac{H}{g}} \boldsymbol{\omega},$$

de sorte que

$$\operatorname{curl}_{X',z'}^{\mu,\gamma}\mathbf{U}^{\mu}=\mu\boldsymbol{\omega}'.$$

Nous omettons les primes dans la suite. Les équations des vagues deviennent

$$\begin{cases}
\left(\partial_t \zeta - \frac{1}{\mu} \underline{\mathbf{U}}^{\mu} \cdot N^{\mu,\gamma} = 0, \\
\left(\partial_t \psi + \zeta + \frac{\varepsilon}{2} \left| \mathbf{U}_{/\!/}^{\mu} \right|^2 - \frac{\varepsilon}{2\mu} \left(1 + \varepsilon^2 \mu \left| \nabla_X^{\gamma} \zeta \right|^2 \right) \underline{\mathbf{w}}^2 + \varepsilon \Delta_X^{-1} \nabla_X \cdot \left[\left(\underline{\omega} \cdot N^{\mu,\gamma} + \frac{1}{\mathrm{Ro}} \right) \underline{\mathbf{V}}^{\perp} \right] = -P, \quad (1.31) \\
\left(\partial_t \boldsymbol{\omega} + \frac{\varepsilon}{\mu} \left(\mathbf{U}^{\mu} \cdot \nabla_{X,z}^{\mu,\gamma} \right) \boldsymbol{\omega} = \frac{\varepsilon}{\mu} \left(\boldsymbol{\omega} \cdot \nabla_{X,z}^{\mu,\gamma} \right) \mathbf{U}^{\mu} + \frac{\varepsilon}{\mu \mathrm{Ro}} \partial_z \mathbf{U}^{\mu}, \, \mathrm{dans} \, \Omega_t,
\end{cases}$$

où $\mathbf{U}^{\mu}:=\mathbf{U}^{\mu}[\varepsilon\zeta,\beta b](\psi,\pmb{\omega})$ est solution de

$$\begin{cases} \operatorname{curl}_{X,z}^{\mu,\gamma} \mathbf{U}^{\mu} = \mu \boldsymbol{\omega} \operatorname{dans} \Omega_{t}, \\ \operatorname{div}_{X,z}^{\mu,\gamma} \mathbf{U}^{\mu} = 0 \operatorname{dans} \Omega_{t}, \\ U_{\mathbb{I}}^{\mu} = \nabla^{\gamma} \psi + \nabla_{X}^{\gamma \perp} (\Delta_{X}^{\gamma})^{-1} (\underline{\omega} \cdot N^{\mu,\gamma}), \\ U_{b}^{\mu} \cdot N_{b}^{\mu,\gamma} = 0. \end{cases}$$
(1.32)

Dans [34], Castro et Lannes montrent l'existence locale du système (1.31) dans le cas d'un fond plat, d'une pression constante à la surface et sans la force de Coriolis. Ils obtiennent un temps d'existence de la forme $\frac{T}{\varepsilon}$. Dans la section 4.3, nous traitons le cas d'un fond non plat et nous obtenons un temps d'existence de la forme $\frac{T}{\max(\varepsilon,\beta,\frac{\varepsilon}{\text{Ro}})}$ dans le cas où P est constante (Théorème 4.3.6). Nous donnons aussi un temps d'existence dans le cas où P n'est pas constante.

1.2 Modèles asymptotiques pour les équations des vagues

Les systèmes (1.17) et (1.31) sont en général trop compliqués pour étudier la propagation de vagues. Nous allons donc les simplifier en supposant que certains des paramètres sans dimension ε , β , γ , μ et $\frac{\varepsilon}{\text{Ro}}$ sont petits. On parle alors de *régime asymptotique*. Nous donnons dans la suite un état de l'art de différents modèles asymptotiques que nous étudions dans ce manuscrit. Notons que dans cette partie, sauf mention du contraire, la pression à la surface est supposée constante et le fond fixe.

1.2.1 Régime d'eaux peu profondes

1.2.1.1 Équations de Saint-Venant

Lorsque μ est petit, on parle de régime d'eaux peu profondes. Ce régime est bien connu des physiciens. Il revient à supposer que la longueur typique de notre phénomène L_x est très grande devant la hauteur d'eau typique H. On peut alors simplifier les équations des vagues en ne négligeant que les termes d'ordre $\mathcal{O}(\mu)$. On obtient les équations dites de Saint-Venant (ou "Nonlinear Shallow Water equations" en anglais). C'est un système d'équations sur la vitesse horizontale moyennée sur la profondeur, noté $\overline{\mathbf{V}}$, et la surface ζ . Dans le cas irrotationnel, Barré de Saint Venant dérive ces équations pour la première fois en 1871 ([13], [12]). Leur justification mathématique est obtenue un siècle plus tard par Ovsjannikov ([115], [116]) puis par Kano et Nishida ([74]). Notons que, dans ces travaux, les auteurs utilisent des données initiales analytiques. Alvarez-Samaniego et Lannes ([9]) et Iguchi ([70]) traitent des données initiales avec régularité Sobolev. Iguchi ([71]) considère aussi le cas d'un fond mobile. Enfin, Castro et Lannes ([34]) justifient ces équations pour un écoulement faiblement cisaillé dans le cas où le fond est plat. Dans la section 4.5, nous généralisons ce dernier résultat en ajoutant la force de Coriolis, un fond non plat et une pression non constante à la surface.

1.2.1.2 Équations de Green-Naghdi

En pratique, les équations de Saint-Venant ne sont pas adaptées à tous les types de phénomènes liés aux eaux peu profondes. Ce sont des équations non dispersives (au moins pour d = 1) contrairement aux équations des vagues. Pour gagner en précision, nous allons alors un cran plus loin dans le développement en ne négligeant que les termes d'ordre $\mathcal{O}(\mu^2)$ dans les équations des vagues. On obtient les équations de Green-Naghdi, dérivées pour la première fois par Green et Naghdi en 1976 dans le cas irrotationnel ([62]). On parle aussi des équations de Serre, dérivées vingt ans plus tôt par Serre ([131]) dans le cas où d = 1. La première justification des équations de Green-Naghdi est faite par Makarenko ([94]) pour des données analytiques, dans le cas où d = 1 et pour un fond plat et sur un temps T indépendant de μ . Li ([88]) généralise ce résultat à des données à régularité Sobolev et sur un temps T indépendant de μ . Enfin, Alvarez-Samaniego et Lannes ([9], [81]) traitent le cas d = 1 et 2 avec fond quelconque grâce à un schéma de Nash-Moser pour prouver l'existence locale des équations de Green-Naghdi et justifient les équations de Green-Naghdi sur un temps $\frac{T}{\mu}$. Notons aussi les travaux de Israwi ([72]) qui permettent de se passer du schéma de Nash-Moser lorsque d = 1. Dans [33], Castro et Lannes étendent les équations de Green-Naghdi pour d = 1 (vorticité scalaire) et d = 2 pour un écoulement faiblement cisaillé lorsque β de l'ordre de $\sqrt{\mu}$. Dans la section 5.4, nous généralisons cette dérivation en ajoutant la force de Coriolis et en nous plaçant dans le cas où γ est de l'ordre $\mathcal{O}(\mu^2)$ et β de l'ordre $\mathcal{O}(\mu)$. Précisons que les résultats de la section 5.4 et de [33] ne sont que des dérivations. À notre connaissance, il n'y a aucune justification complète des équations de Green-Naghdi ou

de Serre dans le cas d'un fluide faiblement cisaillé. Il manque une preuve du caractère bien posé de ce système pour pouvoir totalement le justifier.

1.2.2 Régime faiblement non linéaire

Dans cette sous-section nous parlerons seulement du cas irrotationnel.

1.2.2.1 Équations linéaires des vagues

Lorsque ε est petit et μ de l'ordre de 1, on parle d'un régime faiblement non linéaire pour des eaux dites profondes. Si ε est très petit, un premier modèle asymptotique simple consiste à supposer que tous les termes d'ordre $\mathcal{O}(\varepsilon)$ sont négligeables. On obtient les équations linéaires des vagues. Ce modèle fait apparaître l'opérateur non local $G_{\mu}[0, \beta b]$. On peut alors se demander si des effets dispersifs sont possibles sur ce modèle. Dans la section 2.3.3, nous montrons que, pour un fond plat et lorsque d = 1, ce modèle est bien un système dispersif et nous donnons des estimations de décroissance de la norme L^{∞} (Proposition 2.3.15). Notons que Mésognon-Gireau ([106]) généralise ce résultat en améliorant la régularité demandée pour la condition initiale et que Bulut ([31]) traite le cas d'une profondeur infinie.

1.2.2.2 Équations de Saut-Xu

En pratique, les équations linéaires des vagues sont trop simplistes dans de nombreuses situations. Si nous souhaitons être plus précis, nous allons un cran plus loin dans le développement en ne négligeant uniquement que les termes d'ordre $\mathcal{O}(\varepsilon^2)$ dans les équations des vagues. Le premier modèle asymptotique obtenu dans un tel régime a été découvert par Matsuno pour d = 1 et pour un fond plat ([97]) et un fond faiblement variable ([98]) puis dans le cas où d = 2 pour des vagues faiblement transverses (γ de l'ordre de ε) et un fond plat ([99]). Choi ([38]) étend le résultat de [99] à des vagues quelconques (voir aussi le papier de Smith [132]). Bonneton et Lannes ([22]) donnent une formulation des équations de Matsuno lorsque d = 1 et d = 2 dans le cas d'un fond non plat de faible amplitude (β d'ordre $\mathcal{O}(\varepsilon)$). Les inconnues de ces équations sont la vitesse horizontale à la surface et la surface ζ . Notons que tous les résultats précédents ne sont que des dérivations. A notre connaissance, nous se savons toujours pas montrer si les équations de Matsuno sont bien posées (voir l'article de Ambrose, Bona et Nicholls [10] à ce sujet). Pour pallier cette difficulté, Saut et Xu ([125]) développent un modèle équivalent aux équations de Matsuno (avec la même précision) dans le cas d'un fond plat et ils montrent l'existence locale de leur système. Ainsi, en utilisant les résulats d'Alvarez-Samaniego et Lannes ([9]), ils justifient mathématiquement leur système comme un modèle asymptotique des équations des vagues sur un temps $\mathcal{O}\left(\frac{1}{\varepsilon}\right)$ avec une précision de l'ordre de $\mathcal{O}(\varepsilon)$. Dans le chapitre 3, nous généralisons leur résultat dans le cas où d = 1 en rajoutant un fond (Système (3.5)) et nous proposons un schéma numérique pour résoudre les équations de Saut-Xu (Section 3.3). Notre schéma est basé sur un splitting entre les termes locaux et non locaux. Nous justifions la convergence de ce dernier (Théorème 3.4.6) et nous l'utilisons pour étudier le comportement d'un soliton de KdV lorsque le paramètre μ augmente (Sous-section 3.5.3) et l'effet d'homogénéisation d'un fond fortement oscillant sur la propagation des vagues (Sous-section 3.5.4).

1.2.3 Régime d'ondes longues

1.2.3.1 Équation des ondes

Lorsque ε et μ sont petits et du même ordre, on parle de régime d'ondes longues. Un premier modèle consiste à ne négliger que les termes d'ordre $\mathcal{O}(\max(\varepsilon, \mu))$. Dans le cas irrotationnel, on obtient une équation des ondes satisfaite par la surface ζ . Lagrange ([77]) est le premier à dériver cette équation dans le cas d'un fond plat. Dans le chapitre 2, nous proposons différents modèles pour ce régime lorsque d = 1 (avec en plus γ de l'ordre de $\mathcal{O}(\varepsilon)$). Dans le cas d'un écoulement faiblement cisaillé, si nous supposons que $\frac{\varepsilon}{R_0}$ est de l'ordre de 1 et que β est de l'ordre de $\mathcal{O}(\varepsilon)$, nous obtenons un système d'équations sur la vitesse moyennée sur la profondeur $\overline{\mathbf{V}}$ et la surface ζ . Ces équations jouent un rôle important dans la littérature physique puisque qu'elles permettent de modéliser les ondes de Poincaré, appelées aussi ondes de Sverdrup ([134]), et les ondes de Kelvin. Majda étudie ces ondes dans [93] (voir aussi [117], [61] et [84]). Dans les sections 4.5.4 (pour d = 2) et 5.3.2 (pour d = 1), nous montrons que ces équations sont dispersives et nous justifions mathématiquement les équations linéaires des vagues comme modèle asymptotique des équations des vagues.

1.2.3.2 Équation de Boussinesq

Les modèles donnés dans le paragraphe précédent sont des modèles linéaires donc souvent trop simplistes. Si nous allons un cran plus loin dans le développement en ne négligeant que les termes d'ordre $\mathcal{O}(\max(\varepsilon,\mu)^2)$ dans les équations des vagues, nous obtenons les équations dites de Boussinesq quand β est de l'ordre $\mathcal{O}(\mu)$ et de Boussinesq-Peregrine dans le cas où β est de l'ordre de 1. Dans le cas irrotationnel, Boussinesq les dérive pour la première fois ([25], [26]) dans le cas d'un fond plat et lorsque d = 1. Puis Peregrine ([119]) les généralise pour un fond quelconque. Notons aussi le livre de Whitham [149] qui propose un reformulation de ces équations qui est plus agréable pour prouver l'existence locale. Les premiers travaux pour justifier mathématiquement les équations de Boussinesq sont dus à Craig [43] dans le cas d'un fond plat, lorsque d = 1, pour de petites données initiales et sur un temps long $(\frac{T}{\mu}$ indépendant de μ) et à Nishida et Kano [75] pour des données initiales quelconques lorsque d = 1 et sur un temps court (T indépendant de μ). Alvarez-Samaniego et Lannes ([9]) généralisent ces résultats pour des données initiales quelconques, lorsque d = 1 et d = 2, et justifient que les équations de Boussinesq (pour β de l'ordre $\mathcal{O}(\mu)$ approchent les équations des vagues avec une précision en $\mathcal{O}(\mu)$ sur un temps $\frac{T}{\mu}$. Mésognon-Gireau ([104]) étudie le cas où β n'est pas de l'ordre de μ (fond avec une grande amplitude) et justifie sur un temps $\frac{T}{\mu}$ des équations de Boussinesq-Peregrine modifiées ayant la même précision que les équations de Boussinesq-Peregrine. Citons aussi les nombreux résultats sur les système Boussinesq (voir par exemple [112], [17], [18], [19], [127], [126]), qui sont des équations modifiées ayant la même précision que les équations de Boussinesq. Dans la section 5.2, nous généralisons et justifions les équations de Boussinesq à un écoulement faiblement cisaillé en présence de Coriolis et pour γ de l'ordre de $\mathcal{O}(\mu^2)$. Nous montrons l'existence de solutions pour les équations de Boussinesq sur un temps $\frac{T}{\max\left(\varepsilon,\beta,\frac{\varepsilon\sqrt{\mu}}{\text{Ro}}\right)}$ (Théorème 5.2.15).

1.2.3.3 Équations de KdV et d'Ostrovsky

Le régime d'ondes longues (on parle aussi de régime Boussinesq) permet de dériver des équations scalaires. Nous supposons dans ce paragraphe que γ et β sont de l'ordre de $\mathcal{O}(\mu^2)$. Dans le cas irrotationnel, divers auteurs montrent que l'équation de Korteweig de Vries est un bon modèle asymptotique des équations des vagues ([43], [75], [129] et [19]). Nous justifions dans la soussection 5.3.4, qu'en présence de la force de Coriolis, l'équation de Korteweig de Vries approche les équations des vagues sur un temps $\frac{T}{\mu}$ avec une précision de l'ordre de $\mathcal{O}(\mu)$, si $\frac{\varepsilon}{\text{Ro}}$ est de l'ordre de $\mathcal{O}(\sqrt{\mu})$, Germain et Renouard [60] montrent formellement que l'équation d'Ostrovsky

$$\partial_{\xi} \left(\partial_{\tau} f + \frac{3}{2} f \partial_{\xi} f + \frac{1}{6} \partial_{\xi}^3 f \right) = \frac{1}{2} f \tag{1.33}$$

est un bon modèle asymptotique. Cette équation, dérivée pour la première fois par Ostrovsky ([113]), généralise l'équation de Korteweig de Vries dans le cas d'un force de Coriolis relativement faible. Dans la sous-section 5.3.3, nous justifions mathématiquement l'équation d'Ostrovsky sur un temps $\frac{T}{\sqrt{\mu}}$ avec une précision de l'ordre de $\mathcal{O}(\mu)$ lorsque $\frac{\varepsilon}{\text{Ro}}$ est de l'ordre de $\mathcal{O}(\sqrt{\mu})$. Ce résultat est le premier travail qui justifie mathématiquement léquation d'Ostrovsky. Notons cependant que nous ne justifions pas cette équation sur un temps $\frac{T}{\mu}$ (voir Section 1.5.3 et Chapitre 5).

1.3 Résonance de Proudman

Dans cette section nous nous intéressons à la résonance dite de Proudman mise en évidence par ce dernier en 1929 ([120]). Elle correspond à une élévation localisée du niveau des eaux due aux déplacements de perturbations atmosphériques. Ce manuscrit, et en particulier le chapitre 2, a pour but d'étudier en détail cette résonance grâce à des modèles asymptotiques des équations des vagues.

1.3.1 Mécanisme de la résonance de Proudman

Commençons par expliquer la résonance de Proudman dans un cas très simple. Considérons une dépression atmosphérique (une tempête, un grain orageux ou une bombe météorologique) ou un anti-cyclone se déplacant à une vitesse fixe U au dessus d'un océan initialement au repos. Nous supposons que la longueur de la perturbation météorologique est très grande devant la hauteur d'eau moyenne (μ petit) et que l'amplitude des vagues est très faible devant la hauteur d'eau moyenne (ε petit). Nous nous intéressons au cas où d = 1 (pas d'effets transverses). Un modèle simple pour étudier la propagation des vagues dans un tel régime est l'équation des ondes 1d satisfaite par la surface ζ

$$\partial_t^2 \zeta - \partial_x^2 \zeta = \partial_x^2 P(t, x),$$

où P est la pression à la surface modélisant notre perturbation atmosphérique. Comme nous nous intéressons à une perturbation se déplacant à une vitesse fixe U, nous pouvons écrire P de la forme P(t, x) = P(x - Ut). Nous résolvons explicitement cette équation par la formulation de Duhamel et nous obtenons la solution particulière suivante

$$\zeta(t,x) = \begin{cases} \frac{1}{1-U^2} \left(P(x-t) - P(x-Ut) \right) & \text{si } U \neq 1, \\ -\frac{t}{2} P'(x-t) & \text{si } U = 1. \end{cases}$$
(1.34)

Nous voyons donc que si la vitesse de la tempête est proche de 1, une amplification importante est possible. Notons que la vitesse 1 provient de l'adimensionnement et correspond à la vitesse typique \sqrt{gH} .

Historiquement, Proudman ([120]) met en avant cette amplification pour montrer que les vents et les variations atmosphériques peuvent être à l'origine de vagues de grandes tailles malgré leur faible puissance. Rabinovich, Vilibić, Montserrat et coauteurs (voir par exemple [109], [144] ou [73]) reprennent les travaux de Proudman pour expliquer les météotsunamis (ou tsunamis météorologiques). Dans [109], les auteurs montrent que la résonance de Proudman est le principal mécanisme à l'origine des météotsunamis.



Figure 1.2: mécanismes d'un météotsunami

1.3.2 Tsunamis météorologiques

Les tsunamis météorologiques sont des phénomènes assez rares. Ils sont le résultat de trois mécanismes bien distincts. Ils se forment à cause d'une très large perturbation atmosphérique (de l'ordre de la dizaine voire de la centaine de kilomètres). Le premier mécanisme est la résonance de Proudman qui se déroule loin des côtes. À cause de cette résonance, des vagues de plus en plus hautes se forment. Elles se déplacent à la même vitesse que la perturbation atmosphérique. Quand ces vagues arrivent près des côtes, un deuxième mécanisme entre en scène, le shoaling. La longueur d'onde des vagues diminue et leur amplitude augmente du fait de la remontée de la bathymétrie. Cet effet est bien connu des océanographes et ne se limite pas qu'aux météotsunamis. Après le shoaling, les vagues générées sont souvent relativement peu destructrices. Un troisième effet peut cependant se produire. Si les vagues rentrent dans un port ou une baie (donc un domaine semi-clos), une résonance est possible : la résonance de Helmholtz (voir par exemple [108]). Cette résonance est basée sur les modes de vibrations propres du domaine semi-clos. Si la fréquence des vagues correspond à ces modes propres, leurs amplitudes sont amplifiées. Ainsi, un météotsunami est en général la combinaison de trois phénomènes : la résonance de Proudman, le shoaling et la résonance de Helmholtz, d'où sa rareté. La figure 1.2 résume ces trois mécanismes en donnant un ordre de grandeur typique de l'amplitude des vagues à chaque étape. Pour une étude plus détaillée sur les météotsunamis nous référons à [109] et [86].

Comme la résonance de Helmholtz est souvent nécessaire pour obtenir un météotsunami, les océanographes ont répertorié les lieux comportant un risque. Dans la suite, nous détaillons quelques tsunamis météorologiques qui se sont produits ces dernières années.

La baie de Nagasaki est un des lieux les plus connus pour ses seiches générées par des perturbations atmosphériques. Les habitants parlent de *abiki* pour qualifier ce phénomène. Le plus important *abiki* enregistré depuis 100 ans s'est produit en 1979. Des vagues atteignant 5 mètres de haut ont déferlé sur les côtes de Nagasaki. Aucune alerte tsunami n'a été déclenchée car l'origine de ce tsunami n'était pas sismique mais météorologique. Quelques heures plus tôt une immense tempête s'était formée en mer de Chine. Bien que nous n'ayons pas d'estimations précises, de nombreuses simulations numériques ont montré rétrospectivement que cette tempête faisait approximativement 200 kilomètres de long sur 300 kilomètres de large créant une variation de pression de 3 hectopascal. Elle s'est propagée pendant environ 3 heures jusqu'à atteindre la baie de Nagasaki et a multiplié l'amplitude des vagues créées par 5. Ce météotsunami sert aujourd'hui de référence. Il est étudié en détail dans [67] (voir aussi [109]).

Un autre lieu bien connu pour ses seiches destructrices est le port de Ciutadella (île Baléares, mer méditerranée). Les habitants parlent de *rissaga*. En 2006, une importante élévation du niveau des eaux s'est produite, provoquant la destruction d'une grande partie du port de Ciutadella. Des vagues de 5 mètres de haut ont été observées. Ce phénomène semble provenir de la propagation d'un énorme anti-cyclone. Une augmentation brusque de la pression atmosphérique de 10 hectopascal a été mesurée au moment du phénomène. Pour plus d'informations voir par exemple [73] (voir aussi [109]).

Un exemple intéressant est la série de deux météotsunamis en 1954 dans le lac Michigan. Des vagues de 3 mètres de haut ont été observées sur les berges du lac. Ce phénomène est étudié en détail dans [14]. Notons que ces deux météotsunamis ont pour particularité de n'être que d'origine météorologique (il n'y a pas eu de résonance de Helmholtz). Citons enfin le météotsunami de 1978 à Vela Luka (Croatie, mer Adriatique) qui est considéré comme le plus grand météotsunami d'Europe. Des vagues de 6 mètres de haut ont été observées dans la baie de Vela Luka. Voir [145] pour une étude détaillée.

1.3.3 Tsunamis par glissements de terrain

Dans ce manuscrit, et en particulier dans le chapitre 2, nous étudions un autre phénomène océanographique : les tsunamis par glissements de terrain sous-marins. Leurs formations sont assez similaires aux météotsunamis. Un glissement de terrain sous-marin peut amplifier la vague qu'il crée si sa vitesse est proche de \sqrt{gH} où H est la profondeur d'eau typique. On observe une amplification similaire à la résonance de Proudman. Dans le chapitre 2, nous comparons les tsunamis météorologiques et les tsunamis par glissement de terrain. Nous donnons deux différences entre ces deux phénomènes : la durée et la taille. Alors qu'une perturbation atmosphérique peut se propager jusqu'aux côtes sur plusieurs heures sans trop se déformer, un glissement de terrain sous-marin se propage au maximum pendant une dizaine de minutes. De plus, un tsunami par glissement de terrain a tendance à avoir une amplitude typique en haute mer beaucoup plus importante qu'un météotsunami. Les tsunamis par glissement de terrain sont très similaires aux tsunamis classiques. Une perturbation assez rapide du fond crée une grande vague qui se propage ensuite jusqu'au côtes. Pour plus de détail sur les tsunamis par glissements de terrain voir par exemple [86].

La taille du glissement de terrain est un paramètre important pour créer un tsunami. Les tsunamis par glissement de terrain sous-marin sont donc assez rares. Ils proviennent en général de l'accumulation de sédiments sur des centaines voire des milliers d'années. Diverses études montrent par exemple que des glissements de terrain ont pu avoir eu lieu en mer du Nord au large de la Norvège il y a plus de 8000 ans (voir par exemple [30] pour une étude détaillée). Ils sont communément appelés glissements de terrain de Storegga dans la littérature physique. Citons aussi la zone de fracture d'Owen (frontière de plaque entre l'Arabie et l'Inde) qui est considéré comme un lieu propice à de potentiels glissements de terrain destructeurs (voir par exemple [122]).

Les glissements de terrain sous-marins sont parfois un apport supplémentaire d'énergie et amplifient une vague déjà existante comme par exemple lors du tsunami en Papouasie-Nouvelle-Guinée de 1998. Un tremblement de terre de magnitude 7.1 en plus de générer un tsunami, déstabilisa les fonds marins des environs et créa une série de glissements de terrain sous-marins. Ces glissements de terrain amplifièrent alors les vagues créées par le tremblement de terre (voir par exemple [137] pour une étude détaillée). Le tsunami de 2011 dans la région de Tohoku au Japon semble aussi avoir été amplifié par un glissement de terrain sous-marin (voir [136]), expliquant ainsi la catastrophe de Fukushima. L'amplification due au glissement de terrain aurait fourni suffisamment d'énergie aux vagues pour qu'elles puissent passer par dessus la digue qui protégeait la centrale nucléaire.

1.3.4 Résultats obtenus

Toutes les études menées sur les météotsunamis et les tsunamis par glissement de terrain sousmarin reposent sur l'hypothèse que la longueur d'onde de la perturbation est très grande devant la hauteur d'eau (μ petit) et utilisent les équations de Saint-Venant comme modèle mathématique. Dans la sous-section 2.3.3, nous proposons une étude de la résonance de Proudman sans cette hypothèse. Nous montrons qu'une résonance est encore possible dans des eaux profondes (μ de l'ordre de 1) mais avec un facteur d'amplification moins important que dans le cas d'eaux peu profondes. Supposer que μ n'est pas petit ne nous permet plus de négliger les termes dispersifs des équations des vagues. Ces derniers perturbent la résonance et tendent à l'atténuer. Ce phénomène de résonance en eau profonde n'avait pas été mise en évidence auparavant. Dans la sous-section 2.3.2.2, nous proposons une étude de la résonance de Proudman quand le fond n'est plus supposé relativement plat (β petit). Nous montrons que, pour des fonds tendant suffisament rapidement vers un fond plat, une résonance est encore possible. De plus, même si le travail originel de Proudman [120] prend en compte l'effet Coriolis, il est souvent admis par la communauté physique que ces effets sont négligeables (voir par exemple [144]). Dans les sous-sections 4.5.4 et 5.3.2, nous étudions la résonance de Proudman quand l'effet Coriolis n'est pas négligeable. Dans la sous-section 4.5.4, nous traitons le cas d = 2 et nous montrons que la résonance de Proudman n'est pas possible à cause des effets dispersifs dus à la force de Coriolis. Dans la sous-section 5.3.2, nous étudions le cas d = 1. Nous montrons qu'une résonance est possible pour un profil de pression particulier mais par pour des tempêtes se déplaçant à une vitesse fixe.

1.4 La résonance de Proudman au delà du cadre classique, effets nonlinéaires et dispersion non locale

L'étude de la résonance de Proudman que nous proposons dans ce manuscrit est essentiellement linéaire. Pourtant, des effets non linéaires peuvent se produire lors de la propagation de vagues, en particulier à l'approche des côtes. On peut alors distinguer deux situations. Soit les vagues que nous étudions sont très longues et nous sommes dans un régime d'eaux peu profondes, soit la longueur typique de nos vagues est comparable à la profondeur d'eau et nous sommes dans un régime d'eaux profondes. Lorsque que nous travaillons dans un régime d'eaux peu profondes, nous avons de nombreux modèles asymptotiques à notre disposition (voir Section 1.2). Pour étudier les effets non linéaires sur la résonance de Proudman, les équations de Saint-Venant fournissent une bonne approximation. Pour obtenir des informations quantitatives sur ces équations, nous allons alors adopter un point de vue numérique. Vilibic ([144]) propose une étude numérique de la résonance de Proudman grâce à ces équations. Il montre alors que les effets non linéaires ne perturbent pas la résonance (voir en particulier la figure 5 dans [144]). Nous donnons un aperçu des travaux de [144] dans la section 2.3.4.1 en utilisant un schéma numérique basé sur les travaux de Bouchut ([23]).

Lorsque nous sommes dans un régime d'eaux profondes, nous avons vu à la section 1.2 que les

équations de Saut-Xu peuvent être utilisées pour prendre en compte les effets non linéaires et la forte dispersion (de caractère non local) propre à ce régime. Ainsi, afin d'obtenir des résultats qualitatifs, nous avons développé un schéma numérique pour ces équations. Il est étudié au chapitre 3. Notre schéma est basé sur un splitting entre les termes locaux et non locaux (voir Section 3.3) et nous justifions la convergence de ce dernier (Théorème 3.4.6). Nous l'utilisons alors pour étudier la résonance de Proudman dans un régime d'eaux profondes (voir Section 2.3.4.2). Il s'agit d'un régime pertinent physiquement (surtout pour les tsunamis générés par glissement de terrain). Nous montrons que, malgré les effets non linéaires, une amplification est encore possible. Ce phénomène de résonance en eau profonde n'avait pas été mise en évidence auparavant.

1.5 Perspectives de recherches

Dans cette partie nous proposons différentes perspectives de recherches qui pourraient prolonger ce manuscrit.

1.5.1 Vers un temps d'existence plus long dans le cas irrotationnel

Dans [105], Mésognon-Gireau donne un temps d'existence pour les équations des vagues (1.17) (cas irrotationnel) de la forme $\frac{T}{\varepsilon}$ dans le cas d'un fond fixe de grande amplitude (différent de [9]), d'une pression constante à la surface et avec tension de surface. La méthode utilisée pour obtenir ce résultat est inspirée de Bresch-Métivier [28]. Dans la section 2.2, nous donnons un temps d'existence pour (1.17) de la forme $\frac{T}{\sqrt{\max(\varepsilon,\beta)}}$ (Théorème 2.2.4). Ainsi, en utilisant les travaux de Mésognon-Gireau, nous pouvons espérer améliorer notre temps d'existence en $\frac{T}{\sqrt{\varepsilon}}$.

Cette conjecture est motivée par le fait suivant : une solution (ζ, \overline{V}) assez régulière des équations de Saint-Venant

$$\begin{cases} \partial_t \zeta + \nabla_X \left([1 + \varepsilon \zeta - \beta b] \overline{V} \right) = \partial_t b, \\ \partial_t \overline{V} + \varepsilon \left(\overline{V} \cdot \nabla_X \right) \overline{V} + \nabla \zeta = -\nabla_X P_t \end{cases}$$

existe sur un intervalle de temps de taille $\frac{1}{\sqrt{\varepsilon}}$. La preuve de ce fait repose sur des estimations d'énergies des dérivées temporelles de nos inconnues. Nous commençons par changer l'échelle de la variable temporelle. Nous posons $\tau = \frac{t}{\sqrt{\varepsilon}}$ et nous obtenons

$$\begin{cases} \partial_{\tau}\zeta + \frac{1}{\sqrt{\varepsilon}}\nabla_{X}\left([1+\varepsilon\zeta - \beta b]\overline{V}\right) = \partial_{\tau}b(\sqrt{\varepsilon}\tau, \cdot), \\ \partial_{\tau}\overline{V} + \sqrt{\varepsilon}\left(\overline{V}\cdot\nabla_{X}\right)\overline{V} + \frac{1}{\sqrt{\varepsilon}}\nabla\zeta = -\frac{1}{\sqrt{\varepsilon}}\nabla_{X}P(\sqrt{\varepsilon}\tau, \cdot). \end{cases}$$

Nous remarquons alors que l'on peut appliquer la méthode fournie par Bresch et Métivier [28]. Nous définissons l'énergie

$$\mathcal{E}^{N}(\zeta,\overline{V}) = \sum_{|(\alpha,k)| \le N} \left| (\sqrt{\varepsilon})^{k} \partial_{t}^{k} \partial^{\alpha} \zeta \right|_{L^{2}}^{2} + \left([1 + \varepsilon \zeta - \beta b] (\sqrt{\varepsilon})^{k} \partial_{t}^{k} \partial^{\alpha} \overline{V}, (\sqrt{\varepsilon})^{k} \partial_{t}^{k} \partial^{\alpha} \overline{V} \right)_{L^{2}},$$

et nous montrons que

$$\frac{d}{dt}\mathcal{E}^{N}(\zeta,\overline{V}) \leq C(P,b)\left(\sqrt{\varepsilon}\mathcal{E}^{N}(\zeta,\overline{V})^{\frac{3}{2}} + \mathcal{E}^{N}(\zeta,\overline{V}) + \frac{1}{\varepsilon}\sqrt{\mathcal{E}^{N}(\zeta,\overline{V})}\right).$$

Ainsi en utilisant la stratégie énoncée à la fin de la sous-section 1.1.1.2, nous obtenons que (ζ, \overline{V}) existe sur un intervalle de temps de taille $\frac{1}{\sqrt{\varepsilon}}$. Nous espérons reproduire cette méthode sur le système complet (1.17) dans le futur. Rappelons cependant qu'en contrepartie de ce temps d'existence, l'énergie $\mathcal{E}^N(\zeta, \overline{V})$ n'est plus bornée uniformément par rapport au paramètre ε .

1.5.2 Dispersion pour les équations linéaires des vagues

Dans la section 2.3.3, Nous montrons que les équations linéaires des vagues sont des équations dispersives dans le cas où d = 1 et nous donnons des estimations de décroissances L^{∞} (Proposition 2.3.15). En collaboration avec Maxime Gazeau (university of Toronto), nous voudrions étendre ce résultat à des fonds périodiques. Toute l'étude de la section 2.3.3 repose sur des méthodes de phases stationnaires et la transformée de Fourier. Dans le cas périodique, la transformée de Bloch généralise la transformée de Fourier et toutes les méthodes de phases stationnaires sont encore applicables. Nous voulons nous inspirer des méthodes de Cuccagna sur l'équation de Schrödinger à potentiel périodique ([49, 50]) pour obtenir un estimation de dispersion de type L^{∞} dans ce contexte.

1.5.3 Vers un temps d'existence plus long dans le cas rotationnel

Dans la section 4.3, nous donnons un temps d'existence de la forme $\frac{T}{\max(\varepsilon,\beta,\frac{\varepsilon}{R_0})}$ pour le système (1.31) dans le cas où P est constante (Théorème 4.3.6). Dans la section 5.2, nous étudions les équations de Boussinesq avec vorticité et nous montrons l'existence de solutions sur un temps $\frac{T}{\max(\varepsilon,\beta,\frac{\varepsilon\sqrt{\mu}}{R_0})}$ (Proposition 5.2.15). Nous espérons améliorer le théorème 4.3.6 et obtenir une temps d'existence similaire aux équations de Boussinesq. Cela nous permettrait de justifier l'équation d'Ostrovsky sur un temps plus long $\left(\text{en } \frac{T}{\mu} \text{ au lieu de } \frac{T}{\sqrt{\mu}}, \text{ voir Théorème 5.3.10} \right)$.

Mais le temps d'existence que nous donnons dépend du paramètre de Coriolis. Dans l'optique de justifier des modèles asymptotiques sur des temps plus longs, il pourrait être intéressant de trouver un temps d'existence pour le système (1.31) qui soit indépendant du paramètre de Coriolis. En collaboration avec Stefano Scrobogna (IMB), nous voulons nous attaquer à cette problématique. Pour commencer, nous souhaitons obtenir un temps d'existence indépendant du paramètre de Coriolis sur les équations de Navier-Stokes à surface libre sans fond (profondeur infinie) et avec une viscosité ν non nulle fixe,

$$\begin{cases} \partial_t \mathbf{U} + (\mathbf{U} \cdot \nabla_{X,z}) \mathbf{U} + \mathbf{f} \times \mathbf{U} - \nu \Delta_{X,z} \mathbf{U} = -\frac{1}{\rho} \nabla_{X,z} \mathcal{P} - g \mathbf{e}_{\mathbf{z}} \operatorname{dans} \{ z < \zeta(t, X) \}, \\ \operatorname{div}_{X,z} \mathbf{U} = 0 \operatorname{dans} \{ z < \zeta(t, X) \}, \end{cases}$$
(1.35)

auxquelles nous rajoutons les conditions aux bords suivantes

$$\begin{cases} \partial_t \zeta - \underline{\mathbf{U}} \cdot N = 0, \text{ on } z = \zeta, \\ \mathcal{P}N - \nu \left(\nabla \mathbf{U} + \nabla \mathbf{U}^t \right) N = g \zeta N, \text{ on } z = \zeta. \end{cases}$$
(1.36)

Un premier travail a été fait par Thai ([140]) pour des données initiales petites. Nous voulons traiter le cas de données initiales quelconques.

Ce travail nous permettrait aussi d'étudier les fluides géophysiques dans le cas d'une surface libre et de justifier le modèle quasi-géostrophique dans ce contexte (voir par exemple [37] ou [59] dans le cas d'une surface plate fixe).

Chapter 2

A mathematical study of meteorological and landslide tsunamis : The Proudman resonance

Sommaire

2.1	2.1 Introduction		
	2.1.1	Presentation of the problem	32
	2.1.2	Notations for this chapter	34
2.2	Loca	l existence of the water waves equations	35
	2.2.1	The model	35
	2.2.2	Notations for this section and statement of the main results	37
	2.2.3	Quasilinearization	39
	2.2.4	Local existence	43
	2.2.5	The Rayleigh-Taylor coefficient	47
	2.2.6	Hamiltonian system	48
2.3	Asyr	mptotic models	50
	2.3.1	A shallow water model for very small topography variation $\ldots \ldots$	51
	2.3.2	A shallow water model when the topography is not small $\ \ldots \ \ldots$.	54
	2.3.3	Linear asymptotic and resonance in intermediate depths	59
	2.3.4	Toward nonlinear asymptotic models	66

Ce chapitre est basé sur l'article [102]. Nous avons ajouté la sous-section 2.2.5 où nous expliquons le lien entre la pression et le coefficient de Rayleigh-Taylor. Nous avons aussi ajouté la sous-section 2.3.4 où nous étudions la résonance de Proudman dans le cas non linéaire. Nous avons amélioré le résultat de la sous-section 2.2.6 en prenant ψ dans un espace de type Beppo Levi (et non Sobolev). Enfin, nous avons étoffé la sous-section 2.3.2.2 et nous avons prouvé qu'une résonance peut se produire dans le cas d'un fond non plat pour des glissements de terrain qui se déplacent à vitesse 1.

2.1 Introduction

2.1.1 Presentation of the problem

In this chapter, we want to understand the Proudman resonance. It is a resonant respond in shallow waters of a water body on a traveling atmospheric disturbance when the speed of the disturbance is close to the typical water wave velocity. We show here that the same kind of resonance exists for landslide tsunamis and we propose a mathematical approach to investigate these phenomena based on the derivation, justification and analysis of relevant asymptotic models. This approach allows us to investigate more complex phenomena that are not dealt with in the physics literature such as the influence of a variable bottom or the generalization of the Proudman resonance in deeper waters.

A tsunami is popularly an elevation of the sea level due to an earthquake. However, tsunamis induced by seismic sources represent only 80 % of the tsunamis. 6% are due to landslides and 3% to meteorological effects ([86]). Big traveling storms for instance can give energy to the sea and lead to an elevation of the surface. In some cases, this amplification is important and this phenomenon is called the Proudman resonance ([120]) in the physics literature. Similarly, submarine landslides can significantly increase the level of the sea and we talk about landslide tsunamis. In this chapter, we study mathematically these two phenomena.

We model the sea by an irrotational and incompressible ideal fluid bounded from below by the seabed and from above by a free surface. We suppose that the seabed and the surface are graphs above the still water level. We model an underwater landslide by a moving seabed (moving bottom) and the meteorological effects by a non constant pressure at the surface (air-pressure disturbance). Therefore, we suppose that $b(t, X) = b_0(X) + b_m(t, X)$, where b_0 represents a fixed bottom and b_m the variation of the bottom because of the landslide. Similarly, the pressure at the surface is of the form $P + P_{\text{ref}}$, where P_{ref} is a constant which represents the pressure far from the meteorological disturbance, and P(t, X) models the meteorological disturbance (we assume that the pressure at the surface is known). We denote by d the horizontal dimension, which is equal to 1 or 2. $X \in \mathbb{R}^d$ stands for the horizontal variable and $z \in \mathbb{R}$ is the vertical variable. H is the typical water depth. The water occupies a moving domain $\Omega_t := \{(X, z) \in \mathbb{R}^{d+1}, -H + b(t, X) < z < \zeta(t, X)\}$. The water is homogeneous (constant density ρ), inviscid, irrotational with no surface tension. We denote by U the velocity and Φ the velocity potential. We have $\mathbf{U} = \nabla_{X,z} \Phi$. The law governing the irrotational fluids is the Bernoulli law

$$\partial_t \Phi + \frac{1}{2} |\nabla_{X,z} \Phi|^2 + gz = \frac{1}{\rho} \left(P_{\text{ref}} - \mathcal{P} \right) \text{ in } \Omega_t, \qquad (2.1)$$

where \mathcal{P} is the pressure in the fluid domain. Changing Φ if necessary, it is possible to assume that $P_{\text{ref}} = 0$. Furthermore, the incompressibility of the fluid implies that

$$\Delta_{X,z} \Phi = 0 \text{ in } \Omega_t. \tag{2.2}$$

We suppose also that the fluid particles do not cross the bottom or the surface. We denote by \mathbf{n} the unit normal vector, pointing upward and $\partial_{\mathbf{n}}$ the upward normal derivative. Then, the boundary conditions are

$$\partial_t \zeta - \sqrt{1 + |\nabla \zeta|^2} \partial_\mathbf{n} \Phi = 0 \text{ on } \{ z = \zeta(t, X) \},$$
(2.3)

and

$$\partial_t b - \sqrt{1 + |\nabla b|^2} \partial_\mathbf{n} \Phi = 0 \text{ on } \{ z = -H + b(t, X) \}.$$

$$(2.4)$$

In 1968, Zakharov (see [153]) showed that the water waves problem is a Hamiltonian system and that ψ , the trace of the velocity potential at the surface ($\psi = \Phi_{|z=\zeta}$), and the surface ζ are canonical variables. Then, Craig, Sulem and Sulem (see [47] and [48]) formulate this remark into a system of two non local equations. We follow their construction to formulate our problem. Using the fact that Φ satisfies (2.2) and (2.4), we can characterize Φ thanks to ζ and $\psi = \Phi_{|z=\zeta}$

$$\begin{cases} \Delta_{X,z} \Phi = 0 \text{ in } \Omega_t, \\ \Phi_{|z=\zeta} = \psi , \sqrt{1 + |\nabla b|^2} \partial_{\mathbf{n}} \Phi_{|z=-H+b} = \partial_t b. \end{cases}$$
(2.5)

We decompose this equation in two parts, the surface contribution and the bottom contribution

$$\Phi = \Phi^S + \Phi^B,$$

such that

$$\begin{cases} \Delta_{X,z} \Phi^S = 0 \text{ in } \Omega_t, \\ \Phi^S{}_{|z=\zeta} = \psi , \sqrt{1 + |\nabla b|^2} \partial_{\mathbf{n}} \Phi^S{}_{|z=-H+b} = 0, \end{cases}$$
(2.6)

and

$$\begin{cases} \Delta_{X,z} \Phi^B = 0 \text{ in } \Omega_t, \\ \Phi^B_{|z=\zeta} = 0, \ \sqrt{1+|\nabla b|^2} \partial_{\mathbf{n}} \Phi^B_{|z=-H+b} = \partial_t b. \end{cases}$$
(2.7)

In the purpose of expressing (2.3) with ζ and ψ , we introduce two operators. The first one is the Dirichlet-Neumann operator

$$G[\zeta, b]: \psi \mapsto \sqrt{1 + |\nabla \zeta|^2} \partial_{\mathbf{n}} \Phi^S_{|z=\zeta}, \qquad (2.8)$$

where Φ^S satisfies (2.6). The second one is the Neumann-Neumann operator

$$G^{NN}[\zeta, b]: \partial_t b \mapsto \sqrt{1 + |\nabla \zeta|^2} \partial_{\mathbf{n}} \Phi^B_{|z=\zeta}, \qquad (2.9)$$

where Φ^B satisfies (2.7). Then, we can reformulate (2.3) as

$$\partial_t \zeta - G[\zeta, b](\psi) = G^{NN}[\zeta, b](\partial_t b).$$
(2.10)

Furthermore thanks to the chain rule, we can express $(\partial_t \Phi)_{|z=\zeta}$, $(\nabla_{X,z} \Phi)_{|z=\zeta}$ and $(\partial_z \Phi)_{|z=\zeta}$ in terms of ψ , ζ , $G[\zeta, b](\psi)$ and $G^{NN}[\zeta, b](\partial_t b)$. Then, we take the trace at the surface of (2.1) (since there is no surface tension we have $\mathcal{P}_{|z=\zeta} = P$) and we obtain a system of two scalar equations that reduces to the standard Zakharov/Craig-Sulem formulation when $\partial_t b = 0$ and P = 0,

$$\begin{cases} \partial_t \zeta - G[\zeta, b](\psi) = G^{NN}[\zeta, b](\partial_t b), \\ \partial_t \psi + g\zeta + \frac{1}{2} |\nabla \psi|^2 - \frac{1}{2} \frac{\left(G[\zeta, b](\psi) + G^{NN}[\zeta, b](\partial_t b) + \nabla \zeta \cdot \nabla \psi\right)^2}{(1 + |\nabla \zeta|^2)} = -\frac{P}{\rho}. \end{cases}$$
(2.11)

In the following, we work with a nondimensionalized version of the water waves equations with small parameters ε , β and μ (see section 2.2.1). The wellposedness of the water waves problem with a constant pressure and a fixed bottom was studied by many people. Wu ([150] and [151]) proved it in the case of an infinite depth without nondimensionalization. Then, Lannes ([78]) treated the case of a finite bottom without nondimensionalization, Iguchi ([70]) proved a local wellposedness result for μ small enough in order to justify shallow water approximations for water waves, and Alvarez-Samaniego and Lannes ([9]) showed, in the case of the nondimensionalized equations, that we can find an existence time $T = \frac{T_0}{\max(\varepsilon,\beta)}$ where T_0 does not depend on ε , β and μ . More recently, Mésognon-Gireau ([105]) improved the case of a moving bottom in order to justify asymptotic models for tsunamis. Finally, Alazard, Burq and Zuily study the optimal regularity for the initial data ([4]) and more recently, Alazard, Baldi and Han-Kwan ([2]) show that a well-chosen non constant external pressure can create any small amplitude two-dimensional gravity-capillary water waves (see also [1]).

We organize this chapter in two sections. Firstly in Section 2.2, we prove two local existence theorems for the water waves problem with a moving bottom and a non constant pressure at the surface by differentiating and "quasilinearizing" the water waves equations and we pay attention to the dependence of the time of existence and the size of the solution with respect to the parameters ε , β , λ and μ . This theorem extends the result of Iguchi ([71]) and Lannes (Chapter 4 in [80]). We also prove that the water waves problem can be viewed as a Hamiltonian system. All this part use results about elliptic problems, the Dirichlet-Neumann and the Neumann-Neumann operators that can be found in Appendix A. Secondly in Section 2.3, we justify some linear asymptotic models and study the Proudman resonance. First, in Section 2.3.1 we study the case of small topography variations in shallow waters, approximation used in the Physics literature to investigate the Proudman resonance; then in Section 2.3.2 we derive a model when the topography is not small in the shallow water approximation; and in Section 2.3.3 we study the linear water waves equations in order to extend the Proudman resonance in deep water with a small fixed topography.

2.1.2 Notations for this chapter

A good framework for the velocity in the Euler equations is the Sobolev spaces H^s . But we do not work with **U** but with ψ , the trace of Φ , and $\mathbf{U} = \nabla_{X,z} \Phi$. It will be too restrictive to take ψ in a Sobolev space. A good idea is to work with the Beppo Levi spaces (see [53]). For $s \ge 0$, the Beppo Levi spaces are

$$\dot{H}^{s}(\mathbb{R}^{d}) := \left\{ \psi \in L^{2}_{\text{loc}}(\mathbb{R}^{d}), \, \nabla \psi \in H^{s-1}(\mathbb{R}^{d}) \right\}.$$

In this chapter, C is a constant and for a function f in a normed space $(X, |\cdot|)$ or a parameter γ , $C(|f|, \gamma)$ is a constant depending on |f| and γ whose exact value has non importance. The norm $|\cdot|_2$ is the L^2 -norm and $|\cdot|_{\infty}$ is the L^{∞} -norm in \mathbb{R}^d . Let $f \in \mathcal{C}^0(\mathbb{R}^d)$ and $m \in \mathbb{N}$ such that $\frac{f}{1+|x|^m} \in L^{\infty}(\mathbb{R}^d)$. We define the Fourier multiplier $f(D): H^m(\mathbb{R}^d) \to L^2(\mathbb{R}^d)$ as

$$\forall u \in H^m(\mathbb{R}^d), \ \widehat{f(D)u}(\xi) = f(\xi)\widehat{u}(\xi).$$

In \mathbb{R}^d we denote the gradient operator by ∇ and in Ω or $S = \mathbb{R}^d \times (-1, 0)$ the gradient operator is denoted $\nabla_{X,z}$. Finally, we denote by $\Lambda := \sqrt{1+|D|^2}$ with $D = -i\nabla$. In this chapter, (,) is the standard $L^2(\mathbb{R}^d)$ scalar product.

2.2 Local existence of the water waves equations

This part is devoted to the wellposedness of the water waves equations (Theorems 2.2.3 and 2.2.4). We carefully study the dependence on the parameters ε , β , λ and μ on the existence time and on the size of the solution. Contrary to [80] and [71], we exhibit the nonlinearities of the water waves equations in order to obtain a better existence time in the presence of a source term.

2.2.1 The model

In this part, we present a nondimensionalized version of the water waves equations. In order to derive some asymptotic models to the water waves equations we introduce some dimensionless parameters linked to the physical scales of the system. The first one is the ratio between the typical free surface amplitude a and the water depth H. We define $\varepsilon := \frac{a}{H}$, called the nonlinearity parameter. The second one is the ratio between H and the characteristic horizontal scale L. We define $\mu := \frac{H^2}{L^2}$, called the shallowness parameter. The third one is the ratio between the order of bottom bathymetry amplitude a_{bott} and H. We define $\beta := \frac{a_{\text{bott}}}{H}$, called the bathymetric parameter. Finally, we denote by λ the ratio of the typical landslide amplitude $a_{\text{bott,m}}$ and a_{bott} . We also nondimensionalize the variables and the unknowns. We introduce (see also Figure 2.1)

$$\begin{cases} X' = \frac{X}{L}, \ z' = \frac{z}{H}, \ \zeta' = \frac{\zeta}{a}, \ b' = \frac{b}{a_{\text{bott}}}, \ b'_0 = \frac{b_0}{a_{\text{bott}}}, \ b'_m = \frac{b_m}{a_{\text{bott,m}}}, \ t' = \frac{\sqrt{gH}}{L}t, \\ \left(\Phi^S\right)' = \frac{H}{aL\sqrt{gH}}\Phi^S, \ \left(\Phi^B\right)' = \frac{L}{Ha_{\text{bott,m}}\sqrt{gH}}\Phi^B, \ \psi' = \frac{H}{aL\sqrt{gH}}\psi, \ P' = \frac{P}{a\rho g}. \end{cases}$$
(2.12)

Then,

$$\Omega'_t = \{ (X', z') \in \mathbb{R}^{d+1} , -1 + \beta b'(t', X') < z' < \varepsilon \zeta'(t', X') \}$$

Remark 2.2.1. It is worth noting that the nondimensionalization of Φ^S , ψ and t comes from the linear wave theory (in shallow water regime, the characteristic speed is \sqrt{gH}). See paragraph **1.3.2** in [80]. Let us explain the nondimensionalization of Φ^B . Consider the linear case

$$\begin{cases} \Delta_{X,z} \Phi^B = 0, \ -H < z < 0, \\ \Phi^B_{|z=0} = 0, \ \partial_z \Phi^B_{|z=-H} = \partial_t b. \end{cases}$$

A straightforward computation gives $\Phi^B = \frac{\sinh(z|D|)}{|D|\cosh(H|D|)} \partial_t b$. If the typical wavelength is L, the typical wave number is $\frac{2\pi}{L}$. Furthermore, the typical order of magnitude of $\partial_t b$ is $\frac{a_{bott,m}\sqrt{gH}}{L}$. Then, the order of magnitude of Φ^B in the shallow water case is

$$\frac{L}{2\pi} \frac{\sqrt{gH} a_{bott,m}}{L} \frac{\sinh(2\pi \frac{H}{L})}{\cosh(2\pi \frac{H}{L})} \sim \frac{\sqrt{gH} a_{bott,m} H}{L}.$$



Figure 2.1: Typical scales

For the sake of clarity, we omit the primes. We can now nondimensionalize the water waves problem. Using the notation

 $\nabla^{\mu}_{X,z} := (\sqrt{\mu} \nabla_X \ , \partial_z \)^t \text{ and } \Delta^{\mu}_{X,z} \ := \mu \Delta_X \ + \partial_z^2,$

the water waves equations (2.11) become in dimensionless form

$$\begin{cases} \partial_t \zeta - \frac{1}{\mu} G_\mu[\varepsilon\zeta,\beta b](\psi) = \frac{\beta\lambda}{\varepsilon} G_\mu^{NN}[\varepsilon\zeta,\beta b](\partial_t b), \\ \\ \partial_t \psi + \zeta + \frac{\varepsilon}{2} |\nabla\psi|^2 - \frac{\varepsilon}{2\mu} \frac{\left(G_\mu[\varepsilon\zeta,\beta b](\psi) + \frac{\lambda\beta\mu}{\varepsilon} G_\mu^{NN}[\varepsilon\zeta,\beta b](\partial_t b) + \mu\varepsilon\nabla\zeta\cdot\nabla\psi\right)^2}{(1+\varepsilon^2\mu|\nabla\zeta|^2)} = -P. \end{cases}$$
(2.13)

In the following $\partial_{\mathbf{n}}$ is the upward conormal derivative

$$\partial_{\mathbf{n}} \Phi^{S} = \mathbf{n} \cdot \begin{pmatrix} \sqrt{\mu} I_{d} \ 0 \\ 0 \ 1 \end{pmatrix} \nabla^{\mu}_{X,z} \Phi^{S}_{|\partial\Omega}.$$

Then, The Dirichlet-Neumann operator $G_{\mu}[\varepsilon\zeta,\beta b]$ is

$$G_{\mu}[\varepsilon\zeta,\beta b](\psi) := \sqrt{1+\varepsilon^2 |\nabla\zeta|^2} \partial_{\mathbf{n}} \Phi^S_{|z=\varepsilon\zeta} = -\mu\varepsilon\nabla\zeta\cdot\nabla_X \Phi^S_{|z=\varepsilon\zeta} + \partial_z \Phi^S_{|z=\varepsilon\zeta} , \qquad (2.14)$$

where Φ^S satisfies

$$\begin{cases} \Delta_{X,z}^{\mu} \Phi^{S} = 0 \text{ in } \Omega_{t} ,\\ \Phi^{S}_{|z=\varepsilon\zeta} = \psi , \partial_{\mathbf{n}} \Phi^{S}_{|z=-1+\beta b} = 0, \end{cases}$$
(2.15)

while the Neumann-Neumann operator $G_{\mu}^{N\!N}[\varepsilon\zeta,\beta b]$ is

$$G^{NN}_{\mu}[\varepsilon\zeta,\beta b](\partial_t b) := \sqrt{1+\varepsilon^2 |\nabla\zeta|^2} \partial_{\mathbf{n}} \Phi^B_{\ |z=\varepsilon\zeta} = -\mu\nabla(\varepsilon\zeta) \cdot \nabla_X \Phi^B_{\ |z=\varepsilon\zeta} + \partial_z \Phi^B_{\ |z=\varepsilon\zeta} , \quad (2.16)$$

where Φ^B satisfies
$$\begin{cases} \Delta^{\mu}_{X,z} \Phi^{B} = 0 \text{ in } \Omega_{t} , \\ \Phi^{B}_{|z=\varepsilon\zeta} = 0 , \sqrt{1+\beta^{2}|\nabla b|^{2}} \partial_{\mathbf{n}} \Phi^{B}_{|z=-1+\beta b} = \partial_{t} b. \end{cases}$$

$$(2.17)$$

Remark 2.2.2. We adimensionalized the Dirichlet-Neumann and the Neumann-Neumann operators as follows

$$G[\zeta,b](\psi) = \frac{aL\sqrt{gH}}{H^2}G_{\mu}[\varepsilon\zeta^{'},\beta b^{'}](\psi^{'}), \ G^{NN}[\zeta,b](\partial_{t}b) = \frac{a_{bott,m}\sqrt{gH}}{L}G_{\mu}^{NN}[\varepsilon\zeta^{'},\beta b^{'}](\partial_{t^{'}}b^{'}).$$

We add two classical assumptions. First, we assume some constraints on the nondimensionalized parameters and we suppose there exist $\rho_{\text{max}} > 0$ and $\mu_{\text{max}} > 0$, such that

$$0 < \varepsilon, \beta, \beta \lambda \le 1$$
, $\frac{\beta \lambda}{\varepsilon} \le \rho_{\max}$ and $\mu \le \mu_{\max}$. (2.18)

Furthermore, we assume that the water depth is bounded from below by a positive constant

$$\exists h_{\min} > 0, \, \varepsilon \zeta + 1 - \beta b \ge h_{\min}. \tag{2.19}$$

In order to quasilinearize the water waves equations, we have to introduce the vertical speed at the surface \underline{w} and horizontal speed at the surface \underline{V} . We define

$$\underline{w} := \underline{w}[\varepsilon\zeta,\beta b] \left(\psi,\frac{\beta\lambda}{\varepsilon}\partial_t b\right) = \frac{G_{\mu}[\varepsilon\zeta,\beta b](\psi) + \mu\frac{\beta\lambda}{\varepsilon}G_{\mu}^{NN}[\varepsilon\zeta,\beta b](\partial_t b) + \varepsilon\mu\nabla\zeta\cdot\nabla\psi}{1 + \varepsilon^2\mu|\nabla\zeta|^2}, \quad (2.20)$$

and

$$\underline{V} := \underline{V}[\varepsilon\zeta,\beta b] \left(\psi,\frac{\beta\lambda}{\varepsilon}\partial_t b\right) = \nabla\psi - \varepsilon \underline{w}[\varepsilon\zeta,\beta b] \left(\psi,\frac{\beta\lambda}{\varepsilon}\partial_t b\right)\nabla\zeta.$$
(2.21)

2.2.2 Notations for this section and statement of the main results

In this chapter, d = 1 or 2, $t_0 > \frac{d}{2}$, $N \in \mathbb{N}$ and $s \ge 0$. The constant $T \ge 0$ represents a final time. The pressure P and the bottom b are given functions. We suppose that $b \in W^{3,\infty}(\mathbb{R}^+; H^N(\mathbb{R}^d))$ and $P \in W^{1,\infty}(\mathbb{R}^+; \dot{H}^{N+1}(\mathbb{R}^d))$. We denote by M_N a constant of the form

$$M_N = C\left(\frac{1}{h_{\min}}, \mu_{\max}, \varepsilon |\zeta|_{H^{\max(t_0+2,N)}}, \beta |b|_{L_t^{\infty} H_X^{\max(t_0+2,N+1)}}\right).$$
(2.22)

We denote by $U := (\zeta, \psi)^t$ the unknowns of our problem. We want to express (2.11) as a quasilinear system. It is well-known that the good energy for the water waves problem is

$$\mathcal{E}^{N}(U) = |\mathfrak{P}\psi|_{H^{\frac{3}{2}}}^{2} + \sum_{\alpha \in \mathbb{N}^{d}, |\alpha| \le N} \left(|\zeta_{(\alpha)}|_{2}^{2} + |\mathfrak{P}\psi_{(\alpha)}|_{2}^{2} \right),$$
(2.23)

where $\zeta_{(\alpha)} := \partial^{\alpha} \zeta$, $\psi_{(\alpha)} := \partial^{\alpha} \psi - \varepsilon \underline{w} \partial^{\alpha} \zeta$ and $\mathfrak{P} := \frac{|D|}{\sqrt{1 + \sqrt{\mu} |D|}}$. This energy is motivated by the linearization of the system around the rest state (see Part **4.1** in [80]). \mathfrak{P} acts as the square root of the Dirichlet-Neumann operator (see Proposition A.2.4). Here, $\zeta_{(\alpha)}$ and $\psi_{(\alpha)}$ are the *Alinhac's good unknowns of the system* (see [7] and [6] in the case of the standard water waves problem).

We define $U_{(\alpha)} := (\zeta_{(\alpha)}, \psi_{(\alpha)})^t$. We can introduce an associated energy space. Considering a $T \ge 0$, we define

$$E_T^N := \{ U \in \mathcal{C}([0,T]; H^{t_0+2}(\mathbb{R}^d) \times \dot{H}^2(\mathbb{R}^d)) , \mathcal{E}^N(U) \in L^\infty([0,T]) \}.$$
(2.24)

Our main results are the following theorems. We give two existence results. The first theorem extends the result of Iguchi (Theorem 2.4 in [71]) since we give a control of the dependence of the solution with respect to the parameters ε , β and μ and we add a non constant pressure at the surface and also extends the result of Lannes (Theorem 4.16 in [80]), since we improve the regularity of the initial data and we add a non constant pressure pressure at the surface and a moving bottom. Notice that we explain later what is Condition (2.29) (it corresponds to the positivity of the so called Rayleigh-Taylor coefficient).

Theorem 2.2.3. Let A > 0, $t_0 > \frac{d}{2}$, $N \ge \max(1, t_0) + 3$, $U^0 \in E_0^N$, $b \in W^{3,\infty}(\mathbb{R}^+; H^{N+1}(\mathbb{R}^d))$ and $P \in W^{1,\infty}(\mathbb{R}^+; \dot{H}^{N+1}(\mathbb{R}^d))$ such that

$$\mathcal{E}^{N}\left(U^{0}\right) + \frac{\beta\lambda}{\varepsilon} \left|\partial_{t}b\right|_{L_{t}^{\infty}H_{X}^{N}} + \left|\nabla P\right|_{L_{t}^{\infty}H_{X}^{N}} \leq A.$$

We suppose that the parameters ε , β , μ , λ satisfy (2.18) and that (2.19) and (2.29) are satisfied initially. Then, there exists T > 0 and a unique solution $U \in E_T^N$ to (2.13) with initial data U^0 . Moreover, we have

$$T = \min\left(\frac{T_0}{\max(\varepsilon,\beta)}, \frac{T_0}{\frac{\beta\lambda}{\varepsilon} |\partial_t b|_{L_t^{\infty} H_X^N} + |\nabla P|_{L_t^{\infty} H_X^N}}\right) , \frac{1}{T_0} = c^1 \text{ and } \sup_{t \in [0,T]} \mathcal{E}^N\!(U) = c^2,$$

with $c^j = C\left(A, \frac{1}{h_{\min}}, \frac{1}{\mathfrak{a}_{\min}}, \mu_{\max}, \rho_{\max}, |b|_{W_t^{3,\infty} H_X^{N+1}}, |\nabla P|_{W_t^{1,\infty} H_X^N}\right).$

Notice that if $\partial_t b$ and P are of size $\max(\varepsilon, \beta)$, we find the same existence time that in Theorem 4.16 in [80]. The second result shows that it is possible to go beyond the time scale of the previous theorem. Although the norm of the solution is not uniformly bounded in terms of ε and β , we are able to make this dependence precise. This theorem will be used to justify some of the asymptotic models derived in Section 2.3 over large time scales when the pressure at the surface and the moving bottom are not supposed small. We introduce $\delta := \max(\varepsilon, \beta^2)$.

Theorem 2.2.4. Under the assumptions of the previous theorem, there exists $T_0 > 0$ such that $U \in E^N_{\frac{T_0}{\sqrt{\alpha}}}$. Moreover, for all $\alpha \in [0, \frac{1}{2}]$, we have

$$\frac{1}{T_0} = c^1 \text{ and } \sup_{t \in [0, \frac{T_0}{\delta \alpha}]} \mathcal{E}^N(U) \le \frac{c^3}{\delta^{2\alpha}}$$

where $c^j = C\left(A, \frac{1}{h_{\min}}, \frac{1}{\mathfrak{a}_{\min}}, \mu_{\max}, \rho_{\max}, |b|_{W^{3,\infty}_t H^{N+1}_x}, |\partial_t \nabla P|_{W^{1,\infty}_t H^N_x}
ight).$

w

Notice that when $\partial_t b$ and P are of size $\max(\varepsilon, \beta)$, the existence time of Theorem 2.2.3 is better than the one of Theorem 2.2.4. Theorem 2.2.4 is only useful when $\partial_t b$ and P are not small. Notice finally that Condition (2.29) is satisfied if ε is small enough. Hence, since in Section 2.3, we suppose that ε is small, it is reasonable to assume it.

2.2.3 Quasilinearization

Firstly, we give some controls of $|\mathfrak{P}\psi|_{H^s}$ and $|\mathfrak{P}\psi|_{(\alpha)}|_{H^s}$ with respect to the energy $\mathcal{E}^N(U)$.

Proposition 2.2.5. Let T > 0, $t_0 > \frac{d}{2}$ and $N \ge 2 + \max(1, t_0)$. Consider $U \in E_T^N$, $b \in W^{1,\infty}(\mathbb{R}^+; H^{N+1}(\mathbb{R}^d))$, such that ζ and b satisfy Condition (2.19) for all $0 \le t \le T$. We assume also that μ satisfies (2.18). Then, for $0 \le t \le T$, for $\alpha \in \mathbb{N}^d$ with $|\alpha| \le N - 1$ and for $0 \le s \le N - \frac{1}{2}$,

$$|\partial^{\alpha}\mathfrak{P}\psi|_{2} + |\mathfrak{P}\psi_{(\alpha)}|_{H^{1}} + |\mathfrak{P}\psi|_{H^{s}} \leq M_{N}\mathcal{E}^{N}(U)^{\frac{1}{2}} + \frac{\beta\lambda}{\varepsilon}M_{N}\left|\partial_{t}b\right|_{L^{\infty}_{t}H^{N}_{X}}.$$

Proof. For the first inequality, we have thanks to Proposition B.1.1,

$$\begin{aligned} |\partial^{\alpha} \mathfrak{P} \psi|_{2} &\leq |\mathfrak{P} \psi_{(\alpha)}|_{2} + \varepsilon |\mathfrak{P}(\underline{w} \partial^{\alpha} \zeta)|_{2}, \\ &\leq |\mathfrak{P} \psi_{(\alpha)}|_{2} + \frac{\varepsilon}{\mu^{\frac{1}{4}}} |\underline{w} \partial^{\alpha} \zeta|_{H^{\frac{1}{2}}}. \end{aligned}$$

Since $\psi \in \dot{H}^2(\mathbb{R}^d)$, by Proposition A.2.17, $\underline{w} \in H^1(\mathbb{R}^d)$, and thanks to Proposition B.2.1, we obtain

$$|\partial^{\alpha}\mathfrak{P}\psi|_{2} \leq |\mathfrak{P}\psi_{(\alpha)}|_{2} + C\varepsilon \left|\frac{\underline{w}}{\mu^{\frac{1}{4}}}\right|_{H^{1}} |\zeta|_{H^{N}} \leq |\mathfrak{P}\psi_{(\alpha)}|_{2} + M_{N}\left(|\mathfrak{P}\psi|_{H^{\frac{3}{2}}} + \frac{\beta\lambda}{\varepsilon}|\partial_{t}b|_{H^{1}}\right).$$

The other inequalities follow with the same arguments, see for instance Lemma 4.6 in [80]. \Box

The following statement is a first step to the quasilinearization of the water waves equations. It is essentially Proposition 4.5 in [80] and Lemma 6.2 in [71]. However, we improve the minimal regularity of U (we decrease the minimal value of N to 4 when d = 1) and we provide the dependence on $\partial_t b$ contrary to [71]. For those reasons, we give a proof of this proposition.

Proposition 2.2.6. Let $t_0 > \frac{d}{2}$, T > 0, $N \ge \max(t_0, 1) + 3$, $b \in W^{1,\infty}(\mathbb{R}^+; H^{N+1}(\mathbb{R}^d))$ and $U \in E_T^N$, such that ζ and b satisfy Condition (2.19) for all $0 \le t \le T$. We assume also that μ satisfies (2.18). Then, for all $\alpha \in \mathbb{N}^d$, $1 \le |\alpha| \le N$, we have,

$$\begin{split} \partial^{\alpha} \bigg(\frac{1}{\mu} G_{\mu} [\varepsilon\zeta, \beta b](\psi) + \frac{\lambda\beta}{\varepsilon} G_{\mu}^{NN} [\varepsilon\zeta, \beta b](\partial_{t}b) \bigg) = & \frac{1}{\mu} G_{\mu} [\varepsilon\zeta, \beta b](\psi_{(\alpha)}) + \frac{\beta\lambda}{\varepsilon} G_{\mu}^{NN} [\varepsilon\zeta, \beta b](\partial^{\alpha}\partial_{t}b) \\ & - \varepsilon \mathbf{1}_{\{|\alpha|=N\}} \nabla \cdot (\zeta_{(\alpha)} \underline{V}) + R_{\alpha}. \end{split}$$

Furthermore R_{α} is controlled

$$|R_{\alpha}|_{2} \leq M_{N} |(\varepsilon\zeta,\beta b)|_{H^{N}} \mathcal{E}^{N}(U)^{\frac{1}{2}} + \frac{\beta\lambda}{\varepsilon} M_{N} |\partial_{t}b|_{L^{\infty}_{t}H^{N}_{X}}$$

Proof. We adapt and follow the proof of Proposition 4.5 in [80]. See also Proposition 6.4 in [71]. Using Proposition A.2.20, we obtain

$$\begin{aligned} \partial^{\alpha} \bigg(\frac{1}{\mu} G_{\mu} [\varepsilon\zeta, \beta b](\psi) + \frac{\lambda\beta}{\varepsilon} G_{\mu}^{NN} [\varepsilon\zeta, \beta b](\partial_{t}b) \bigg) &= \frac{1}{\mu} G_{\mu} [\varepsilon\zeta, \beta b](\psi_{(\alpha)}) + \frac{\beta\lambda}{\varepsilon} G_{\mu}^{NN} [\varepsilon\zeta, \beta b](\partial^{\alpha}\partial_{t}b) \\ &- \varepsilon \mathbf{1}_{\{|\alpha|=N\}} \nabla \cdot (\zeta_{(\alpha)} \underline{V}) - \beta G_{\mu}^{NN} [\varepsilon\zeta, \beta b] \left(\nabla \cdot \left(\partial^{\alpha} b \, \widetilde{\underline{V}} \right) \right) \\ &+ R_{\alpha}, \end{aligned}$$

where $\underline{\widetilde{V}} = \underline{\widetilde{V}}[\varepsilon\zeta,\beta b](\psi,\partial_t b)$ is the horizontal velocity at the bottom and is defined in Equation (2.21) and R_{α} is a sum of terms of the form

$$A_{j,\iota,\nu} := d^j \left(\frac{1}{\mu} G_\mu(\partial^\nu \psi) + \frac{\beta \lambda}{\varepsilon} G^{NN}_\mu(\partial^\nu \partial_t b) \right) . (\partial^{\iota^1} \zeta, ..., \partial^{\iota^j} \zeta; \partial^{\iota^1} b, ..., \partial^{\iota^j} b),$$

where j is an integer and $\iota^1, ..., \iota^j$ and ν are multi-index, and

$$\sum_{1 \le l \le j} |\iota^l| + |\nu| = N,$$

with $(j, |\iota^{l_0}|, |\nu|) \neq (1, N, 0)$ and (0, 0, N). Here ι^{l_0} is such that $\max_{1 \leq l \leq j} |\iota^l| = |\iota^{l_0}|$. In particular, $1 \leq |\iota^{l_0}| \leq N$. First, using Propositions A.2.13, A.2.18 and Product estimates B.2.1, we get

$$\left|\beta G^{NN}_{\mu}[\varepsilon\zeta,\beta b]\left(\nabla\cdot\left(\partial^{\alpha}b\,\underline{\widetilde{V}}\right)\right)\right|_{2} \leq \beta M_{N}\left(|\mathfrak{P}\psi|_{H^{2}}+|\partial_{t}b|_{H^{1}}\right)$$

Then, we distinguish several cases.

a)
$$|\iota^{l_0}| + |\nu| \le N - 2$$
 and $|\iota^{l_0}| \le N - 3$ or $|\iota^{l_0}| + |\nu| \le N$, $|\iota^{l_0}| \le N - 3$ and $|\nu| \le N - 2$

Applying the second point of Proposition A.2.23 and the first point of Proposition A.2.24 with $s = \frac{1}{2}$ and $t_0 = \min(t_0, \frac{3}{2})$, we get that

$$|A_{j,\iota,\nu}|_2 \leq M_N \prod_l |(\varepsilon \partial^{\iota^l} \zeta, \beta \partial^{\iota^l} b)|_{H^3} \left[|\mathfrak{P} \partial^{\nu} \psi|_{H^1} + \frac{\beta \lambda}{\varepsilon} |\partial^{\nu} \partial_t b|_2 \right],$$

and the result follows by Proposition 2.2.5.

b)
$$|\iota^{l_0}| = N - 2$$
 and $|\nu| = 0$, 1 or 2:

We apply the fourth point of Proposition A.2.23 and the second point of Proposition A.2.24 with $s = \frac{1}{2}$ and $t_0 = \max(t_0, 1)$,

$$|A_{j,\iota,\nu}|_2 \leq M_N |(\varepsilon \partial^{\iota^{l_0}} \zeta, \beta \partial^{\iota^{l_0}} b)|_{H^{\frac{3}{2}} \prod_{l \neq l_0}} |(\varepsilon \partial^{\iota^l} \zeta, \beta \partial^{\iota^l} b)|_{H^{N-2}} \Big[|\mathfrak{P} \partial^{\nu} \psi|_{H^{N-2}} + \frac{\beta \lambda}{\varepsilon} |\partial^{\nu} \partial_t b|_{H^{N-2}} \Big].$$

Then, we get the result thanks to Proposition 2.2.5.

c)
$$\iota^1 = \iota$$
 with $|\iota| = N - 1, |\nu| = j = 1$:

We proceed as in Proposition 4.5 in [80], using Proposition A.2.12 and Propositions A.2.20, A.2.13, A.2.17.

d)
$$|\iota^{\iota_0}| = N - 1$$
 and $|\nu| = 0$:

Here j = 2. For instance we consider that $l_0 = 1$ and $|\iota^2| = 1$. Using the second inequality of Proposition A.2.24 we have

$$\left| d^2 G^{NN}_{\mu}(\partial_t b) . (\partial^{\iota^1} \zeta, \partial^{\iota^2} \zeta; \partial^{\iota^1} b, \partial^{\iota^2} b) \right|_2 \leq M_N \left| \left(\varepsilon \partial^{\iota^1} \zeta, \beta \partial^{\iota^1} b \right) \right|_{H^1} \left| \left(\varepsilon \partial^{\iota^2} \zeta, \beta \partial^{\iota^2} b \right) \right|_{H^2} |\partial_t b|_{H^2}.$$

Furthermore, using two times Proposition A.2.20, we get

$$\begin{split} \frac{1}{\mu} d^2 G_{\mu}(\psi).(\partial^{\iota^1} \zeta, \partial^{\iota^2} \zeta; \partial^{\iota^1} b, \partial^{\iota^2} b) &= -\frac{\varepsilon}{\sqrt{\mu}} dG_{\mu}[\varepsilon \zeta, \beta b] \left(\partial^{\iota^1} \zeta \frac{1}{\sqrt{\mu}} \underline{w}(\psi, 0) \right).(\partial^{\iota^2} \zeta, 0) \\ &- \frac{\varepsilon}{\sqrt{\mu}} G_{\mu}[\varepsilon \zeta, \beta b] \left(\partial^{\iota^1} \zeta \frac{1}{\sqrt{\mu}} d\underline{w}(\psi, 0).(\partial^{\iota^2} \zeta, 0) \right) \\ &- \varepsilon \nabla \cdot \left(\partial^{\iota^1} \zeta d\underline{V}(\psi, 0).(\partial^{\iota^2} \zeta, 0) \right) \\ &+ \beta dG_{\mu}^{NN}[\varepsilon \zeta, \beta b] \left(\partial^{\iota^1} b \ \underline{\widetilde{V}}(\psi, 0).(0, \partial^{\iota^2} b) \right) \\ &+ \beta G_{\mu}^{NN}[\varepsilon \zeta, \beta b] \left(\partial^{\iota^1} b \ d\underline{\widetilde{V}}(\psi, 0).(0, \partial^{\iota^2} b) \right). \end{split}$$

The control follows from the first inequality of Proposition A.2.12, Proposition A.2.17, and Propositions A.2.22, A.2.17 and A.2.18.

e) $|\nu| = N - 1$ and $|\iota^{l_0}| = 1$:

Here, j = 1. It is clear that

$$\left|\frac{\beta\lambda}{\varepsilon} dG^{NN}_{\mu}(\partial^{\nu}\partial_{t}b).(\partial^{\iota^{1}}\zeta;\partial^{\iota^{1}}b)\right|_{2} \leq \frac{\beta\lambda}{\varepsilon} M_{N} |\partial_{t}b|_{H^{N}}.$$

Furthermore,

$$\frac{1}{\mu}dG_{\mu}(\partial^{\nu}\psi).(\partial^{\iota^{1}}\zeta;\partial^{\iota^{1}}b) = \frac{1}{\mu}dG_{\mu}(\psi_{(\nu)}).(\partial^{\iota^{1}}\zeta;\partial^{\iota^{1}}b) + \frac{1}{\sqrt{\mu}}dG_{\mu}\left(\frac{\varepsilon}{\sqrt{\mu}}\underline{w}\partial^{\nu}\zeta\right).(\partial^{\iota^{1}}\zeta;\partial^{\iota^{1}}b).$$

Then, using Proposition A.2.12 and Proposition 2.2.5, we get the result.

This Proposition enables to quasilinearize the first equation of the water waves equations. For the second equation, it is the purpose of the following proposition.

Proposition 2.2.7. Let $t_0 > \frac{d}{2}$, T > 0, $N \ge \max(t_0, 1) + 3$, $b \in W^{1,\infty}(\mathbb{R}^+; H^{N+1}(\mathbb{R}^d))$ and $U \in E_T^N$, such that ζ and b satisfy (2.19) for all $0 \le t \le T$. We assume also that μ satisfies (2.18). Then, for all $\alpha \in \mathbb{N}^d$, $1 \le |\alpha| \le N$, we have,

$$\partial^{\alpha} \left[\frac{\varepsilon}{2} |\nabla \psi|^2 - \frac{\varepsilon}{2\mu} (1 + \varepsilon^2 \mu |\nabla \zeta|^2) \underline{w}^2 \right] = \varepsilon \underline{V} \cdot (\nabla \psi_{(\alpha)} + \varepsilon \partial^{\alpha} \zeta \nabla \underline{w}) - \frac{\varepsilon}{\mu} \underline{w} \, \partial^{\alpha} G_{\mu}(\psi) \\ - \beta \lambda \underline{w} \, \partial^{\alpha} G_{\mu}^{NN}(\partial_t b) + S_{\alpha}.$$

Furthermore S_{α} is controlled

$$|\mathfrak{P}S_{\alpha}|_{2} \leq \varepsilon M_{N} \mathcal{E}^{N}(U) + C \left(M_{N}, \frac{\beta \lambda}{\varepsilon} \left| \partial_{t} b \right|_{L^{\infty}_{t} H^{N}_{X}} \right) \varepsilon \mathcal{E}^{N}(U)^{\frac{1}{2}} + \varepsilon M_{N} \left(\frac{\beta \lambda}{\varepsilon} \left| \partial_{t} b \right|_{L^{\infty}_{t} H^{N}_{X}} \right)^{2}.$$

Proof. The proof of this Proposition is similar to the proof of Proposition 4.10 in [80] expect we use Propositions A.2.17 and A.2.20. See also Proposition 6.4 in [71]. \Box

Thanks to this linearization, we can "quasilinarize" equations (2.13). It is the purpose of the next proposition. Let us introduce, the Rayleigh-Taylor coefficient

$$\underline{\mathbf{\mathfrak{g}}} := \underline{\mathbf{\mathfrak{g}}}(U,\beta b) = 1 + \varepsilon \partial_t \left(\underline{w}[\varepsilon\zeta,\beta b] \left(\psi, \frac{\beta\lambda}{\varepsilon} \partial_t b \right) \right) + \varepsilon^2 \underline{V}[\varepsilon\zeta,\beta b] \left(\psi, \frac{\beta\lambda}{\varepsilon} \partial_t b \right) \cdot \nabla \left(\underline{w}[\varepsilon\zeta,\beta b] \left(\psi, \frac{\beta\lambda}{\varepsilon} \partial_t b \right) \right).$$
(2.25)

This quantity plays an important role and is linked to the pressure in the fluid domain (see Part 2.2.5). We also introduce two new operators,

$$\mathcal{A}[U,\beta b] := \begin{pmatrix} 0 & -\frac{1}{\mu}G_{\mu}[\varepsilon\zeta,\beta b] \\ \underline{\mathfrak{a}}(U,\beta b) & 0 \end{pmatrix}$$
(2.26)

and

$$\mathcal{B}[U,\beta b] := \begin{pmatrix} \varepsilon \nabla \cdot (\cdot \underline{V}) & 0\\ 0 & \varepsilon \underline{V} \cdot \nabla \end{pmatrix}.$$
(2.27)

We can now quasilinearize the water waves equations. We use the same arguments as in Proposition 4.10 in [80] and part **6** in [71]. Notice that we give here a precise estimate with respect to $\partial_t b$ and P of the residuals R_{α} and S_{α} and that the minimal value of N, regularity of U, is smaller than in Proposition 4.10 in [80].

Proposition 2.2.8. Let $t_0 > \frac{d}{2}$, T > 0, $N \ge \max(t_0, 1) + 3$, $U \in E_T^N$ satisfies (2.19) for all $0 \le t \le T$ and solving (2.13), $b \in W^{2,\infty}(\mathbb{R}^+; H^{N+1}(\mathbb{R}^d))$, $P \in L^{\infty}(\mathbb{R}^+; \dot{H}^{N+1}(\mathbb{R}^d))$. We assume also that μ satisfies (2.18). Then, for all $\alpha \in \mathbb{N}^d$, $1 \le |\alpha| \le N$, we have,

$$\partial_t U_{(\alpha)} + \mathcal{A}[U,\beta b](U_{(\alpha)}) + \mathbf{1}_{\{|\alpha|=N\}} \mathcal{B}[U,\beta b](U_{(\alpha)}) = \left(\frac{\lambda\beta}{\varepsilon} G^{NN}_{\mu}[0,0](\partial^{\alpha}\partial_t b), -\partial^{\alpha} P\right)^t + \left(\widetilde{R_{\alpha}}, S_{\alpha}\right)^t.$$
(2.28)

Furthermore, $\widetilde{R_{\alpha}}$ and S_{α} satisfy

$$\begin{cases} |\widetilde{R}_{\alpha}|_{2} \leq M_{N}|(\varepsilon\zeta,\beta b)|_{H^{N}}\mathcal{E}^{N}(U)^{\frac{1}{2}} + \frac{\beta\lambda}{\varepsilon}M_{N}|\partial_{t}b|_{L_{t}^{\infty}H_{X}^{N}}, \\ |\mathfrak{P}S_{\alpha}|_{2} \leq \varepsilon M_{N}\mathcal{E}^{N}(U) + C\left(M_{N},\frac{\beta\lambda}{\varepsilon}|\partial_{t}b|_{L_{t}^{\infty}H_{X}^{N}}\right)\varepsilon\mathcal{E}^{N}(U)^{\frac{1}{2}} + \varepsilon M_{N}\left(\frac{\beta\lambda}{\varepsilon}|\partial_{t}b|_{L_{t}^{\infty}H_{X}^{N}}\right)^{2}. \end{cases}$$

Proof. Thanks to Proposition A.2.24, we get

$$\begin{split} \left| G^{NN}_{\mu} [\varepsilon\zeta,\beta b] (\partial^{\alpha}\partial_{t}b) - G^{NN}_{\mu} [0,0] (\partial^{\alpha}\partial_{t}b) \right|_{2} &\leq \int_{0}^{1} \left| dG^{NN}_{\mu} [z\varepsilon\zeta,z\beta b] (\partial^{\alpha}\partial_{t}b).(\zeta,b) \right|_{2} dz \\ &\leq M_{N} |(\varepsilon\zeta,\beta b)|_{H^{N}} \left| \partial^{\alpha}\partial_{t}b \right|_{L^{\infty}_{t}H^{N}_{X}}. \end{split}$$

Then, denoting $\widetilde{R}_{\alpha} = R_{\alpha} + G_{\mu}^{NN}[\varepsilon\zeta,\beta b](\partial^{\alpha}\partial_t b) - G_{\mu}^{NN}[0,0](\partial^{\alpha}\partial_t b)$, we obtain the first equation thanks to Proposition 2.2.6. For the second equation, using Proposition 2.2.7 and the first equation of the water waves problem, we have

$$\begin{split} \partial_t \partial^{\alpha} \psi &= -\partial^{\alpha} \zeta - \varepsilon \underline{V} \cdot (\nabla \psi_{(\alpha)} + \varepsilon \partial^{\alpha} \zeta \nabla \underline{w}) + \frac{\varepsilon}{\mu} \underline{w} \, \partial^{\alpha} G_{\mu}(\psi) + \beta \underline{w} \, \partial^{\alpha} G_{\mu}^{NN}(\partial_t b) - \partial^{\alpha} P + S_{\alpha} \\ &= -\partial^{\alpha} \zeta - \varepsilon \underline{V} \cdot (\nabla \psi_{(\alpha)} + \varepsilon \partial^{\alpha} \zeta \nabla \underline{w}) + \varepsilon \underline{w} \partial_t \partial^{\alpha} \zeta - \partial^{\alpha} P + S_{\alpha} \\ &= -\partial^{\alpha} \zeta (1 + \varepsilon \partial_t \underline{w} + \varepsilon^2 \underline{V} \cdot \nabla \underline{w}) - \varepsilon \underline{V} \cdot \nabla \psi_{(\alpha)} + \varepsilon \partial_t (\underline{w} \partial^{\alpha} \zeta) - \partial^{\alpha} P + S_{\alpha} \\ &= -\underline{\mathfrak{a}} \partial^{\alpha} \zeta - \varepsilon \underline{V} \cdot \nabla \psi_{(\alpha)} + \varepsilon \partial_t (\underline{w} \partial^{\alpha} \zeta) - \partial^{\alpha} P + S_{\alpha}, \end{split}$$

and the result follows.

In the case of a constant pressure at the surface and a fixed bottom, it is well-known that system (2.28) is symmetrizable if

$$\exists \mathfrak{a}_{min} > 0 , \, \underline{\mathfrak{a}}(U,\beta b) \ge \mathfrak{a}_{\min}.$$

$$(2.29)$$

Then, we introduce the symmetrizer

$$\mathcal{S}[U,\beta b] := \begin{pmatrix} \underline{\mathfrak{a}}(U,\beta b) & 0\\ 0 & \frac{1}{\mu}G_{\mu}[\varepsilon\zeta,\beta b] \end{pmatrix}.$$
(2.30)

This symmetrization has an associated energy

$$\mathcal{F}^{\alpha}(U) = \frac{1}{2} \left(\mathcal{S}[U,\beta b](U_{(\alpha)}), U_{(\alpha)} \right), \text{ if } \alpha \neq 0,$$

$$\mathcal{F}^{0}(U) = \frac{1}{2} |\zeta|^{2}_{H^{\frac{3}{2}}} + \frac{1}{2} \left(\Lambda^{\frac{3}{2}} \psi, \frac{1}{\mu} G_{\mu}[\varepsilon \zeta, \beta b](\Lambda^{\frac{3}{2}} \psi) \right),$$

$$\mathcal{F}^{[N]}(U) = \sum_{|\alpha| \leq N} \mathcal{F}^{\alpha}(U).$$

(2.31)

As in Lemma 4.27 in [80], it can be shown that $\mathcal{F}^{[N]}$ and $\mathcal{E}^{[N]}$ are equivalent in the following sense.

Proposition 2.2.9. Let T > 0, $N \in \mathbb{N}$, $U \in E_T^N$ satisfying (2.19) and (2.29) for all $0 \le t \le T$. Then, for all $0 \le k \le N$, $\mathcal{F}^{[k]}$ is comparable to \mathcal{E}^k

$$\frac{1}{|\underline{\mathfrak{a}}(U,\beta b)|_{L^{\infty}} + M_N} \mathcal{F}^{[k]}[U,b] \le \mathcal{E}^k(U) \le \left(M_N + \frac{1}{\mathfrak{a}_{min}}\right) \mathcal{F}^{[k]}[U,b].$$
(2.32)

Remark 2.2.10. We said that the water waves equations (2.13) are symmetrizable under the assumption that the Rayleigh-Taylor coefficient is positive. In fact, Ebin ([56]) shown that if this condition is not satisfied, the water waves equations are illposed.

2.2.4 Local existence

The water water equations can be written as follow :

$$\partial_t U + \mathcal{N}(U) = (0, -P)^t, \qquad (2.33)$$

with $\mathcal{N}(U) = (\mathcal{N}_1(U), \mathcal{N}_2(U))^t$ and

$$\mathcal{N}_{1}(U) := -\frac{1}{\mu} G_{\mu}[\varepsilon\zeta,\beta b](\psi) - \frac{\beta\lambda}{\varepsilon} G_{\mu}^{NN}[\varepsilon\zeta,\beta b](\partial_{t}b),$$

$$\mathcal{N}_{2}(U) := \zeta + \frac{\varepsilon}{2} |\nabla\psi|^{2} - \frac{\varepsilon}{2\mu} \left(1 + \varepsilon^{2}\mu |\nabla\zeta|^{2}\right) \left(\underline{w}[\varepsilon\zeta,\beta b]\left(\psi,\frac{\beta\lambda}{\varepsilon}\partial_{t}b\right)\right)^{2}.$$

(2.34)

According to our quasilinearization, we need that \underline{a} be a positive real number. Therefore, we have to express \underline{a} without partial derivative with respect to t, particularly when t = 0. It is easy to check that

$$\underline{\mathfrak{a}}(U,\beta b) = 1 + \varepsilon^{2} \underline{V}[\varepsilon\zeta,\beta b] \left(\psi,\frac{\beta\lambda}{\varepsilon}\partial_{t}b\right) \cdot \nabla \left[\underline{w}[\varepsilon\zeta,\beta b] \left(\psi,\frac{\beta\lambda}{\varepsilon}\partial_{t}b\right)\right] + \varepsilon d\underline{w} \left(\psi,\frac{\beta\lambda}{\varepsilon}\partial_{t}b\right) \cdot \left(-\mathcal{N}_{1}(U),\partial_{t}b\right) + \varepsilon \underline{w}[\varepsilon\zeta,\beta b] \left(-P - \mathcal{N}_{2}(U),\frac{\beta\lambda}{\varepsilon}\partial_{t}^{2}b\right).$$

$$(2.35)$$

The following Proposition gives estimates for $\underline{\mathfrak{a}}(U,\beta b)$ and is adapted from Proposition 6.6 in [71].

Proposition 2.2.11. Let T > 0, $t_0 > \frac{d}{2}$, $N \ge \max(t_0, 1) + 3$, $(\zeta, \psi) \in E_T^N$ is a solution of the water waves equations (2.13), $P \in L^{\infty}(\mathbb{R}^+; \dot{H}^{N+1}(\mathbb{R}^d))$ and $b \in W^{2,\infty}(\mathbb{R}^+; H^{N+1}(\mathbb{R}^d))$, such that Condition (2.19) is satisfied. We assume also that μ satisfies (2.18). Then, for all $0 \le t \le T$,

$$\begin{aligned} |\underline{\mathfrak{a}}(U,\beta b) - 1|_{H^{t_0}} \leq & C \Big(M_N, \max(\beta\lambda,\beta) |\partial_t b|_{L^{\infty}_t H^N_X}, \varepsilon \mathcal{E}^N(U)^{\frac{1}{2}} \Big) \, \varepsilon \mathcal{E}^N(U)^{\frac{1}{2}} \\ &+ \varepsilon M_N \bigg(|\nabla P|_{L^{\infty}_t H^N_X} + \frac{\beta\lambda}{\varepsilon} |\partial_t^2 b|_{L^{\infty}_t H^N_X} \bigg). \end{aligned}$$

Furthermore, if $\partial_t^3 b \in L^{\infty}(\mathbb{R}^+; H^N(\mathbb{R}^d))$ and $\partial_t P \in L^{\infty}(\mathbb{R}^+; \dot{H}^N(\mathbb{R}^d))$, then,

$$\begin{split} |\partial_t(\underline{\mathfrak{a}}(U,\beta b))|_{H^{t_0}} \leq & C \Big(M_N, \max(\beta\lambda,\beta) \, |\partial_t b|_{W^{1,\infty}_t H^N_X} \,, |\nabla P|_{L^{\infty}_t H^N_X} \,, \varepsilon \mathcal{E}^N(U)^{\frac{1}{2}} \Big) \, \varepsilon \mathcal{E}^N(U)^{\frac{1}{2}} \\ & + \varepsilon C \Big(M_N, \max(\beta\lambda,\beta) \, |\partial_t b|_{L^{\infty}_t H^N_X} \Big) \bigg(|\nabla P|_{W^{1,\infty}_t H^N_X} + \frac{\beta\lambda}{\varepsilon} \, |\partial_t^2 b|_{W^{1,\infty}_t H^N_X} \Big). \end{split}$$

Proof. Using the first point of Proposition A.2.17 and Product estimate B.2.1 we have

$$|\underline{V}[\varepsilon\zeta,\beta b](\varepsilon\psi,\beta\lambda\partial_t b)\cdot\nabla[\underline{w}[\varepsilon\zeta,\beta b](\varepsilon\psi,\beta\lambda\partial_t b)]|_{H^{t_0}} \leq M_N \left(|\mathfrak{P}\varepsilon\psi|_{H^{t_0+\frac{1}{2}}}+\beta\lambda|\partial_t b|_{L^{\infty}_t H^{t_0}_X}\right)^2$$

Furthermore, thanks to the first point of Proposition A.2.24 and the first point of Proposition A.2.23 we obtain

$$\left|\varepsilon d\underline{w}\left(\psi,\frac{\beta\lambda}{\varepsilon}\partial_t b\right).\left(-\mathcal{N}_1(U),\partial_t b\right)\right|_{H^{t_0}} \leq M_N |(\varepsilon \mathcal{N}_1(U),\beta\partial_t b)|_{H^{t_0+1}} \left(|\mathfrak{P}\varepsilon\psi|_{H^{t_0+\frac{1}{2}}} + \beta\lambda \left|\partial_t b\right|_{L^\infty_t H^{t_0}_X}\right).$$

Then, the first inequality follows easily from Proposition A.2.17, Proposition 2.2.5 and Product estimate B.2.1. The second inequality can be proved similarly. $\hfill \Box$

Remark 2.2.12. Notice that in the previous proposition, we only use the fact that $\partial_t^2 b$ belongs to $L^{\infty}(\mathbb{R}^+; H^{t_0+1}(\mathbb{R}^d))$ and that $\partial_t^3 b$ belongs to $L^{\infty}(\mathbb{R}^+; H^{t_0}(\mathbb{R}^d))$.

We can now prove Theorems 2.2.3 and 2.2.4. We recall that $\delta := \max(\varepsilon, \beta^2)$.

Proof. We slice up this proof in three parts. First we regularize and symmetrize the equations, then we find some energy estimates and finally we conclude by convergence. We only give the energy estimates in this chapter and a carefully study of the nonlinearities of the water waves equations is done. We refer to the proof of Theorem 4.16 in [80] for the regularization, the convergence and the uniqueness (see also part 7 in [71]). For Theorem 2.2.3 (respectively Theorem 2.2.4), we assume that U solves (2.13) on [0, T] (respectively on $\left[0, \frac{T}{\sqrt{\delta}}\right]$) and that (2.19) and (2.29) are satisfied for $\frac{h_{min}}{2}$ and $\frac{a_{min}}{2}$ on [0, T] (respectively on $\left[0, \frac{T}{\sqrt{\delta}}\right]$) for some T > 0.

a) $|\alpha| = 0$, The 0 - energy

We proceed as in Subsection 4.3.4.3 in [80] and part 6 in [71]. We have

$$\frac{d}{dt}\mathcal{F}^{0}(U) = \frac{1}{2\mu} \left(dG_{\mu}[\varepsilon\zeta,\beta b](\Lambda^{\frac{3}{2}}\psi).(\partial_{t}\zeta,\partial_{t}b),\Lambda^{\frac{3}{2}}\psi \right) + \frac{\beta\lambda}{\varepsilon} \left(\Lambda^{\frac{3}{2}}G_{\mu}^{NN}[\varepsilon\zeta,\beta b](\partial_{t}b),\Lambda^{\frac{3}{2}}\zeta \right) \\
- \left(\Lambda^{\frac{3}{2}}\left(\mathcal{N}_{2}(U)-\zeta\right),\frac{1}{\mu}G_{\mu}[\varepsilon\zeta,\beta b](\Lambda^{\frac{3}{2}}\psi)\right) - \left(\frac{1}{\mu}G_{\mu}[\varepsilon\zeta,\beta b](\Lambda^{\frac{3}{2}}\psi),\Lambda^{\frac{3}{2}}P\right).$$
(2.36)

We have to control all the term in the r.h.s.

* Control of $\frac{\beta\lambda}{\varepsilon} \left(\Lambda^{\frac{3}{2}} G^{NN}_{\mu} [\varepsilon\zeta, \beta b](\partial_t b), \Lambda^{\frac{3}{2}} \zeta \right).$

Using Proposition A.2.13, we get

$$\left|\frac{\beta\lambda}{\varepsilon} \left(\Lambda^{\frac{3}{2}} G^{NN}_{\mu}[\varepsilon\zeta,\beta b](\partial_t b),\Lambda^{\frac{3}{2}}\zeta\right)\right| \leq M_N \frac{\beta\lambda}{\varepsilon} \left|\partial_t b\right|_{L^{\infty}_t H^N_X} \mathcal{E}^N(U)^{\frac{1}{2}}.$$

* Control of $\left(\Lambda^{\frac{3}{2}}\left(\mathcal{N}_{2}(U)-\zeta\right),\frac{1}{\mu}G_{\mu}[\varepsilon\zeta,\beta b](\Lambda^{\frac{3}{2}}\psi)\right).$

Using Proposition 2.2.5 and Proposition A.2.17, we get

$$\begin{split} \left| \left(\Lambda^{\frac{3}{2}} (\mathcal{N}_2(U) - \zeta), \frac{1}{\mu} G_{\mu}[\varepsilon\zeta, \beta b](\Lambda^{\frac{3}{2}} \psi) \right) \right| &\leq \left| \mathcal{N}_2(U) - \zeta \right|_{H^{\frac{3}{2}}} \left| \frac{1}{\mu} G_{\mu}[\varepsilon\zeta, \beta b](\Lambda^{\frac{3}{2}} \psi) \right|_2, \\ &\leq \varepsilon M_N \mathcal{E}^N(U)^{\frac{3}{2}} + M_N \left(\frac{\beta \lambda}{\varepsilon} \left| \partial_t b \right|_{L^{\infty}_t H^N_X} \right)^2 \mathcal{E}^N(U). \end{split}$$

* Control of $\left(\frac{1}{\mu}G_{\mu}[\varepsilon\zeta,\beta b](\Lambda^{\frac{3}{2}}\psi),\Lambda^{\frac{3}{2}}P\right)$.

We get, using Propositions A.2.2 and A.2.4,

$$\left| \left(\frac{1}{\mu} G_{\mu} [\varepsilon \zeta, \beta b] (\Lambda^{\frac{3}{2}} \psi), \Lambda^{\frac{3}{2}} P \right) \right| \leq M_N E^N (U)^{\frac{1}{2}} |\nabla P|_{L_t^{\infty} H_X^N}.$$

* Control of $\frac{1}{2\mu} \left(dG_{\mu}[\varepsilon\zeta,\beta b](\Lambda^{\frac{3}{2}}\psi).(\partial_t\zeta,\partial_t b),\Lambda^{\frac{3}{2}}\psi \right).$

Using the second point of Proposition A.2.12, Proposition A.2.13 and Proposition 2.2.5, we get

$$\left|\frac{1}{\mu} \left(dG_{\mu}[\varepsilon\zeta,\beta b](\Lambda^{\frac{3}{2}}\psi).(\partial_{t}\zeta,\partial_{t}b),\Lambda^{\frac{3}{2}}\psi \right) \right| \leq M_{N} \left| \left(\varepsilon\mathcal{N}_{1}(U),\beta\partial_{t}b\right) \right|_{H^{N-2}} \left|\mathfrak{P}\psi\right|_{H^{\frac{3}{2}}}^{2} \\ \leq M_{N}\varepsilon\mathcal{E}^{N}(U)^{\frac{3}{2}} + \max(\beta,\beta\lambda) \left|\partial_{t}b\right|_{L_{t}^{\infty}H_{X}^{N}} \mathcal{E}^{N}(U).$$

Finally, gathering all the previous estimates, we get that

$$\frac{d}{dt}\mathcal{F}^{0}(U) \leq \varepsilon M_{N}\mathcal{E}^{N}(U)^{\frac{3}{2}} + M_{N}C\left(\rho_{\max}, |\partial_{t}b|_{L_{t}^{\infty}H_{X}^{N}}\right) \max(\varepsilon, \beta)\mathcal{E}^{N}(U)
+ M_{N}\sqrt{E^{N}(U)}\left(|\nabla P|_{L_{t}^{\infty}H_{X}^{N}} + \frac{\beta\lambda}{\varepsilon}|\partial_{t}b|_{L_{t}^{\infty}H_{X}^{N}}\right).$$
(2.37)

b) $|\alpha| > 0$, the higher orders energies

We proceed as in Subsection 4.3.4.3 in [80] and part 6 in [71]. A simple computation gives

$$\frac{d}{dt}(\mathcal{F}^{\alpha}(U)) = -\varepsilon \mathbf{1}_{\{|\alpha|=N\}} \left(\underline{\mathfrak{a}}\zeta_{(\alpha)}, \nabla \cdot \left(\zeta_{(\alpha)} \underline{V} \right) \right) + \left(\underline{\mathfrak{a}}\zeta_{(\alpha)}, \frac{\beta\lambda}{\varepsilon} G^{NN}_{\mu}[0,0](\partial_{t}\partial^{\alpha}b) + \widetilde{R_{\alpha}} \right)
- \varepsilon \mathbf{1}_{\{|\alpha|=N\}} \left(\frac{1}{\mu} G_{\mu}[\varepsilon\zeta,\beta b](\psi_{(\alpha)}), \underline{V} \cdot \nabla \psi_{(\alpha)} \right) + \left(\frac{1}{\mu} G_{\mu}[\varepsilon\zeta,\beta b](\psi_{(\alpha)}), S_{\alpha} - \partial^{\alpha}P \right)
+ \frac{1}{2} \left(\partial_{t} \underline{\mathfrak{a}}\zeta_{(\alpha)}, \zeta_{(\alpha)} \right) + \frac{1}{2} \left(\frac{1}{\mu} dG_{\mu}[\varepsilon\zeta,\beta b](\psi_{(\alpha)}).(\partial_{t}\zeta,\partial_{t}b), \psi_{(\alpha)} \right).$$
(2.38)

We have to control all the term in the r.h.s.

 $\star \text{ Control of } \big(\partial_t \underline{\mathfrak{a}} \zeta_{(\alpha)}, \zeta_{(\alpha)} \big).$

Using the second point of Proposition 2.2.11 we get

$$\begin{split} \left| \left(\partial_t \underline{\mathfrak{a}} \zeta_{(\alpha)}, \zeta_{(\alpha)} \right) \right| &\leq M_N C \left(\rho_{\max}, \left| \partial_t b \right|_{W_t^{1,\infty} H_X^N}, \left| \nabla P \right|_{L_t^{\infty} H_X^N}, \varepsilon \mathcal{E}^N(U)^{\frac{1}{2}} \right) \varepsilon \mathcal{E}^N(U)^{\frac{3}{2}} \\ &+ C \Big(M_N, \beta \lambda \left| \partial_t b \right|_{L_t^{\infty} H_X^N} \Big) \left(\left| \nabla P \right|_{W_t^{1,\infty} H_X^N} + \frac{\beta \lambda}{\varepsilon} \left| \partial_t^2 b \right|_{W_t^{1,\infty} H_X^N} \right) \varepsilon \mathcal{E}^N(U). \end{split}$$

 $\star \text{ Control of } \Big(\underline{\mathfrak{a}}\zeta_{(\alpha)}, \frac{\beta\lambda}{\varepsilon}G^{NN}_{\mu}[0,0](\partial_t\partial^{\alpha}b)\Big).$

We get, thanks to Proposition 2.2.11 and A.2.13, $\,$

$$\left| \left(\underline{\mathfrak{a}}\zeta_{(\alpha)}, \frac{\beta\lambda}{\varepsilon} G^{NN}_{\mu}[0,0](\partial_t \partial^{\alpha} b) \right) \right| \leq C \left(\rho_{\max}, \mu_{\max}, |b|_{W^{2,\infty}_t H^N_X}, |\nabla P|_{L^{\infty}_t H^N_X}, \varepsilon \mathcal{E}^N(U)^{\frac{1}{2}} \right) \frac{\beta\lambda}{\varepsilon} |\partial_t b|_{L^{\infty}_t H^N_X} \mathcal{E}^N(U)^{\frac{1}{2}}$$

 $\star \text{ Controls of } \varepsilon \mathbf{1}_{\{|\alpha|=N\}} \left(\underline{\mathfrak{a}}\zeta_{(\alpha)}, \nabla \cdot \left(\zeta_{(\alpha)}\underline{V}\right)\right).$

A simple computation gives, using Proposition 2.2.11 and Proposition A.2.17,

$$\begin{split} \left| \varepsilon \big(\underline{\mathfrak{a}} \zeta_{(\alpha)}, \nabla \cdot \big(\zeta_{(\alpha)} \underline{V} \big) \big) \right| &= \left| \varepsilon \left(\underline{\mathfrak{a}} \zeta_{(\alpha)} \nabla \cdot \underline{V}, \zeta_{(\alpha)} \right) \right| \\ &\leq C \Big(\rho_{\max}, \mu_{\max}, |b|_{W_t^{2,\infty} H_X^N}, |\nabla P|_{L_t^{\infty} H_X^N}, \delta \mathcal{E}^N(U) \Big) \varepsilon \Big[\mathcal{E}^N(U)^{\frac{3}{2}} + \mathcal{E}^N(U) \Big]. \end{split}$$

* Controls of the terms $\left(\frac{1}{\mu}G_{\mu}[\varepsilon\zeta,\beta b](\psi_{(\alpha)}),S_{\alpha}-\partial^{\alpha}P\right), \left(\frac{1}{\mu}dG_{\mu}[\varepsilon\zeta,\beta b](\psi_{(\alpha)}).(\partial_{t}\zeta,\partial_{t}b),\psi_{(\alpha)}\right)$ and $\left(\underline{\mathfrak{a}}\zeta_{(\alpha)},\widetilde{R_{\alpha}}\right).$

We can use the same arguments as in the third and the fourth point of part a) and Propositions 2.2.8 and 2.2.11.

* Control of $\left(\frac{1}{\mu}G_{\mu}[\varepsilon\zeta,\beta b](\psi_{(\alpha)}),\underline{V}\cdot\nabla\psi_{(\alpha)}\right).$

Thanks to Proposition A.2.19, we get the control.

Gathering the previous estimates and using Proposition 2.2.9, we obtain that

$$\frac{d}{dt}\mathcal{F}^{N}(U) \leq C\left(\rho_{\max}, \frac{1}{h_{\min}}, \mu_{\max}, \frac{1}{\mathfrak{a}_{\min}}, |b|_{W^{3,\infty}_{t}H^{N}_{x}}, |\nabla P|_{W^{1,\infty}_{t}H^{N}_{x}}, \varepsilon \mathcal{F}^{N}(U)^{\frac{1}{2}}\right) \times \left(\varepsilon \mathcal{F}^{N}(U)^{\frac{3}{2}} + \max(\varepsilon, \beta)\mathcal{F}^{N}(U) + \mathcal{F}^{N}(U)^{\frac{1}{2}}\left[\frac{\beta\lambda}{\varepsilon} |\partial_{t}b|_{L^{\infty}_{t}H^{N}_{x}} + |\nabla P|_{L^{\infty}_{t}H^{N}_{x}}\right]\right).$$

$$(2.39)$$

Then, we easily prove Theorem 2.2.3, using the same arguments as Subsection 4.3.4.4 in [80]. Furthermore, for $\alpha \in [0, \frac{1}{2}]$, defining $\widetilde{\mathcal{F}^N}(U)(\tau) = \delta^{2\alpha} \mathcal{F}^N(U)(\frac{\tau}{\delta^{\alpha}})$, we get

$$\frac{d}{d\tau}\widetilde{\mathcal{F}^{N}}(U) \leq C\left(\rho_{\max},\mu_{\max},\frac{1}{\mathfrak{a}_{\min}},\frac{1}{h_{\min}},|b|_{W^{3,\infty}_{t}H^{N}_{X}},|\nabla P|_{W^{1,\infty}_{t}H^{N}_{X}},\widetilde{\mathcal{F}^{N}}(U)\right)$$

We can also apply the same arguments as Subsection 4.3.4.4 in [80] and Theorem 2.2.4 follows. \Box

2.2.5 The Rayleigh-Taylor coefficient

The classical Rayleigh-Taylor criterion ([138]) is the fact that

$$\inf_{\mathbb{D}^d} [(-\partial_z \mathcal{P})_{|z=\varepsilon\zeta}] > 0,$$

where $\mathcal{P} = \mathcal{P}(x, z, t)$ is here the pressure in the domain Ω_t given by the Euler equation. In this part, we express the Rayleigh-Taylor coefficient $\underline{\mathbf{a}}$ in terms of the pressure and we show that the positivity of $\underline{\mathbf{a}}$ is linked to the Rayleigh-Taylor criterion. Recall that

$$\underline{\mathfrak{a}} := 1 + \varepsilon \partial_t \underline{w} + \varepsilon^2 \underline{V} \cdot \nabla \underline{w}.$$

We nondimensionalize \mathcal{P} by $\mathcal{P}' = \frac{\mathcal{P}}{\rho Hg}$ and we forget the prime in the following. Notice that $\mathcal{P}_{|z=\varepsilon\zeta} = \varepsilon P$.

Proposition 2.2.13. The Rayleigh-Taylor coefficient \mathfrak{a} defined in (2.25) can be expressed as

$$\underline{\mathfrak{a}} = -\varepsilon(\partial_z \mathcal{P})_{z=\varepsilon\zeta}$$

Proof. This proof is very similar to the one in Subsection 4.3.5 in [80]. We consider the adimensionalization of the velocity

$$\mathbf{U} = \frac{aL\sqrt{gH}}{H^2} \nabla^{\mu} \left(\Phi^S + \frac{\beta\mu\lambda}{\varepsilon} \Phi^B \right)$$

and $(V, w) := \mathbf{U}$. Then, (V, w) satisfy the nondimensionalized Euler equations

$$\begin{cases} \partial_t V + \varepsilon (V \cdot \nabla_X + \frac{1}{\mu} \partial_z) V = -\nabla_X \mathcal{P}, \\ \partial_t w + \varepsilon (V \cdot \nabla_X + \frac{1}{\mu} \partial_z) w = -(\partial_z \mathcal{P} + \frac{1}{\varepsilon}) \end{cases}$$

We take the trace at the surface of these equations denoted $\mathbf{U}_{|z=\varepsilon\zeta} := (\underline{V}, \underline{w})$. A straightforward computation gives

$$\begin{cases} \partial_t \underline{V} + \varepsilon \underline{V} \cdot \nabla_X \underline{V} + \varepsilon (\partial_t \zeta + \varepsilon \underline{V} \cdot \nabla \zeta - \frac{1}{\mu} \underline{w}) (\partial_z \underline{V})_{|z=\varepsilon\zeta} = -(\nabla_X \mathcal{P})_{|z=\varepsilon\zeta}, \\ \partial_t \underline{w} + \varepsilon \underline{V} \cdot \nabla_X \underline{w} + \varepsilon (\partial_t \zeta + \varepsilon \underline{V} \cdot \nabla \zeta - \frac{1}{\mu} \underline{w}) (\partial_z \underline{w})_{|z=\varepsilon\zeta} = -(\partial_z \mathcal{P})_{|z=\varepsilon\zeta} - \frac{1}{\varepsilon}. \end{cases}$$

Then, remarking that the first equation of the water waves equations (2.13) can be written as

$$\partial_t \zeta + \varepsilon \underline{V} \cdot \nabla \zeta - \frac{1}{\mu} \underline{w} = 0,$$

we obtain that

$$\partial_t \underline{w} + \varepsilon \underline{V} \cdot \nabla_X \underline{w} = -(\partial_z \mathcal{P})_{|z=\varepsilon\zeta} - \frac{1}{\varepsilon}.$$

2.2.6 Hamiltonian system

In this section we prove that the water waves problem (2.13) is a Hamiltonian system. This extends the classical result of Zakharov ([153]) to the case where the bottom is moving and the atmospheric pressure is not constant (see also [44]). In the case of a moving bottom, Guyenne and Nicholls already pointed out it in [66] ⁽¹⁾. We have to introduce the Dirichlet-Dirichlet and the Neumann-Dirichlet operators

$$\begin{cases} G^{DD}_{\mu}[\varepsilon\zeta,\beta b](\psi) = \left(\Phi^{S}\right)_{|z=-1+\beta b},\\ G^{ND}_{\mu}[\varepsilon\zeta,\beta b](\partial_{t}b) = \left(\Phi^{B}\right)_{|z=-1+\beta b}, \end{cases}$$
(2.40)

where Φ^S is defined in (2.15) and Φ^B is defined in (2.17). We postpone the study of these operators to appendix A (Section A.2).

Remark 2.2.14. If we denote $\Phi := \Phi^S + \frac{\beta \lambda \mu}{\varepsilon} \Phi^B$, Φ satisfies

$$\begin{cases} \Delta^{\mu}_{X,z} \Phi = 0 \ in \ \Omega_t \ ,\\ \Phi_{|z=\varepsilon\zeta} = \psi \ , \ \sqrt{1+\beta^2 |\nabla b|^2} \partial_n \Phi_{|z=-1+\beta b} = \frac{\beta \lambda \mu}{\varepsilon} \partial_t b. \end{cases}$$

Then

$$\sqrt{1+\varepsilon^2 |\nabla\zeta|^2} \partial_n \Phi_{|z=\varepsilon\zeta} = G_\mu[\varepsilon\zeta,\beta b](\psi) + \frac{\beta\mu\lambda}{\varepsilon} G_\mu^{NN}[\varepsilon\zeta,\beta b](\partial_t b), \tag{2.41}$$

¹It seems that there is a typo in their hamiltonian; " $-\zeta v$ " should read " $+\zeta v$ ".

and

$$\Phi_{|z=-1+\beta b} = G^{DD}_{\mu}[\varepsilon\zeta,\beta b](\psi) + \frac{\beta\mu\lambda}{\varepsilon}G^{ND}_{\mu}[\varepsilon\zeta,\beta b](\partial_t b).$$
(2.42)

We also have to introduce the dual spaces of the Beppo Levi spaces $\dot{H}^{-s}(\mathbb{R}^d)$ which are the spaces

$$H^s_*(\mathbb{R}^d) := \left\{ u \in H^s(\mathbb{R}^d), \ \exists \underline{u} \in H^{s+1}(\mathbb{R}^d), \ u = |D|\underline{u} \right\},\$$

endowed with norm $|\cdot|_{H^s_*} := \left|\frac{\cdot}{|D|}\right|_{H^{s+1}}$. In the end of Subsection A.1.1, we give different properties of these spaces. We can now show that the water waves equations (2.13) is a Hamiltonian system.

Theorem 2.2.15. Let T > 0, $t_0 > \frac{d}{2}$. We assume that $\zeta, b \in \mathcal{C}^0([0,T]; H^{t_0+1}(\mathbb{R}^d))$, that $\psi \in \mathcal{C}^0([0,T]; \dot{H}^2(\mathbb{R}^d))$, that $\partial_t b \in \mathcal{C}^0([0,T]; H^1_*(\mathbb{R}^d))$ and that $P \in \mathcal{C}^0([0,T]; L^2(\mathbb{R}^d))$. We also suppose that (ζ, ψ) is a solution of (2.13) and verified Condition (2.19). Then, if we define $H = H(\zeta, \psi) = \mathcal{T}(\zeta, \psi) + \mathcal{U}(\zeta, \psi)$, where $\mathcal{T}(\zeta, \psi) = \mathcal{T}$ is

$$\mathcal{T} = \frac{1}{2\mu} \int_{\Omega_t} \left| \nabla^{\mu}_{X,z} \left(\Phi^S + \frac{\beta \lambda \mu}{\varepsilon} \Phi^B \right) \right|^2 + \int_{\mathbb{R}^d} \frac{\beta \lambda}{\varepsilon} \partial_t b \left(G^{DD}_{\mu} [\varepsilon \zeta, \beta b](\psi) + \frac{\beta \lambda \mu}{\varepsilon} G^{ND}_{\mu} [\varepsilon \zeta, \beta b](\partial_t b) \right), \quad (2.43)$$

and $\mathcal{U}(\zeta, \psi) = \mathcal{U}$ is

$$\mathcal{U} = \frac{1}{2} \int_{\mathbb{R}^d} \zeta^2 dX + \int_{\mathbb{R}^d} \zeta P dX, \qquad (2.44)$$

the water waves equations (2.13) take the form

$$\partial_t \begin{pmatrix} \zeta \\ \psi \end{pmatrix} = \begin{pmatrix} 0 & I \\ -I & 0 \end{pmatrix} \begin{pmatrix} \partial_{\zeta} H \\ \partial_{\psi} H \end{pmatrix}.$$

Remark 2.2.16. \mathcal{T} is the sum of the kinetic energy and the moving bottom contribution and \mathcal{U} the sum of the potential energy and the pressure contribution. Using Green's formula and Remark 2.2.14 we obtain that

$$\begin{split} \mathcal{T} = & \frac{1}{2} \int_{\mathbb{R}^d} \psi \left(\frac{1}{\mu} G_{\mu} [\varepsilon \zeta, \beta b](\psi) + \frac{\beta \lambda}{\varepsilon} G_{\mu}^{NN} [\varepsilon \zeta, \beta b](\partial_t b) \right) dX \\ &+ \frac{1}{2} \int_{\mathbb{R}^d} \frac{\beta \lambda}{\varepsilon} \partial_t b \left(G_{\mu}^{DD} [\varepsilon \zeta, \beta b](\psi) + \frac{\beta \lambda \mu}{\varepsilon} G_{\mu}^{ND} [\varepsilon \zeta, \beta b](\partial_t b) \right) dX, \end{split}$$

Remark 2.2.17. The fact that $\partial_t b$ has to be in $H^1_*(\mathbb{R}^d)$ is useful to make sense to \mathcal{T} (see Proposition A.2.8). Furthermore, notice that if $b \in \mathcal{C}^0([0,T]; L^1(\mathbb{R}^d))$ and $\partial_t b \in \mathcal{C}^0([0,T]; H^1_*(\mathbb{R}^d))$, Remark A.1.15 shows that the quantity $\int_{\mathbb{R}^d} b(t, \cdot) dX$ is constant. It means that the mass of the seabed stays constant during the movement which is a quite reasonable assumption for submarine landslides.

Proof. Using the linearity of the Dirichlet-Neumann and the Dirichlet-Dirichlet operators with respect to ψ and the fact that the adjoint of $G_{\mu}^{NN}[\varepsilon\zeta,\beta b]$ is $G_{\mu}^{DD}[\varepsilon\zeta,\beta b]$ (see Proposition A.2.8), we get that

$$\partial_{\psi} H = \frac{1}{\mu} G_{\mu} [\varepsilon \zeta, \beta b](\psi) + \frac{\beta \lambda}{\varepsilon} G_{\mu}^{NN} [\varepsilon \zeta, \beta b](\partial_t b).$$

Applying Proposition A.2.20 (which provides explicit expressions for shape derivatives) and remark 2.2.16, we obtain that

$$\begin{split} &2\partial_{\zeta}H = -\frac{\varepsilon}{\mu}G_{\mu}[\varepsilon\zeta,\beta b](\psi)\underline{w} + \varepsilon\nabla\psi\cdot\underline{V} - \varepsilon\frac{\beta\lambda}{\varepsilon}G_{\mu}^{NN}[\varepsilon\zeta,\beta b](\partial_{t}b)\underline{w} + 2P + 2\zeta, \\ &= -\frac{\varepsilon}{\mu}G_{\mu}[\varepsilon\zeta,\beta b](\psi)\underline{w} + \varepsilon\nabla\psi\cdot\nabla\psi - \varepsilon^{2}\underline{w}\nabla\psi\cdot\nabla\zeta - \varepsilon\frac{\beta\lambda}{\varepsilon}G_{\mu}^{NN}[\varepsilon\zeta,\beta b](\partial_{t}b)\underline{w} + 2P + 2\zeta, \\ &= \varepsilon\left|\nabla\psi\right|^{2} - \frac{\varepsilon}{\mu}\underline{w}^{2}\left(1 + \varepsilon^{2}\mu|\nabla\zeta|^{2}\right) + 2P + 2\zeta, \end{split}$$

which ends the proof.

2.3 Asymptotic models

In this part, we derive some asymptotic models in order to model two different types of tsunamis. The most important phenomenon that we want to catch is the Proudman resonance ([120], see also [109] or [144] for numerical simulations) and the submarine landslide tsunami phenomenon (see [86], [141] or [142] for numerical simulations). The resonance occurs in a linear case. The duration of the resonance depends on the phenomenon. For a meteotsunami, the duration of the resonance corresponds to the time the meteorological disturbance takes to reach the coast (see [109]). However, for a landslide tsunami, the duration of the resonance corresponds to the time the meteorological disturbance corresponds to the duration of the landslide (which depends on the size of the slope, see [86] or [141]). If the landslide is offshore, it is unreasonable to assume that the duration of the landslide is the time the water waves take to reach the coast. A variation of the pressure of 1 hPa creates a water wave of 1 cm whereas a moving bottom of 1 cm tends to create a water wave of 1 cm. Therefore we assume in the following that $a_{\text{bott,m}} = a$ (and hence $\beta \lambda = \varepsilon$). However, it is quite ordinary that a submarine landslide have a thickness of 1 m. Typically, a storm makes a variation of few Hpa, and the thickness of a submarine landslide is few dm (we refer to [86]).

In this part, we only study the propagation of such phenomena. Therefore, we take d = 1. In the following, we give three linear asymptotic models of the water waves equations and we give examples of pressures and moving bottoms that create a resonance. The pressure at the surface P and the moving bottom b_m move from the left to the right. We consider that the system is initially at rest. We start this part by giving an asymptotic expansion with respect to μ and $\max(\varepsilon, \beta)$ of $G_{\mu}[\varepsilon\zeta, \beta b]$ and $G_{\mu}^{NN}[\varepsilon\zeta, \beta b]$.

Proposition 2.3.1. Let $t_0 > \frac{d}{2}$, ζ and $b \in H^{t_0+2}(\mathbb{R}^d)$ such that Condition (2.19) is satisfied. We suppose that the parameters ε , β and μ satisfy (2.18). Then, for all $B \in H^{s-\frac{1}{2}}(\mathbb{R}^d)$ with $0 \le s \le t_0 + \frac{3}{2}$, we have

$$\left|G^{NN}_{\mu}[\varepsilon\zeta,\beta b](B) - G^{NN}_{\mu}[0,0](B)\right|_{H^{s-\frac{1}{2}}} \le M_0 |(\varepsilon\zeta,\beta b)|_{H^{t_0+2}} \left|B\right|_{H^{s-\frac{1}{2}}}$$

and

$$\left|G_{\mu}^{NN}[0,0](B) - B\right|_{H^{s-\frac{1}{2}}} \le C\mu \left|B\right|_{H^{s+\frac{3}{2}}}$$

Proof. The first inequality follows from Proposition A.2.24 and the second from Remark A.2.1. \Box

In the same way, using Proposition A.2.23 and Remark A.2.1, we have the following proposition.

Proposition 2.3.2. Let $t_0 > \frac{d}{2}$, ζ and $b \in H^{t_0+2}(\mathbb{R}^d)$ such that Condition (2.19) is satisfied. We suppose that the parameters ε , β and μ satisfy (2.18). Then, for all $\psi \in \dot{H}^{s+1}(\mathbb{R}^d)$ with $0 \le s \le t_0 + \frac{3}{2}$, we have

$$|G_{\mu}[\varepsilon\zeta,\beta b](\psi) - G_{\mu}[0,0](\psi)|_{H^{s-\frac{1}{2}}} \leq \mu M_0 |(\varepsilon\zeta,\beta b)|_{H^{t_0+2}} |\mathfrak{P}\psi|_{H^{s+\frac{1}{2}}}$$

and

$$\left|\frac{1}{\mu}G_{\mu}\!\left[0,0\right]\!(\psi)\!+\!\Delta\psi\right|_{H^{s-\frac{1}{2}}}\!\leq \mu C\left|\nabla\psi\right|_{H^{s+\frac{5}{2}}}$$

We denote by \overline{V} the vertically averaged horizontal component,

$$\overline{V} = \overline{V}[\varepsilon\zeta,\beta b](\psi,\partial_t b) = \frac{1}{1+\varepsilon\zeta-\beta b} \int_{-1+\beta b}^{\varepsilon\zeta} \nabla_X(\Phi[\varepsilon\zeta,\beta b](\psi,\partial_t b)(\cdot,z)) dz, \qquad (2.45)$$

where $\Phi = \Phi[\varepsilon\zeta, \beta b](\psi, \partial_t b)$ satisfies

$$\begin{cases} \Delta^{\mu}_{X,z} \Phi = 0, \ -1 + \beta b \le z \le \varepsilon \zeta, \\ \Phi_{|z=\varepsilon\zeta} = \psi \ , \ \sqrt{1 + \beta^2 |\nabla b|^2} \partial_{\mathbf{n}} \Phi_{|z=-1+\beta b} = \mu \partial_t b. \end{cases}$$

The following Proposition is Remark 3.36 and a small adaptation of Proposition 3.37 and Lemma 5.4 in [80] (see also Subsection A.5.5 in [80]).

Proposition 2.3.3. Let T > 0, $t_0 > \frac{d}{2}$, $0 \le s \le t_0$ and $\zeta, b \in W^{1,\infty}([0,T]; H^{t_0+2}(\mathbb{R}^d))$ such that Condition (2.19) is satisfied on [0,T]. We suppose that the parameters ε , β and μ satisfy (2.18). We also assume that $\psi \in W^{1,\infty}([0,T]; \dot{H}^{s+3}(\mathbb{R}^d))$. Then,

$$G_{\mu}[\varepsilon\zeta,\beta b](\psi) + \mu G_{\mu}^{NN}[\varepsilon\zeta,\beta b](\partial_t b) = -\mu\nabla\cdot\left((1+\varepsilon\zeta-\beta b)\overline{V}\right) + \mu\partial_t b,$$

and

$$\begin{cases} \left|\overline{V} - \nabla\psi\right|_{H^{s}} \leq \mu C\left(\frac{1}{h_{\min}}, \mu_{\max}, \varepsilon |\zeta|_{H^{t_{0}+2}}, \beta |b|_{L_{t}^{\infty} H_{X}^{t_{0}+2}}\right) \max\left(\left|\nabla\psi\right|_{H^{s+2}}, \left|\partial_{t}b\right|_{L_{t}^{\infty} H_{X}^{s+1}}\right), \\ \left|\partial_{t}\overline{V} - \nabla\partial_{t}\psi\right|_{H^{s}} \leq \mu C\left(\frac{1}{h_{\min}}, \mu_{\max}, |\zeta|_{H^{t_{0}+2}}, \left|\partial_{t}\zeta\right|_{H^{t_{0}+2}}, \left|b\right|_{W_{t}^{2,\infty} H_{X}^{t_{0}+2}}, \left|\nabla\psi\right|_{H^{s+2}}, \left|\partial_{t}\nabla\psi\right|_{H^{s+2}}\right). \end{cases}$$

In this part, we will consider symmetrizable linear hyperbolic systems of the first order. We refer to [16] for more details about the wellposedness. In the following, we will only give the energy associated to the symmetrization.

2.3.1 A shallow water model for very small topography variation

2.3.1.1 Linear asymptotic

We consider the case that ε , β , μ are small. Physically, this means that we consider small amplitudes for the surface and the bottom (compared to the mean depth) and waves with large wavelengths (compared to the mean depth). The asymptotic regime (in the sense of Definition 4.19 in [80]) is

$$\mathcal{A}_{LW} = \{ (\varepsilon, \beta, \lambda, \mu), \, 0 < \mu, \, \varepsilon, \, \beta \le \delta_0, \, \beta \lambda = \varepsilon \} \,, \tag{2.46}$$

with $\delta_0 \ll 1$.

Proposition 2.3.4. Let $t_0 > \frac{d}{2}$, $N \ge \max(1, t_0) + 3$, $U^0 \in E_0^N$, $P \in W^{1,\infty}(\mathbb{R}^+; \dot{H}^{N+1}(\mathbb{R}^d))$ and $b \in W^{3,\infty}(\mathbb{R}^+; H^{N+1}(\mathbb{R}^d))$. We suppose (2.19) and (2.29) are satisfied initially. Then, there exists T > 0, such that for all $(\varepsilon, \beta, \lambda, \mu) \in \mathcal{A}_{LW}$, there exists a solution $U = (\zeta, \psi) \in E_{\frac{T}{\sqrt{\delta_0}}}^N$ to the water waves equations with initial data U^0 and this solution is unique. Furthermore, for all $\alpha \in [0, \frac{1}{3})$,

$$\left|\zeta - \widetilde{\zeta}\right|_{L^{\infty}\left(\left[0, \frac{T}{\delta_{0}^{\alpha}}\right]; H^{N-4}(\mathbb{R}^{d})\right)} + \left|\nabla\psi - \nabla\widetilde{\psi}\right|_{L^{\infty}\left(\left[0, \frac{T}{\delta_{0}^{\alpha}}\right]; H^{N-2}(\mathbb{R}^{d})\right)} \le T\delta_{0}^{1-3\alpha}\widetilde{C}$$

where

$$\widetilde{C} = C\left(\mathcal{E}^{N}\left(U^{0}\right), \frac{1}{h_{\min}}, \frac{1}{\mathfrak{a}_{\min}}, |b|_{W^{3,\infty}_{t}H^{N}_{X}}, |\nabla P|_{W^{1,\infty}_{t}H^{N}_{X}}\right),$$

and with, $(\widetilde{\zeta}, \widetilde{\psi})$ solution of the waves equation

$$\begin{cases} \partial_t \widetilde{\zeta} + \Delta_X \widetilde{\psi} = \partial_t b, \\ \partial_t \widetilde{\psi} + \widetilde{\zeta} = -P, \end{cases}$$
(2.47)

with initial data U^0 .

Proof. First, the system (2.47) is wellposed since it can be symmetrized thanks to the energy

$$\mathcal{E}(t) = \left| \widetilde{\zeta} \right|_2^2 + \left| \nabla \widetilde{\psi} \right|_2^2.$$

Using Theorem 2.2.4 we get a uniform time of existence $\frac{T}{\sqrt{\delta_0}} > 0$ for the water waves equation and for all parameters in \mathcal{A}_{LW} . Then, using Proposition 2.3.1, Proposition 2.3.2, Proposition A.2.17 and B.1.1 and standard controls we get that

$$\begin{cases} \partial_t \zeta + \Delta_X \psi = \partial_t b + R_1, \\ \partial_t \psi + \zeta = -P + R_2, \end{cases}$$
(2.48)

with

$$\begin{cases} |R_1|_{H^{N-4}} \leq C\Big(\varepsilon|\zeta|_{H^N}, |b|_{L^{\infty}_t H^N_X}\Big)(|(\varepsilon\zeta, \beta b)|_{H^N} + \mu) \max\Big(|\mathfrak{P}\psi|_{H^{N-\frac{1}{2}}}, |\partial_t b|_{H^N}\Big) \\ |R_2|_{H^{N-1}} \leq \varepsilon C\Big(\varepsilon|\zeta|_{H^N}, |b|_{L^{\infty}_t H^N_X}\Big) \max\Big(|\mathfrak{P}\psi|_{H^{N-\frac{1}{2}}}^2, |\partial_t b|_{H^N}^2\Big). \end{cases}$$

If we denote $\zeta_1 = \zeta - \widetilde{\zeta}$ and $\psi_1 = \psi - \widetilde{\psi}$, we see that (ζ_1, ψ_1) satisfies

$$\begin{cases} \partial_t \zeta_1 + \Delta_X \psi_1 = R_1, \\ \partial_t \psi_1 + \zeta_1 = R_2. \end{cases}$$

Differentiating the energy

$$\mathcal{E}^{N}(t) = \frac{1}{2} \left| \zeta_{1} \right|_{H^{N-4}}^{2} + \frac{1}{2} \left| \nabla \psi_{1} \right|_{H^{N-2}}^{2}$$

we get the estimate thanks to Proposition 2.2.5 and energy estimate in Theorem 2.2.4. $\hfill \Box$

This model is well-known in the physics literature ([120] when $\partial_t b \equiv 0$, [141] when $P \equiv 0$ and [86]).

2.3.1.2 Resonance in shallow waters for very small topography variation

We consider the equation (2.47) for d = 1. We transform it in order to have a unique equation for $h := \tilde{\zeta} - b$,

$$\begin{cases} \partial_t^2 h - \partial_X^2 h = \partial_X^2 (P+b), \\ h_{|t=0} = -b(0,.), \\ \partial_t h_{|t=0} = 0. \end{cases}$$
(2.49)

We denote f(t, X) := (P + b)(t, X), which represents a disturbance. We want to understand the resonance for landslide and meteo tsunamis. In both cases, it is a linear respond, in the shallow water case, of a body of water due to a moving pressure or a moving bottom, when the speed of the storm or the landslide is close to the typical wave celerity (here 1). We can compute h thanks to the d'Alembert's formula

$$\begin{split} h(t,X) = \underbrace{-\frac{1}{2} \left(b(0,X-t) + b(0,X+t) \right)}_{h_T(t,X)} + \underbrace{\frac{1}{2} \int_0^t \partial_X f(\tau,X+t-\tau) d\tau}_{:=h_L(t,X)} \\ - \underbrace{\frac{1}{2} \int_0^t \partial_X f(\tau,X-t+\tau) d\tau}_{:=h_R(t,X)}. \end{split}$$

We are interesting in disturbances f moving from the left to the right (propagation to a coast). Therefore, we study only h_R . The following Proposition shows that a disturbance moving with a speed equal to 1 makes appear a resonance.

Proposition 2.3.5. Let $f \in L^{\infty}(\mathbb{R}^+; H^1(\mathbb{R}^d))$ and $\partial_X f \in L^{\infty}_{t \times X}(\mathbb{R} \times \mathbb{R}^d)$. Then, for all $X \in \mathbb{R}$, t > 0,

$$|h_R(t,X)| \le \frac{t}{2} \left|\partial_X f\right|_{\infty}$$

Furthermore, if $f(t, X) = f_0(X - t)$, $f_0 \in H^1(\mathbb{R}^d)$ and $|f'_0(X_0 - t_0)| = |f'|_{\infty}$ the equality holds for (t_0, X_0) . If $f(t, X) = f_0(X - Ut)$ with $f_0 \in H^1(\mathbb{R}^d)$ and $U \neq 1$,

$$|h_R|_{\infty} \le \min\left(\frac{|f_0|_{\infty}}{|1-U|}, \frac{t}{2} |f_0'|_{\infty}\right).$$

Proof. If $f(t, X) = f_0(X - Ut)$,

$$h_R(t,X) = -\frac{1}{2} \int_0^t f_0'(X - t + (1 - U)\tau) d\tau,$$

and the result follows.

This Proposition corresponds to the historical work of Proudman ([120]). We rediscover the fact that the resonance occurs if the speed of the disturbance is 1. For a disturbance with a speed different from 1, we notice a saturation effect (also pointed out in [141]). The graph in Figure



Figure 2.2: Evolution of the maximum of h, solution of equation (2.49), with different values of the speed U.

2.2, gives the typical evolution of $|h(t, \cdot)|_{\infty}$ with respect to the time t for different values of the speed. We can see the saturation effect. We compute h with a finite difference method and we take $f(t, X) = e^{-\frac{1}{2}(X-Ut)^2}$. We also see that the landslide resonance and the Proudman resonance have the same effects. There are however two important differences between these two phenomena. The first one is the duration of the resonance. A landslide is quicker than a meteorological effect. The second one, is the fact that the typical size of the landslide (few dm) is bigger than the size of a storm (few hPa). For instance, for a moving storm which creates a variation of the pressure of 3 hPa during $15t_0$, the final wave can reach a amplitude of 13 cm (it is for example the case of the meteotsunami in Nagasaki in 1979, see [109]). Conversely, an offshore landslide with a thickness of 1 m that lasts t_0 , can create a wave of 50 cm (which corresponds to the results in [141]). Therefore, we see that the principal difference between an offshore landslide and a moving storm is the size.

2.3.2 A shallow water model when the topography is not small

2.3.2.1 Linear asymptotic

In this case, we suppose only that ε and μ are small. We recall that $\beta b(t, X) = \beta b_0(X) + \beta \lambda b_m(t, X)$. Then, we assume also that $1 - b_0 \ge h_{\min} > 0$. In the following, we denote $h_0 := 1 - \beta b_0$. The asymptotic regime is

$$\mathcal{A}_{LVW} = \{ (\varepsilon, \beta, \lambda, \mu), \, 0 < \varepsilon, \, \mu \le \delta_0, \, 0 < \beta \le 1, \, \beta \lambda = \varepsilon \} \,, \tag{2.50}$$

with $\delta_0 \ll 1$. We can now give a asymptotic model.

Proposition 2.3.6. Let $t_0 > \frac{d}{2}$, $N \ge \max(1, t_0) + 4$, $b \in W^{3,\infty}(\mathbb{R}^+; H^{N+1}(\mathbb{R}^d))$, $U^0 = (\zeta_0, \psi_0) \in E_0^N$, and $P \in W^{1,\infty}(\mathbb{R}^+; \dot{H}^{N+1}(\mathbb{R}^d))$. We suppose that (2.19) and (2.29) are satisfied initially. We suppose also that $b_0 \in H^N(\mathbb{R}^d)$ and that $h_0 = 1 - \beta b_0 \ge h_{\min}$. Then, there exists T > 0, such that for all $(\varepsilon, \beta, \lambda, \mu) \in \mathcal{A}_{LVW}$, there exists a unique solution $U = (\zeta, \psi) \in E_T^N$ to the water waves equations with initial data U^0 . Furthermore, for \overline{V} as in (2.45),

$$\left|\zeta-\zeta_{1}\right|_{L^{\infty}([0,T];H^{N-4}(\mathbb{R}^{d}))}+\left|\overline{V}-\overline{V}_{1}\right|_{L^{\infty}([0,T];H^{N-4}(\mathbb{R}^{d}))}\leq T\delta_{0}\widetilde{C},$$

where

$$\widetilde{C} = C \bigg(\mathcal{E}^N \left(U^0 \right), \frac{1}{h_{\min}}, \frac{1}{\mathfrak{a}_{\min}}, |b|_{W_t^{3,\infty} H_X^N}, |\nabla P|_{W_t^{1,\infty} H_X^N} \bigg),$$

and $(\zeta_1, \overline{V}_1)$ solution of the waves equation

$$\begin{cases} \partial_t \zeta_1 + \nabla \cdot \left(h_0 \overline{V}_1\right) = \partial_t b_m, \\ \partial_t \overline{V}_1 + \nabla \zeta_1 = -\nabla P, \\ (\zeta_1)_{|t=0} = \zeta_0, \ (V_1)_{|t=0} = \overline{V} \left[\varepsilon \zeta_0, \beta b_{|t=0}\right] \left(\psi_0, (\partial_t b)_{|t=0}\right). \end{cases}$$
(2.51)

Proof. The system (2.51) is wellposed since it can be symmetrized thanks to the energy

$$\mathcal{E}(t) = \frac{1}{2} \left| \zeta_1 \right|_2^2 + \frac{1}{2} \left(h_0 \overline{V}_1, \overline{V}_1 \right).$$

For the inequality, we proceed as in Proposition 2.3.4, differentiating the energy

$$\mathcal{E}^{N}(t) = \frac{1}{2} \left| \zeta_{2} \right|_{H^{N-4}}^{2} + \frac{1}{2} \left(h_{0} \Lambda^{N-4} \overline{V}_{2}, \Lambda^{N-4} \overline{V}_{2} \right),$$

with $\zeta_2 = \zeta - \zeta_1$ and $\overline{V}_2 = \overline{V} - \overline{V}_1$. Using Gronwall's Lemma, Proposition 2.3.3 and standard controls, we get result.

This model is well-known in the physics literature to investigate the landslide tsunami phenomenon (see [141]).

2.3.2.2 Amplification in shallow waters when β is large

In this part, d = 1 and we suppose that P = 0. The same study can be done for a non constant pressure. For the sake of simplicity, we assume also that initially the velocity of the landslide is zero and hence that $(\partial_t b_m)_{|t=0} = 0$ (the bottom does not move at the beginning). We transform the system (2.51) in order to get an equation for ζ_1 only. We obtain that ζ_1 satisfies

$$\partial_t^2 \zeta_1 - \partial_X \left(h_0 \partial_X \zeta_1 \right) = \partial_t^2 b_m, \tag{2.52}$$

with $(\zeta_1)_{|t=0} = 0$ and $(\partial_t \zeta_1)_{|t=0} = 0$. Notice that this equation can not be transformed into a system of two transport equation with opposite speeds. Hence, there are no analytical solutions. We wonder now if we can catch an elevation of the sea level with this asymptotic model. Therefore, we are looking for solutions of the form

$$\zeta_2(t, X) = t\zeta_3(t, X). \tag{2.53}$$

The following proposition gives example of such solutions for bounded moving bottoms (with finite energy).

Proposition 2.3.7. Suppose that $h_0 \ge h_{\min} > 0$ with $h_0 \in H^1(\mathbb{R})$. Let $(\zeta_3, \overline{V}_3)$ be a solution of

$$\begin{cases} \partial_t \zeta_3 + \partial_X \left(h_0 \overline{V}_3 \right) = 0\\ \partial_t \overline{V}_3 + \partial_X \zeta_3 = 0, \end{cases}$$

with $(\zeta_3, \overline{V}_3)_{|t=0} = (0, f')$ with $f \in H^1(\mathbb{R})$. Then, $\zeta_1(t, X) = t\zeta_3(t, X)$ is a non trivial solution of (2.52) with

$$b_m(t,X) = 2\int_0^t \zeta_3(s,X)ds,$$
(2.54)

and $b_m(t, \cdot)$ is bounded in $L^2(\mathbb{R}^d)$ and in $L^{\infty}(\mathbb{R}^d)$ uniformly with respect to t

$$|b_m(t,\cdot)|_2 + |b_m(t,\cdot)|_{L^{\infty}} \le C,$$

where C is independent on t.

Proof. Plugging the expression of ζ and b_m in (2.52), we get the first result. We have to show that $\zeta_3 \in L^1(\mathbb{R}^+; L^2(\mathbb{R}^d))$. Consider the linear hyperbolic equation

$$\begin{cases} \partial_t \eta + \partial_X \left(h_0 W \right) = 0, \\ \partial_t W + \partial_X \eta = 0, \end{cases}$$

with $(\eta, W)_{|t=0} = (-f, 0)$. This system has a unique solution $(\eta, W) \in \mathcal{C}^0(\mathbb{R}; H^1(\mathbb{R}))$. Furthermore, $(\partial_t \eta, \partial_t W) \in \mathcal{C}^0(\mathbb{R}; L^2(\mathbb{R}))$, and $(\partial_t \eta, \partial_t W)$ satisfies the same linear hyperbolic system as $(\zeta_3, \overline{V}_3)$. By uniqueness, $\zeta_3 = \partial_t \eta$ and

$$b_m(t,X) = 2\eta(t,X) + 2f(X).$$

Since, for all t,

$$\int_{\mathbb{R}} \eta(t,X)^2 + h_0(X)W(t,X)^2 dX = \int_{\mathbb{R}} f(X)^2 dX$$

and $h_0 \ge h_{\min} > 0$, we get the control of $|b_m(t, \cdot)|_2$. Finally, η satisfies the waves equation

$$\partial_t^2 \eta - \partial_X \left(h_0 \partial_X \eta \right) = 0,$$

with $(\eta, \partial_t \eta)_{|t=0} = (-f, 0) \in H^1(\mathbb{R}^d)$. Then, for all t,

$$\int_{\mathbb{R}} \left| \partial_t \eta(t, X) \right|^2 + h_0(X) \left| \partial_X \eta(t, X) \right|^2 dX = \int_{\mathbb{R}} h_0(X) f'(X)^2 dX.$$

Therefore, $|\eta|_{H^1}$ (and $|\eta|_{L^{\infty}}$ by Sobolev embedding) is controlled uniformly with respect to t. \Box

In the following, we compute numerically some solutions of Equations (2.52) of the form (2.53) with a finite difference method. We take $b_0(X) = -\tanh(X)$, $\beta = \frac{1}{2}$ and $(\partial_t \zeta_3)_{|t=0} = (4X^2 - 2)e^{-X^2}$. The figure 2.3 is the evolution of the maximum of ζ_1 . The figure 2.4 is the graph at different times of the waves and the landslide. The dashed curves are the landslide, the solid curves are the waves and the dotted curve is the slope. Therefore, we see that an important elevation of the sea level is possible even if we do not consider that the seabed is flat. This is what happened during the meteotsunami in Nagasaki bay in 1979. The shelf in the East China sea gradually decreases from 20 meters to 200 meters over 500 kilometers (see [67], [144]) and hence is favourable for an amplification.

Remark 2.3.8. In order to simplify, we consider that the system is initially at rest. But our study can easily be extended to waves with non trivial initial data. In particular, we can study a wave amplified by a landslide. This is what seemed to happen during the 2011 Tohoku tsunami, responsible of the Fukushima nuclear disaster. Indeed, no models and numerical simulations validated the run-up heights of up to 40 meters measured along the coast of the north east part of Honshu Island. Hence, in [136], they proposed that a landslide amplified the tsunami wave



Figure 2.3: Evolution of the maximum of ζ_1 , solution of (2.52), for a non flat bottom b_0 .



Figure 2.4: Evolution of the surface ζ_1 (solid line), solution of (2.52), and the landslide b_m (dashed line).



Figure 2.5: Evolution of the maximum of h, solution of (2.52), with non trivial initial data and with b_m like in Figure 2.4.

and they validated this assumption thanks to numerical simulations and physical considerations. We compute numerically this amplification. We consider a wave moving with a speed equal to 1 (typical speed in the sea after nondimensionalization) that is amplified by a landslide. Figure 2.5 represents the evolution of the maximum of this wave. We can see that an amplification occurs relatively quickly.

Remark 2.3.9. We saw in Subsection 2.3.1 that the main mathematical argument that explains the Proudman resonance is the non L^{∞} -dispersion of the 1D wave equation. It allows us to show that a resonance could occur with a factor of amplification of t (linear amplification). When we add a bottom, our model is a variable coefficient 1D wave equation. In our knowledge, there are no mathematical results about non L^{∞} -dispersion for the variable 1D wave equation.

We wonder now if a landslide with a speed equal to 1 can also create a resonance. The following results show that it is true and that the factor of amplification is of order t.

Lemma 2.3.10. Let $f \in H^3(\mathbb{R}^d)$ such that, there exists $\alpha > 1$, for $k \in \{0, 1, 2, 3\}$,

$$\left|f^{(k)}(X)\right| \leq \frac{C}{(1+|X|)^{\alpha}}$$

Let $h_0 \in W^{1,\infty}(\mathbb{R})$, such that $h_0 \ge h_{\min} > 0$ and

$$|1 - h_0(X)| \le \frac{C}{(1 + |X|)^{\alpha + 1}}$$
 and $|h'_0(X)| \le \frac{C}{(1 + |X|)^{\alpha + 1}}$

Then, if we denote $\zeta(t, X) = tf'(X - t) - f(X - t)$ and v(t, X) = tf'(X - t), (ζ, v) is solution of

$$\begin{cases} \partial_t \zeta + \partial_X \left(h_0 v \right) = g + 2f'(X - t), \\ \partial_t v + \partial_X \zeta = 0, \end{cases}$$

where g satisfies

$$|g(t,\cdot)|_{L^2_X} \le \frac{C}{(1+|t|)^{\alpha-1}} \text{ and } |\partial_t g(t,\cdot)|_{L^2_X} \le \frac{C}{(1+|t|)^{\alpha-1}}.$$

Proof. we compute g

$$g(t,X) = t\left((h_0(X) - 1)f''(X - t) + h'_0(X)f'(X - t)\right).$$

The result follows from Peetre's inequality.

We also have an integrability result for the variable wave equation.

Lemma 2.3.11. Let $h_0 \in W^{1,\infty}(\mathbb{R})$, such that $h_0 \ge h_{\min} > 0$ and let $u = (\zeta, v)$ be a solution of

$$\begin{cases} \partial_t \zeta + \partial_X \left(h_0 v \right) = g, \\ \partial_t v + \partial_X \zeta = 0. \end{cases}$$

with initial data equal to 0 and where $g, \partial_t g \in L^1_t(\mathbb{R}^+; L^2_X(\mathbb{R}))$ and $g \in L^\infty_t(\mathbb{R}^+; L^2_X(\mathbb{R}))$. Then, $u \in L^\infty_{(t,X)}(\mathbb{R}^2)$.

Proof. We denote $E(t) = (\zeta, \zeta)_2 + (h_0 v, v)_2 + (\partial_t \zeta, \partial_t \zeta)_2 + (h_0 \partial_t v, \partial_t v)_2$. Using the equation we get

$$\frac{d}{dt}E(t) \le C. \left(|g(t, \cdot)|_{L^2_X} + |\partial_t g(t, \cdot)|_{L^2_X} \right) \sqrt{E(t)}.$$

Since $|g(t, \cdot)|_{L^2_X}$ and $|\partial_t g(t, \cdot)|_{L^2_X}$ are integrable, the energy E is bounded. Finally, using the equations, we can control the H^1 -norm of u and we get the result.

Gathering these two results we obtain the following proposition.

Proposition 2.3.12. Let $(\zeta_0, v_0) \in H^1(\mathbb{R})$, $f \in H^3(\mathbb{R}^d)$ such that, there exists $\alpha > 2$, for $k \in \{0, 1, 2, 3\}$,

$$\left|f^{(k)}(X)\right| \le \frac{C}{(1+|X|)^{\alpha}}$$

Let $h_0 \in W^{1,\infty}(\mathbb{R})$, such that $h_0 \ge h_{\min} > 0$ and

$$|1 - h_0(X)| \le \frac{C}{(1 + |X|)^{\alpha + 1}}$$
 and $|h'_0(X)| \le \frac{C}{(1 + |X|)^{\alpha + 1}}$.

Consider the solution (ζ, v) of

$$\begin{cases} \partial_t \zeta + \partial_X (h_0 v) = 2f'(X - t), \\ \partial_t v + \partial_X \zeta = 0, \\ (\zeta, v)_{|t=0} = (\zeta_0, v_0). \end{cases}$$

Then, there exists a constant C > 0, such that $|(\zeta, v)(t, \cdot)|_{L^{\infty}_{v}} \ge C(t-1)$.

Hence, for slopes which converge quickly enough to a flat bottom, we get a resonance for a landslide with a speed equal to 1. This fact was conjectured by physicists thanks to various numerical simulations ([57]).

Remark 2.3.13. We saw that we can not expect analytical solution for Equation (2.52). In order to get some, physicists commonly assume that the slope varies slowly. They consider a slope with h_0 of the form $h_0(x) = h_0(\alpha x)$ where $\alpha\beta$ is small. By neglecting all the terms of order $\mathcal{O}(\alpha\beta)$, we obtain

$$\partial_t^2 \zeta - (1 - h_0) \partial_X^2 \zeta = \partial_t^2 b_m,$$

which can be viewed of the composition of two transport equations with opposite speeds thanks to an appropriate change of variables. Then, it was shown in this setting that a resonance is possible for landslides which adapt their speeds with the slope (see for instance [54] and [55]).

2.3.3 Linear asymptotic and resonance in intermediate depths

In this case, we consider only that ε , β are small. Physically, this means that we consider small amplitudes for the surface and the bottom (compared to the mean depth) and that the depth is comparable to wavelength of the waves. In this part, we generalize the Proudman resonance in deeper waters. The asymptotic regime is

$$\mathcal{A}_{LWW} = \{(\varepsilon, \beta, \lambda, \mu), 0 < \varepsilon, \beta \le \delta_0, \beta \lambda = \varepsilon \text{ and } 0 < \mu \le \mu_{\max}\}, \qquad (2.55)$$

with $\delta_0 \ll 1$ and $0 < \mu_{\text{max}}$. Using the energy

$$\mathcal{E}(t) = \frac{1}{2} |\zeta|_2^2 + \frac{1}{2} \left(\frac{1}{\mu} G_{\mu}[0,0](\psi), \psi \right),$$

and proceeding as in Proposition 2.3.4 (we need also Proposition A.2.4), we get a new asymptotic model.

Proposition 2.3.14. Let $t_0 > \frac{d}{2}$, $N \ge \max(1, t_0) + 3$, $b \in W^{3,\infty}(\mathbb{R}^+; H^{N+1}(\mathbb{R}^d))$, $U^0 = (\zeta_0, \psi_0) \in E_0^N$ and $P \in W^{1,\infty}(\mathbb{R}^+; \dot{H}^{N+1}(\mathbb{R}^d))$. We suppose that (2.19) and (2.29) are satisfied initially. Then, there exists T > 0, such that for all $(\varepsilon, \beta, \lambda, \mu) \in \mathcal{A}_{LWW}$, there exists a unique solution $U = (\zeta, \psi) \in E_{\frac{T}{\sqrt{\delta_0}}}^N$ to the water waves equations with initial data U^0 . Furthermore, for all $\alpha \in [0, \frac{1}{3})$,

$$\left|\zeta - \widetilde{\zeta}\right|_{L^{\infty}\left(\left[0, \frac{T}{\delta_{0}^{\alpha}}\right]; H^{N-2}(\mathbb{R}^{d})\right)} + \left|\frac{|D|}{\sqrt{1+|D|}} \left(\psi - \widetilde{\psi}\right)\right|_{L^{\infty}\left(\left[0, \frac{T}{\delta_{0}^{\alpha}}\right]; H^{N-2}(\mathbb{R}^{d})\right)} \le T\delta_{0}^{1-3\alpha}\widetilde{C},$$

where

$$\widetilde{C} = C\left(\mathcal{E}^{N}\left(U^{0}\right), \frac{1}{h_{\min}}, \frac{1}{\mathfrak{a}_{\min}}, \mu_{\max}, |b|_{W^{3,\infty}_{t}H^{N}_{X}}, |\nabla P|_{W^{1,\infty}_{t}H^{N}_{X}}\right),$$

where $\left(\widetilde{\zeta},\widetilde{\psi}\right)$ is a solution of the waves equation

$$\begin{cases} \partial_t \widetilde{\zeta} - \frac{1}{\mu} G_\mu[0,0](\widetilde{\psi}) = G_\mu^{NN}[0,0](\partial_t b), \\ \partial_t \widetilde{\psi} + \widetilde{\zeta} = -P, \end{cases}$$
(2.56)

with initial data U^0 .

The Proudman resonance is a phenomenon which occurs in shallow water regime. We wonder if there is also a resonance in deeper waters. In this part, we only work with a non constant pressure and hence $\partial_t b = 0$. The same study can be done for a moving bottom. We consider the equation (2.56) for d = 1. Since, the initial data does not affect the possible resonance, we suppose in the following that $U^0 = 0$. We transform the system (2.56) in order to have a unique equation for $\tilde{\zeta}$ (in the following we denote $\tilde{\zeta}$ by ζ to simplify the notation)

$$\begin{cases} \partial_t^2 \zeta + \frac{1}{\mu} G_{\mu}[0,0](\zeta) = -\frac{1}{\mu} G_{\mu}[0,0](P), \\ \zeta_{|t=0} = 0, \ \partial_t \zeta_{|t=0} = 0. \end{cases}$$

We can solve explicitly the previous equation, we get that

$$\begin{split} \widehat{\zeta}(t,\xi) &= \underbrace{\frac{i}{2} \int_{0}^{t} \xi \sqrt{\frac{\tanh(\sqrt{\mu}|\xi|)}{\sqrt{\mu}|\xi|}} \widehat{P}(\tau,\xi) e^{i(t-\tau)\xi \sqrt{\frac{\tanh(\sqrt{\mu}|\xi|)}{\sqrt{\mu}|\xi|}} d\tau}_{:=\widehat{\zeta}_{L}(t,\xi)} \\ &- \underbrace{\frac{i}{2} \int_{0}^{t} \xi \sqrt{\frac{\tanh(\sqrt{\mu}|\xi|)}{\sqrt{\mu}|\xi|}} \widehat{P}(\tau,\xi) e^{i(\tau-t)\xi \sqrt{\frac{\tanh(\sqrt{\mu}|\xi|)}{\sqrt{\mu}|\xi|}} d\tau}_{:=\widehat{\zeta}_{R}(t,\xi)}. \end{split}$$

In order to find a resonant pressure, we suppose that P has the form $e^{-ita(D)}P_0$, where a is a real smooth odd function which is sublinear, there exists C > 0 such that $|a(\xi)| \leq C|\xi|$. We also suppose that the phase velocity of the disturbance is positive, $\frac{a(\xi)}{\xi} \geq 0$. P_0 is a smooth function in a Sobolev space. We denote $\omega(\xi) = \sqrt{\frac{\tanh(\xi)}{\xi}}$. A simple computation gives that

$$|\zeta_L(t,\cdot)| \le |\widehat{\zeta_L}(t,\cdot)|_{L^1} \le \left|\widehat{P_0}\right|_{L^1}.$$

Furthermore, we have

$$\begin{split} |\widehat{\zeta_R}(t,\xi)| &= \frac{1}{2} \left| \int_0^t \xi \omega(\sqrt{\mu}\xi) \widehat{P}_0(\xi) e^{i\tau \left(\xi \omega(\sqrt{\mu}\xi) - a(\xi)\right)} d\tau \right| \\ &\leq \frac{t}{2} \left| \xi \omega(\sqrt{\mu}\xi) \widehat{P}_0(\xi) \right|, \end{split}$$



Figure 2.6: Profile of ϕ'' .

with an equality if and only if $a(\xi) = \xi \omega(\sqrt{\mu}\xi)$. Hence, it is natural to consider that

$$\widehat{P}(t,\xi) = e^{-it\xi\omega(\sqrt{\mu}\xi)}P_0(\xi).$$
(2.57)

A simple computation gives

$$\zeta_R(t,X) = -\frac{it}{2} \int_{\mathbb{R}} \xi \omega(\sqrt{\mu}\xi) \widehat{P_0}(\xi) e^{-it\xi\omega(\sqrt{\mu}\xi)} e^{iX\xi} d\xi.$$
(2.58)

We wonder now if a resonance occurs. We need a dispersion estimate for the linear water waves equation.

Proposition 2.3.15. Let $f \in W^{1,1}(\mathbb{R})$ such that $\widehat{f}(0) = 0$. Then,

$$\left|\int_{\mathbb{R}} e^{-it\xi\omega(\sqrt{\mu}\xi)} e^{iX\xi} \widehat{f}(\xi) d\xi\right| \leq \frac{C}{\sqrt{t}} \left(\frac{1}{\sqrt{\mu}} \left|\frac{1}{\sqrt{|\xi|}} \left(\widehat{f}\right)'\right|_{L^{1}(\mathbb{R})} + \mu^{\frac{1}{8}} \left||\xi|^{\frac{3}{4}} \left(\widehat{f}\right)'\right|_{L^{1}(\mathbb{R})}\right).$$

Remark 2.3.16. In [106], Mésognon-Gireau improved this result by only assuming f and xf in $H^1(\mathbb{R})$ (no condition on $\widehat{f}(0)$) and he got a dispersion of order $\frac{1}{t^{\frac{1}{3}}}$. Noticed that the result of Mésognon-Gireau is more convenient if one wants to apply this dispersion property to the water waves equations (2.13) (see [107] for instance). We also refer to [31] for the case of an infinite depth.

Proof. We denote I(t),

$$\begin{split} I(t) &:= \int_{\mathbb{R}} e^{-it\xi\omega(\sqrt{\mu}\xi)} e^{iX\xi} \widehat{f}(\xi) d\xi \\ &= \frac{1}{\sqrt{\mu}} \int_{\mathbb{R}} e^{-i\frac{t}{\sqrt{\mu}} \left(y\omega(y) - \frac{X}{t}y\right)} \widehat{f}\left(\frac{y}{\sqrt{\mu}}\right) dy. \end{split}$$

We denote ϕ ,

$$\phi(y) = y\omega(y) - \frac{X}{t}y,$$

and y_0 the unique minimum of ϕ'' . Figure 2.6 represents ϕ'' on $[0, +\infty[$. To estimate I(t) we decompose I(t) into four parts.

$$\begin{split} I_1(t) &= \frac{1}{\sqrt{\mu}} \int_0^{y_0} e^{-i\frac{t}{\sqrt{\mu}}\phi(y)} \widehat{f}\left(\frac{y}{\sqrt{\mu}}\right) dy \\ &= \frac{1}{\sqrt{\mu}} \int_0^{y_0} -\frac{d}{dy} \left(\int_y^{y_0} e^{-i\frac{t}{\sqrt{\mu}}\phi(z)} dz\right) \widehat{f}\left(\frac{y}{\sqrt{\mu}}\right) dy \\ &= \frac{1}{\mu} \int_0^{y_0} \int_y^{y_0} e^{-i\frac{t}{\sqrt{\mu}}\phi(z)} dz \left(\widehat{f}\right)' \left(\frac{y}{\sqrt{\mu}}\right) dy. \end{split}$$

Then, using Van der Corput's Lemma (see [133]) and the fact that for $z \in [y, y_0]$, $|\phi''(z)| \ge |\phi''(y)|$ and $|\phi''(z)| \ge Cz$,

$$|I_1(t)| \le \frac{C}{\mu^{\frac{3}{4}}\sqrt{t}} \int_0^{y_0} \left| \frac{1}{\sqrt{y}} \left(\widehat{f} \right)' \left(\frac{y}{\sqrt{\mu}} \right) \right| dy$$
$$\le \frac{C}{\sqrt{\mu}\sqrt{t}} \int_0^{+\infty} \left| \frac{1}{\sqrt{\xi}} \left(\widehat{f} \right)' (\xi) \right| d\xi.$$

Furthermore, for $M > y_0$ large enough,

$$\begin{split} I_2(t) &= \frac{1}{\sqrt{\mu}} \int_{y_0}^M e^{-i\frac{t}{\sqrt{\mu}}\phi(y)} \widehat{f}\left(\frac{y}{\sqrt{\mu}}\right) dy \\ &= \frac{1}{\sqrt{\mu}} \int_{y_0}^M \frac{d}{dy} \left(\int_{y_0}^y e^{-i\frac{t}{\sqrt{\mu}}\phi(z)} dz\right) \widehat{f}\left(\frac{y}{\sqrt{\mu}}\right) dy \\ &= \int_{y_0}^M e^{-i\frac{t}{\sqrt{\mu}}\phi(z)} \frac{dz}{\sqrt{\mu}} \widehat{f}\left(\frac{M}{\sqrt{\mu}}\right) - \frac{1}{\mu} \int_{y_0}^M \int_{y_0}^y e^{-i\frac{t}{\sqrt{\mu}}\phi(z)} dz \left(\widehat{f}\right)' \left(\frac{y}{\sqrt{\mu}}\right) dy. \end{split}$$

Then, using Van der Corput's Lemma and the fact that for $z \in [y_0, y]$, $|\phi''(z)| \ge |\phi''(y)|$ and $|\phi''(z)| \ge Cz^{-\frac{3}{2}}$,

$$\begin{aligned} |I_{2}(t)| &\leq \left|\frac{M}{\sqrt{\mu}}\widehat{f}\left(\frac{M}{\sqrt{\mu}}\right)\right| + \frac{C}{\mu^{\frac{3}{4}}\sqrt{t}}\int_{y_{0}}^{M}\left|y^{\frac{3}{4}}\left(\widehat{f}\right)'\left(\frac{y}{\sqrt{\mu}}\right)\right|dy\\ &\leq \left|\widehat{f}'\left(\frac{M}{\sqrt{\mu}}\right)\right| + \frac{C\mu^{\frac{1}{8}}}{\sqrt{t}}\int_{0}^{+\infty}\left|\xi^{\frac{3}{4}}\left(\widehat{f}\right)'(\xi)\right|.\end{aligned}$$

Tending M to $+\infty$ we get the result. The control for $\xi < 0$ is similar.

Therefore, in the linear case, we have also a resonance.

Corollary 2.3.17. Let $P_0 \in H^3(\mathbb{R}) \cap W^{2,1}(\mathbb{R})$ such that $XP_0 \in H^3(\mathbb{R})$ and let $0 < \mu \leq \mu_{\max}$. We consider

$$\zeta_R(t,X) = -\frac{it}{2} \int_{\mathbb{R}} \xi \omega(\sqrt{\mu}\xi) \widehat{P}_0(\xi) e^{-it\xi\omega(\sqrt{\mu}\xi)} e^{iX\xi} d\xi.$$

Then,

$$|\zeta_R(t,\cdot)|_{\infty} \le C(\mu_{\max})\sqrt{\frac{t}{\mu}} (|P_0|_{H^3} + |P_0|_{L^1} + |XP_0|_{H^3}),$$

and

$$\lim_{t \to +\infty} \left| \frac{1}{\sqrt{t}} \zeta_R(t, \cdot) \right|_{\infty} \ge C(P_0) > 0.$$

Proof. We take $\widehat{f}(\xi) = \xi \omega(\sqrt{\mu}\xi) \widehat{P}_0(\xi)$. Then,

$$\left| \left(\widehat{f} \right)'_{(\xi)} \right| \le (1 + \sqrt{\mu} |\xi|) \left| \widehat{P}_0(\xi) \right| + |\xi| \left| \left(\widehat{P}_0 \right)'_{(\xi)} \right|,$$

and the first inequality follows from the previous Proposition. For the second inequality, we use a stationary phase approximation. We denote $\phi(\xi) = \xi \omega(\xi)$. Let $\xi_0 > 0$, such that $\left|\xi_0 \widehat{P_0}(\xi_0)\right| = \left|\widehat{\xi P_0}\right|_{L^{\infty}}$, and $X_{\mu} < 0$, such that $\phi'(\sqrt{\mu}\xi_0) = X_{\mu}$. Then, we have,

$$\lim_{t \to +\infty} \left| \frac{1}{\sqrt{t}} \zeta_R(t, tX_\mu) \right| = \lim_{t \to +\infty} \frac{\sqrt{t}}{2\mu} \left| \int_{\mathbb{R}} \xi \omega(\xi) \widehat{P_0}\left(\frac{\xi}{\sqrt{\mu}}\right) e^{-i\frac{t}{\sqrt{\mu}}\xi(\omega(\xi) - X_\mu)} d\xi \right|$$
$$= \frac{\sqrt{2\pi}}{2\mu^{\frac{1}{4}}} \left| \frac{\omega(\xi_0 \sqrt{\mu})\xi_0 \widehat{P_0}(\xi_0)}{\sqrt{|\phi''(\xi_0 \sqrt{\mu})|}} \right|.$$

Since $|\phi''(\xi)| \leq C|\xi|$ and $\omega(\xi_0\sqrt{\mu}) \geq C(\xi_0)\sqrt{\mu}$, we get the result.

Remark 2.3.18. *Notice that for all* $s \in \mathbb{R}$ *,*

$$\left| \zeta_R(t, \cdot) + \frac{t}{2} P'_0(\cdot - t) \right|_{H^s} \le \sqrt{\mu} t^2 \, |\nabla P_0|_{H^{s+2}} \, .$$

Hence, by tending formally μ to 0, we rediscover the result we get in the shallow water case (section 2.3.1).

Remark 2.3.19. Notice that for a general pressure term P(t, X) we can show that the amplitude ζ satisfying

$$\begin{split} \widehat{\zeta}(t,\xi) &= \frac{i}{2} \int_0^t \xi \sqrt{\frac{\tanh(\sqrt{\mu}|\xi|)}{\sqrt{\mu}|\xi|}} \widehat{P}(\tau,\xi) e^{i(t-\tau)\xi \sqrt{\frac{\tanh(\sqrt{\mu}|\xi|)}{\sqrt{\mu}|\xi|}}} d\tau \\ &- \frac{i}{2} \int_0^t \xi \sqrt{\frac{\tanh(\sqrt{\mu}|\xi|)}{\sqrt{\mu}|\xi|}} \widehat{P}(\tau,\xi) e^{i(\tau-t)\xi \sqrt{\frac{\tanh(\sqrt{\mu}|\xi|)}{\sqrt{\mu}|\xi|}}} d\tau, \end{split}$$

 $satisfies \ also$

$$|\zeta(t,\cdot)|_{\infty} \leq C(\mu_{\max})\sqrt{\frac{t}{\mu}}\left(|P|_{L^{\infty}(\mathbb{R}^+;L^1(\mathbb{R}^d))} + |P|_{L^{\infty}(\mathbb{R}^+;H^3(\mathbb{R}^d)} + |XP|_{L^{\infty}(\mathbb{R}^+;H^3(\mathbb{R}^d))}\right).$$

Hence, contrary to the shallow water case, we can not hope a linear amplification with respect to the time t. Corollary 2.3.17 also shows that the factor of amplification of \sqrt{t} is optimal.

Hence, we observe that in intermediate water depths, a resonance can occur but with a factor of amplification of \sqrt{t} and not t. But we saw that in the shallow water case, the resonance occurs for a moving pressure with a speed equal to 1, $P(t, X) = P_0(X - t)$. We wonder if this pressure can create a resonance. The following Proposition shows that the previous pressure can create a resonance with a factor of amplification of $t^{\frac{1}{3}}$.

Proposition 2.3.20. Let $0 < \mu \leq \mu_{\max}$. Let $P_0 \in L^1(\mathbb{R}) \cap H^1(\mathbb{R})$ such that $\widehat{P_0}(0) \neq 0$. We consider the amplitude ζ_R created by $P(t, X) = P_0(X - t)$

$$\widehat{\zeta_R}(t,\xi) = -\frac{i}{2}\xi\omega(\sqrt{\mu}\xi)\widehat{P_0}(\xi)e^{-it\xi}\int_{-t}^0 e^{is\xi\left(\omega(\sqrt{\mu}\xi)-1\right)}ds.$$
(2.59)

Then,

$$|\zeta_R(t,\cdot)|_{\infty} \le C(\mu_{\max}) \left(\frac{t^{\frac{1}{3}}}{\mu} |P_0|_{L^1} + \mu^{\frac{1}{4}} |P_0|_{H^1} \right).$$

Furthermore,

$$\lim_{t \to +\infty} \left| \frac{1}{t^{\frac{1}{3}}} \zeta_R(t, \cdot) \right|_{\infty} \ge \frac{C}{\mu^{\frac{2}{3}}} \left| \widehat{P}_0(0) \right|.$$

Proof. We have

$$\begin{aligned} \zeta_R(t,X) &= -\frac{i}{2} \int_{\mathbb{R}} \xi \omega(\sqrt{\mu}\xi) \widehat{P}_0(\xi) e^{-it\xi} \int_{-t}^0 e^{is\xi \left(\omega(\sqrt{\mu}\xi) - 1\right)} e^{iX\xi} ds d\xi \\ &= -\frac{i}{2} \frac{1}{\mu} \int_{\mathbb{R}} \xi \omega(\xi) \widehat{P}_0\left(\frac{\xi}{\sqrt{\mu}}\right) e^{-i\frac{t}{\sqrt{\mu}}\xi} \int_{-t}^0 e^{i\frac{s}{\sqrt{\mu}}\xi(\omega(\xi) - 1)} e^{i\frac{X}{\sqrt{\mu}}\xi} ds d\xi \end{aligned}$$

We decompose this integral into 3 parts.

$$\begin{split} |I_1(t)| &= \left| \frac{1}{\mu} \int_{|\xi| \le t^{-\frac{1}{3}}} \xi \omega(\xi) \widehat{P_0}\left(\frac{\xi}{\sqrt{\mu}}\right) e^{-i\frac{t}{\sqrt{\mu}}\xi} \int_{-t}^0 e^{i\frac{s}{\sqrt{\mu}}\xi(\omega(\xi)-1)} e^{i\frac{X}{\sqrt{\mu}}\xi} d\xi ds \right| \\ &\leq \frac{t^{\frac{1}{3}}}{\mu} \left| \widehat{P_0} \right|_{\infty}. \end{split}$$

Furthermore, since $|\omega(\xi) - 1| \ge C\xi^2$ for $0 \le |\xi| \le 1$, we have

$$\begin{split} |I_2(t)| &= \left| \frac{1}{\mu} \int_{t^{-\frac{1}{3}} \le |\xi| \le 1} \xi \omega(\xi) \widehat{P_0} \left(\frac{\xi}{\sqrt{\mu}} \right) e^{-i\frac{t}{\sqrt{\mu}}\xi} \int_{-t}^{0} e^{i\frac{s}{\sqrt{\mu}}\xi(\omega(\xi)-1)} e^{i\frac{X}{\sqrt{\mu}}\xi} d\xi ds \right| \\ &= \left| \frac{1}{\sqrt{\mu}} \int_{t^{-\frac{1}{3}} \le |\xi| \le 1} e^{i\frac{X}{\sqrt{\mu}}\xi} \frac{\omega(\xi)}{\omega(\xi)-1} \widehat{P_0} \left(\frac{\xi}{\sqrt{\mu}} \right) \left(e^{-i\frac{t}{\sqrt{\mu}}\xi} - e^{-i\frac{t}{\sqrt{\mu}}\xi\omega(\xi)} \right) d\xi \right| \\ &\leq C \frac{t^{\frac{1}{3}}}{\sqrt{\mu}} \left| \widehat{P_0} \right|_{\infty}. \end{split}$$

Finally,

$$\begin{split} |I_{3}(t)| &= \left| \frac{1}{\mu} \int_{|\xi| \ge 1} \xi \omega(\xi) \widehat{P_{0}} \left(\frac{\xi}{\sqrt{\mu}} \right) e^{-i\frac{t}{\sqrt{\mu}}\xi} \int_{-t}^{0} e^{i\frac{s}{\sqrt{\mu}}\xi \left(\omega(\sqrt{\mu}\xi) - 1 \right)} e^{i\frac{x}{\sqrt{\mu}}\xi} d\xi ds \right| \\ &= \left| \frac{1}{\sqrt{\mu}} \int_{|\xi| \ge 1} e^{i\frac{x}{\sqrt{\mu}}\xi} \frac{\omega(\xi)}{\omega(\xi) - 1} \widehat{P_{0}} \left(\frac{\xi}{\sqrt{\mu}} \right) \left(e^{-i\frac{t}{\sqrt{\mu}}\xi} - e^{-i\frac{t}{\sqrt{\mu}}\xi\omega(\xi)} \right) d\xi \right| \\ &\leq C \int_{|\xi| \ge \frac{1}{\sqrt{\mu}}} \left| \widehat{P_{0}}(\xi) \right| d\xi, \\ &\leq C \mu^{\frac{1}{4}} \left| P_{0} \right|_{H^{1}}, \end{split}$$

and the first inequality follows. For the second inequality, we use a stationary phase approximation. We denote $\phi(\xi) := \xi(\omega(\xi) - 1)$. We recall that $\phi(\xi) = -\frac{1}{6}\xi^3 + o(\xi^3)$. Using a generalization of Morse Lemma at the order 3, there exists a > 0 and $\psi \in \mathcal{C}^{\infty}([-a, a])$, such that for all $|y| \leq a$,

$$\phi(\psi(y)) = \frac{1}{6}\phi'''(0)y^3, \psi(0) = 0 \text{ and } \psi'(0) = 1.$$

Then,

$$\begin{split} I(s) &:= \int_{\mathbb{R}} \omega(\xi) \xi \widehat{P_0}\left(\frac{\xi}{\sqrt{\mu}}\right) e^{i\frac{s}{\sqrt{\mu}}\xi(\omega(\xi)-1)} d\xi \\ &= \int_{-a}^{a} \psi'(y) \omega(\psi(y)) \psi(y) \widehat{P_0}\left(\frac{\psi(y)}{\sqrt{\mu}}\right) e^{i\frac{s}{6\sqrt{\mu}}y^3} dy + o(s^{-\frac{2}{3}}) \\ &= \left(\frac{6\sqrt{\mu}}{s}\right)^{\frac{2}{3}} \widehat{P_0}(0) \int_{z \in \mathbb{R}} z e^{iz^3} dz + o(s^{-\frac{2}{3}}). \end{split}$$

Therefore,

$$\lim_{t \to +\infty} \left| \frac{1}{t^{\frac{1}{3}}} \zeta_R(t, t) \right| = \frac{C}{\mu^{\frac{2}{3}}} \left| \widehat{P}_0(0) \right|.$$

		_	
- 1		1	
- 1			
- 1			

Then, in intermediate water depths, a traveling pressure with a constant speed equal to 1 is also resonant. It takes more time to obtain a significant elevation of the level of the sea compare to the shallow water case. In the following, we compute numerically some solutions. We take $P_0(X) = -e^{-X^2}$ and $\mu = 1$. The figure 2.7 displays the evolution of a water wave because of a pressure of the form (2.57). The solid curve is the wave and the dashed curve is the moving pressure. The figure 2.8 displays the evolution is a water wave when the pressure moves with a speed 1. The figure 2.9 compares the evolution of the maximum of the resonant case and the case when the speed is equal to 1.

Remark 2.3.21. In our work, we neglect the Coriolis effect. However, in view of the duration of the meteotsunami phenomenon, it would be more realistic to consider it. It will be studied in Chapter 4.



Figure 2.7: Evolution of the surface elevation ζ_R in (2.58) (solid line) because of a resonant moving pressure P in (2.57) (dashed line).



Figure 2.8: Evolution of the surface elevation ζ_R in (2.59) (solid line) because of a moving pressure P with a speed of 1 (dashed line).



Figure 2.9: Evolution of the maximum of ζ_R in the resonant case (solid line) and the moving pressure with a speed of 1 (dashed line).

2.3.4 Toward nonlinear asymptotic models

Previously, we gave linear asymptotic models to understand the Proudman resonance. However, sometime, the nonlinear effects can not be neglected. We propose in this part numerical simulations to understand how the nonlinear effects perturb the resonance. The first model is the nonlinear shallow water equation and the second one is the Saut-Xu equation. Since our goal is to catch the nonlinear effects, we consider that the bottom is flat and we only work with a non constant pressure. We show in both cases that an amplification due to the source term is

possible but some shocks can occur and create a saturation of the L^{∞} -norm.

2.3.4.1 The nonlinear shallow water equation

The linear asymptotic model that we studied in Part 2.3.1 is a shallow water model and we saw that a Proudman resonance is possible for pressure with a speed equal to 1 and the factor of amplification of t. In order to study the impact of nonlinear effects, we can use the so-called nonlinear shallow water equations

$$\begin{cases} \partial_t \zeta + \partial_x ((1 + \varepsilon \zeta)v) = 0\\ \partial_t v + \partial_x \zeta + \varepsilon v \partial_x v = -P. \end{cases}$$
(2.60)

To simulate numerically this system, we use a well balanced finite volume scheme with a rusanov flux. We refer to [11] and [23] for more details about this scheme. We take P of the form $P(t,x) = P_0(X - t)$ with $P_0(X) = \exp(-\frac{1}{3}X^2)$ and we take $\varepsilon = 1$. Figure 2.10 shows the evolution of the surface in the linear and nonlinear case. Figure 2.11 compares the maximum of the two cases. In Figure 2.10, we see that in the nonlinear case, the water wave is about to break. It creates a saturation of the L^{∞} -norm. This effect is well-known for water waves without a forcing term (see for instance the numerical simulations in [23]).

In [144], Vilibic also studied the Proudman resonance thanks to the nonlinear shallow water equations. He shown that an amplification due to the source term of the amplitude of the surface is possible (see Figure 2 and figure 5 in [144]).

Remark 2.3.22. In [118] (see also [63]), Pelinovsky et al. use the forced KdV equation to model water waves generated by atmospheric disturbances moving with long wave speeds. It could be interesting to study the Proudman resonance in this framework.

2.3.4.2 The Saut-Xu equations

We studied in Part 2.3.3 the Proudman resonance in deep water, i.e typically for $\mu = 1$. The idea to get a weakly nonlinear model in deep water is to forget all the terms of order $\mathcal{O}(\varepsilon^2)$ in the water waves equations (2.13). Then, we get the following system

$$\begin{cases} \partial_t \zeta - \mathcal{H}v + \varepsilon \left(\mathcal{H} \left(\zeta \partial_x \mathcal{H}v \right) + \partial_x \left(\zeta v \right) \right) = 0\\ \partial_t v + \partial_x \zeta + \varepsilon v \partial_x v - \varepsilon \sqrt{\mu} \partial_x \zeta \mathcal{H} \partial_x \zeta = -P, \end{cases}$$
(2.61)

where $\zeta = \zeta(t, x)$ is the free surface, v = v(t, x) is the horizontal velocity at the surface and \mathcal{H} is the Fourier multiplier, $\mathcal{H} = -\frac{\tanh(D)}{D}\partial_x$. This model was derived by Matsuno ([97]) and is now called the Matsuno equations. It is important to notice that this model is only derived. In our knowledge, the wellposedness of the Matsuno equations is still an open problem (see the paper of Ambrose, Bona and Nicholls [10]). To avoid this difficulty, Saut and Xu ([125]) developed an equivalent problem to the Matsuno system which is consistent to the water waves problem and with the same accuracy ($\mathcal{O}(\varepsilon^2)$). Then, they proved that this new system is wellposed. The Saut-Xu equations are

$$\begin{cases}
\partial_t \zeta - \mathcal{H}_{\mu} v + \varepsilon \sqrt{\mu} \left(\frac{1}{2} v \partial_x \zeta + \frac{1}{2} \mathcal{H}_{\mu} \left(v \partial_x \mathcal{H}_{\mu} \zeta \right) + \mathcal{H}_{\mu} \left(\zeta \partial_x \mathcal{H}_{\mu} v \right) + \zeta \partial_x v \right) = 0, \\
\partial_t v + \partial_x \zeta + \frac{3\varepsilon \sqrt{\mu}}{2} v \partial_x v - \frac{\varepsilon \sqrt{\mu}}{2} \partial_x \zeta \mathcal{H}_{\mu} \partial_x \zeta - \frac{\varepsilon \sqrt{\mu}}{2} v \mathcal{H}_{\mu}^2 \partial_x v = -P.
\end{cases}$$
(2.62)



Figure 2.10: Evolution of the surface in the nonlinear case (blue line), the linear case (red line). The dashed in the corresponding pressure.



Figure 2.11: Comparison of the maximum of the surface between the linear case (blue line) and the nonlinear case (red line).

In Section 3.3, we introduce a splitting scheme to solve numerically the Saut-Xu system and we show that this scheme converges (Theorem 3.4.6). We also refer to Section 3.2 for more details about the Saut-Xu equations. In the following we give different numerical simulations to understand how the nonlinear effect can perturb the Proudman resonance. We recall that if the pressure P satisfies

$$\widehat{P}(t,\xi) = e^{-it\xi\sqrt{\frac{\tanh(\xi)}{\xi}}}P_0(\xi), \qquad (2.63)$$

we have a resonance with a factor of amplification of \sqrt{t} . In the first simulation, we compare the linear case and the nonlinear case. We take P as in (2.63) with $P_0(X) = \exp(-\frac{1}{3}X^2)$ and we take $\varepsilon = 0.1$. Figure 2.12 displays the evolution of the surface in the linear and nonlinear cases. Figure 2.13 compares the maximum of the two cases. We notice that the nonlinear effects can increase the maximum at the beginning (compared to the linear situation) and an amplification



Figure 2.12: Evolution of the surface in the nonlinear case (red line), the linear case (blue line). The dashed line is the corresponding pressure.



Figure 2.13: Comparison of the maximum of the surface between the linear case (red line) and the nonlinear case (blue line).

occurs, but, after a while, a saturation of the L^{∞} -norm is possible since the wave breaks. This situation is similar to the one in the shallow water case. The only difference, is the fact that the Saut-Xu system is dispersive and hence the wave breaking is delayed.

In the following simulation, we take $\varepsilon=0.1$ and

$$P(t,X) = P_0(X-t)$$
 and $\zeta_0(X) = v_0(X) = -P_0(X) = \left(\operatorname{sech}\left(\frac{\sqrt{3}}{2}x\right)\right)^2$. (2.64)

Figure 2.14 displays the evolution of the surface and Figure 2.15 displays the evolution the maximum of the surface. We see that an amplification due to the source term is possible even in the nonlinear case.



Figure 2.14: Evolution of the surface with initial data (2.64). The dashed line is the corresponding pressure in (2.64).



Figure 2.15: Evolution of the maximum of the the surface with initial data (2.64).

2.3.4.3 Conclusion

We studied numerically nonlinear effects on the Proudman resonance thanks to two different systems. We saw that, in both cases, the source term can amplify the amplitude of the water wave. However, if the water wave is too big, wave breaking can occur. This leads to a saturation of the L^{∞} -norm. It could be interesting to study mathematically if we actually get a shock.

Chapter 3

A splitting method for deep water with bathymetry

Sommaire

3.1	Introduction				
	3.1.1	Presentation of the problem	72		
	3.1.2	Notations and assumptions	74		
3.2	The	Saut-Xu system	74		
3.3	A sp	litting scheme	79		
	3.3.1	The transport equation	79		
	3.3.2	The dispersive equation	79		
3.4	Erro	or estimates	80		
	3.4.1	The local error estimate	80		
	3.4.2	Global error estimate	84		
3.5	Nun	nerical experiments	86		
	3.5.1	Example 1: Convergence curve	86		
	3.5.2	Example 2: Linear versus nonlinear	86		
	3.5.3	Example 3: Boussinesq regime	87		
	3.5.4	Example 4: Rapidly varying topographies	90		

Ce chapitre est l'article [24]. Il a été écrit en collaboration avec Afaf Bouharguane (IMB).

3.1 Introduction

3.1.1 Presentation of the problem

In this chapter we derive and prove the wellposedness of a deep water model that generalizes the Saut-Xu system for nonflat bottoms. Then, we present a new numerical method based on a splitting approach for studying this system. The advantage of this method is that it does not require any low pass filter to avoid spurious oscillations. We prove a local error estimate and we show that our scheme represents a good approximation of order one in time. Then, we perform some numerical experiments which confirm our theoretical result and we study two physical phenomena : the behaviour of a KdV soliton when the shallowness parameter increases; the homogenization effect of rapidly varying topographies on water waves.

The study of the influence of the topography on water waves is an important issue in oceanography. Many phenomena are linked to the variation of the topography : shoaling, rip currents, diffraction, Bragg reflection. Since the direct study on the Euler equations is quite involved, several authors derived and justified asymptotic models according to different small parameters. A usual way to derive asymptotic models is to start from the Zakharov/Craig-Sulem-Sulem formulation [153, 47, 48], which is a good formulation for irrotational water waves, and to expand the Dirichlet-Neumann operator. Then, in the shallow water regime for example, several models were obtained like the Saint-Venant equations or the Green-Naghdi or Boussinesq equations, see [9, 80], [70] for instance. The present chapter addresses the influence of the bathymetry in deep water, in the sense explained below.

In this chapter, a denotes the typical amplitude of the water waves, L the typical length, Hthe typical height and a_{bott} the typical amplitude of the bathymetry. Then, we introduce three parameters : $\varepsilon = \frac{a}{H}$ the nonlinearity parameter, $\mu = \frac{H^2}{L^2}$ the shallowness parameter and $\beta = \frac{a_{bott}}{H}$ the bathymetric parameter. We denote by d the horizontal dimension, which is equal to 1 or 2. We recall that assuming μ small leads to shallow water models. In deep water, which is typically the case when μ is of order 1, it is quite common to assume that the steepness parameter $\varepsilon \sqrt{\mu} = \frac{a}{T}$ is small. The first nonlinear asymptotic model with a small steepness assumption was derived by Matsuno, when d = 1 for a flat bottom ([97]) and a slowly varying bottom ([98]), and when d = 2 for weakly transverse water waves (γ is of order $\mathcal{O}(\varepsilon)$) and a flat bottom ([99]). Then, Choi ([38]) extended [99] for general water waves (see also the work of Smith [132]). Finally, Bonneton and Lannes ([22]) gave a formulation of the Matsuno equations when d = 1 and d = 2 in the case of a small non-flat bottom (β of order $\mathcal{O}(\varepsilon)$). It is important to notice that these models are only formally derived. It is proven in [9] that smooth enough solutions to these models are close to the solutions of the water waves equations but, to our knowledge, the wellposedness of the Matsuno equations, even in the case of a flat bottom, is still an open problem. This system could be illposed (see Ambrose, Bona and Nicholls [10]). To avoid this difficulty, Saut and Xu ([125]) developed an equivalent problem to the Matsuno system which is consistent with the water waves problem and with the same accuracy. Then, they proved that this new system is wellposed. However, this model is for a flat bottom. In this chapter, we shall derive, use, and prove the wellposedness of a generalization of the Saut-Xu system with a non-flat bottom.

Many authors developed numerical approaches to study the impact of the bottom on water waves, see for instance [100], [91], [132], [65], [35], [20], [21]). However to our knowledge, when
one works with deep water models, there is no convergence result in the literature. After the original work of Craig and Sulem ([47]) and the paper of Craig et al. ([45]), Guyenne and Nicholls ([66]) developed a numerical method based on a pseudospectral method and a fourth-order Runge-Kutta scheme for the time integration. The linear terms are solved exactly whereas the nonlinear terms are viewed as source terms. Their approach has been developed for the whole water waves equations but we could easily adapt it to our system. However with their scheme, we observe spurious oscillations in the wave profile that lead to instabilities. These errors seem to appear when the nonlinear part is evaluated via the Fourier transform. This is the aliasing phenomenon. Guyenne and Nicholls also observe these oscillations and, to fix it, they apply at every time step a low-pass filter. The scheme that we propose in this chapter avoids this low-pass filter.

We present a new numerical method based on a splitting approach for studying nonlinear water waves in the presence of a bottom. We remark that the Saut-Xu system contains a dispersive part and a nonlinear transport part. Thus, the splitting method becomes an interesting alternative to solve the system since this approach is commonly used to split different physical terms, see for instance [123]. We also motivate our decomposition by the fact that, due to the pseudodifferential operator, some terms in the dispersive part may be computed efficiently using the fast Fourier transform. The transport part is computed thanks a Lax-wendroff method. Various versions of the splitting method have been developed for instance for the nonlinear Schrodinger, the viscous Burgers equation, Korteweg-de-Vries equations [32, 68, 92, 124, 135]. Thanks to this splitting, we only use a pseudospectral method for the nonlocal terms (contrary to [47, 66]), which limits the aliasing phenomenon and allows us to avoid a low-pass filter.

We denote by Φ^t the nonlinear flow associated to the Saut-Xu system (3.6), $\Phi^t_{\mathcal{A}}$ and $\Phi^t_{\mathcal{D}}$, respectively, the evolution operator associated with the transport part (see equation (3.11)) and with the dispersive part (see equation (3.12)). We consider the Lie formula defined by

$$\mathcal{Y}^t = \Phi^t_{\mathcal{A}} \circ \Phi^t_{\mathcal{D}}.\tag{3.1}$$

Since the Saut-Xu system (3.6) is a quasilinear system, we have derivatives losses in the proof of the convergence of our splitting scheme. In Theorem 3.4.6, we show that the numerical solution converges to the solution of the Saut-Xu system (3.6) in the $H^{N+\frac{1}{2}} \times H^N$ -norm for initial data in $H^{N+\frac{1}{2}-4} \times H^{N-4}(\mathbb{R})$.

Notice that it is not painful to generalize the present work to the Lie formula $\Phi_{\mathcal{D}}^t \circ \Phi_{\mathcal{A}}^t$. We also make the choice to prove a convergence result for a Lie splitting but our proof can be adapted to a Strang splitting or a more complex one. Finally, notice that our scheme can be used for others equations (see Remark 3.4.7).

The chapter is organised as follows. In the next section, we extend the Saut-Xu system by adding a topography effect and we prove a local wellposedness result. We also show that the flow map Φ^t is uniformly Lipschitzean. In section 3.3, we split the problem and we give some estimates on $\Phi^t_{\mathcal{A}}$ and $\Phi^t_{\mathcal{D}}$. In Section 3.4, we prove a local error estimate and we show that the Lie method represents a good approximation of order one in time (Theorem 3.4.6). Finally, in Section 3.5, we perform some numerical experiments which confirm our theoretical result and we illustrate two physical phenomena : the behaviour of a KdV soliton when the shallowness parameter increases and the homogenization effect of rapidly varying topographies on water waves.

3.1.2 Notations and assumptions

- x denotes the horizontal variable and z the vertical variable. In this chapter, we only study the case d = 1 ($x \in \mathbb{R}$).
- We assume that

$$0 \le \varepsilon, \beta \le 1, \ \exists \mu_{\max} > \mu_{\min} > 0, \ \mu_{\max} \ge \mu \ge \mu_{\min}.$$

$$(3.2)$$

We explain in Remark 3.2.1 our assumption on μ .

- We denote $\delta = \max(\varepsilon, \beta)$.
- Let $f \in \mathcal{C}^0(\mathbb{R})$ and $m \in \mathbb{N}$ such that $\frac{f}{1+|x|^m} \in L^\infty(\mathbb{R})$. We define the Fourier multiplier $f(D): H^m(\mathbb{R}) \to L^2(\mathbb{R})$ as

$$\forall u \in H^m(\mathbb{R}), \ \widehat{f(D)u}(\xi) = f(\xi)\widehat{u}(\xi).$$

- D denotes the Fourier multiplier corresponding to $\frac{\partial_x}{i}$.
- We denote by $C(c_1, c_2, ...)$ a generic positive constant, strictly positive, which depends on parameters $c_1, c_2, ...$.

3.2 The Saut-Xu system

In this part, we extend the Saut-Xu system ([125]) for a non-flat bottom. Then, we give a wellposedness result that generalizes the one of Saut-Xu.

The Matsuno system, which is a full dispersion model for deep water, is an asymptotic model of the water waves equations with an accuracy of order $\mathcal{O}(\delta^2)$. Bonneton and Lannes [22] formulated it in the following way in the presence of a non flat topography

$$\begin{cases} \partial_t \zeta - \frac{1}{\sqrt{\mu\nu}} \mathcal{H}_\mu v + \frac{\varepsilon}{\nu} \left(\mathcal{H}_\mu \left(\zeta \partial_x \mathcal{H}_\mu v \right) + \partial_x \left(\zeta v \right) \right) = \frac{\beta}{\nu} \partial_x \left(B_\mu v \right) \\ \partial_t v + \partial_x \zeta + \frac{\varepsilon}{\nu} v \partial_x v - \varepsilon \sqrt{\mu} \partial_x \zeta \mathcal{H}_\mu \partial_x \zeta = 0, \end{cases}$$
(3.3)

where $\zeta = \zeta(t, x)$ is the free surface, v = v(t, x) is the horizontal velocity at the surface, $\nu = \frac{\tanh(\sqrt{\mu})}{\sqrt{\mu}}$ and \mathcal{H}_{μ} and B_{μ} are Fourier multipliers,

$$\mathcal{H}_{\mu} = -\frac{\tanh(\sqrt{\mu}D)}{D}\partial_x \text{ and } B_{\mu} = \operatorname{sech}(\sqrt{\mu}D) \left(b \operatorname{sech}(\sqrt{\mu}D) \cdot\right),$$

and b is the topography. It is important to notice that in this context, the fluid domain is

$$\Omega_t := \{ (x, z) \in \mathbb{R}^2 , -1 + \beta b(x) < z < \varepsilon \zeta(t, x) \}.$$

In [9], Alvarez-Samaniego and Lannes show that the Matsuno system (3.3) is consistent with the Zakharov/Craig-Sulem-Sulem formulation when $\beta = 0$ and it is not painful to generalize their result to the case where $\beta \neq 0$. In [125], Saut and Xu obtained a new model with the same accuracy than the Matsuno system thanks to a nonlinear change of variables in the case of a flat bottom. The advantage of this model is that they proved a local wellposedness on large time for this new model. We follow their approach. We define new variables

$$\widetilde{v} = v + \frac{\varepsilon \sqrt{\mu}}{2} v \mathcal{H}_{\mu} \partial_x \zeta \text{ and } \widetilde{\zeta} = \zeta - \frac{\varepsilon \sqrt{\mu}}{4} v^2.$$
 (3.4)

Then, up to terms of order $\mathcal{O}(\delta^2)$, $\tilde{\zeta}$ and \tilde{v} satisfy (we omit the tildes for the sake of simplicity)

$$\begin{cases} \partial_t \zeta + \left(\frac{\varepsilon}{\nu} - \frac{\varepsilon \sqrt{\mu}}{2}\right) v \partial_x \zeta - \frac{1}{\sqrt{\mu}\nu} \mathcal{H}_\mu v + \frac{\varepsilon}{\nu} \left(\frac{1}{2} \mathcal{H}_\mu \left(v \partial_x \mathcal{H}_\mu \zeta\right) + \mathcal{H}_\mu (\zeta \partial_x \mathcal{H}_\mu v) + \zeta \partial_x v\right) = \frac{\beta}{\nu} \partial_x \left(B_\mu v\right) \\ \partial_t v + \left(\frac{\varepsilon}{\nu} + \frac{\varepsilon \sqrt{\mu}}{2}\right) v \partial_x v + \partial_x \zeta - \frac{\varepsilon \sqrt{\mu}}{2} \partial_x \zeta \mathcal{H}_\mu \partial_x \zeta - \frac{\varepsilon}{2\nu} v \mathcal{H}_\mu^2 \partial_x v = 0. \end{cases}$$
(3.5)

Since our motivation is the study of water waves in deep water (μ of order $\mathcal{O}(1)$), we assume that $\nu = \frac{1}{\mu}$. Hence, we study the following system, which is the system studied by Saut and Xu,

$$\begin{cases} \partial_t \zeta - \mathcal{H}_\mu v + \varepsilon \sqrt{\mu} \left(\frac{1}{2} v \partial_x \zeta + \frac{1}{2} \mathcal{H}_\mu (v \partial_x \mathcal{H}_\mu \zeta) + \mathcal{H}_\mu (\zeta \partial_x \mathcal{H}_\mu v) + \zeta \partial_x v \right) = \beta \sqrt{\mu} \partial_x (B_\mu v) \\ \partial_t v + \partial_x \zeta + \frac{3\varepsilon \sqrt{\mu}}{2} v \partial_x v - \frac{\varepsilon \sqrt{\mu}}{2} \partial_x \zeta \mathcal{H}_\mu \partial_x \zeta - \frac{\varepsilon \sqrt{\mu}}{2} v \mathcal{H}_\mu^2 \partial_x v = 0, \end{cases}$$
(3.6)

In the following, we denote $\mathbf{U} = (\zeta, v)^t$ and we define the energy of the system for $N \in \mathbb{N}$ by

$$\mathcal{E}^{N}(\mathbf{U}) = \frac{1}{\sqrt{\mu}} \left| \Lambda^{N} \zeta \right|_{2}^{2} + \left| |D|^{\frac{1}{2}} \Lambda^{N} \zeta \right|_{2}^{2} + |v|_{H^{N}}^{2} .$$
(3.7)

We also denote by E^N_{μ} the energy space related to this norm.

Remark 3.2.1. Notice that if μ satisfies condition (3.2), the energy \mathcal{E}^N is equivalent to the $H^{N+\frac{1}{2}} \times H^N$ -norm. Furthermore, the assumption $\mu_{\min} \leq \mu$ is essential for the Theorem 3.2.2.

The main result of this section is the following local wellposedness result. We recall that $\delta = \max(\varepsilon, \beta)$.

Theorem 3.2.2. Let $N \ge 2$, $\mathbf{U}_0 \in H^{N+\frac{1}{2}}(\mathbb{R}) \times H^N(\mathbb{R})$ and $b \in L^{\infty}(\mathbb{R})$. We assume that ε, β, μ satisfy Condition (3.2) and

$$|\mathbf{U}_0|_{H^{N+\frac{1}{2}} \times H^N} + |b|_{L^{\infty}} \le M.$$

Then, there exists a time $T_0 = T_0\left(M, \frac{1}{\mu_{\min}}, \mu_{\max}\right)$ independent of ε , μ and β and a unique solution $\mathbf{U} \in \mathcal{C}\left(\left[0, \frac{T_0}{\delta}\right], E^N_{\mu}\right)$ of the system (3.6) with initial data \mathbf{U}_0 . Furthermore, we have the following energy estimate, for all $t \in \left[0, \frac{T_0}{\delta}\right]$,

$$\mathcal{E}^{N}(\mathbf{U}(t,\cdot)) \leq e^{\delta C_{0}t} \mathcal{E}^{N}(\mathbf{U}_{0}),$$

where $C_0 = C\left(M, \frac{1}{\mu_{\min}}, \mu_{\max}\right)$.

Proof. We refer to Paragraph IV in [125] for a complete proof and we focus only on the bottom contribution. We quasilinearize System (3.6). For $0 \leq \alpha \leq N$, we denote $\mathbf{U}^{(\alpha)} = (\partial^{\alpha} \zeta, \partial^{\alpha} v)$. Then, applying ∂^{α} to System (3.6), we get

$$\partial_t \mathbf{U}^{(\alpha)} + \mathcal{L} \mathbf{U}^{(\alpha)} + \frac{\varepsilon \sqrt{\mu}}{2} \mathbb{1}_{\{\alpha \neq 0\}} \mathcal{B}[\mathbf{U}] \mathbf{U}^{(\alpha)} = \beta \sqrt{\mu} \left(\partial_x \partial^\alpha \left(B_\mu v \right), 0 \right)^t + \varepsilon \sqrt{\mu} \mathcal{G}^\alpha,$$

where

$$\begin{split} \mathcal{L} &= \begin{pmatrix} 0 & -\frac{1}{\sqrt{\mu\nu}} \mathcal{H}_{\mu} \\ \partial_{x} & 0 \end{pmatrix} \\ \mathcal{B}[\mathbf{U}] &= \begin{pmatrix} \mathcal{H}_{\mu} \left(v \mathcal{H}_{\mu} \partial_{x} \cdot \right) + v \partial_{x} & \mathcal{H}_{\mu} \left(\cdot \mathcal{H}_{\mu} \partial_{x} \zeta \right) - \partial_{x} \zeta \mathcal{H}_{\mu}^{2} \\ -\partial_{x} \zeta \mathcal{H}_{\mu} \partial_{x} - \mathcal{H}_{\mu} \partial_{x} \zeta \partial_{x} & 3v \partial_{x} - v \mathcal{H}_{\mu}^{2} \partial_{x} \end{pmatrix}, \end{split}$$

and $\mathcal{G}^{\alpha} = (\mathcal{G}^{\alpha}_1, \mathcal{G}^{\alpha}_2)^t$ with

$$\begin{aligned} \mathcal{G}_{1}^{\alpha} &= \partial_{x}^{\alpha} g(\zeta, v) - \frac{1}{2} \sum_{1 \leq \gamma \leq \alpha - 1} C_{\alpha}^{\gamma} \left(\mathcal{H}_{\mu} (\partial_{x}^{\gamma} v \mathcal{H}_{\mu} \partial_{x}^{1 + \alpha - \gamma} \zeta) + \partial_{x}^{\gamma} v \partial_{x}^{1 + \alpha - \gamma} \zeta \right) - \frac{1}{2} \partial_{x} \zeta (\mathcal{H}_{\mu}^{2} + 1) \partial_{x}^{\alpha} v \\ \mathcal{G}_{2}^{\alpha} &= \frac{1}{2} \sum_{1 \leq \gamma \leq \alpha - 1} C_{\alpha}^{\gamma} \partial_{x}^{1 + \gamma} \zeta \mathcal{H}_{\mu} \partial_{x}^{1 + \alpha - \gamma} \zeta + \sum_{1 \leq \gamma \leq \alpha} C_{\alpha}^{\gamma} \left(-\frac{3}{2} \partial_{x}^{\gamma} v \partial_{x}^{1 + \alpha - \gamma} v + \frac{1}{2} \partial_{x}^{\gamma} v \mathcal{H}_{\mu}^{2} \partial_{x}^{1 + \alpha - \gamma} v \right) \end{aligned}$$

where

$$g(\zeta, v) = -[\mathcal{H}_{\mu}, \zeta]\mathcal{H}_{\mu}\partial_{x}v - \zeta(\mathcal{H}_{\mu}^{2} + 1)\partial_{x}v.$$

Then we can show, as in Paragraph IV. B in [125], that

$$\left|\mathcal{G}^{\alpha}\right|_{2}+\left|\left|D\right|^{\frac{1}{2}}\mathcal{G}^{\alpha}\right|_{2}\leq C\left(\frac{1}{\mu_{\min}}\right)\mathcal{E}^{N}(\mathbf{U})$$

As Saut and Xu, we define a symmetrizer for ${\cal L}$

$$\mathcal{S} = \begin{pmatrix} \frac{D}{\tanh(\sqrt{\mu}D)} & 0\\ 0 & 1 \end{pmatrix}.$$

Notice that $\sqrt{(\mathcal{S},\cdot)}$ is a norm equivalent to $\sqrt{\mathcal{E}^0}$. For the bottom contribution, we easily get

$$\left| \left(\frac{D}{\tanh(\sqrt{\mu}D)} \partial^{\alpha} \zeta, \partial_{x} \partial^{\alpha} \operatorname{sech}\left(\sqrt{\mu}D\right) \left(b \operatorname{sech}\left(\sqrt{\mu}D\right) v \right) \right) \right| \leq C \left(\frac{1}{\mu_{\min}} \right) \left| b \right|_{\infty} \mathcal{E}^{N}(U) \,.$$

Then, we obtain,

$$\mathcal{E}^{N}(\mathbf{U}) \leq \mathcal{E}^{N}(\mathbf{U}_{0}) + \delta C\left(\frac{1}{\mu_{\min}}, \mu_{\max}\right) \int_{0}^{t} \left(\mathcal{E}^{N}(U)^{\frac{3}{2}} + \mathcal{E}^{N}(U)\right)(s) ds,$$

and there exists a time T > 0, such that, for all $t \in \left[0, \frac{T}{\delta}\right]$,

$$\mathcal{E}^{N}(\mathbf{U}(t,\cdot)) \leq C\left(\frac{1}{\mu_{\min}},\mu_{\max},\mathcal{E}^{N}(\mathbf{U}_{0})\right)$$

The energy estimate follows from the Gronwall Lemma.

In order to use a Lady Windermere's fan argument to prove the convergence of the numerical scheme, we need a Lipschitz property for the flow of the Saut-Xu system (3.6). We first give a control of the differential of the flow with respect to the initial datum.

Proposition 3.2.3. Let $N \geq 2$, $\mathbf{V}_0 \in H^{N+\frac{1}{2}}(\mathbb{R}) \times H^N(\mathbb{R})$, $\mathbf{U}_0 \in H^{N+1+\frac{1}{2}}(\mathbb{R}) \times H^{N+1}(\mathbb{R})$, and $b \in L^{\infty}(\mathbb{R})$. We assume that ε, β, μ satisfy Condition (3.2) and

$$|\mathbf{V}_0|_{H^{N+\frac{1}{2}} \times H^N} + |\mathbf{U}_0|_{H^{N+1+\frac{1}{2}} \times H^{N+1}} + |b|_{L^{\infty}} \le M.$$

Then, there exists a time $T = T\left(M, \frac{1}{\mu_{\min}}, \mu_{\max}\right)$ independent of ε , μ and β such that $(\Phi^t)'(\mathbf{U}_0) \cdot (\mathbf{V}_0)$ exists on $\left[0, \frac{T}{\delta}\right]$. Furthermore, we have, for all $0 \le t \le \frac{T}{\delta}$,

$$\left| \left(\Phi^t \right)' \left(\mathbf{U}_0 \right) \cdot \left(\mathbf{V}_0 \right) \right|_{H^{N+\frac{1}{2}} \times H^N} \le C \left(\frac{1}{\mu_{\min}}, \mu_{\max}, M \right) \left| \mathbf{V}_0 \right|_{H^{N+\frac{1}{2}} \times H^N}.$$

Proof. This proof is similar to the one in Paragraph IV in [125]. We denote $\mathbf{U}(t) = (\zeta(t), v(t))$ the solution of the Saut-Xu system (3.6) with initial data \mathbf{U}_0 . We denote also $(\eta(t), w(t)) = (\Phi^t)'(\mathbf{U}_0) \cdot (\mathbf{V}_0)$. Then, (η, w) satisfy the following system

$$\partial_t \begin{pmatrix} \eta \\ w \end{pmatrix} = \mathcal{L} \begin{pmatrix} \eta \\ w \end{pmatrix} + \varepsilon \sqrt{\mu} \mathcal{N}[(\zeta, v)] \partial_x \begin{pmatrix} \eta \\ w \end{pmatrix} + \varepsilon \sqrt{\mu} \mathcal{N}[(\eta, w)] \partial_x \begin{pmatrix} \zeta \\ v \end{pmatrix} = \beta \sqrt{\mu} \left(\partial_x \left(B_\mu w \right), 0 \right)^t, \quad (3.8)$$

where

$$\mathcal{L} = \begin{pmatrix} 0 & -\frac{1}{\sqrt{\mu\nu}} \mathcal{H}_{\mu} \\ \partial_x & 0 \end{pmatrix}$$
$$\mathcal{N}[(\zeta, v)] = \begin{pmatrix} \frac{1}{2} \mathcal{H}_{\mu} (v \mathcal{H}_{\mu} \cdot) + \frac{1}{2} v & \mathcal{H}_{\mu} (\zeta \mathcal{H}_{\mu} \cdot) + \zeta \\ -\frac{1}{2} \partial_x \zeta \mathcal{H}_{\mu} & \frac{3}{2} v - \frac{1}{2} v \mathcal{H}_{\mu}^2 \end{pmatrix}.$$

We quasilinearize System (3.8). For $0 \le \alpha \le N$, we denote $\mathbf{V}^{(\alpha)} = (\partial^{\alpha} \eta, \partial^{\alpha} w)$. Then, applying ∂^{α} to System (3.6), we get

$$\partial_t \mathbf{V}^{(\alpha)} + \mathcal{L} \mathbf{V}^{(\alpha)} + \frac{\varepsilon \sqrt{\mu}}{2} \mathbb{1}_{\{\alpha \neq 0\}} \left(\mathcal{B}[\mathbf{U}] \mathbf{V}^{(\alpha)} + \mathcal{B}[\mathbf{V}] \partial^{\alpha} \mathbf{U} \right) = \sqrt{\mu} \beta \begin{pmatrix} \partial_x \partial^{\alpha} \left(B_{\mu} w \right) \\ 0 \end{pmatrix} + \varepsilon \sqrt{\mu} \mathcal{G}^{\alpha},$$

where

$$\mathcal{B}[\mathbf{U}] = \begin{pmatrix} \mathcal{H}_{\mu} \left(v \mathcal{H}_{\mu} \partial_{x} \cdot \right) + v \partial_{x} & \mathcal{H}_{\mu} \left(\cdot \mathcal{H}_{\mu} \partial_{x} \zeta \right) - \partial_{x} \zeta \mathcal{H}_{\mu}^{2} \\ -\partial_{x} \zeta \mathcal{H}_{\mu} \partial_{x} - \mathcal{H}_{\mu} \partial_{x} \zeta \partial_{x} & 3v \partial_{x} - v \mathcal{H}_{\mu}^{2} \partial_{x} \end{pmatrix}$$

Then, we can show, as in Paragraph IV. B in [125], that

$$\left|\mathcal{G}^{\alpha}\right|_{2} + \left|\left|D\right|^{\frac{1}{2}} \mathcal{G}^{\alpha}\right|_{2} \le \varepsilon \sqrt{\mu} C\left(\frac{1}{\mu_{\min}}\right) \mathcal{E}^{N}(\mathbf{U}).$$

$$(3.9)$$

We recall that we can symmetrize ${\mathcal L}$ thanks to

$$\mathcal{S} = \begin{pmatrix} \frac{D}{\tanh(\sqrt{\mu}D)} & 0\\ 0 & 1 \end{pmatrix}$$

We define the energy associated to this symmetrizer

$$F^{\alpha}\left(\mathbf{V}\right) = \left|\sqrt{\frac{D}{\tanh(\sqrt{\mu}D)}}\partial^{\alpha}\eta\right|_{2}^{2} + \left|\partial^{\alpha}w\right|_{2}^{2},$$

$$F^{N}\left(\mathbf{V}\right) = \sum_{0 \le \alpha \le N} F^{\alpha}\left(\mathbf{V}\right).$$

We have, for $\alpha \neq 0$,

$$\frac{d}{dt}F^{\alpha}\left(\mathbf{V}\right) = \left(\mathcal{G}^{\alpha}, \mathcal{S}\mathbf{V}^{(\alpha)}\right) - \frac{\varepsilon\sqrt{\mu}}{2}\left(\mathcal{B}[\mathbf{U}]\mathbf{V}^{(\alpha)}, \mathcal{S}\mathbf{V}^{(\alpha)}\right) - \frac{\varepsilon\sqrt{\mu}}{2}\left(\mathcal{B}[\mathbf{V}]\partial^{\alpha}\mathbf{U}, \mathcal{S}\mathbf{V}^{(\alpha)}\right) \\
+ \beta\sqrt{\mu}\left(\partial_{x}\partial^{\alpha}\left(B_{\mu}v\right), \mathcal{S}\mathbf{V}^{(\alpha)}\right) \\
= I + II + III + IIII.$$

We can estimate I and II as in Paragraph IV. B in [125] thanks to estimate (3.9). For IIII, we can proceed as in the previous theorem. For the term III, we get, thanks to Proposition B.4.1,

$$|III| \le \varepsilon \sqrt{\mu} \left| (\zeta, v) \right|_{H^{N+1+\frac{1}{2}} \times H^{N+1}} \left| (\eta, w) \right|_{H^{N+\frac{1}{2}} \times H^{N}}$$

Then, we get

$$\frac{d}{dt}F^{N}\left(\mathbf{V}\right) \leq \delta\sqrt{\mu}C(M)\left(F^{N}\left(\mathbf{V}\right) + \sqrt{F^{N}\left(\mathbf{V}\right)}\right),$$

and the result follows.

We give now a Lipschitz property for the flow of the Saut-Xu system (3.6).

Proposition 3.2.4. Let $N \geq 2$, $\mathbf{U}_0, \mathbf{V}_0 \in H^{N+1+\frac{1}{2}} \times H^{N+1}(\mathbb{R})$ and $b \in L^{\infty}(\mathbb{R})$. We assume that ε, β, μ satisfy Condition (3.2) and

$$|\mathbf{V}_0|_{H^{N+1+\frac{1}{2}} \times H^{N+1}} + |\mathbf{U}_0|_{H^{N+1+\frac{1}{2}} \times H^{N+1}} + |b|_{L^{\infty}} \le M.$$

Then, there exists a time T independent of ε , μ and β and two unique solutions \mathbf{U}, \mathbf{V} of the system (3.6) on $[0, \frac{T}{\delta}]$ with initial data \mathbf{U}_0 and \mathbf{V}_0 . Furthermore, we have the following lipschitz estimate, for all $0 \le t \le \frac{T}{\delta}$,

$$\left| \mathbf{U}(t, \cdot) - \mathbf{V}(t, \cdot) \right|_{H^{N+\frac{1}{2}} \times H^{N}} \le K \left| \mathbf{U}_{0} - \mathbf{V}_{0} \right|_{H^{N+\frac{1}{2}} \times H^{N}}, \tag{3.10}$$

where $K = C\left(\frac{1}{\mu_{\min}}, \mu_{\max}, M\right)$.

Proof. The existence of \mathbf{U}, \mathbf{V} and T follow from the previous theorem. Furthermore, we have

$$\mathbf{U}(t) - \mathbf{V}(t) = \int_{s=0}^{1} \left(\Phi^{t} \right)' \left(\mathbf{V}_{0} + s \left(\mathbf{U}_{0} - \mathbf{V}_{0} \right) \right) \cdot \left(\mathbf{U}_{0} - \mathbf{V}_{0} \right).$$

The result follows from Proposition 3.2.3.

and

3.3 A splitting scheme

In this section, we split the Saut-Xu system (3.6) and we give some estimates for the subproblems. We consider, separately, the transport part

$$\begin{cases} \partial_t \zeta + \frac{\varepsilon \sqrt{\mu}}{2} \left(\left(\mathcal{H}^2_{\mu} + 1 \right) v \right) \partial_x \zeta = 0 \\ \partial_t v + \frac{3\varepsilon \sqrt{\mu}}{2} v \partial_x v = 0, \end{cases}$$
(3.11)

and the dispersive part

$$\begin{cases} \partial_t \zeta - \mathcal{H}_{\mu} v + \varepsilon \sqrt{\mu} \left(\frac{1}{2} \mathcal{H}_{\mu} \left(v \partial_x \mathcal{H}_{\mu} \zeta \right) + \mathcal{H}_{\mu} \left(\zeta \partial_x \mathcal{H}_{\mu} v \right) + \zeta \partial_x v - \frac{1}{2} \partial_x \zeta \mathcal{H}_{\mu}^2 v \right) = \beta \sqrt{\mu} \partial_x \left(B_{\mu} v \right) \\ \partial_t v + \partial_x \zeta - \frac{\varepsilon \sqrt{\mu}}{2} \partial_x \zeta \mathcal{H}_{\mu} \partial_x \zeta - \frac{\varepsilon \sqrt{\mu}}{2} v \mathcal{H}_{\mu}^2 \partial_x v = 0. \end{cases}$$
(3.12)

We denote by $\Phi_{\mathcal{A}}^t$ the flow of System (3.11) and by $\Phi_{\mathcal{D}}^t$ the flow of System (3.12).

Remark 3.3.1. Notice that we keep the term $\zeta \partial_x v$ in the first equation and we decompose $v \partial_x \zeta$ as $v \partial_x \zeta = \partial_x \zeta \left(\mathcal{H}^2_\mu + 1\right) v - \partial_x \zeta \mathcal{H}^2_\mu v$. This will be useful for the local wellposedness of the dispersive part.

In the following, we prove the local existence on large time for Systems (3.11) and (3.12).

3.3.1 The transport equation

The system (3.11) is a transport equation. Then, it is easy to get the following result.

Proposition 3.3.2. Let $s_1 \ge 0$, $s_2 > \frac{3}{2}$ and M > 0. We assume that ε, μ satisfies Condition (3.2). Then, there exists a time $T_1 = T_1(M, \mu_{\max}) > 0$, such that if

$$|\zeta_0|_{H^{s_1}} + |v_0|_{H^{s_2}} \le M,$$

we have a unique solution $(\zeta, v) \in \mathcal{C}\left(\left[0, \frac{T_1}{\varepsilon}\right], H^{s_1}(\mathbb{R}) \times H^{s_2}(\mathbb{R})\right)$, to System (3.11) with initial data (ζ_0, v_0) . Furthermore, we have, for all, $t \leq \frac{T_1}{\varepsilon}$,

$$|\zeta(t,\cdot)|_{H^{s_1}} + |v(t,\cdot)|_{H^{s_2}} \le e^{C_1 t} |U_0|_{H^{s_1} \times H^{s_2}}, \tag{3.13}$$

where $C_1 > 0$ depends on M and μ_{max} .

Proof. The proof follows from the fact that the quasilinear system (3.11) is symmetric. Thanks to the Coifman-Meyer estimate (see Lemma B.3.1), we get

$$\frac{d}{dt} \left(|\zeta(t,\cdot)|^2_{H^{s_1}} + |v(t,\cdot)|^2_{H^{s_2}} \right) \le \varepsilon \sqrt{\mu} \left(|\zeta(t,\cdot)|^2_{H^{s_1}} + |v(t,\cdot)|^2_{H^{s_2}} \right)^{\frac{3}{2}}.$$

Then, we see that the energy is bounded uniformly with respect to ε and μ and applying the Gronwall lemma, we get the result.

3.3.2 The dispersive equation

The system (3.12) contains all the dispersive terms of the Saut-Xu system. We have the following estimate for the flow.

Proposition 3.3.3. Let $N \ge 2$, and $b \in L^{\infty}(\mathbb{R})$. We assume that ε, β, μ satisfy Condition (3.2). Then, there exists a time $T_2 = T_2\left(M, \frac{1}{\mu_{\min}}, \mu_{\max}\right)$ such that if

$$|\zeta_0|_{H^{N+\frac{1}{2}}} + |v_0|_{H^N} + |b|_{L^{\infty}} \le M,$$

we have a unique solution $(\zeta, v) \in \mathcal{C}\left(\left[0, \frac{T_2}{\delta}\right], H^{N+\frac{1}{2}}(\mathbb{R}) \times H^N(\mathbb{R})\right)$ to the system (3.12) with initial data (ζ_0, v_0) . Furthermore, we have, for all $t \leq \frac{T_2}{\delta}$,

$$|\zeta(t,\cdot)|_{H^{N+\frac{1}{2}}} + |v(t,\cdot)|_{H^N} \le e^{C_2 t} |\mathbf{U}_0|_{H^{N+1/2} \times H^N}, \tag{3.14}$$

where C_2 is a positive constant which depends on $\mu_{\max}, \frac{1}{\mu_{\min}}, M$.

Proof. The proof is a small adaptation of the proof of Theorem 3.2.2 and part IV in [125]. We notice that in the proof of Saut and Xu, the transport part can be treated separately and does not influence the control of the others terms. Hence, we can use the same estimate and we get that

$$\frac{d}{dt}\mathcal{E}^N(\zeta,v) \leq C\left(\frac{1}{\mu_{\min}}\right) \left(\frac{\varepsilon}{\nu}\mathcal{E}^N(\zeta,v)^{\frac{3}{2}} + \frac{\beta}{\nu}\mathcal{E}^N(\zeta,v)\right)$$

Since μ is bounded from above and from below, there exists a constant $C = C\left(\mu_{\max}, \frac{1}{\mu_{\min}}\right)$ such that

$$\frac{1}{C}\mathcal{E}^{N}(\zeta, v) \le |\zeta|_{H^{N+\frac{1}{2}}}^{2} + |v|_{H^{N}}^{2} \le C\mathcal{E}^{N}(\zeta, v) + C\mathcal{E}^{N}(\zeta, v$$

and we see that the energy is bounded uniformly with respect to ε and μ . Applying the Gronwall lemma, we get the result.

Remark 3.3.4. Under the assumption of Proposition 3.3.3, we get from relations (3.13) and (3.14) that, there exists a time $T_3 > 0$, such that for all $t \in [0, \frac{T_3}{\delta}]$,

$$|\mathcal{Y}^t \mathbf{U}_0|_{H^{N+1/2} \times H^N} \le e^{C_3 \delta t} |\mathbf{U}_0|_{H^{N+1/2} \times H^N},$$

where T_3 and C_3 depend only on M, μ_{\max} and $\frac{1}{\mu_{\min}}$.

3.4 Error estimates

The goal of this part is to prove the main result of this chapter (Theorem 3.4.6). Our analysis is based on energy estimates.

3.4.1 The local error estimate

The local error is the following quantity

$$e(t, \mathbf{U}_0) = \Phi^t(\mathbf{U}_0) - \mathcal{Y}^t(\mathbf{U}_0).$$
(3.15)

Our approach is similar to the one developed in [36] (see Lemma 3.4 in [36]). We use the fact that $\Phi^t(\mathbf{U}_0)$ satisfies a symmetrizable system. Therefore, e satisfies this system up to a remainder and then, we can control e thanks to energy estimates. In the following we give different technical lemmas in order to control the local error. The main result of this part is Proposition 3.4.5. We recall that the transport operator is the operator \mathcal{A}

$$\mathcal{A}(\zeta, v) = -\frac{\varepsilon\sqrt{\mu}}{2} \left(\begin{pmatrix} \left(\left(\mathcal{H}^2_{\mu} + 1\right)v\right) \partial_x \zeta \\ 3v \partial_x v \end{pmatrix} \right).$$

The following proposition gives an estimate of the differential of the transport operator.

Lemma 3.4.1. Let $s_1, s_2 \ge 0$ and ε, μ satisfying Condition (3.2). Then,

$$|\mathcal{A}'(\zeta, v).(\eta, w)|_{H^{s_1} \times H^{s_2}} \le \varepsilon C(\mu_{\max}) |(\zeta, v)|_{H^{s_1+1} \times H^{s_2+1}} |(\eta, w)|_{H^{s_1+1} \times H^{s_2+1}}.$$

Proof. We have

$$\mathcal{A}'(\zeta, v).(\eta, w) = -\frac{\varepsilon\sqrt{\mu}}{2} \begin{pmatrix} \left(\left(\mathcal{H}^2_{\mu} + 1\right)v\right) \partial_x \eta + \left(\left(\mathcal{H}^2_{\mu} + 1\right)w\right) \partial_x \zeta \\ 3w \partial_x v + 3v \partial_x w \end{pmatrix},$$

and the estimate follows from the product estimate B.2.1.

We can do the same for the dispersive part (using also Proposition B.4.1). We recall that the dispersive operator is the operator \mathcal{D}

$$\mathcal{D}(\zeta, v) = \begin{pmatrix} \mathcal{H}_{\mu}v + \varepsilon\sqrt{\mu} \left(\frac{1}{2}\mathcal{H}_{\mu} \left(v\partial_{x}\mathcal{H}_{\mu}\zeta\right) + \mathcal{H}_{\mu} \left(\zeta\partial_{x}\mathcal{H}_{\mu}v\right) + \zeta\partial_{x}v - \frac{1}{2}\partial_{x}\zeta\mathcal{H}_{\mu}^{2}v\right) - \beta\sqrt{\mu}\partial_{x}\left(B_{\mu}v\right) \\ -\partial_{x}\zeta + \frac{\varepsilon\sqrt{\mu}}{2}\partial_{x}\zeta\mathcal{H}_{\mu}\partial_{x}\zeta + \frac{\varepsilon\sqrt{\mu}}{2}v\mathcal{H}_{\mu}^{2}\partial_{x}v \end{pmatrix}$$

Lemma 3.4.2. Let s > 0, ε, β, μ satisfying Condition (3.2) and $b \in L^{\infty}(\mathbb{R})$. Then,

$$|\mathcal{D}'(\zeta,v).(\eta,w)|_{H^s\times H^s} \leq C(\mu_{\max}) \left(1 + \beta \left|b\right|_{L^{\infty}} + \varepsilon \left|(\zeta,v)\right|_{H^{s+1}\times H^{s+1}}\right) |(\eta,w)|_{H^{s+1}\times H^{s+1}}.$$

Furthermore, we have to control the derivative of the flow $\Phi_{\mathcal{A}}^t$ with respect to the initial data. We denote it by $(\Phi_{\mathcal{A}}^t)'$. We recall that $\delta = \max(\varepsilon, \beta)$.

Lemma 3.4.3. Let $s_1, s_2 \ge 0$, M > 0, ε, β, μ satisfying Condition (3.2) and $b \in L^{\infty}(\mathbb{R})$. Let $(\zeta_0, v_0) \in H^{s_1+1} \times H^{s_2+1}(\mathbb{R}^d)$ such that,

$$|(\zeta_0, v_0)|_{H^{s_1+1} \times H^{s_2+1}} \le M.$$

Then, there exists a time $T = T(M, \mu_{\max})$, such that $(\Phi^t_{\mathcal{A}})'(\zeta_0, v_0) \cdot (\eta_0, w_0)$ exists for all $t \in [0, \frac{T}{\delta}]$ and if we denote

$$\begin{pmatrix} \eta \\ w \end{pmatrix} = \left(\Phi_{\mathcal{A}}^t \right)' \left(\zeta_0, v_0 \right) \cdot \left(\eta_0, w_0 \right),$$

for all $0 \le t \le \frac{T}{\delta}$,

$$|(\eta, w)(t, \cdot)|_{H^{s_1} \times H^{s_2}} \le |(\eta_0, w_0)|_{H^{s_1} \times H^{s_2}} C(\mu_{\max}, M).$$

Proof. The quantity (η, w) satisfies the following linear system

$$\begin{cases} \partial_t \eta + \frac{\varepsilon \sqrt{\mu}}{2} \left(\mathcal{H}^2_{\mu} + 1 \right) v \partial_x \eta + \frac{\varepsilon \sqrt{\mu}}{2} \left(\mathcal{H}^2_{\mu} + 1 \right) w \partial_x \zeta = 0, \\ \partial_t w + \frac{3\varepsilon \sqrt{\mu}}{2} v \partial_x w + \frac{3\varepsilon \sqrt{\mu}}{2} w \partial_x v = 0, \end{cases}$$

where $(\zeta, v) = \Phi_{\mathcal{A}}^t(\zeta_0, v_0)$. The result follows from energy estimates, the Gronwall lemma and Proposition 3.3.2.

In the following, we use the fact $\Phi_{\mathcal{A}}^t \circ \Phi_{\mathcal{D}}^t$ satisfies the Saut-Xu system (3.6) up to a remainder. The following lemma is the key point for the control of this remainder.

Lemma 3.4.4. Let $N \geq 2$, M > 0, ε, β, μ satisfying Condition (3.2) and $b \in L^{\infty}(\mathbb{R})$. Let $\mathbf{U} = (\zeta, v) \in H^{N+\frac{1}{2}} \times H^{N}(\mathbb{R}^{d})$ such that,

$$|b|_{L^{\infty}} + |\mathbf{U}|_{H^{N+\frac{1}{2}} \times H^{N}(\mathbb{R})} \leq M.$$

Then, there exists a time $T = T\left(M, \mu_{\max}, \frac{1}{\mu_{\min}}\right) > 0$, such that $\Phi_{\mathcal{A}}^{t}(\mathbf{U})$ exists for all $0 \le t \le \frac{T}{\delta}$, and furthermore,

$$\left| \left(\Phi_{\mathcal{A}}^{t} \right)' (\mathbf{U}) \cdot \mathcal{D} (\mathbf{U}) - \mathcal{D} \left(\Phi_{\mathcal{A}}^{t} (\mathbf{U}) \right) \right|_{H^{N-2} \times H^{N-2}} \leq \varepsilon C \left(M, \mu_{\max}, \frac{1}{\mu_{\min}} \right) t$$

Proof. The existence of T follows from Proposition 3.3.2. Then, we notice that

$$\left(\Phi_{\mathcal{A}}^{t}\right)'(\mathbf{U}) \cdot \mathcal{D}(\mathbf{U}) - \mathcal{D}\left(\Phi_{\mathcal{A}}^{t}(\mathbf{U})\right) = \int_{0}^{t} \mathcal{A}'(\Phi_{\mathcal{A}}^{s}(\mathbf{U})) \cdot \left(\left(\Phi_{\mathcal{A}}^{s}\right)'(\mathbf{U}) \cdot \mathcal{D}(\mathbf{U})\right) - \mathcal{D}'(\Phi_{\mathcal{A}}^{s}(\mathbf{U})) \cdot \mathcal{A}\left(\Phi_{\mathcal{A}}^{s}(\mathbf{U})\right).$$

Using Lemmas 3.4.1, 3.4.2 and Proposition 3.3.2, we get,

$$\left| \left(\Phi_{\mathcal{A}}^{t} \right)'(\mathbf{U}) \cdot \mathcal{D}(\mathbf{U}) - \mathcal{D}\left(\Phi_{\mathcal{A}}^{t}(\mathbf{U}) \right) \right|_{H^{N-2} \times H^{N-2}} \leq C\left(\mu_{\max}, M \right) \int_{0}^{t} \varepsilon \left| \left(\Phi_{\mathcal{A}}^{s} \right)'(\mathbf{U}) \cdot \mathcal{D}(\mathbf{U}) \right|_{H^{N-1} \times H^{N-1}} + \left| \mathcal{A}\left(\Phi_{\mathcal{A}}^{s}(\mathbf{U}) \right) \right|_{H^{N-1} \times H^{N-1}} \right|_{H^{N-1} \times H^{N-1}}$$

Then, using Lemma 3.4.3, the product estimate B.2.1 and the expression of \mathcal{A} , we obtain

$$\left| \left(\Phi_{\mathcal{A}}^{t} \right)' (\mathbf{U}) \cdot \mathcal{D}(\mathbf{U}) - \mathcal{D} \left(\Phi_{\mathcal{A}}^{t}(\mathbf{U}) \right) \right|_{H^{N-2} \times H^{N-2}} \leq \varepsilon C(\mu_{\max}, M) \int_{0}^{t} \left| \mathcal{D}(\mathbf{U}) \right|_{H^{N-1} \times H^{N-1}} + \left| \Phi_{\mathcal{A}}^{s}(\mathbf{U}) \right|_{H^{N} \times H^{N}}^{2}.$$

Finally, the result follows from the expression of \mathcal{D} , the product estimate B.2.1 and Proposition B.4.1.

We can now give the main result of this part, the local error estimate. We recall that $\delta = \max(\varepsilon, \beta)$.

Proposition 3.4.5. Let $N \ge 4$, M > 0, ε, β, μ satisfying Condition (3.2) and $b \in L^{\infty}(\mathbb{R})$. Let $\mathbf{U}_0 = (\zeta_0, v_0)$ such that,

$$|b|_{L^{\infty}} + |\mathbf{U}_0|_{H^{N+\frac{1}{2}} \times H^N} \le M.$$

Then, there exists a time $T_4 = T_4\left(M, \frac{1}{\mu_{\min}}, \mu_{\max}\right) > 0$, such that the local error $e(t, \mathbf{U})$ defined in (3.15) exists for all $0 \le t \le \frac{T_4}{\delta}$, and furthermore,

$$|e(t, \mathbf{U}_0)|_{H^{N-4+\frac{1}{2}} \times H^{N-4}} \le \delta C_4 t^2,$$

where $C_4 = C\left(\frac{1}{\mu_{\min}}, \mu_{\max}, M\right)$.

Proof. From Propositions 3.3.2 and 3.3.3, we obtain the existence of T. We denote

$$\mathbf{U}(t) = \begin{pmatrix} \zeta(t) \\ v(t) \end{pmatrix} = \Phi^t \left(\mathbf{U}_0 \right) \text{ and } \mathbf{V}(t) = \begin{pmatrix} \eta(t) \\ w(t) \end{pmatrix} = \Phi^t_{\mathcal{A}} \left(\Phi^t_{\mathcal{D}} \left(\mathbf{U}_0 \right) \right)$$

Then, from Theorem 3.2.2 and Propositions 3.3.2 and 3.3.3 we also have, for all $0 \le t \le \frac{T}{\delta}$,

$$\left|\mathbf{U}(t,\cdot)\right|_{H^{N+\frac{1}{2}} \times H^{N}} + \left|\mathbf{V}(t,\cdot)\right|_{H^{N+\frac{1}{2}} \times H^{N}} \le C\left(\frac{1}{\mu_{\min}},\mu_{\max},M\right).$$
(3.16)

We know that (ζ, v) satisfy the Saut-Xu system (3.6). Furthermore, (η, w) satisfy also the Saut-Xu system (3.6) up to a remainder

$$\partial_t \begin{pmatrix} \eta \\ w \end{pmatrix} = \mathcal{A}(\eta, w) + \mathcal{D}(\eta, w) + \mathcal{R}(t),$$

where $\mathcal{R}(t) = (\Phi_{\mathcal{A}}^{t})'(\Phi_{\mathcal{D}}^{t}(\mathbf{U}_{0})) \cdot \mathcal{D}(\Phi_{\mathcal{D}}^{t}(\mathbf{U}_{0})) - \mathcal{D}(\Phi_{\mathcal{A}}^{t}(\Phi_{\mathcal{D}}^{t}(\mathbf{U}_{0}))))$. Therefore, the local error *e* satisfies the following system

$$\partial_t e = \begin{pmatrix} 0 & H_\mu \\ -\partial_x & 0 \end{pmatrix} e + \begin{pmatrix} 0 & \beta \sqrt{\mu} B_\mu \\ 0 & 0 \end{pmatrix} e + \mathcal{T}_\mu \left(\left(\zeta, v \right), (\eta, w) \right) - \mathcal{R}(t), \tag{3.17}$$

where the operator $\mathcal{T}_{\mu}(\mathbf{U}, \mathbf{V})$ is quadratic and satisfies the following estimate, for $0 \leq s \leq N-1$,

$$\left|\mathcal{T}_{\mu}\left(\left(\zeta, v\right), \left(\eta, w\right)\right)\right|_{H^{s} \times H^{s}} \leq \varepsilon C\left(\frac{1}{\mu_{\min}}, \mu_{\max}, M\right) \left|e\right|_{H^{s+1} \times H^{s+1}}.$$
(3.18)

Then, since $e_{|t=0} = 0$,

$$e(t,\cdot) = \int_0^t \partial_t e(s,\cdot) ds,$$

and since e satisfies (3.17), we obtain, thanks to Estimates (3.16), (3.18) and Lemma 3.4.4,

$$|e(t,\cdot)|_{H^{N-2}\times H^{N-2}} \le C\left(\frac{1}{\mu_{\min}},\mu_{\max},M\right)t.$$
(3.19)

Furthermore, we recall that the Saut-Xu system (3.6) is symmetrizable thanks to the symmetrizer (see Theorem 3.2.2)

$$S = \begin{pmatrix} \frac{D}{\tanh(\sqrt{\mu}D)} & 0\\ 0 & 1 \end{pmatrix}.$$

Therefore, applying S to the system (3.17), and using the fact that $\sqrt{(S,\cdot)}$ is a norm equivalent to the $H^{\frac{1}{2}} \times L^2$ -norm, we obtain, thanks to estimates (3.16), (3.18) and (3.19) and Lemma 3.4.4,

$$\frac{d}{dt}\mathcal{F}(e) \le C\left(\frac{1}{\mu_{\min}}, \mu_{\max}, M\right) \left(\beta \mathcal{F}(e) + \varepsilon t \sqrt{\mathcal{F}(e)}\right),$$

where $\mathcal{F}(e) = \sum_{|\alpha| \le N-4} (S \partial^{\alpha} e, \partial^{\alpha} e)$. Then, we get

$$\mathcal{F}(e)(t) \le \delta C\left(\frac{1}{\mu_{\min}}, \mu_{\max}, M\right) \int_0^t \mathcal{F}(e)(s) + s\sqrt{\mathcal{F}(e)(s)} ds$$

Denoting $\mathcal{M}(t) = \max_{[0,t]} \sqrt{\mathcal{F}(e)(t)}$, we have

$$\mathcal{M}(t) \leq \delta C\left(\frac{1}{\mu_{\min}}, \mu_{\max}, M\right) \int_0^t \mathcal{M}(s) + s ds,$$

and the result follows from the Grönwall's lemma.

3.4.2 Global error estimate

In this part, we prove our main result. We denote by

$$\mathbf{U}_{k} = \left(\mathcal{Y}^{\Delta t}\right)^{k} \left(\mathbf{U}_{0}\right)$$

the numerical solution and by $\mathbf{U}(t_k) := \Phi^{k\Delta t} (\mathbf{U}_0)$ the exact solution at the time $t_k = k\Delta t$. We recall that $\delta = \max(\varepsilon, \beta)$.

Theorem 3.4.6. Let $N \ge 4$, M > 0, ε, β, μ satisfying Condition (3.2) and $b \in L^{\infty}(\mathbb{R})$. Let $\mathbf{U}_0 = (\zeta_0, v_0)$ such that,

$$|b|_{L^{\infty}} + |\mathbf{U}_0|_{H^{N+\frac{1}{2}} \times H^N} \le M.$$

Then, there exist a time $T = T\left(M, \mu_{\max}, \frac{1}{\mu_{\min}}\right) > 0$ and constants $\gamma, \nu, \Delta t_0, C_0 > 0$ such that for all $\Delta t \in]0, \Delta t_0]$ and for all $n \in \mathbb{N}$ such that $0 \le n\Delta t \le \frac{T}{\delta}$,

$$\left|\mathbf{U}_{n}\right|_{H^{N+\frac{1}{2}}\times H^{N}} \leq \nu \text{ and } \left|\Phi^{n\Delta t}\left(\mathbf{U}_{0}\right) - \left(\mathcal{Y}^{\Delta t}\right)^{n}\left(\mathbf{U}_{0}\right)\right|_{H^{N-4+\frac{1}{2}}\times H^{N-4}} \leq \gamma\Delta t$$

Proof. The proof is based on a Lady's Windermere's fan argument and is similar to the one in [32]. In order to simplify the notations, we forget the dependence on $\frac{1}{\mu_{\min}}$ and μ_{\max} in all the constants. We denote by X^N the following space

$$X^N = H^{N + \frac{1}{2}} \times H^N.$$

Thanks to Theorem 3.2.2, we know that there exists a time $T_0(M) > 0$, such that $\Phi^t \mathbf{U}_0$ exists for all $t \in \left[0, \frac{T_0(M)}{\delta}\right]$ and there exists ρ such that, for all $t \in \left[0, \frac{T_0(M)}{\delta}\right]$,

$$\left|\mathbf{U}(t_k)\right|_{X^N} = \left|\Phi^t\left(\mathbf{U}_0\right)\right|_{X^N} \le \rho.$$

We prove by induction that there exists $\Delta t_0, \gamma, \nu$ such that if $0 < \Delta t \leq \Delta t_0$, for all $n \in \mathbb{N}$ with $n\Delta t \leq \frac{T}{\delta}$,

(i)
$$|\mathbf{U}_n|_{X^N} \leq 2M$$
,
(ii) $|\mathbf{U}_n|_{X^N} \leq e^{C_3(2M)\delta n\Delta t} |\mathbf{U}_0|_{X^N}$,
(iii) $|\mathbf{U}_n - \mathbf{U}(t_n)|_{X^{N-4}} \leq \gamma \Delta t$,

with

$$\begin{split} T &= \min\left(T_0(M), \frac{\ln(2)}{C_3(2M)}\right), \, \Delta t_0 = \min\left(T_0(2M), T_3(2M), \frac{\ln(2)}{C_0(2M)}\right), \\ \gamma &= T \max(K, 1) C_4(\max(2M, \rho)), \end{split}$$

and where K = K(4M) is a constant from Inequality (3.10) and C_0 , T_0 , C_3 , T_3 , C_4 are constants from Theorem 3.2.2, Remark 3.3.4 and Proposition 3.4.5. The above properties are satisfied for n = 0. Let $n \ge 1$, and suppose that the induction assumption is true for $0 \le k \le n - 1$. First, using Remark 3.3.4 and the induction assumption, we have, since $\Delta t \le T_3(2M)$,

$$|\mathbf{U}_{n}|_{X^{N}} = \left| \mathcal{Y}^{\Delta t} \left(\mathbf{U}_{n-1} \right) \right|_{X^{N}} \le e^{\delta C_{3}(2M)\Delta t} \left| \mathbf{U}_{n-1} \right|_{X^{N}} \le e^{C_{3}(2M)\delta n\Delta t} \left| \mathbf{U}_{0} \right|_{X^{N}}.$$

Inequality (i) follows from the fact that $\delta n \Delta t \leq T \leq \frac{\ln(2)}{C_3(2M)}$. Furthermore, we have the following telescopic series (see [68] or [32])

$$\mathbf{U}_{n} - \mathbf{U}(t_{n}) = \sum_{0 \le k \le n-1} \Phi^{(n-k-1)\Delta t} \circ \mathcal{Y}^{\Delta t} \left(\mathbf{U}_{k}\right) - \Phi^{(n-k-1)\Delta t} \circ \Phi^{\Delta t} \left(\mathbf{U}_{k}\right).$$
(3.20)

For $k \leq n-2$, since $\mathcal{Y}^{\Delta t}\mathbf{U}_k = \mathbf{U}_{k+1}$, using the induction assumption, we have

$$\left|\mathcal{Y}^{\Delta t}\left(\mathbf{U}_{k}\right)\right|_{X^{N-3}} \leq 2M,$$

and from Theorem 3.2.2, since $\Delta t \leq \min\left(T_0(2M), \frac{\ln(2)}{C_0(2M)}\right)$, we get

$$\left|\Phi^{\Delta t}\left(\mathbf{U}_{k}\right)\right|_{X^{N-3}} \leq e^{C_{0}(2M)\Delta t} |\mathbf{U}_{k}|_{X^{N-3}} \leq 4M,$$

Therefore, from Proposition 3.2.4 and up to replacing K = K(4M) with $\max(K, 1)$, we obtain, for $k \leq n-1$ and $n\Delta t \leq \frac{T}{\delta}$,

$$\left|\Phi^{(n-k-1)\Delta t}\circ\mathcal{Y}^{\Delta t}\left(\mathbf{U}_{k}\right)-\Phi^{(n-k-1)\Delta t}\circ\Phi^{\Delta t}\left(\mathbf{U}_{k}\right)\right|_{X^{N-4}}\leq K\left|\mathcal{Y}^{\Delta t}\left(\mathbf{U}_{k}\right)-\Phi^{\Delta t}\left(\mathbf{U}(t_{k})\right)\right|_{X^{N-4}}.$$

Then, using Proposition 3.4.5 and Inequality (i), we infer

$$\left|\Phi^{(n-k-1)\Delta t}\circ\mathcal{Y}^{\Delta t}\left(\mathbf{U}_{k}\right)-\Phi^{(n-k-1)\Delta t}\circ\Phi^{\Delta t}\left(\mathbf{U}_{k}\right)\right|_{X^{N-4}}\leq\delta C_{4}(\max(2M,\rho))K(\Delta t)^{2}.$$

Therefore, using the telescopic series (3.20), we get

$$|\mathbf{U}_n - \mathbf{U}(t_n)|_{X^{N-4}} \le nC_4(\max(2M,\rho))K\delta(\Delta t)^2 \le C_4(\max(2M,\rho))KT\Delta t.$$

Remark 3.4.7. The method proposed in this chapter can be used for others equations. For instance, we can extend the work [68] by considering equations of the form

$$\partial_t u = iP(D)u + \varepsilon u \partial_x u + \varepsilon u ig(D)u, \qquad (3.21)$$

where u is a real function, P is a real polynomial and g a smooth real function satisfying, for all $\xi \in \mathbb{R}$,

$$P(\xi) = -P(-\xi), \ g(\xi) = -g(-\xi), \ \exists C > 0, \ |g'(\xi)| \le C.$$
(3.22)

We split (3.21) in two parts : the transport part,

$$\partial_t u = \varepsilon u \partial_x u, \tag{3.23}$$

that we can compute using a Lax Wendroff scheme, and the non local part,

$$\partial_t u = iP(D)u + \varepsilon uig(D)u, \tag{3.24}$$

that we can compute using a pseudo-spectral method. Thanks to Conditions (3.22), energy estimates and a commutator estimate for g(D) similar than Proposition B.4.2 (see for instance [79] for such a result), we can show the wellposedness of Equations (3.21) and (3.24) over a time $\frac{T}{\varepsilon}$. Then, proceeding as in this section, we can obtain a similar result than Theorem 3.4.6.

3.5 Numerical experiments

The aim of this section is to numerically verify the Lie method convergence rate in $\mathcal{O}(\Delta t)$ for the Saut-Xu system (3.6) and to illustrate some physical phenomena. To solve the dispersive equation (3.12), discrete Fourier transform is used and for the transport equation (3.11), we consider a Lax-Wendroff scheme. In the both cases, we use a Euler method for the time integration. For the latter problem, we have to be careful of the numerical instability, and that why the time and the space steps are chosen in a way that the classical CFL condition is satisfied.

In others works and particularly on the whole water waves problem (see for example [47], [66], [111] and references therein), several authors use a discrete Fourier transform even for the transport part. They observe spurious oscillations in the wave profile that lead to instabilities. These errors seem to appear when they evaluate the nonlinear part via Fourier transform because additional terms appear in the approximation, this is the aliasing phenomenon. To fix this problem, they apply at every time step a low-pass filter. The main interest of our scheme is that we do not need one.

3.5.1 Example 1: Convergence curve

In this example, we consider the following initial data:

$$\zeta_0(x) = \operatorname{sech}\left(\frac{\sqrt{3}}{2}x\right), \quad v_0 = \zeta_0. \tag{3.25}$$

with two different bathymetries: a bump and a ripple bottom. Note that in order to avoid numerical reflections due to the boundaries and justify of the use of the Fast Fourier Transform, we decide to take rapidly decreasing initial data. Figures 3.1 and 3.2 display the evolution for different times of the free surface ζ for these two test cases. We take $\varepsilon = 0.1, \mu = 1, \beta = \frac{1}{2}$ and the final time T = 10.

Figures 3.3 displays the convergence curve for this example. We plot the logarithm of the error (in norm $H^1 \times L^2$) in function of the logarithm of the time step Δt . The convergence numerical order is then given by the slope of this curve. For reference, a small line (the dashed line) of slope one is added in this figure. We see that the numerical rate of convergence is greater than 1.

3.5.2 Example 2: Linear versus nonlinear

In this example, we compare the effect of the nonlinear terms on the evolution of the free surface. The linear case corresponds to $\varepsilon = 0$ and for the nonlinear, we take $\varepsilon = 0.1$. We take $\mu = 1$ and the initial data (3.25). Figure 3.4 displays the evolution of the free surface for T = 15. We notice that the leading wave in the nonlinear solution is higher than in the linear one. This fact is also noticed in [66] (Section 4.5) when the authors study the propagation of a tsunami.



Figure 3.1: Upper: Evolution of the free surface for different times. Lower: bottom topography and initial condition.



Figure 3.2: Upper: Evolution of the free surface for different times. Lower: bottom topography and initial condition.

3.5.3 Example 3: Boussinesq regime

In the shallow water regime (μ small), there is a huge literature for asymptotic models (see for instance [80]). Among all these asymptotic models, we have the KdV equation. It is a model obtained under the Boussinesq regime, i.e. when $\varepsilon = \mu$, $\beta = 0$ and μ small. In the following, we formally derive a KdV equation from the Saut-Xu equations and we give numerical simulations in this setting.



Figure 3.3: Convergence curve for the Lie method for two bottoms: bump (red line) and ripple bottom (blue line).



Figure 3.4: Evolution of the free surface for $\varepsilon = 0$ (red) and $\varepsilon = 0.1$ (blue).

We recall that, without the assumption $\nu = \frac{1}{\mu}$, the Saut-Xu equations are given by the system (3.5). Notice also that

$$\mathcal{H}_{\mu} = -\sqrt{\mu}\partial_x - \frac{1}{3}\mu^{\frac{3}{2}}\partial_x^3 + \mathcal{O}(\mu^2).$$
(3.26)

Then if we assume that $\mu = \varepsilon$, $\nu = 1$ (since $\nu \sim 1$ if μ is small) and we drop all the terms of order $\mathcal{O}(\mu^2)$ in System (3.5), we obtain the following equations



Figure 3.5: Evolution of the soliton at different times t = 0, 3, 6, 9 ($\varepsilon = 0.01$).

$$\begin{cases} \partial_t \zeta + \partial_x v + \mu v \partial_x \zeta - \frac{1}{2} \mu^{\frac{3}{2}} v \partial_x \zeta + \frac{1}{3} \mu \partial_x^3 v + \mu \zeta \partial_x v = 0, \\ \partial_t v + \partial_x \zeta + \mu v \partial_x v + \mu^{\frac{3}{2}} \frac{1}{2} v \partial_x v = 0. \end{cases}$$
(3.27)

Formally, the solutions of this system are close to the solutions of (3.5) with an accuracy of order $\mathcal{O}(\mu^2)$. Notice that this system is not a standard Boussinesq system (in the sense of [17] or [80]) because of our nonlinear change of variables (3.4). Using the approach developed in [129], [19], [9] (see also Part 7.1.1 in [80]) we can check that, formally, the following KdV equation is an asymptotic model of the system (3.27)

$$\partial_{\tau}f + \frac{3}{2}f\partial_{\xi}f + \frac{1}{6}\partial_{\xi}^{3}f = 0.$$
 (3.28)

This means that if we solve (3.27) with the initial data (f_0, f_0) and (3.28) with the initial datum f_0 , the solution $(\zeta, v) (t, x)$ of (3.27) is close to $(f, f) (\mu t, x - t)$. Furthermore, if we take $f_0(x) = \alpha \operatorname{sech}^2 \left(\sqrt{\frac{3}{4}\alpha x}\right)$, the solution f of the KdV equation with this initial datum is the soliton $f(\tau, \xi) = f_0(\xi - c\tau)$ with $c = \frac{\alpha}{2}$. Hence, in this case, the solution of (3.27) and (3.5) are close to a soliton.

In the following we check that the solution to (3.6) is indeed close to the KdV solution when μ is small. We simulate one soliton. We take $\varepsilon = \mu = 0.01$, $\alpha = 1$ and the final time is T = 10. Figure 3.5 represents the evolution of this soliton at different times.

In deep water (μ not small), the KdV approximation ceases to be a good approximation. In order to get some insight on the range of validity of the KdV approximation, we compare in Figure 3.6 the solution of (3.6) to the exact soliton after a time T = 10 for various values of μ . We notice that even for $\mu = 0.1$ and a final time $T = \frac{1}{\mu}$, the KdV approximation remains a good approximation of the Saut-Xu system.

Remark 3.5.1. Notice that in Section 3.3, we crucially use the fact that μ is bounded from below. Therefore, we do not have a proof of the convergence of our scheme in the shallow water regime. However, we see that our scheme also works in this context.



Figure 3.6: Difference after a time T = 10 between a real soliton and a soliton generated by our scheme with the same initial data for different values of $\varepsilon = \mu$. Abscissa : value of ε ; Ordinate : difference after a final time T = 10.

3.5.4 Example 4: Rapidly varying topographies

In this example we study the evolution of water waves over a rapidly varying periodic bottom. We assume that $\mu = 1$. This problem is linked to the Bragg reflection phenomenon (see for instance [100], [91], [66]). We take

$$\zeta_0 = v_0 = \left(\operatorname{sech}\left(\frac{\sqrt{3}}{2}x\right)\right)^2 \text{ and } b(x) = \cos(\alpha x).$$
(3.29)

Figure 3.7 compares the evolution of water waves when we take $\alpha = 10$ (blue line) and when we take b(x) = 0 (blue line). Figure 3.8 displays the difference between the case of a flat bottom and the case of a bottom of the form $b(x) = \cos(\alpha x)$ for different values of α . We observe an homogenization effect when α is large. It seems that when α goes to infinity, the solution of the Saut-Xu equations converges to the solution of the Saut-Xu equations with a flat bottom (corresponding to the mean of b). Notice that this result is different from what we could see in the literature (for instance [40] or [46]), since we take a bottom of the form $b(x) = \cos(\alpha x)$ and not of the form $b(x) = \frac{1}{\alpha} \cos(\alpha x)$. Our numerical simulations suggest therefore a homogenization effect for large amplitude bottom variations that has not been investigated so far.



Figure 3.7: Comparison between the evolution of a water wave (blue line) over a bottom of the form $b(x) = \cos(10x)$ (dashed line) and the evolution of a water wave over a flat bottom (red line). $\varepsilon = 0.05$, $\beta = 0.5$.



Figure 3.8: Difference between a water wave over a rapidly varying topography $b(x) = \cos(\alpha x)$ and a water wave over a flat bottom. Abscissa : value of α ; Ordinate : difference after a final time T = 10.

Chapter 4

Coriolis effect on water waves

Sommaire

4.1	Intro	oduction
	4.1.1	Presentation of the problem
	4.1.2	Existence result
	4.1.3	Notations for this chapter
4.2	The	div-curl problem
	4.2.1	Notations for this part
	4.2.2	Existence and uniqueness
	4.2.3	The transformed div-curl problem
	4.2.4	Time derivatives and few remarks about the good unknown 108
4.3	Well	-posedness of the water waves equations
	4.3.1	Framework
	4.3.2	A priori estimate for the vorticity
	4.3.3	Quasilinearization and a priori estimates
	4.3.4	Existence result
4.4	Fron	n the Castro-Lannes system to the Euler equations, recon-
	strue	ction of the pressure
4.5	The	nonlinear shallow water equations
	4.5.1	The context
	4.5.2	WKB expansion with respect to μ
	4.5.3	Rigorous derivation
	4.5.4	The Proudman resonance, linear properties of the shallow water equa-
		tions $\ldots \ldots \ldots$

Ce chapitre est basé sur l'article [101]. Nous avons ajouté la section 4.4 où nous montrons qu'à partir d'une solution régulière de la formulation de Castro-Lannes nous pouvons reconstruire la pression à l'intérieur du fluide et retrouver la solution du système d'Euler à surface libre correspondante. Nous avons aussi ajouté la soussection 4.5.4 où nous montrons que, lorsque d = 2 et en présence de la force de Coriolis, la résonance de Proudman n'est pas possible à cause des effets dispersives.

4.1 Introduction

4.1.1 Presentation of the problem

This chapter is devoted to the study of water waves under the influence of the gravity and the Coriolis force. It is quite common in the physical literature that the rotating shallow water equations are used to study such water waves. We prove a local wellposedness theorem for the water waves equations with vorticity and Coriolis force, taking into account the dependence on various physical parameters and we justify rigorously the shallow water model. We also consider a possible non constant pressure at the surface that can be used to describe meteorological disturbances such as storms or pressure jumps for instance.

There has been a lot of interest on the Cauchy problem for the irrotational water waves problem since the work of Wu ([150] and [151]). More relevant for our present work is the Eulerian approach developed by Lannes ([78]) in the presence of a bottom. Another program initiated by Craig ([43]) consists in justifying the use of the many asymptotic models that exist in the physical literature to describe the motion of water waves. This requires a local wellposedness result that is uniform with respect to the small parameters involved (typically, the shallow water parameter). This was achieved by Alvarez-Samaniego and Lannes (9) for many regimes; other references in this direction are ([128], [129], [70]). The irrotational framework is however not always the relevant one. When dealing with wave-current interactions or, at larger scales, if one wants to take into account the Coriolis force. The latter configuration motivates the present study. Several authors considered the local wellposedness theory for the water waves equations in the presence of vorticity ([42], [90], [155], [156]). Recently, Castro and Lannes proposed a generalization of the Zakharov-Craig-Sulem formulation (see [153], [47], [48], [80], [3] for an explanation of this formulation), and gave a system of three equations that allow for the presence of vorticity. Then, they used it to derive new shallow water models that describe wave current interactions and more generally the coupling between waves and vorticity effects ([34] and [33]). In this chapter, we base our study on their formulation.

This chapter is organized in four parts : firstly we derive a generalization of the Castro-Lannes formulation (4.20) that takes into account the Coriolis forcing as well as non flat bottoms and a non constant pressure at the surface; secondly, we prove a local wellposedness result taking account the dependence of small parameters; thirdly, we show how we can reconstruct the pressure in the fluid domain from the Castro-Lannes formulation; Finally, we justify that the rotational shallow water model is a good asymptotic model of the rotational water waves equations under a Coriolis forcing.

We model the sea by an incompressible ideal fluid bounded from below by the seabed and from above by a free surface. We suppose that the seabed and the surface are graphs above the still water level. The pressure at the surface is of the form $P + P_{\text{ref}}$ where $P(t, \cdot)$ models a meteorological disturbance and P_{ref} is a constant which represents the pressure far from the

meteorological disturbance. We denote by d the horizontal dimension, which is equal to 1 or 2. The horizontal variable is $X \in \mathbb{R}^d$ and $z \in \mathbb{R}$ is the vertical variable. H is the typical water depth. The water occupies the domain $\Omega_t := \{(X, z) \in \mathbb{R}^{d+1}, -H + b(X) < z < \zeta(t, X)\}$. The water is homogeneous (constant density ρ), inviscid with no surface tension. We denote by **U** the velocity of the fluid, **V** is the horizontal component of the velocity and **w** its vertical component. The water is under the influence of the gravity $g = -ge_z$ and the rotation of the Earth with a rotation vector $\mathbf{f} = \frac{f}{2} \mathbf{e}_z$. Finally, we define the pressure in the fluid domain by \mathcal{P} . The equations governing the motion of the surface of an ideal fluid under the influence of gravity and Coriolis force are the free surface Euler Coriolis equations ⁽¹⁾

$$\begin{cases} \partial_t \mathbf{U} + (\mathbf{U} \cdot \nabla_{X,z}) \mathbf{U} + \mathbf{f} \times \mathbf{U} = -\frac{1}{\rho} \nabla_{X,z} \mathcal{P} - g \mathbf{e}_{\mathbf{z}} \text{ in } \Omega_t, \\ \text{div } \mathbf{U} = 0 \text{ in } \Omega_t, \end{cases}$$
(4.1)

with the boundary conditions

$$\begin{cases} \partial_t \zeta - \underline{U} \cdot N = 0, \\ U_b \cdot N_b = 0, \end{cases}$$
(4.2)

where $N = \begin{pmatrix} -\nabla\zeta\\ 1 \end{pmatrix}$, $N_b = \begin{pmatrix} -\nabla b\\ 1 \end{pmatrix}$, $\underline{\mathbf{U}} = \begin{pmatrix} \underline{\mathbf{V}}\\ \underline{\mathbf{w}} \end{pmatrix} = \mathbf{U}_{|z=\zeta}$ and $\mathbf{U}_b = \begin{pmatrix} \mathbf{V}_b\\ \mathbf{w}_b \end{pmatrix} = \mathbf{U}_{|z=-H+b}$.

The pressure \mathcal{P} can be decomposed as the surface contribution and the internal pressure

$$\mathcal{P}(t, X, z) = P(t, X) + P_{\text{ref}} + \widetilde{\mathcal{P}}(t, X, z),$$

with $\widetilde{\mathcal{P}}_{|z=\zeta} = 0.$

Remark 4.1.1. In this chapter, we identify functions on \mathbb{R}^2 as function on \mathbb{R}^3 . Then, the gradient, the curl and the divergence operators become in the two dimensional case

$$\nabla_{X,z}f = \begin{pmatrix} \partial_x f \\ 0 \\ \partial_z f \end{pmatrix}, \text{ curl } \mathbf{A} = \begin{pmatrix} -\partial_z \mathbf{A}_2 \\ \partial_z \mathbf{A}_1 - \partial_x \mathbf{A}_3 \\ -\partial_x \mathbf{A}_2 \end{pmatrix}, \text{ div } \mathbf{A} = \partial_x \mathbf{A}_1 + \partial_z \mathbf{A}_3.$$

In order to obtain some asymptotic models we nondimensionalize the previous equations. There are five important physical parameters : the typical amplitude of the surface a, the typical amplitude of the bathymetry a_{bott} , the typical horizontal scale L, the characteristic water depth H and the typical Coriolis frequency f. Then we can introduce four dimensionless parameters

$$\varepsilon = \frac{a}{H}, \ \beta = \frac{a_{\text{bott}}}{H}, \ \mu = \frac{H^2}{L^2} \text{ and } \operatorname{Ro} = \frac{a}{fL}\sqrt{\frac{g}{H}},$$

$$(4.3)$$

where ε is called the nonlinearity parameter, β the bathymetric parameter, μ the shallowness parameter and Ro the Rossby number. We also nondimensionalize the variables and the unknowns. We introduce (see Section 2.2.1 and [34] for an explanation of this nondimensionalization)

$$\begin{cases} X' = \frac{X}{L}, \ z' = \frac{z}{H}, \ \zeta' = \frac{\zeta}{a}, \ b' = \frac{b}{a_{\text{bott}}}, \ t' = \frac{\sqrt{gH}}{L}t, \\ \mathbf{V}' = \sqrt{\frac{H}{g}} \frac{\mathbf{V}}{a}, \ \mathbf{w}' = H\sqrt{\frac{H}{g}} \frac{\mathbf{w}}{aL}, \ P' = \frac{P}{\rho ga} \ \text{and} \ \widetilde{\mathcal{P}}' = \frac{\widetilde{\mathcal{P}}}{\rho gH}. \end{cases}$$
(4.4)

¹We consider that the centrifugal potential is constant and included in the pressure term.

In this chapter, we use the following notations

$$\nabla^{\mu}_{X',z'} = \begin{pmatrix} \sqrt{\mu}\nabla_{X'} \\ \partial_{z'} \end{pmatrix}, \operatorname{curl}^{\mu} = \nabla^{\mu}_{X',z'} \times, \operatorname{div}^{\mu} = \nabla^{\mu}_{X',z'} \cdot.$$
(4.5)

We also define

$$\mathbf{U}^{\mu} = \begin{pmatrix} \sqrt{\mu} \mathbf{V}' \\ \mathbf{w}' \end{pmatrix}, \, \boldsymbol{\omega}' = \frac{1}{\mu} \operatorname{curl}^{\mu} \mathbf{U}^{\mu}, \, \underline{\mathbf{U}}^{\mu} = \mathbf{U}^{\mu}_{|z'=\varepsilon\zeta'}, \, \mathbf{U}^{\mu}_{b} = \mathbf{U}^{\mu}_{|z'=-1+\beta b'}, \tag{4.6}$$

and

$$N^{\mu} = \begin{pmatrix} -\varepsilon \sqrt{\mu} \nabla \zeta' \\ 1 \end{pmatrix}, \ N^{\mu}_{b} = \begin{pmatrix} -\beta \sqrt{\mu} \nabla b' \\ 1 \end{pmatrix}.$$
(4.7)

Notice that our nondimensionalization of the vorticity allows us to consider only weakly sheared flows (see [33], [139], [121]). The nondimensionalized fluid domain is

$$\Omega'_{t'} := \{ (X', z') \in \mathbb{R}^{d+1}, -1 + \beta b'(X') < z' < \varepsilon \zeta'(t', X') \}.$$
(4.8)

Finally, if $\mathbf{V} = \begin{pmatrix} V_1 \\ V_2 \end{pmatrix} \in \mathbb{R}^2$, we define \mathbf{V} by $\mathbf{V}^{\perp} = \begin{pmatrix} -V_2 \\ V_1 \end{pmatrix}$. Then, the Euler Coriolis equations (4.1) become

$$\begin{cases} \partial_{t'} \mathbf{U}^{\mu} + \frac{\varepsilon}{\mu} \left(\mathbf{U}^{\mu} \cdot \nabla^{\mu}_{X',z'} \right) \mathbf{U}^{\mu} + \frac{\varepsilon \sqrt{\mu}}{\mathrm{Ro}} \begin{pmatrix} \mathbf{V}'^{\perp} \\ 0 \end{pmatrix} = -\sqrt{\mu} \begin{pmatrix} \nabla P' \\ 0 \end{pmatrix} - \frac{1}{\varepsilon} \nabla^{\mu}_{X',z'} \widetilde{\mathcal{P}}' - \frac{1}{\varepsilon} \mathbf{e}_{\mathbf{z}} \text{ in } \Omega'_{t}, \\ \operatorname{div}^{\mu}_{X',z'} \mathbf{U}^{\mu} = 0 \text{ in } \Omega'_{t}, \end{cases}$$
(4.9)

with the boundary conditions

$$\begin{cases} \partial_{t'}\zeta' - \frac{1}{\mu}\underline{\mathbf{U}}^{\mu} \cdot N^{\mu} = 0, \\ \mathbf{U}_{b}^{\mu} \cdot N_{b}^{\mu} = 0. \end{cases}$$
(4.10)

In the following we omit the primes. In [34], Castro and Lannes derived a formulation of the water waves equations with vorticity. We outline the main ideas of this formulation and extend it to take into account the Coriolis force. Even in absence of Coriolis forcing, our results extend the result of [34] by allowing non flat bottoms. First, applying the curl^{μ} operator to the first equation of (4.9) we obtain an equation on ω

$$\partial_t \boldsymbol{\omega} + \frac{\varepsilon}{\mu} \left(\mathbf{U}^{\mu} \cdot \nabla^{\mu}_{X,z} \right) \boldsymbol{\omega} = \frac{\varepsilon}{\mu} \boldsymbol{\omega} \cdot \nabla^{\mu}_{X,z} \mathbf{U}^{\mu} + \frac{\varepsilon}{\mu \text{Ro}} \partial_z \mathbf{U}^{\mu}.$$
(4.11)

Furthermore, taking the trace at the surface of the first equation of (4.9) we get

$$\partial_t \underline{\mathbf{U}}^{\mu} + \varepsilon \left(\underline{\mathbf{V}} \cdot \nabla_X\right) \underline{\mathbf{U}}^{\mu} + \frac{\varepsilon \sqrt{\mu}}{\mathrm{Ro}} \left(\begin{array}{c} \underline{\mathbf{V}}^{\perp} \\ 0 \end{array} \right) = -\sqrt{\mu} \left(\begin{array}{c} \nabla P \\ 0 \end{array} \right) - \frac{1}{\varepsilon} \left(\begin{array}{c} 0 \\ 1 \end{array} \right) - \frac{1}{\varepsilon} \left(\partial_z \widetilde{\mathcal{P}} \right)_{|z=\varepsilon\zeta} N^{\mu}.$$
(4.12)

Then, in order to eliminate the term $\left(\partial_z \widetilde{\mathcal{P}}\right)_{|z=\varepsilon\zeta} N^{\mu}$, we have to introduce the following quantity. If **A** is a vector field on Ω_t , we define $\mathbf{A}_{\mathbb{M}}$ as

$$\mathbf{A}_{/\!\!/} = \frac{1}{\sqrt{\mu}} \underline{\mathbf{A}}_h + \varepsilon \underline{\mathbf{A}}_v \nabla \zeta, \qquad (4.13)$$

where A_h is the horizontal component of A, A_v its vertical component, $\underline{A} = \mathbf{A}_{|z=\varepsilon\zeta}$ and $A_b = \mathbf{A}_{|z=-1+\beta b}$. Notice that,

$$\underline{\mathbf{A}} \times N^{\mu} = \sqrt{\mu} \begin{pmatrix} -\mathbf{A}_{\mathbb{A}}^{\perp} \\ -\varepsilon \sqrt{\mu} \mathbf{A}_{\mathbb{A}}^{\perp} \cdot \nabla \zeta \end{pmatrix}.$$
(4.14)

Therefore, taking the orthogonal of the horizontal component of the vectorial product of (4.12) with N^{μ} we obtain

$$\partial_t \mathbf{U}^{\mu}_{\mathbb{I}} + \nabla\zeta + \frac{\varepsilon}{2} \nabla \left| \mathbf{U}^{\mu}_{\mathbb{I}} \right|^2 - \frac{\varepsilon}{2\mu} \nabla \left[\left(1 + \varepsilon^2 \mu \left| \nabla \zeta \right|^2 \right) \underline{\mathbf{w}}^2 \right] + \varepsilon \left(\underline{\omega} \cdot N^{\mu} + \frac{1}{\mathrm{Ro}} \right) \underline{\mathbf{V}}^{\perp} = -\nabla P. \quad (4.15)$$

Since $\mathrm{U}_{/\!\!/}^{\mu}$ is a vector field on $\mathbb{R}^2,$ we have the classical Hodge-Weyl decomposition

$$\mathbf{U}_{\mathbb{I}}^{\mu} = \nabla \frac{\nabla}{\Delta} \cdot \mathbf{U}_{\mathbb{I}}^{\mu} + \nabla^{\perp} \frac{\nabla^{\perp}}{\Delta} \cdot \mathbf{U}_{\mathbb{I}}^{\mu}.$$
(4.16)

In the following we denote by $\psi := \frac{\nabla}{\Delta} \cdot U_{\parallel}$ and $\tilde{\psi} := \frac{\nabla^{\perp}}{\Delta} \cdot U_{\parallel}$ ⁽²⁾. Applying the operator $\frac{\nabla}{\Delta}$ to (4.15), we obtain

$$\partial_t \psi + \zeta + \frac{\varepsilon}{2} \left| \mathbf{U}_{/\!/}^{\mu} \right|^2 - \frac{\varepsilon}{2\mu} \left(1 + \varepsilon^2 \mu \left| \nabla \zeta \right|^2 \right) \underline{\mathbf{w}}^2 + \varepsilon \frac{\nabla}{\Delta} \cdot \left[\left(\underline{\omega} \cdot N^{\mu} + \frac{1}{\mathrm{Ro}} \right) \underline{\mathbf{V}}^{\perp} \right] = -P.$$
(4.17)

Moreover, using the following vectorial identity

$$\left(\nabla^{\mu}_{X,z} \times \mathbf{U}^{\mu}\right)_{|z=\varepsilon\zeta} \cdot N^{\mu} = \mu \nabla^{\perp} \cdot \mathbf{U}^{\mu}_{/\!\!/},\tag{4.18}$$

we have

$$\Delta \widetilde{\psi} = (\underline{\omega} \cdot N^{\mu}) \,. \tag{4.19}$$

We can now give the nondimensionalized Castro-Lannes formulation of the water waves equations with vorticity in the presence of Coriolis forcing. It is a system of three equations for the unknowns (ζ, ψ, ω)

$$\begin{cases} \partial_t \zeta - \frac{1}{\mu} \underline{\mathbf{U}}^{\mu} \cdot N^{\mu} = 0, \\ \partial_t \psi + \zeta + \frac{\varepsilon}{2} \left| \mathbf{U}_{\mathscr{I}}^{\mu} \right|^2 - \frac{\varepsilon}{2\mu} \left(1 + \varepsilon^2 \mu \left| \nabla \zeta \right|^2 \right) \underline{\mathbf{w}}^2 + \varepsilon \frac{\nabla}{\Delta} \cdot \left[\left(\underline{\omega} \cdot N^{\mu} + \frac{1}{\mathrm{Ro}} \right) \underline{\mathbf{V}}^{\perp} \right] = -P, \\ \partial_t \omega + \frac{\varepsilon}{\mu} \left(\mathbf{U}^{\mu} \cdot \nabla_{X,z}^{\mu} \right) \omega = \frac{\varepsilon}{\mu} \left(\omega \cdot \nabla_{X,z}^{\mu} \right) \mathbf{U}^{\mu} + \frac{\varepsilon}{\mu \mathrm{Ro}} \partial_z \mathbf{U}^{\mu}, \end{cases}$$
(4.20)

where $\mathbf{U}^{\mu} := \mathbf{U}^{\mu}[\varepsilon\zeta,\beta b](\psi,\boldsymbol{\omega})$ is the unique solution in $H^{1}(\Omega_{t})$ of

$$\begin{cases} \operatorname{curl}^{\mu} \mathbf{U}^{\mu} = \mu \boldsymbol{\omega} \text{ in } \Omega_{t}, \\ \operatorname{div}^{\mu} \mathbf{U}^{\mu} = 0 \text{ in } \Omega_{t}, \\ U^{\mu}_{\mathbb{J}} = \nabla \psi + \frac{\nabla^{\perp}}{\Delta} \left(\underline{\omega} \cdot N^{\mu} \right), \\ U^{\mu}_{b} \cdot N^{\mu}_{b} = 0. \end{cases}$$

$$(4.21)$$

 $^{^{2}}$ We define rigorously these operators in Section 4.2.1.

We add a technical assumption. We assume that the water depth is bounded from below by a positive constant

$$\exists h_{\min} > 0, \, \varepsilon \zeta + 1 - \beta b \ge h_{\min}. \tag{4.22}$$

We also suppose that the dimensionless parameters satisfy

$$\exists \mu_{\max}, \ 0 < \mu \le \mu_{\max}, \ 0 < \varepsilon \le 1, \ 0 < \beta \le 1 \text{ and } \frac{\varepsilon}{Ro} \le 1.$$
(4.23)

Remark 4.1.2. The assumption $\varepsilon \leq Ro$ is equivalent to $fL \leq \sqrt{gH}$. This means that the typical rotation speed due to the Coriolis force is less than the typical water wave celerity. For water waves, this assumption is common (see for instance [117], [61]). Typically for offshore long water waves at mid-latitudes, we have L = 100 km and H = 1 km and $f = 10^{-4} Hz$. Then, $\frac{\varepsilon}{Ro} = 10^{-1}$.

4.1.2 Existence result

In this part, we give our main result. It is a wellposedness result for the system (4.46) which is a straightened system of the Castro-Lannes formulation. This result extends Theorem 4.7 and Theorem 5.1 in [34] by adding a non flat bottom and a Coriolis forcing. We define the energy \mathcal{E}^N in Subsection 4.3.1.

Theorem 4.1.3. Assume that the initial data, b and P are smooth enough and the initial vorticity is divergent free. Assume also that Conditions (4.22) and (4.54) are satisfied initially. Then, there exists T > 0, and a unique solution to the water waves equations (4.46) on [0, T]. Moreover,

$$T = \min\left(\frac{T_0}{\max(\varepsilon, \beta, \frac{\varepsilon}{\operatorname{Ro}})}, \frac{T_0}{|\nabla P|_{L_t^{\infty} H_X^N}}\right) \ , \ \frac{1}{T_0} = c^1 \text{ and } \sup_{t \in [0,T]} \mathcal{E}^N(\zeta(t), \psi(t), \omega(t)) = c^2,$$

where c^{j} is a constant which depends on the initial conditions, P and b.

A full version is given in Subsection 4.3.4. This theorem allows us to investigate the justification of asymptotic models in the presence of Coriolis forcing. In the case of a constant pressure at the surface and without a Coriolis forcing, our existence time is similar to Theorem 3.16 in [80] (see also [9]); without a Coriolis forcing, it is as Theorem 2.2.3 in the second chapter.

4.1.3 Notations for this chapter

- If $\mathbf{A} \in \mathbb{R}^3$, we denote by \mathbf{A}_h its horizontal component and by \mathbf{A}_v its vertical component.

- If
$$\mathbf{V} = \begin{pmatrix} V_1 \\ V_2 \end{pmatrix} \in \mathbb{R}^2$$
, we define the orthogonal of \mathbf{V} by $\mathbf{V}^{\perp} = \begin{pmatrix} -V_2 \\ V_1 \end{pmatrix}$

- In this chapter, $C\left(\cdot\right)$ is a nondecreasing and positive function whose exact value has no importance.

- Consider a vector field **A** or a function **w** defined on Ω . Then, we denote $\mathbf{A} = \mathbf{A} \circ \Sigma$ and $\mathbf{w} = \mathbf{w} \circ \Sigma$, where Σ is defined in (4.40). Furthermore, we denote $\underline{\mathbf{A}} = \mathbf{A}_{|z=\varepsilon\zeta} = \mathbf{A}_{|z=0}$, $\underline{\mathbf{w}} = \mathbf{w}_{|z=\varepsilon\zeta} = \mathbf{w}_{|z=0}$ and $\mathbf{A}_b = \mathbf{A}_{|z=-1+\beta b} = \mathbf{A}_{|z=-1}$, $\mathbf{w}_b = \mathbf{w}_{|z=-1+b} = \mathbf{w}_{|z=-1}$.

- If $s \in \mathbb{R}$ and f is a function on \mathbb{R}^d , $|f|_{H^s}$ is its H^s -norm and $|f|_2$ is its L^2 -norm. The quantity $|f|_{W^{k,\infty}}$ is $W^{k,\infty}(\mathbb{R}^d)$ -norm of f, where $k \in \mathbb{N}^*$, and $|f|_{L^{\infty}}$ its $L^{\infty}(\mathbb{R}^d)$ -norm.

- The operator (,) is the L^2 -scalar product in \mathbb{R}^d .

- If $N \in \mathbb{N}^*$, **A** is defined on Ω and $A = \mathbf{A} \circ \Sigma$, $||A||_{H^N}$ and $||\mathbf{A}||_{H^N}$ are respectively the $H^N(\mathcal{S})$ -norm of A and the $H^N(\Omega)$ -norm of **A**. The L^p -norm are denoted $||\cdot||_p$.

- The norm $||\cdot||_{H^{s,k}}$ is defined in Definition 4.2.10.

- The space $H^s_*(\mathbb{R}^d)$, $\dot{H}^s(\mathbb{R}^d)$ and $H_b(\operatorname{div}_0^{\mu}, \Omega)$ are defined in Subsection 4.2.1.
- If f is a function defined on \mathbb{R}^d , we denote ∇f the gradient of f.

- If \mathbf{w} is a function defined on Ω , $\nabla_{X,z}\mathbf{w}$ is the gradient of \mathbf{w} and $\nabla_X\mathbf{w}$ its horizontal component. We have the same definition for functions defined on \mathcal{S} .

- \mathfrak{P} , Λ and M_N are defined in Subsection 4.2.1.

4.2 The div-curl problem

In [34], A. Castro and D. Lannes study the system (4.21) in the case of a flat bottom (b = 0). The purpose of this part is to extend their results in the case of a non flat bottom.

4.2.1 Notations for this part

In this chapter, we use the Beppo-Levi spaces (see [53])

$$\forall s \ge 0, \, \dot{H}^s(\mathbb{R}^d) = \left\{ f \in L^2_{\text{loc}}(\mathbb{R}^d), \, \nabla f \in H^{s-1}(\mathbb{R}^d) \right\} \text{ and } |\cdot|_{\dot{H}^s} = |\nabla \cdot|_{H^{s-1}}.$$

The dual space of $\dot{H}^{s}(\mathbb{R}^{d})/\mathbb{R}$ is the space (see [29])

$$H_*^{-s}(\mathbb{R}^d) = \left\{ u \in H^{-s}(\mathbb{R}^d), \, \exists v \in H^{-s+1}(\mathbb{R}^d), \, u = |D| \, v \right\} \text{ and } \left| \cdot \right|_{H_*^{-s}} = \left| \frac{\cdot}{|D|} \right|_{H^{-s+1}}.$$

Notice that $\dot{H}^1(\mathbb{R}^d)/\mathbb{R}$ is a Hilbert space (see Proposition A.1.1). Then, we can rigorously define the Hodge-Weyl decomposition and the operators $\frac{\nabla}{\Delta}$ and $\frac{\nabla^{\perp}}{\Delta}$. For $f \in L^2(\mathbb{R}^d)^d$, $u = \frac{\nabla}{\Delta} \cdot f$ is defined as the unique solution, up to a constant, in $\dot{H}^1(\mathbb{R}^d)$ of the variational problem

$$\int_{\mathbb{R}^d} \nabla u \cdot \nabla \phi = \int_{\mathbb{R}^d} f \cdot \nabla \phi \,, \, \forall \phi \in \dot{H}^1(\mathbb{R}^d).$$

The operator $\frac{\nabla^{\perp}}{\Delta}$ can be defined similarly. Then, it is easy to check that the operators $\frac{\nabla^{\perp}}{\Delta}$ and $\frac{\nabla^{\perp}}{\Delta}$ belong to $\mathcal{L}\left(H^{s}(\mathbb{R}^{d})^{d}, \dot{H}^{s+1}(\mathbb{R}^{d})\right)$, for all $s \geq 0$.

The subspace of $L^2(\Omega)^3$ of functions whose rotationnal is in $L^2(\Omega)^3$ is the space

$$H(\operatorname{curl}^{\mu},\Omega) = \left\{ \mathbf{A} \in L^{2}(\Omega)^{3}, \operatorname{curl}^{\mu} \mathbf{A} \in L^{2}(\Omega)^{3} \right\}.$$

The subspace of $L^2(\Omega)^3$ of divergence free vector fields is the space

$$H(\operatorname{div}_{0}^{\mu},\Omega) = \left\{ \mathbf{A} \in L^{2}(\Omega)^{3}, \operatorname{div}^{\mu}\mathbf{A} = 0 \right\}$$

Remark 4.2.1. Notice that $\mathbf{A} \in H(\operatorname{div}_{0}^{\mu}, \Omega)$ implies that $(\mathbf{A}_{|\partial\Omega} \cdot n)$ belongs to $H^{-\frac{1}{2}}(\partial\Omega)$ and $\mathbf{A} \in H(\operatorname{curl}^{\mu}, \Omega)$ implies that $(\mathbf{A}_{|\partial\Omega} \times n)$ belongs to $H^{-\frac{1}{2}}(\partial\Omega)$ (see [51]).

Finally, we define $H_b(\operatorname{div}_0^{\mu}, \Omega)$ as

$$H_b(\operatorname{div}_0^{\mu}, \Omega) = \left\{ \mathbf{A} \in H(\operatorname{div}_0^{\mu}, \Omega), \, A_b \cdot N_b^{\mu} \in H_*^{-\frac{1}{2}}(\mathbb{R}^d) \right\}.$$

Remark 4.2.2. We have a similar equation to (4.18) at the bottom

$$\frac{1}{\mu} \left(\nabla^{\mu}_{X,z} \times \mathbf{U}^{\mu} \right)_{|z=-1+\beta b} \cdot N^{\mu}_{b} = \nabla^{\perp} \cdot \left(\mathbf{V}_{b} + \beta \mathbf{w}_{b} \nabla b \right),$$

hence, in the following, we suppose that $\boldsymbol{\omega} \in H_b(\operatorname{div}_0^{\mu}, \Omega)$.

We define \mathfrak{P} and Λ as the Fourier multiplier in $\mathcal{S}'(\mathbb{R}^d)$,

$$\mathfrak{P} = \frac{|D|}{\sqrt{1 + \sqrt{\mu} |D|}} \text{ and } \Lambda = \sqrt{1 + |D|^2}.$$

Then it is important to notice that, if $\boldsymbol{\omega} \in H_b(\operatorname{div}_0^{\mu}, \Omega)$, the quantity $\frac{1}{\mathfrak{P}}(\omega_b \cdot N_b^{\mu})$ makes sense and belongs to $L^2(\mathbb{R}^d)$.

In the following M_N is a constant of the form

$$M_N = C\left(\mu_{\max}, \frac{1}{h_{\min}}, \varepsilon \left|\zeta\right|_{H^N}, \beta \left|\nabla b\right|_{H^N}, \beta \left|b\right|_{L^{\infty}}\right).$$
(4.24)

4.2.2 Existence and uniqueness

In this part, we forget the dependence on t. First, notice that we can split the problem into two parts. Let $\Phi \in \dot{H}^2(\Omega)$ the unique solution of the Laplace problem (see [80])

$$\begin{cases} \Delta^{\mu}_{X,z} \Phi = 0 \text{ in } \Omega, \\ \Phi_{|z=\varepsilon\zeta} = \psi, \ \left(N^{\mu}_{b} \cdot \nabla^{\mu}_{X,z} \Phi \right)_{|z=-1+\beta b} = 0. \end{cases}$$
(4.25)

Using the vectorial identity

$$\left(\nabla^{\mu}_{X,z}\Phi\right)_{/\!\!/} = \nabla\psi$$

it is easy to check that if \mathbf{U}^{μ} satisfies (4.21), $\widetilde{\mathbf{U}}^{\mu} := \mathbf{U}^{\mu} - \nabla^{\mu}_{X,z} \Phi$ satisfies

$$\begin{cases} \operatorname{curl}^{\mu} \widetilde{\mathbf{U}}^{\mu} = \mu \, \boldsymbol{\omega} \text{ in } \Omega_{t}, \\ \operatorname{div}^{\mu} \widetilde{\mathbf{U}}^{\mu} = 0 \text{ in } \Omega_{t}, \\ \widetilde{\mathbf{U}}_{l}^{\mu} = \frac{\nabla^{\perp}}{\Delta} \left(\underline{\boldsymbol{\omega}} \cdot N^{\mu} \right) \text{ at the surface,} \\ \widetilde{\mathbf{U}}_{b}^{\mu} \cdot N_{b}^{\mu} = 0 \text{ at the bottom.} \end{cases}$$

$$(4.26)$$

In the following we focus on the system (4.26). We give 4 intermediate results in order to get the existence and uniqueness. The first Proposition shows how to control the norm of the gradient of a function with boundary conditions as in (4.26).

Proposition 4.2.3. Let ζ , $b \in W^{2,\infty}(\mathbb{R}^d)$, $\mathbf{A} \in H(\operatorname{div}_0^{\mu}, \Omega) \cap H(\operatorname{curl}^{\mu}, \Omega)$. Then, for all $C \in H^1(\mathbb{R}^d)^3$, we have

$$\int_{\Omega} \nabla_{X,z}^{\mu} \mathbf{A} : \nabla_{X,z}^{\mu} \mathbf{C} = \int_{\Omega} \operatorname{curl}^{\mu} \mathbf{A} : \operatorname{curl}^{\mu} \mathbf{C} + \left\langle l^{\mu} [\varepsilon \zeta] (\underline{A}), \underline{C} \right\rangle_{H^{-\frac{1}{2}} - H^{\frac{1}{2}}} - \left\langle l^{\mu} [\beta b] (\mathbf{A}_{b}), C_{b} \right\rangle_{H^{-\frac{1}{2}} - H^{\frac{1}{2}}},$$
(4.27)

where for $\mathbf{B} \in H^{\frac{1}{2}}\left(\mathbb{R}^{2}\right)^{3}$ and for $\eta \in W^{2,\infty}(\mathbb{R}^{d})$,

$$l^{\mu}[\eta](\mathbf{B}) = \begin{pmatrix} \sqrt{\mu}\nabla\mathbf{B}_{v} - \mu\left(\nabla^{\perp}\eta\cdot\nabla\right)\mathbf{B}_{h}^{\perp} \\ -\sqrt{\mu}\nabla\cdot\mathbf{B}_{h} \end{pmatrix}.$$
(4.28)

Furthermore, if $\widetilde{\psi} \in \dot{H}^{\frac{3}{2}}(\mathbb{R}^d)$ and

$$A_b \cdot N_b^{\mu} = 0$$
 and $A_{\parallel} = \nabla^{\perp} \widetilde{\psi}$,

we have the following estimate

$$\left| \nabla_{X,z}^{\mu} \mathbf{A} \right|_{2}^{2} \leq \left| \left| \operatorname{curl}^{\mu} \mathbf{A} \right| \right|_{2}^{2} + \mu C \left(\varepsilon \left| \zeta \right|_{W^{2,\infty}}, \beta \left| b \right|_{W^{2,\infty}} \right) \left(\left| \underline{A} \right|_{2}^{2} + \left| A_{bh} \right|_{2}^{2} \right) \right.$$

$$+ \mu C \left(\mu_{\max}, \varepsilon \left| \zeta \right|_{W^{2,\infty}}, \beta \left| b \right|_{W^{2,\infty}} \right) \left| \sqrt{1 + \sqrt{\mu} |D|} \nabla \widetilde{\psi} \right|_{2} \left| \sqrt{1 + \sqrt{\mu} |D|} \underline{A}_{h} \right|_{2}.$$

$$(4.29)$$

Proof. Using the Einstein summation convention and denoting $\nabla_{X,z}^{\mu} = (\partial_1^{\mu}, \partial_2^{\mu}, \partial_3^{\mu})^T$, a simple computation gives (see Lemma 3.2 in [34] or Chapter 9 in [51]),

$$\left|\left|\nabla^{\mu}\mathbf{A}\right|\right|_{2}^{2} = \left|\left|\operatorname{curl}^{\mu}\mathbf{A}\right|\right|_{2}^{2} + \left|\left|\operatorname{div}^{\mu}\mathbf{A}\right|\right|_{2}^{2} + \int_{\partial\Omega} n_{i}^{\mu}\mathbf{A}_{j}\partial_{j}^{\mu}\mathbf{A}_{i} - n_{j}^{\mu}\mathbf{A}_{j}\partial_{i}^{\mu}\mathbf{A}_{i}.$$
(4.30)

In this case, $\partial\Omega$ is the union of two surfaces and $\vec{n^{\mu}} = \pm \begin{pmatrix} -\sqrt{\mu}\nabla\eta \\ 1 \end{pmatrix}$, where η is the corresponding surface. Then, one can check that (see also Lemma 3.8 in [34]),

$$\int_{\{z=\eta\}} n_i^{\mu} \mathbf{A}_j \partial_j^{\mu} \mathbf{A}_i - n_j^{\mu} \mathbf{A}_j \partial_i^{\mu} \mathbf{A}_i = \pm \int_{\mathbb{R}^d} \mathbf{A}_{\eta,h} \cdot \left(2\sqrt{\mu} \nabla_X \mathbf{A}_{\eta,v} - \mu \left(\nabla \eta^{\perp} \cdot \nabla \right) \mathbf{A}_{\eta,h}^{\perp} \right),$$
(4.31)

where $\mathbf{A}_{\eta} := \mathbf{A}_{|z=\eta}$. The first part of the Proposition follows by polarization of Equations (4.30) and (4.31) (as quadratic forms). For the second estimate, since $\mathbf{A}_b \cdot N_b^{\mu} = 0$, we get at the bottom that

$$\begin{split} \int_{\{z=-1+\beta b\}} &n_i^{\mu} \mathbf{A}_j \partial_j^{\mu} \mathbf{A}_i - n_j^{\mu} \mathbf{A}_j \partial_i^{\mu} \mathbf{A}_i = -2 \int_{\mathbb{R}^d} & \left(\partial_x b \mathbf{A}_{bx} \partial_y \mathbf{A}_{by} + \partial_y b \mathbf{A}_{by} \partial_x \mathbf{A}_{bx} + \partial_{xy}^2 b \mathbf{A}_{bx} \mathbf{A}_{by} \right) - \mathbf{A}_z \sqrt{\mu} \mathrm{div}_X \mathbf{A}_{bh} \\ &= -\mu \beta \int_{\mathbb{R}^d} & \partial_x^2 b \mathbf{A}_{bx}^2 + \partial_y^2 b \mathbf{A}_{by}^2 + 2 \partial_{xy}^2 b \mathbf{A}_{bx} \mathbf{A}_{by}. \end{split}$$

At the surface, since $\underline{\mathbf{A}}_h = \sqrt{\mu} \nabla^{\perp} \widetilde{\psi} - \varepsilon \sqrt{\mu} \underline{\mathbf{A}}_v \nabla \zeta$ (we use equality (4.14)), we have

Then,

$$\left| 2\varepsilon\mu^{\frac{3}{2}} \int_{\mathbb{R}^{d}} \underline{\mathbf{A}}_{h} \cdot \nabla^{\perp} \left(\nabla\widetilde{\psi} \cdot \nabla\zeta \right) \right| \leq \varepsilon\mu \left| \sqrt{1 + \sqrt{\mu}|D|} \underline{\mathbf{A}}_{h} \right|_{2} \left| \sqrt{\mu} \mathfrak{P} \left(\nabla\widetilde{\psi} \cdot \nabla\zeta \right) \right|_{2}.$$

and estimate (4.29) follows easily from Lemma B.1.2.

The second Proposition gives a control of the L^2 -norm of the trace.

Proposition 4.2.4. Let ζ , $b \in W^{1,\infty}(\mathbb{R}^d)$, $\mathbf{A} \in H(\operatorname{div}_0^{\mu}, \Omega) \cap H(\operatorname{curl}^{\mu}, \Omega)$ and $\widetilde{\psi} \in \dot{H}^1(\mathbb{R}^d)$ such that

$$\mathbf{A}_b \cdot N_b^{\mu} = 0 \text{ and } \mathbf{A}_{/\!\!/} = \nabla^{\perp} \widetilde{\psi}.$$

Then,

$$|\underline{\mathbf{A}}|_{2}^{2} + |\mathbf{A}_{b}|_{2}^{2} \leq C. \left(\mu \left| \nabla \widetilde{\psi} \right|_{2}^{2} + ||\operatorname{curl}^{\mu} \mathbf{A}||_{2} ||\mathbf{A}||_{2} \right).$$

$$(4.32)$$

Proof. Using the fact that $\partial_z \mathbf{A}_h = -\left(\operatorname{curl}^{\mu} \mathbf{A}\right)_h^{\perp} + \sqrt{\mu} \nabla_X \mathbf{A}_v$, we have

$$\begin{split} \int_{\mathbb{R}^d} & |\mathbf{A}_{bh}|^2 = \int_{\mathbb{R}^d} |\mathbf{A}_{bh}|^2 + 2 \int_{\Omega} \partial_z \mathbf{A}_h \cdot \mathbf{A}_h \\ &= \int_{\mathbb{R}^d} |\mathbf{A}_{bh}|^2 - 2 \int_{\Omega} (\operatorname{curl}^{\mu} \mathbf{A})_h^{\perp} \cdot \mathbf{A}_h + 2\sqrt{\mu} \int_{\Omega} \nabla_X \mathbf{A}_v \cdot \mathbf{A}_h \\ &= \int_{\mathbb{R}^d} |\mathbf{A}_{bh}|^2 - 2 \int_{\Omega} (\operatorname{curl}^{\mu} \mathbf{A})_h^{\perp} \cdot \mathbf{A}_h + 2 \int_{\Omega} \partial_z \mathbf{A}_v \mathbf{A}_v + 2\sqrt{\mu} \left(\int_{\mathbb{R}^d} (\nabla b \cdot \mathbf{A}_{bh}) \mathbf{A}_{bv} \\ &- \int_{\mathbb{R}^d} (\varepsilon \nabla \zeta \cdot \underline{\mathbf{A}}_h) \, \underline{\mathbf{A}}_v \right), \end{split}$$

where the third equality is obtained by integrating by parts the third integral and by using the fact that $\operatorname{div}^{\mu} \mathbf{A} = 0$. Furthermore, thanks to the boundary conditions and Equality (4.14), we have

$$\varepsilon \sqrt{\mu} \left(\nabla \zeta \cdot \underline{\mathbf{A}}_h \right) \underline{\mathbf{A}}_v = \sqrt{\mu} \nabla^{\perp} \widetilde{\psi} \cdot \underline{\mathbf{A}}_h - \left| \underline{\mathbf{A}}_h \right|^2 \text{ and } \beta \sqrt{\mu} \left(\nabla b \cdot \mathbf{A}_{bh} \right) \mathbf{A}_{bv} = \mathbf{A}_{bv}^2.$$

Then, we get

$$|\underline{\mathbf{A}}|_{2}^{2} + |\mathbf{A}_{b}|_{2}^{2} = 2\sqrt{\mu} \int_{\mathbb{R}^{d}} \nabla^{\perp} \widetilde{\psi} \cdot \underline{\mathbf{A}}_{h} + 2 \int_{\mathbb{R}^{d}} (\operatorname{curl}^{\mu} \mathbf{A})_{h}^{\perp} \cdot \mathbf{A}_{h}, \qquad (4.33)$$

Follows.

and the inequality follows.

The third Proposition is a Poincaré inequality.

Proposition 4.2.5. Let ζ , $b \in W^{1,\infty}(\mathbb{R}^d)$ and $\mathbf{A} \in H(\operatorname{div}_0^{\mu}, \Omega) \cap H(\operatorname{curl}^{\mu}, \Omega)$ such that

$$A_b \times N_b^{\mu} = 0$$
 and $\underline{A} \cdot N^{\mu} = 0$.

Then,

$$||\mathbf{A}||_{2} \leq C |\varepsilon \zeta - \beta b + 1|_{L^{\infty}} (||\operatorname{curl}^{\mu} \mathbf{A}||_{2} + ||\partial_{z} \mathbf{A}||_{2}).$$

$$(4.34)$$

Proof. We have

$$\left|\mathbf{A}(X,z)\right|^{2} = \left|\mathbf{A}_{b}(X)\right|^{2} + 2\int_{s=-1+\beta b(X)}^{z} \partial_{z}\mathbf{A}(X,s) \cdot \mathbf{A}(X,s) dX ds.$$

Then, the result follows from the following lemma, which is a similar computation to the one in Proposition 4.2.4 (by switching the boundary conditions).

Lemma 4.2.6. Let ζ , $b \in W^{1,\infty}(\mathbb{R}^d)$, $\mathbf{A} \in H(\operatorname{div}_0^{\mu}, \Omega) \cap H(\operatorname{curl}^{\mu}, \Omega)$ such that

$$A_b \times N_b^{\mu} = 0$$
 and $\underline{A} \cdot N^{\mu} = 0$

Then,

$$|\underline{\mathbf{A}}|_{2}^{2} + |\mathbf{A}_{b}|_{2}^{2} \leq C \, ||\operatorname{curl}^{\mu} \mathbf{A}||_{2} \, ||\mathbf{A}||_{2} \,. \tag{4.35}$$

Finally, the fourth Proposition links the regularity of $\widetilde{\psi}$ to the regularity of $\omega_b \cdot N_b^{\mu}$.

Proposition 4.2.7. Let $\zeta, b \in W^{1,\infty}(\mathbb{R}^d)$ be such that Condition (4.22) is satisfied and let $\omega \in H_b(\operatorname{div}_0^{\mu}, \Omega)$. Then, there exists a unique solution $\widetilde{\psi} \in \dot{H}^{\frac{3}{2}}(\mathbb{R}^d)$ to the equation $\Delta \widetilde{\psi} = \underline{\omega} \cdot N^{\mu}$ and we have

$$\left|\nabla \widetilde{\psi}\right|_{2} \leq \sqrt{\mu} C\left(\frac{1}{\mathbf{h}_{\min}}, \varepsilon \left|\zeta\right|_{W^{1,\infty}}, \beta \left|b\right|_{W^{1,\infty}}\right) \left(\left|\left|\boldsymbol{\omega}\right|\right|_{2} + \frac{1}{\sqrt{\mu}} \left|\frac{1}{\mathfrak{P}} \left(\omega_{b} \cdot N_{b}^{\mu}\right)\right|_{2}\right),$$

and

$$\left|\sqrt{1+\sqrt{\mu}|D|}\nabla\widetilde{\psi}\right|_{2} \leq \sqrt{\mu}C\left(\frac{1}{\mathbf{h}_{\min}}, \varepsilon \left|\zeta\right|_{W^{1,\infty}}, \beta \left|b\right|_{W^{1,\infty}}\right) \left(\left|\left|\boldsymbol{\omega}\right|\right|_{2} + \frac{1}{\sqrt{\mu}} \left|\frac{1}{\mathfrak{P}}\left(\omega_{b} \cdot N_{b}^{\mu}\right)\right|_{2}\right).$$

Proof. The proof is a small adaptation of Lemma 3.7 and Lemma 5.5 in [34].

We can now prove an existence and uniqueness result for the system (4.21) and (4.26).

Theorem 4.2.8. Let ζ , $b \in W^{2,\infty}(\mathbb{R}^d)$ such that Condition (4.22) is satisfied, $\psi \in \dot{H}^{\frac{3}{2}}(\mathbb{R}^d)$ and $\omega \in H_b(\operatorname{div}_0^{\mu}, \Omega)$. There exists a unique solution $\mathbf{U}^{\mu} = \mathbf{U}^{\mu}[\varepsilon\zeta, \beta b](\psi, \omega) \in H^1(\Omega)$ to (4.21). Furthermore, $\mathbf{U}^{\mu} = \nabla_{X,z}^{\mu} \Phi + \operatorname{curl}^{\mu} \mathbf{A}$, where Φ satisfies (4.25) and \mathbf{A} satisfies

$$\begin{cases} \operatorname{curl}^{\mu}\operatorname{curl}^{\mu}\mathbf{A} = \mu \boldsymbol{\omega} \text{ in } \Omega_{t}, \\ \operatorname{div}^{\mu}\mathbf{A} = 0 \text{ in } \Omega_{t}, \\ N_{b}^{\mu} \times A_{b} = 0, \\ N^{\mu} \cdot \underline{A} = 0, \\ (\operatorname{curl}^{\mu}\mathbf{A})_{/\!\!/} = \frac{\nabla^{\perp}}{\Delta} (\underline{\omega} \cdot N^{\mu}), \\ N_{b}^{\mu} \cdot (\operatorname{curl}^{\mu}\mathbf{A})_{|z=-1+\beta b} = 0. \end{cases}$$

$$(4.36)$$

Finally, one has

$$\left\| \mathbf{U}^{\mu} \right\|_{2} \leq \sqrt{\mu} C \left(\mu_{\max}, \frac{1}{\mathbf{h}_{\min}}, \varepsilon \left| \zeta \right|_{W^{2,\infty}}, \beta \left| b \right|_{W^{2,\infty}} \right) \left(\sqrt{\mu} \left\| \boldsymbol{\omega} \right\|_{2} + \left| \frac{1}{\mathfrak{P}} \left(\omega_{b} \cdot N_{b}^{\mu} \right) \right|_{2} + \left| \mathfrak{P} \psi \right|_{2} \right), \quad (4.37)$$

and

$$\left|\left|\nabla_{X,z}^{\mu}\mathbf{U}^{\mu}\right|\right|_{2} \leq \mu C\left(\mu_{\max}, \frac{1}{\mathbf{h}_{\min}}, \varepsilon \left|\zeta\right|_{W^{2,\infty}}, \beta \left|b\right|_{W^{2,\infty}}\right) \left(\left|\left|\boldsymbol{\omega}\right|\right|_{2} + \left|\frac{1}{\mathfrak{P}}\left(\omega_{b} \cdot N_{b}^{\mu}\right)\right|_{2} + \left|\mathfrak{P}\psi\right|_{H^{1}}\right).$$
(4.38)

Proof. The uniqueness follows easily from the last Propositions. The existence of Φ and the control of its norm are proved in Section A.1.1. We focus on the existence of a solution of (4.36). The main idea is the following variational formulation for the system (4.36) (we refer to Lemma 3.5 and Proposition 5.3 in [34] for the details). We denote by

$$\mathcal{X} = \left\{ \mathbf{C} \in H^{1}(\Omega), \operatorname{div}^{\mu} \mathbf{C} = 0, \underline{\mathbf{A}} \cdot N^{\mu} = 0 \text{ and } \mathbf{A}_{b} \times N_{b}^{\mu} = 0 \right\}$$

and $\tilde{\psi}$ the unique solution in $\dot{H}^1(\mathbb{R}^d)$ of $\Delta \tilde{\psi} = \underline{\omega} \cdot N^{\mu}$. Then, $A \in \mathcal{X}$ is a variational solution of System (4.36) if

$$\forall \mathbf{C} \in \mathcal{X}, \ \int_{\Omega} \operatorname{curl}^{\mu} \mathbf{A} \cdot \operatorname{curl}^{\mu} \mathbf{C} = \mu \int_{\Omega} \boldsymbol{\omega} \cdot \mathbf{C} + \mu \int_{\mathbb{R}^{d}} \nabla \widetilde{\psi} \cdot C_{/\!\!/}, \tag{4.39}$$

The existence of such a \mathbf{A} follows Lax-Milgram's theorem. In the following we only explain how we get the coercivity. Thanks to a similar computation that we used to prove Estimate (4.29) (by switching the boundary conditions), we get

$$\left\|\left|\nabla_{X,z}^{\mu}\mathbf{A}\right\|\right\|_{2}^{2} \leq \left\|\left|\operatorname{curl}^{\mu}\mathbf{A}\right|\right\|_{2}^{2} + \mu C\left(\varepsilon \left|\nabla\zeta\right|_{W^{2,\infty}}, \beta \left|\nabla b\right|_{W^{2,\infty}}\right)\left(\left|\underline{A}\right|_{2}^{2} + \left|A_{bh}\right|_{2}^{2}\right).$$

Then, thanks to a similar computation that in Proposition 4.2.4 and Proposition 4.2.5, we obtain the coercivity

$$\left|\left|\mathbf{A}\right|\right|_{2} + \left|\left|\nabla_{X,z}^{\mu}\mathbf{A}\right|\right|_{2} \le C\left(\mu_{\max}, \frac{1}{h_{\min}}, \varepsilon \left|\zeta\right|_{W^{2,\infty}}, \beta \left|b\right|_{W^{2,\infty}}\right) \left|\left|\operatorname{curl}^{\mu}\mathbf{A}\right|\right|_{2}.$$

Then, we can easily extend this for all **C** in $\{\mathbf{C} \in H^1(\Omega), \underline{C} \cdot N^{\mu} = 0 \text{ and } \mathbf{C}_b \times N_b^{\mu} = 0\}$ (see Lemma 3.5 in [34]) and thanks to the variational formulation of **A** we get

$$\left|\left|\operatorname{curl}^{\mu}\mathbf{A}\right|\right|_{2} \leq C\left(\mu_{\max}, \frac{1}{h_{\min}}, \varepsilon \left|\zeta\right|_{W^{2,\infty}}, \beta \left|b\right|_{W^{2,\infty}}\right) \left(\mu \left|\left|\boldsymbol{\omega}\right|\right|_{2} + \sqrt{\mu} \left|\nabla \widetilde{\psi}\right|_{2}\right).$$

Using Proposition 4.2.7, we get the first estimate. The second estimate follows from the first estimate, the inequality (4.29), Proposition 2.4, Proposition 2.6 and the following Lemma.

Lemma 4.2.9. Let ζ , $b \in W^{1,\infty}(\mathbb{R}^d)$ be such that Condition (4.22) is satisfied. Then, for all $u \in H^1(\Omega)$,

$$\left|\sqrt{1+\sqrt{\mu}|D|}\underline{u}\right|_2 + \left|\sqrt{1+\sqrt{\mu}|D|}u_b\right|_2 \leq C\left(\frac{1}{h_{\min}},\varepsilon\left|\zeta\right|_{W^{1,\infty}},\beta\left|b\right|_{W^{1,\infty}}\right)\left(\left|\left|\nabla_{X,z}^{\mu}u\right|\right|_2 + \left|\left|u\right|\right|_2\right).$$

Proof. The proof is a small adaptation of Lemma 5.4 in [34].

4.2.3 The transformed div-curl problem

In this section, we transform the div-curl problem in the domain Ω into a variable coefficients problem in the flat strip $S = \mathbb{R}^d \times (-1, 0)$. We introduce the diffeomorphism Σ ,

$$\Sigma := \begin{array}{c} S \to \Omega\\ (X,z) \mapsto (X,z + \sigma(X,z)) \,, \end{array}$$
(4.40)

where

$$\sigma(X,z) := z \left(\varepsilon \zeta(X) - \beta b(X) \right) + \varepsilon \zeta(X).$$

We keep the notations of [34]. We define

$$\mathbf{U}^{\sigma,\mu}\left[\varepsilon\zeta,\beta b\right](\psi,\omega) = \mathbf{U}^{\mu} = \begin{pmatrix}\sqrt{\mu}\mathbf{V}\\\mathbf{w}\end{pmatrix} = \mathbf{U}^{\mu}\circ\Sigma,\,\omega = \boldsymbol{\omega}\circ\Sigma,$$

and

$$\nabla_{X,z}^{\sigma,\mu} = \left(J_{\Sigma}^{-1}\right)^t \nabla_{X,z}^{\mu}, \text{ where } \left(J_{\Sigma}^{-1}\right)^t = \begin{pmatrix} Id_{d\times d} & \frac{-\sqrt{\mu}\nabla\sigma}{1+\partial_z\sigma} \\ 0 & \frac{\Gamma}{1+\partial_z\sigma} \end{pmatrix}$$

Furthermore, for $A = \mathbf{A} \circ \Sigma$,

$$\operatorname{curl}^{\sigma,\mu} \mathbf{A} = (\operatorname{curl}^{\mu} \mathbf{A}) \circ \Sigma = \nabla_{X,z}^{\sigma,\mu} \times \mathbf{A}, \operatorname{div}^{\sigma,\mu} \mathbf{A} = (\operatorname{div}^{\mu} \mathbf{A}) \circ \Sigma = \nabla_{X,z}^{\sigma,\mu} \cdot \mathbf{A}.$$

Finally, if A is vector field on \mathcal{S} ,

$$\underline{\mathbf{A}} = \mathbf{A}_{|z=0}, \, \mathbf{A}_b = \mathbf{A}_{|z=-1} \text{ and } \mathbf{A}_{/\!\!/} = \frac{1}{\sqrt{\mu}} \underline{\mathbf{A}}_h + \varepsilon \underline{\mathbf{A}}_v \nabla \zeta.$$

Then, $\mathbf{U}^{\mu}=\mathbf{U}^{\mu}[\varepsilon\zeta,\beta b](\psi,\omega)$ is the unique solution in $H^{1}(\mathcal{S})$ of

$$\begin{array}{l} \left(\operatorname{curl}^{\sigma,\mu} \mathrm{U}^{\mu} = \mu \, \omega \, \operatorname{in} \, \mathcal{S}, \\ \operatorname{div}^{\sigma,\mu} \mathrm{U}^{\mu} = 0 \, \operatorname{in} \, \mathcal{S}, \\ \mathrm{U}_{/\!\!/}^{\mu} = \nabla \psi + \frac{\nabla^{\perp}}{\Delta} \left(\underline{\omega} \cdot N^{\mu} \right) \, \operatorname{on} \, \left\{ z = 0 \right\}, \\ \mathrm{U}_{b}^{\mu} \cdot N_{b}^{\mu} = 0 \, \operatorname{on} \, \left\{ z = -1 \right\}. \end{array}$$

$$\begin{array}{l} (4.41) \\ \end{array}$$

We also keep the notations in [96]. If $A = \mathbf{A} \circ \Sigma$, we define

$$\partial_i^{\sigma} \mathcal{A} = \partial_i \mathbf{A} \circ \Sigma, \ i \in \{t, x, y, z\}, \ \partial_i^{\sigma} = \partial_i - \frac{\partial_i \sigma}{1 + \partial_z \sigma} \partial_z, \ i \in \{x, y, t\} \ \text{and} \ \partial_z^{\sigma} = \frac{1}{1 + \partial_z \sigma} \partial_z.$$

Then, by a change of variables and Proposition 4.2.3 we get the following variational formulation for U^{μ} . For all $C \in H^1(\mathcal{S})$,

$$\int_{\mathcal{S}} \nabla_{X,z}^{\mu} \mathbf{U}^{\mu} \cdot P\left(\Sigma\right) \nabla_{X,z}^{\mu} \mathbf{C} = \mu \int_{\mathcal{S}} \left(1 + \partial_{z}\sigma\right) \omega \cdot \operatorname{curl}^{\sigma,\mu} \mathbf{C} + \int_{\mathbb{R}^{d}} l^{\mu} [\varepsilon\zeta](\underline{\mathbf{U}}^{\mu}) \cdot \underline{C} - \int_{\mathbb{R}^{d}} l^{\mu} [\beta b](\mathbf{U}_{b}^{\mu}) \cdot C_{b}, \quad (4.42)$$

where $P(\Sigma) = (1 + \partial_z \sigma) J_{\Sigma}^{-1} (J_{\Sigma}^{-1})^t$ and

$$l^{\mu}[\eta] \left(\mathbf{U}_{|z=\eta}^{\mu} \right) = \begin{pmatrix} \sqrt{\mu} \nabla \mathbf{w}_{|z=\eta} - \mu^{\frac{3}{2}} \left(\nabla^{\perp} \eta \cdot \nabla \right) \mathbf{V}_{|z=\eta}^{\perp} \\ -\mu \nabla \cdot \mathbf{V}_{|z=\eta} \end{pmatrix}$$

In order to obtain higher order estimates on U^{μ} , we have to separate the regularity on z and the regularity on X. We use the following spaces.

Definition 4.2.10. We define the spaces $H^{s,k}$

$$H^{s,k}(\mathcal{S}) = \bigcap_{0 \le l \le k} H^l_z(-1,0; H^{s-l}_X(\mathbb{R}^d)) \text{ and } |u|_{H^{s,k}} = \sum_{0 \le l \le k} |\Lambda^{s-j} \partial^j_z u|_2.$$

Furthermore, if $\alpha \in \mathbb{N}^d \setminus \{0\}$, we define the Alinhac's good unknown

$$\psi_{(\alpha)} = \partial^{\alpha} \psi - \varepsilon \underline{w} \partial^{\alpha} \zeta \text{ and } \psi_{(0)} = \psi.$$
(4.43)

This quantities play an important role in the wellposedness of the water waves equations (see [6] or [80]). In fact, more generally, if A is vector field on \mathcal{S} , we denote by

$$\mathbf{A}_{(\alpha)} = \partial^{\alpha} \mathbf{A} - \partial^{\alpha} \sigma \partial_{z}^{\sigma} \mathbf{A} , \ \mathbf{A}_{(0)} = \mathbf{A} , \ \underline{\mathbf{A}}_{(\alpha)} = \partial^{\alpha} \underline{\mathbf{A}} - \varepsilon \partial^{\alpha} \zeta \underline{\partial_{z}^{\sigma} \mathbf{A}} \text{ and } \underline{\mathbf{A}}_{(0)} = \underline{\mathbf{A}}.$$
(4.44)

We can now give high order estimates on U^{μ} . We recall that M_N is defined in (4.24).

Theorem 4.2.11. Let $N \in \mathbb{N}$, $N \geq 5$. Then, under the assumptions of Theorem 4.2.8, for all $0 \leq l \leq 1$ and $0 \leq l \leq k \leq N - 1$, the straightened velocity U^{μ} , satisfies

$$\left\| \nabla_{X,z}^{\mu} \mathbf{U}^{\mu} \right\|_{H^{k,l}} \leq \mu M_N \left(\left| \mathfrak{P}\psi \right|_{H^1} + \sum_{1 < |\alpha| \le k+1} \left| \mathfrak{P}\psi_{(\alpha)} \right|_2 + \left| |\omega| \right|_{H^{k,l}} + \left| \frac{\Lambda^k}{\mathfrak{P}} \left(\omega_b \cdot N_b^{\mu} \right) \right|_2 \right).$$

Proof. We start with l = 0. We follow the proof of Proposition 3.12 and Proposition 5.8 in [34]. Let $k \in [1, N-1], \alpha \in \mathbb{N}^d$ with $|\alpha| \leq k$. We take $C = \partial^{2\alpha} U^{\mu}$ in $(4.42)^{(3)}$ and we get

$$\begin{split} \int_{\mathcal{S}} \nabla^{\mu}_{X,z} \mathbf{U}^{\mu} \cdot P\left(\Sigma\right) \nabla^{\mu}_{X,z} \partial^{2\alpha} \mathbf{U}^{\mu} &= \mu \int_{\mathcal{S}} (1 + \partial_{z} \sigma) \, \omega \cdot \operatorname{curl}^{\sigma,\mu} \partial^{2\alpha} \mathbf{U}^{\mu} + \int_{\mathbb{R}^{d}} l^{\mu} [\varepsilon \zeta] (\underline{\mathbf{U}}^{\mu}) \cdot \partial^{2\alpha} \underline{\mathbf{U}}^{\mu} \\ &- \int_{\mathbb{R}^{d}} l^{\mu} [\beta b] (\mathbf{U}^{\mu}_{b}) \cdot \partial^{2\alpha} \mathbf{U}^{\mu}_{b}. \end{split}$$

We focus on the bottom contribution, which is the last term of the previous equation (see [34] for the other terms). Using the fact that $w_b = \mu \beta \nabla b \cdot V_b$, we have

³A. Castro and D. Lannes explain why we can take such a C in the variational formulation.

$$\begin{split} (-1)^{|\alpha|} \int_{\mathbb{R}^d} l^{\mu} [\beta b] \left(\mathbf{U}_b \right) \cdot \partial^{2\alpha} \mathbf{U}_b &= \int_{\mathbb{R}^d} 2\mu \partial^{\alpha} \nabla \mathbf{w}_b \cdot \partial^{\alpha} \mathbf{V}_b - \mu^2 \beta \partial^{\alpha} \left[\left(\nabla^{\perp} b \cdot \nabla \right) \mathbf{V}_b^{\perp} \right] \cdot \partial^{\alpha} \mathbf{V}_b \\ &= \int_{\mathbb{R}^d} 2\mu^2 \beta \partial^{\alpha} \nabla \left(\nabla b \cdot \mathbf{V}_b \right) \cdot \partial^{\alpha} \mathbf{V}_b - \mu^2 \beta \partial^{\alpha} \left[\left(\nabla^{\perp} b \cdot \nabla \right) \mathbf{V}_b^{\perp} \right] \cdot \partial^{\alpha} \mathbf{V}_b \\ &= \underbrace{\int_{\mathbb{R}^d} 2\mu^2 \beta \left(\nabla b \right)^t \cdot \partial^{\alpha} \nabla \mathbf{V}_b \cdot \partial^{\alpha} \mathbf{V}_b - \beta \mu^2 \left[\left(\nabla^{\perp} b \cdot \nabla \right) \partial^{\alpha} \mathbf{V}_b^{\perp} \right] \cdot \partial^{\alpha} \mathbf{V}_b }_{I_1} \\ &+ \underbrace{\int_{\mathbb{R}^d} 2\mu^2 \beta \left[\partial^{\alpha} \nabla, \nabla b \right] \mathbf{V}_b \cdot \partial^{\alpha} \mathbf{V}_b - \beta \mu^2 \left[\partial^{\alpha}, \left(\nabla^{\perp} b \cdot \nabla \right) \right] \mathbf{V}_b^{\perp} \cdot \partial^{\alpha} \mathbf{V}_b }_{I_2} \end{split}$$

Then, a careful computation gives

$$\begin{split} |I_{1}| &= \left| \mu^{2} \beta \int_{\mathbb{R}^{d}} \partial_{x}^{2} b \left(\partial^{\alpha} \mathbf{V}_{bx} \right)^{2} + \partial_{y}^{2} b \left(\partial^{\alpha} \mathbf{V}_{by} \right)^{2} + 2\mu^{2} \beta \int_{\mathbb{R}^{d}} \partial_{xy}^{2} b \left(\partial^{\alpha} \mathbf{V}_{bx} \right)^{\alpha} \mathbf{V}_{by} \right| \\ &\leq \mu C \left(\delta, \frac{1}{h_{\min}}, \varepsilon \left| \zeta \right|_{W^{1,\infty}}, \beta \left| b \right|_{W^{2,\infty}} \right) \left| \left| \partial^{\alpha} \mathbf{U}^{\mu} \right| \right|_{2}^{2} + \delta \left| \left| \nabla_{X,z}^{\mu} \partial^{\alpha} \mathbf{U}^{\mu} \right| \right|_{2}^{2} \\ &\leq C \left(\delta, \frac{1}{h_{\min}}, \varepsilon \left| \zeta \right|_{W^{1,\infty}}, \beta \left| b \right|_{W^{2,\infty}} \right) \left| \left| \nabla_{X,z}^{\mu} \mathbf{U}^{\mu} \right| \right|_{H^{k-1}}^{2} + \delta \left| \left| \nabla_{X,z}^{\mu} \partial^{\alpha} \mathbf{U}^{\mu} \right| \right|_{2}^{2}, \end{split}$$

where $\delta > 0$ is small enough and where we use the following Lemma.

Lemma 4.2.12. Let ζ , $b \in W^{1,\infty}(\mathbb{R}^d)$, such that Condition (4.22) is satisfied. Then, for all $u \in H^1(S)$ and $\delta > 0$,

$$|\underline{u}|_{2}^{2} + |u_{b}|_{2}^{2} \leq C\left(\delta, \frac{1}{h_{\min}}, \varepsilon |\zeta|_{W^{1,\infty}}, \beta |b|_{W^{1,\infty}}\right) ||u||_{2}^{2} + \delta ||\partial_{z}u||_{2}^{2}.$$

Furthermore, using Lemma B.3.1 and the previous Lemma, we get

$$\begin{aligned} |I_2| &\leq C\mu\beta \left|\nabla b\right|_{H^{k+1}} \left|\mathbf{U}_b^{\mu}\right|_{H^k} \left|\partial^{\alpha} \mathbf{U}_b^{\mu}\right|_2 \\ &\leq \mu C \left(\delta, \frac{1}{h_{\min}}, \varepsilon \left|\zeta\right|_{W^{1,\infty}}, \beta \left|b\right|_{W^{1,\infty}}, \beta \left|\nabla b\right|_{H^{k+1}}\right) \left|\left|\nabla_{X,z}^{\mu} \mathbf{U}^{\mu}\right|\right|_{H^{k-1}}^2 + \delta \left|\left|\nabla_{X,z}^{\mu} \partial^{\alpha} \mathbf{U}^{\mu}\right|\right|_2^2. \end{aligned}$$

For the surface contribution, we can do the same thing as in Proposition 3.12 and Proposition 5.8 in [34], using the previous Lemma to control $\partial^{\alpha} w$. Finally, for the other terms, the main idea is the following Lemma (which is a small adaptation of Lemma 3.13 and Lemma 5.6 in [34]).

Lemma 4.2.13. Let $\tilde{\psi}$ the unique solution of $\Delta \tilde{\psi} = \underline{\omega} \cdot N^{\mu}$ in $\dot{H}^1(\mathbb{R}^d)$. Under the assumptions of the Theorem, we have the following estimate

$$\left|\mathfrak{P}\nabla^{\perp}\widetilde{\psi}\right|_{H^{k}} \leq M_{N}\left(\left|\left|\omega\right|\right|_{H^{k,0}} + \left|\frac{\Lambda^{k}}{\mathfrak{P}}\left(\omega_{b}\cdot N_{b}^{\mu}\right)\right|_{2}\right).$$

Gathering the previous estimates with the estimate without the bottom contribution in Proposition 5.8 in [34], we get

$$\left|\left|\partial^{\alpha}\nabla^{\mu}\mathbf{U}^{\mu}\right|\right|_{2} \leq \mu M_{N}\left(\left|\mathfrak{P}\psi\right|_{H^{1}} + \sum_{1 < |\alpha| \leq k+1} \left|\mathfrak{P}\psi_{(\alpha)}\right| + \left|\left|\omega\right|\right|_{H^{k,0}} + \left|\frac{\Lambda^{k}}{\mathfrak{P}}(\omega_{b} \cdot N_{b}^{\mu})\right|_{2}\right) + M_{N}\left|\left|\Lambda^{k-1}\nabla^{\mu}_{X,z}\mathbf{U}^{\mu}\right|\right|_{2},$$

and the inequality follows by a finite induction on k. If l = 1, we can adapt the proof of Corollary 3.14 in [34] easily.

Remark 4.2.14. Notice that for $k \ge 2$, we have

$$\left|\frac{\Lambda^{k}}{\mathfrak{P}}\left(\omega_{b}\cdot N_{b}^{\mu}\right)\right|_{2} \leq C\left(\frac{1}{h_{\min}},\mu_{\max},\beta\left|\nabla b\right|_{H^{k+1}}\right)\left(\left|\left|\omega\right|\right|_{H^{k,1}}+\left|\frac{1}{\mathfrak{P}}\left(\omega_{b}\cdot N_{b}^{\mu}\right)\right|_{2}\right),$$

thanks to Lemma B.1.4, Lemma 4.2.9 and Lemma B.2.1.

4.2.4 Time derivatives and few remarks about the good unknown

This part is devoted to recall and adapt some results in [34]. Unlike the previous Propositions, adding a non flat bottom is not painful. That is why we do not give proofs. We refer to section 3.5 and 3.6 in [34] for the details. Firstly, in order to obtain an energy estimate of the Castro-Lannes water waves formulation, we need to control $\partial_t U^{\mu}$. This is the purpose of the following result.

Proposition 4.2.15. Let T > 0, $\zeta \in C^1([0,T], W^{2,\infty}(\mathbb{R}^d))$, $b \in W^{2,\infty}(\mathbb{R}^d)$ such that (4.22) is satisfied for $0 \leq t \leq T$, $\psi \in C^1([0,T], \dot{H}^{\frac{3}{2}}(\mathbb{R}^d))$ and $\omega \in C^1([0,T], L^2(\mathcal{S})^{d+1})$ such that $\nabla_{X,z}^{\mu,\sigma} \cdot \omega = 0$ for $0 \leq t \leq T$. Then,

$$\partial_t \left(\mathbf{U}^{\sigma,\mu} [\varepsilon\zeta,\beta b] (\psi,\omega) \right) = \mathbf{U}^{\sigma,\mu} [\varepsilon\zeta,\beta b] \left(\partial_t \psi - \varepsilon \underline{\mathbf{w}} \partial_t \zeta + \varepsilon \sqrt{\mu} \frac{\nabla}{\Delta} \cdot \left(\underline{\omega_h}^{\perp} \partial_t \zeta \right), \partial_t^{\sigma} \omega \right) \\ + \partial_t \sigma \partial_z^{\sigma} \left(\mathbf{U}^{\mu,\sigma} [\varepsilon\zeta,\beta b] (\psi,\omega) \right).$$

Furthermore, for $N \geq 5$, $U^{\mu} = U^{\sigma,\mu}[\varepsilon\zeta,\beta b](\psi,\omega)$ satisfies

$$\begin{split} \sqrt{\mu} \left| \left| \partial_{t} \mathbf{U}^{\mu} \right| \right|_{2} + \left| \left| \partial_{t} \nabla_{X,z}^{\mu} \mathbf{U}^{\mu} \right| \right|_{H^{N-2,0}} &\leq \mu \max\left(M_{N}, \varepsilon \left| \partial_{t} \zeta \right|_{H^{N-1}} \right) \times \\ \left(\left| \mathfrak{P} \partial_{t} \psi \right|_{H^{1}} + \sum_{1 < |\alpha| \leq N-1} \left| \mathfrak{P} \partial_{t} \psi_{(\alpha)} \right|_{2} + \left| \left| \partial_{t} \omega \right| \right|_{H^{N-2,0}} + \left| \frac{\Lambda^{N-2}}{\mathfrak{P}} \left(\partial_{t} \omega_{b} \cdot N_{b}^{\mu} \right) \right|_{2} \right. \\ \left. + \left| \mathfrak{P} \psi \right|_{H^{1}} + \sum_{1 < |\alpha| \leq N} \left| \mathfrak{P} \psi_{(\alpha)} \right|_{2} + \left| \left| \omega \right| \right|_{H^{N-1,1}} + \left| \frac{1}{\mathfrak{P}} \left(\omega_{b} \cdot N_{b}^{\mu} \right) \right|_{2} \right). \end{split}$$

Secondly, in the context of water waves, the Alinhac's good unknown plays a crucial role. Masmoudi and Rousset remarked in [96] that the Alinhac's good unknown $U^{\mu}_{(\alpha)}$ is almost incompressible and Castro and Lannes showed that the curl^{σ,μ} $U^{\mu}_{(\alpha)}$ is also well controlled. This is the purpose of the following Proposition. We recall that $U^{\mu}_{(\alpha)}$ is defined in (4.44).
Proposition 4.2.16. Let $N \geq 5$, $\zeta \in H^N(\mathbb{R}^d)$, $b \in L^{\infty} \cap \dot{H}^{N+1}(\mathbb{R}^d)$ such that Condition (4.22) is satisfied and $\omega \in H^{N-1}(\mathcal{S})$ such that $\nabla^{\sigma,\mu} \cdot \omega = 0$. Then if we denote by $U^{\mu} = U^{\mu,\sigma}[\varepsilon\zeta,\beta b]$, we have for $1 \leq |\alpha| \leq N$,

$$\begin{split} \left\| \left\| \nabla_{X,z}^{\sigma,\mu} \cdot \mathbf{U}_{(\alpha)}^{\mu} \right\|_{2}^{2} + \left\| \left\| \nabla_{X,z}^{\sigma,\mu} \times \mathbf{U}_{(\alpha)}^{\mu} - \mu \partial^{\alpha} \omega \right\|_{2} \\ & \leq \mu \left| (\varepsilon \zeta, \beta b) \right|_{H^{N}} M_{N} \left(\left| \mathfrak{P} \psi \right|_{H^{1}} + \sum_{1 < |\alpha'| \le |\alpha|} \left| \mathfrak{P} \psi_{(\alpha')} \right|_{2}^{2} + \left| |\omega| \right|_{H^{\max(|\alpha|-1,1)}} + \left| \frac{1}{\mathfrak{P}} \left(\omega_{b} \cdot N_{b}^{\mu} \right) \right|_{2} \right), \end{split}$$

and

$$\left|\mathfrak{P}\psi_{(\alpha)}\right|_{2} \leq M_{N} \left(\left|\mathfrak{P}\psi\right|_{H^{3}} + \frac{1}{\sqrt{\mu}} \sum_{1 < |\alpha'| \leq |\alpha| - 1} \left|\left|\nabla_{X} \mathbf{U}_{(\alpha')}^{\mu}\right|\right|_{2} + \left||\omega|\right|_{H^{N-1}} + \left|\frac{1}{\mathfrak{P}} \left(\omega_{b} \cdot N_{b}^{\mu}\right)\right|_{2}\right).$$

Furthermore, we can control $|\mathfrak{P}\psi|_{H^3}$ by $U^{\sigma,\mu}[\varepsilon\zeta,\beta b](\psi,\omega)$ and ω .

Proposition 4.2.17. Let $N \geq 5$, $\zeta \in H^N(\mathbb{R}^d)$, $b \in L^{\infty} \cap \dot{H}^{N+1}(\mathbb{R}^d)$ such that Condition (4.22) is satisfied and $\omega \in H^{2,1}(\mathcal{S})$ such that $\nabla^{\sigma,\mu} \cdot \omega = 0$. Then,

$$\left|\mathfrak{P}\psi\right|_{H^{3}} \leq M_{N}\left(\frac{1}{\sqrt{\mu}}\left|\left|\Lambda^{3}\mathbf{U}^{\sigma,\mu}[\varepsilon\zeta,\beta b]\left(\psi,\omega\right)\right|\right|_{2} + \left|\left|\omega\right|\right|_{H^{2,1}} + \left|\frac{1}{\mathfrak{P}}\left(\omega_{b}\cdot N_{b}^{\mu}\right)\right|_{2}\right).$$

Proof. The proof is a small adaptation of Lemma 3.23 in [34].

Finally, we give a result that is useful for the energy estimate. Since the proof is a little different from Corollary 3.21 in [34], we give it. Notice that the main difference with Corollary 3.21 in [34] is the fact that we do not have a flat bottom.

Proposition 4.2.18. Let $N \geq 5$, $\zeta \in H^N(\mathbb{R}^d)$, $b \in L^{\infty} \cap \dot{H}^{N+1}(\mathbb{R}^d)$ and $\omega \in H^{N-1}(\mathcal{S})$ such that $\nabla^{\sigma,\mu} \cdot \omega = 0$. Then, for $k = x, y, |\gamma| \leq N - 1$, α such that $\partial^{\alpha} = \partial_k \partial^{\gamma}$ and $\varphi \in H^{\frac{1}{2}}(\mathbb{R}^d)$, we have

where we denote by $U^{\mu} = U^{\sigma,\mu}[\varepsilon\zeta,\beta b].$

Proof. Notice that when $\gamma \neq 0$,

$$\partial_k \mathbf{U}^{\mu}_{(\gamma)} = \mathbf{U}^{\mu}_{(\alpha)} - \partial^{\gamma} \sigma \partial_k \partial_z^{\sigma} \mathbf{U}^{\mu}.$$

Then, using Lemma 4.2.9, it is easy to check that

$$\left(\varphi, \underline{\partial^{\gamma} \sigma \partial_{k} \partial_{z}^{\sigma} \mathbf{U}^{\mu}} \cdot N^{\mu}\right) \leq M_{N} \left| \frac{1}{\sqrt{1 + \sqrt{\mu} |D|}} \varphi \right|_{2} \left| \left| \nabla_{X, z}^{\mu} \mathbf{U}^{\mu} \right| \right|_{H^{2}}.$$

Furthermore, using the Green identity we get

$$\left(\varphi, \mathbf{U}_{(\alpha)}^{\mu} \cdot N^{\mu}\right) = \int_{\mathcal{S}} (1 + \partial_{z}\sigma) \varphi^{\dagger} \nabla_{X,z}^{\sigma,\mu} \cdot \mathbf{U}_{(\alpha)}^{\mu} + \int_{\mathcal{S}} (1 + \partial_{z}\sigma) \mathbf{U}_{(\alpha)}^{\mu} \cdot \nabla_{X,z}^{\sigma,\mu} \varphi^{\dagger} + \left(\varphi_{b}^{\dagger}, \left(\mathbf{U}_{(\alpha)}^{\mu}\right)_{b} \cdot N_{b}^{\mu}\right),$$

where $\varphi^{\dagger} = \chi \left(z \sqrt{\mu} |D| \right) \varphi$ and χ is an even positive compactly supported function equal to 1 near 0. Then, using the fact that $U_b^{\mu} \cdot N_b^{\mu} = 0$ and the trace Lemma, we get

$$\begin{pmatrix} \varphi_b^{\dagger} , \left(\mathbf{U}_{(\alpha)}^{\mu} \right)_b \cdot N_b^{\mu} \end{pmatrix} = (\chi(\sqrt{\mu}|D|)\varphi , \partial^{\alpha}\mathbf{U}_b^{\mu} \cdot N_b^{\mu} - \beta \partial^{\alpha}b \left(\partial_z^{\sigma}\mathbf{U}^{\mu}\right)_b \cdot N_b^{\mu})$$

= $(\chi(\sqrt{\mu}|D|)\varphi , \mu\beta [\nabla b, \partial^{\alpha}] \cdot \mathbf{V}_b - \beta \partial^{\alpha}b \left(\partial_z^{\sigma}\mathbf{U}^{\mu}\right)_b \cdot N_b^{\mu})$
 $\leq M_N \left(\sqrt{\mu} ||\mathbf{U}^{\mu}||_{H^N} + ||\mathbf{U}^{\mu}||_{H^{2,2}}\right) |\chi(\sqrt{\mu}|D|)\varphi|_2 .$

Therefore, using Proposition 4.2.16, Theorem 4.2.11 and the following Lemma (Lemma 2.20 and Lemma 2.34 in [80]) we get the control.

Lemma 4.2.19. Let $\varphi \in H^{\frac{1}{2}}(\mathbb{R}^d)$ and χ an even positive compactly supported function equal to 1 near 0. Then,

$$\left|\left|\chi\left(z\sqrt{\mu}|D|\right)\varphi\right|\right|_{2} \leq C \left|\frac{1}{\sqrt{1+\sqrt{\mu}|D|}}\varphi\right|_{2} \text{ and } \left|\left|\nabla_{X,z}^{\mu}\left(\chi\left(z\sqrt{\mu}|D|\right)\varphi\right)\right|\right|_{2} \leq C\sqrt{\mu}\,|\mathfrak{P}\varphi|_{2}\,.$$

4.3 Well-posedness of the water waves equations

4.3.1 Framework

In this section, we prove a local well-posedness result of the water waves equations. We improve the result of [34] by adding a non flat bottom, a non constant pressure at the surface and a Coriolis forcing. In order to work on a fixed domain, we seek a system of 3 equations on ζ , ψ and $\omega = \boldsymbol{\omega} \circ \Sigma$. We keep the first and the second equations of the Castro-Lannes formulation (4.20). It is easy to check that ω satisfies

$$\partial_t^{\sigma}\omega + \frac{\varepsilon}{\mu} \left(\mathbf{U}^{\mu} \cdot \nabla_{X,z}^{\sigma,\mu} \right) \omega = \frac{\varepsilon}{\mu} \left(\omega \cdot \nabla_{X,z}^{\sigma,\mu} \right) \mathbf{U}^{\mu} + \frac{\varepsilon}{\mu \mathrm{Ro}} \partial_z^{\sigma} \mathbf{U}^{\mu}, \tag{4.45}$$

where $U^{\mu} = U^{\sigma,\mu}[\varepsilon\zeta,\beta b](\psi,\omega)$. Then, in the following the water waves equations will be the system

$$\begin{cases} \partial_t \zeta - \frac{1}{\mu} \underline{U}^{\mu} \cdot N^{\mu} = 0, \\ \partial_t \psi + \zeta + \frac{\varepsilon}{2} \left| U_{/\!\!/}^{\mu} \right|^2 - \frac{\varepsilon}{2\mu} \left(1 + \varepsilon^2 \mu \left| \nabla \zeta \right|^2 \right) \underline{w}^2 + \varepsilon \frac{\nabla}{\Delta} \cdot \left[\left(\underline{\omega} \cdot N^{\mu} + \frac{1}{\text{Ro}} \right) \underline{V}^{\perp} \right] = -P, \\ \partial_t^{\sigma} \omega + \frac{\varepsilon}{\mu} \left(U^{\mu} \cdot \nabla_{X,z}^{\sigma,\mu} \right) \omega = \frac{\varepsilon}{\mu} \left(\omega \cdot \nabla_{X,z}^{\sigma,\mu} \right) U^{\mu} + \frac{\varepsilon}{\mu \text{Ro}} \partial_z^{\sigma} U^{\mu}. \end{cases}$$
(4.46)

The following quantity is the energy that we will use to get the local wellposedness

$$\mathcal{E}^{N}(\zeta,\psi,\omega) = \frac{1}{2} \left| \zeta \right|_{H^{N}}^{2} + \frac{1}{2} \left| \mathfrak{P}\psi \right|_{H^{3}}^{2} + \frac{1}{2} \sum_{1 \le |\alpha| \le N} \left| \mathfrak{P}\psi_{(\alpha)} \right|_{2}^{2} + \frac{1}{2} \left| |\omega| \right|_{H^{N-1}}^{2} + \frac{1}{2} \left| \frac{1}{\mathfrak{P}} \left(\omega_{b} \cdot N_{b}^{\mu} \right) \right|_{2}^{2},$$

where we recall that $\psi_{(\alpha)}$ is defined in (4.43). For $T \ge 0$, we also introduce the energy space

$$\mathbf{E}_T^N = \left\{ (\zeta, \psi, \omega) \in \mathcal{C}\left([0, T], H^2(\mathbb{R}^d) \times \dot{H}^2(\mathbb{R}^d) \times H^2(\mathcal{S}) \right), \mathcal{E}^N(\zeta, \psi, \omega) \in L^\infty([0, T]) \right\}.$$

We also recall that M_N is defined in (4.24). We organize this section in three parts. First, we give an a priori estimate for the vorticity. Then, we explain briefly how we can quasilinearize the system and how we obtain a priori estimates for the full system. The last part of this section is devoted to the proof of the main result.

4.3.2 A priori estimate for the vorticity

In this part, we give a priori estimate for the straightened equation of the vorticity.

Proposition 4.3.1. Let $N \geq 5$, T > 0, $b \in L^{\infty} \cap \dot{H}^{N+1}(\mathbb{R}^d)$ and $(\zeta, \psi, \omega) \in \mathbb{E}_T^N$ such that (4.45) and Condition (4.22) hold on [0, T]. We also assume that on [0, T]

$$\partial_t \zeta - \frac{1}{\mu} \mathbf{U}^{\sigma,\mu} [\varepsilon \zeta, \beta b] \cdot N^{\mu} = 0.$$

Then,

$$\frac{d}{dt}\left(\left|\left|\omega\right|\right|_{H^{N-1}}^{2}+\left|\frac{1}{\mathfrak{P}}\left(\omega_{b}\cdot N_{b}^{\mu}\right)\right|_{2}^{2}\right)\leq M_{N}\left(\varepsilon\mathcal{E}^{N}(\zeta,\psi,\omega)^{\frac{3}{2}}+\max\left(\varepsilon,\frac{\varepsilon}{\mathrm{Ro}}\right)\mathcal{E}^{N}(\zeta,\psi,\omega)\right).$$

Proof. We denote $U^{\sigma,\mu}[\varepsilon\zeta,\beta b] = U^{\mu} = \begin{pmatrix} \sqrt{\mu}V \\ W \end{pmatrix}$. We can reformulate Equation (4.45) as $\partial_t \omega + \varepsilon (V \cdot \nabla_X) \omega + \frac{\varepsilon}{\mu} a \partial_z \omega = \frac{\varepsilon}{\mu} \left(\omega \cdot \nabla_{X,z}^{\sigma,\mu} \right) U^{\mu} + \frac{\varepsilon}{\mu \text{Ro}} \partial_z^{\sigma} U^{\mu},$

where

$$a = \frac{1}{1 + \partial_z \sigma} \left(\mathbf{U}^{\mu} \cdot \begin{pmatrix} -\sqrt{\mu} \nabla_X \sigma \\ 1 \end{pmatrix} - (z+1) \underline{\mathbf{U}}^{\mu} \cdot N^{\mu} \right).$$

Notice that $\underline{a} = a_b = 0$. Then, we get

$$\partial_t ||\omega||_2^2 = \varepsilon \int_S \left(\nabla_X \cdot \mathbf{V} + \frac{1}{\mu} \partial_z a \right) \omega^2 + \frac{2}{\mu} \left(\omega \cdot \nabla_{X,z}^{\sigma,\mu} \right) \mathbf{U}^{\mu} \cdot \omega + \frac{1}{\mathrm{Ro}} \partial_z^{\sigma} \mathbf{U}^{\mu} \cdot \omega,$$

and

$$\begin{split} \partial_t \left| \left| \omega \right| \right|_2^2 &\leq \frac{\varepsilon}{\mu} C \bigg(\frac{1}{h_{\min}}, \varepsilon \left| \zeta \right|_{W^{1,\infty}}, \beta \left| b \right|_{W^{1,\infty}} \bigg) \bigg(\left[\left| \left| \nabla_{X,z}^{\mu} \mathbf{U}^{\mu} \right| \right|_{\infty} + \sqrt{\mu} \left| \left| \mathbf{U}^{\mu} \right| \right|_{\infty} \right] \left| \left| \omega \right| \right|_2^2 \\ &+ \frac{1}{\text{Ro}} \left| \left| \nabla_{X,z}^{\mu} \mathbf{U}^{\mu} \right| \right|_{\infty} \left| \left| \omega \right| \right|_2 \bigg), \end{split}$$

where we use the fact that

$$|\underline{\mathbf{U}}^{\mu} \cdot N^{\mu}|_{L^{\infty}} \leq C\left(\varepsilon \left|\zeta\right|_{W^{1,\infty}}, \beta \left|b\right|_{W^{1,\infty}}\right)\left(\left|\left|\partial_{z}\mathbf{U}^{\mu}\right|\right|_{\infty} + \sqrt{\mu}\left|\left|\mathbf{U}^{\mu}\right|\right|_{\infty}\right).$$

The estimate for the L^2 -norm of ω follows thanks to Theorem, 4.2.8, Theorem 4.2.11 and Remark 4.2.14. For the high order estimates, we differentiate Equation (4.45) and we easily obtain the

control thanks to Theorem 4.2.11 and Remark 4.2.14 (see the proof of Proposition 4.1 in [34]). Finally, taking the trace at the bottom of the vorticity equation in System (4.20), we get the following equation for $\omega_b \cdot N_b^{\mu}$,

$$\partial_t \left(\omega_b \cdot N_b^{\mu}\right) + \varepsilon \nabla \cdot \left(\left[\omega_b \cdot N_b^{\mu} + \frac{1}{\text{Ro}} \right] \mathbf{V}_b \right) = 0, \tag{4.47}$$

and then,

$$\partial_t \left| \frac{1}{\mathfrak{P}} \left(\omega_b \cdot N_b^{\mu} \right) \right|_2^2 \le 2\varepsilon \left| \sqrt{1 + \sqrt{\mu} |D|} \left(\left[\omega_b \cdot N_b^{\mu} + \frac{1}{\mathrm{Ro}} \right] \mathrm{V}_b \right) \right|_2 \left| \frac{1}{\mathfrak{P}} \left(\omega_b \cdot N_b^{\mu} \right) \right|_2.$$

The control follows easily thanks to and Lemma 4.2.9, Theorem 4.2.8, Theorem 4.2.11 and Remark 4.2.14. $\hfill \Box$

Remark 4.3.2. Notice that we can also take the trace at the surface of the vorticity equation and we obtain a transport equation for $\underline{\omega} \cdot N^{\mu}$,

$$\partial_t \left(\underline{\omega} \cdot N^{\mu}\right) + \varepsilon \nabla \cdot \left(\left[\underline{\omega} \cdot N^{\mu} + \frac{1}{\text{Ro}} \right] \underline{V} \right) = 0.$$
 (4.48)

4.3.3 Quasilinearization and a priori estimates

In this part, we quasilinearize the system (4.20). We introduce the Rayleigh-Taylor coefficient

$$\mathfrak{a} := \mathfrak{a}[\varepsilon\zeta,\beta b](\psi,\omega) = 1 + \varepsilon \left(\partial_t + \varepsilon \underline{\mathbf{V}}[\varepsilon\zeta,\beta b](\psi,\omega) \cdot \nabla\right) \underline{\mathbf{w}}[\varepsilon\zeta,\beta b](\psi,\omega).$$
(4.49)

It is well-known that the positivity of this quantity is essential for the wellposedness of the water waves equations (see for instance Remark 4.17 in [80] or [56]). Thanks to Equation (4.12), we can easily adapt Part 2.2.5 and check that the positivity of \mathfrak{a} is equivalent to the classical Rayleigh-Taylor criterion ([138])

$$\inf_{\mathbb{R}^d} \left(-\partial_z \mathcal{P}_{|z=\varepsilon\zeta} \right) > 0,$$

where we recall that \mathcal{P} is the pressure in the fluid domain. We can now give a quasilinearization of (4.46). We recall that the notation $\underline{U}^{\mu}_{(\alpha)}$ is defined in (4.44) and $\psi_{(\alpha)}$ is defined in (4.43).

Proposition 4.3.3. Let $N \geq 5$, T > 0, $b \in L^{\infty} \cap \dot{H}^{N+1}(\mathbb{R}^d)$, $P \in L_t^{\infty}(\mathbb{R}^+; \dot{H}_X^{N+1}(\mathbb{R}^d))$ and $(\zeta, \psi, \omega) \in E_T^N$ solution of the system (4.46) such that (ζ, b) satisfy Condition (4.22) on [0, T]. Then, for $\alpha, \gamma \in \mathbb{N}^d$ and for $k \in \{x, y\}$ such that $\partial^{\alpha} = \partial_k \partial^{\gamma}$ and $|\gamma| \leq N - 1$, we have the following quasilinearization

$$(\partial_t + \varepsilon \underline{\mathbf{V}} \cdot \nabla) \,\partial^\alpha \zeta - \frac{1}{\mu} \partial_k \underline{\mathbf{U}}^\mu_{(\gamma)} \cdot N^\mu = R^1_\alpha, (\partial_t + \varepsilon \underline{\mathbf{V}} \cdot \nabla) \left(\underline{\mathbf{U}}^\mu_{(\gamma)/\!\!/} \cdot e_{\mathbf{k}} \right) + \mathfrak{a} \partial^\alpha \zeta = -\partial^\alpha P + R^2_\alpha,$$
(4.50)

where

$$\left|R_{\alpha}^{1}\right|_{2}+\left|R_{\alpha}^{2}\right|_{2}+\left|\mathfrak{P}R_{\alpha}^{2}\right|_{2}\leq M_{N}\left(\max\left(\varepsilon,\frac{\varepsilon}{\mathrm{Ro}}\right)\mathcal{E}^{N}(\zeta,\psi,\omega)+\frac{\varepsilon}{\mathrm{Ro}}\sqrt{\mathcal{E}^{N}(\zeta,\psi,\omega)}\right).$$
(4.51)

Before proving this result, we introduce the following notation. For $\alpha \in \mathbb{N}^d$, $f, g \in H^{|\alpha|-1}(\mathbb{R}^d)$, we define the symmetric commutator

$$[\partial^{\alpha}, f, g] = \partial^{\alpha} (fg) - g \partial^{\alpha} f - f \partial^{\alpha} g.$$

Proof. Firstly, we apply ∂^{α} to the first equation of (4.46)

$$\partial_t \partial^\alpha \zeta + \varepsilon \underline{\mathbf{V}} \cdot \nabla \partial^\alpha \zeta + \varepsilon \partial^\alpha \underline{\mathbf{V}} \cdot \nabla \zeta - \frac{1}{\mu} \partial^\alpha \underline{\mathbf{w}} + \varepsilon \left[\partial^\alpha, \underline{\mathbf{V}}, \nabla \zeta \right] = 0.$$

Using Theorem 4.2.11 and the trace Lemma 4.2.12, we get the first equality. For the second equality we get, after applying ∂_k to the second equation of (4.20),

$$\begin{split} \partial_t \partial_k \psi + \partial_k \zeta + \varepsilon \underline{\mathbf{V}} \cdot \left(& \left(\partial_k \nabla \psi - \varepsilon \underline{\mathbf{w}} \nabla \partial_k \zeta \right) + \partial_k \nabla^\perp \widetilde{\psi} \right) - \frac{\varepsilon}{\mu} \underline{\mathbf{w}} \partial_k \left(\underline{\mathbf{U}}^\mu \cdot N^\mu \right) \\ & - \varepsilon \partial_k \frac{\nabla^\perp}{\Delta} \cdot \left(\left(\underline{\omega} \cdot N^\mu + \frac{1}{\mathrm{Ro}} \right) \underline{\mathbf{V}} \right) = -\partial_k P. \end{split}$$

Then, applying ∂^{γ} and using Lemma 4.3 in [34] (we can easily adapt it thanks to Theorem 4.2.11 and Lemma 4.2.13) we get

$$\begin{split} \partial_t \partial^\alpha \psi + \partial^\alpha \zeta + \varepsilon \underline{\mathbf{V}} \cdot \left((\partial^\alpha \nabla \psi - \varepsilon \underline{\mathbf{w}} \nabla \partial^\alpha \zeta) + \partial^\alpha \nabla^\perp \widetilde{\psi} \right) \\ &- \frac{\varepsilon}{\mu} \underline{\mathbf{w}} \partial^\alpha \left(\underline{\mathbf{U}}^\mu \cdot N^\mu \right) - \varepsilon \partial^\alpha \frac{\nabla^\perp}{\Delta} \cdot \left(\left(\underline{\omega} \cdot N^\mu + \frac{1}{\mathrm{Ro}} \right) \underline{\mathbf{V}} \right) = -\partial^\alpha P + \widetilde{R_\alpha^2}, \end{split}$$

where $\widetilde{R_{\alpha}^2}$ is controlled

$$\left|\widetilde{R_{\alpha}^{2}}\right|_{2} + \left|\mathfrak{P}\widetilde{R_{\alpha}^{2}}\right|_{2} \le \varepsilon M_{N} \mathcal{E}^{N}(\zeta, \psi, \omega).$$

$$(4.52)$$

Using the first equation of (4.20) and the fact that $\Delta \tilde{\psi} = \underline{\omega} \cdot N^{\mu}$, we obtain

$$\begin{split} \partial_t \psi_{(\alpha)} + \mathfrak{a} \partial^\alpha \zeta + \varepsilon \underline{\mathbf{V}} \cdot \nabla \psi_{(\alpha)} + \frac{\varepsilon}{\mathrm{Ro}} \partial^\alpha \frac{\nabla^\perp}{\Delta} \cdot \underline{\mathbf{V}} + \partial^\alpha P &= \varepsilon \partial^\alpha \frac{\nabla^\perp}{\Delta} \cdot (\underline{\omega} \cdot N^\mu \underline{\mathbf{V}}) \\ &\quad - \varepsilon \underline{\mathbf{V}} \cdot \nabla^\perp \partial^\alpha \widetilde{\psi} + \widetilde{R}^2_\alpha \\ &= \varepsilon \sum_{k \in \{1,2\}} (-1)^{k+1} \Big[\partial^\alpha \frac{\partial_k}{\Delta}, \underline{\mathbf{V}}_{3-k} \Big] \left(\underline{\omega} \cdot N^\mu \right) \\ &\quad + \widetilde{R}^2_\alpha \\ &:= \widetilde{R}^3_\alpha + \widetilde{R}^2_\alpha, \end{split}$$

where $\partial_1 = \partial_x$ and $\partial_2 = \partial_y$. Then, using Theorem 3 in [79], Lemma B.1.2 and Lemma 4.2.9 we get

$$\left|\widetilde{R_{\alpha}^{3}}\right|_{2} + \left|\mathfrak{P}\widetilde{R_{\alpha}^{3}}\right|_{2} \leq \varepsilon M_{N} \left|\left|\mathbf{V}\right|\right|_{H^{N,1}} \left|\left|\omega\right|\right|_{H^{N-1,1}} + \varepsilon \left|\mathfrak{P}\frac{\nabla^{\perp}}{\Delta} \cdot \left(\underline{\omega} \cdot N^{\mu}\partial^{\alpha}\underline{\mathbf{V}}\right)\right|_{2}.$$

Furthermore,

$$\begin{split} \left| \mathfrak{P} \frac{\nabla^{\perp}}{\Delta} \cdot (\underline{\omega} \cdot N^{\mu} \partial^{\alpha} \underline{\mathbf{V}}) \right|_{2} &\leq \left| \frac{1}{\sqrt{1 + \sqrt{\mu} |D|}} \left(\underline{\omega} \cdot N^{\mu} \partial^{\alpha} \underline{\mathbf{V}} \right) \right|_{2}, \\ &\leq \left| \frac{1}{\sqrt{1 + \sqrt{\mu} |D|}} \left(\partial_{k} \left(\underline{\omega} \cdot N^{\mu} \right) \partial^{\gamma} \underline{\mathbf{V}} \right) \right|_{2} + \left| \mathfrak{P} \left(\underline{\omega} \cdot N^{\mu} \partial^{\gamma} \underline{\mathbf{V}} \right) \right|_{2}, \\ &\leq C \left(\varepsilon \left| \zeta \right|_{H^{N}} \right) \left| \underline{\omega} \right|_{H^{N-2}} \left(\left| \underline{\mathbf{V}} \right|_{H^{N-1}} + \left| \mathfrak{P} \partial^{\gamma} \underline{\mathbf{V}} \right|_{2} \right), \end{split}$$

where we use Lemma B.1.3. The first term is controlled thanks to the trace Lemma 4.2.12 and Theorem 4.2.11. For the second term, we have

$$\partial^{\gamma}\underline{\mathbf{V}} = \nabla\partial^{\gamma}\psi - \varepsilon\underline{\mathbf{w}}\nabla\partial^{\gamma}\zeta - \varepsilon\partial^{\gamma}\underline{\mathbf{w}}\nabla\zeta + \nabla^{\perp}\partial^{\gamma}\widetilde{\psi} - \varepsilon\left[\partial^{\gamma},\underline{\mathbf{w}},\nabla\zeta\right],$$

and the control follows from Lemma B.1.2, Lemma 4.2.9, Theorem 4.2.11 and Lemma 4.2.13. Then, we obtain

$$\partial_t \psi_{(\alpha)} + \mathfrak{a} \partial^\alpha \zeta + \varepsilon \underline{\mathbf{V}} \cdot \nabla \psi_{(\alpha)} + \frac{\varepsilon}{\mathrm{Ro}} \partial^\alpha \frac{\nabla^\perp}{\Delta} \cdot \underline{\mathbf{V}} + \partial^\alpha P = \widetilde{\widetilde{R_\alpha}^2},$$

where $\widetilde{\widetilde{R}_{\alpha}^2}$ satisfied also the estimate (4.52). Finally, we can adapt Lemma 4.4 in [34] thanks to Remark 4.3.2, Theorem 4.2.11 and Proposition 4.2.15 and we get

$$\partial_t \psi_{(\alpha)} = \partial_t \left(\mathbf{U}^{\mu}_{(\gamma)/\!\!/} \cdot \mathbf{e}_k \right) + \widetilde{R_{\alpha}},$$

where \widetilde{R}_{α} satisfies the same estimate as R_2 in (4.51). The third equality is a direct consequence of Proposition 4.3.1.

In order to establish an a priori estimate we need to control the Rayleigh-Taylor coefficient \mathfrak{a} . The following Proposition is adapted from Proposition 2.2.11.

Proposition 4.3.4. Let T > 0, $t_0 > \frac{d}{2}$, $N \ge 5$, $(\zeta, \psi, \omega) \in E_T^N$ is a solution of the water waves equations (4.46), $P \in L^{\infty}(\mathbb{R}^+; \dot{H}^{N+1}(\mathbb{R}^d))$ and $b \in L^{\infty} \cap \dot{H}^{N+1}(\mathbb{R}^d)$, such that Condition (4.22) is satisfied. We assume also that $\varepsilon, \beta, Ro, \mu$ satisfy (4.23). Then, for all $0 \le t \le T$,

$$\|\mathfrak{a}-1\|_{W^{1,\infty}} \leq C\left(M_N, \varepsilon\sqrt{\mathcal{E}^N(\zeta,\psi,\omega)}\right)\varepsilon\sqrt{\mathcal{E}^N(\zeta,\psi,\omega)} + \varepsilon M_N \left|\nabla P\right|_{L^{\infty}_t H^N_X}$$

Furthermore, if $\partial_t P \in L^{\infty}(\mathbb{R}^+; \dot{H}^N(\mathbb{R}^d))$, then,

$$\left|\partial_t \mathfrak{a}\right|_{L^{\infty}} \leq C \left(M_N, \left|\nabla P\right|_{L^{\infty}_t H^N_X}, \varepsilon \sqrt{\mathcal{E}^N(\zeta, \psi, \omega)} \right) \varepsilon \sqrt{\mathcal{E}^N(\zeta, \psi, \omega)} + \varepsilon M_N \left|\nabla P\right|_{W^{1,\infty}_t H^N_X}.$$

Proof. Using Proposition 4.2.15 we get that

$$\mathfrak{a}[\varepsilon\zeta,\beta b](\psi,\omega) = 1 + \varepsilon^{2}\underline{\mathbf{V}}\cdot\nabla\underline{\mathbf{w}} + \varepsilon\partial_{t}\zeta\underline{\partial_{z}^{\sigma}}\underline{\mathbf{w}} + \varepsilon\underline{\mathbf{w}}[\varepsilon\zeta,\beta b]\left(\partial_{t}\psi - \varepsilon\underline{\mathbf{w}}[\varepsilon\zeta,\beta b](\psi,\omega)\partial_{t}\zeta + \varepsilon\sqrt{\mu}\frac{\nabla}{\Delta}\cdot\left(\underline{\omega_{h}}^{\perp}\partial_{t}\zeta\right),\partial_{t}^{\sigma}\omega\right).$$
(4.53)

Then, using the equations satisfied by (ζ, ψ, ω) , Theorems 4.2.8 and 4.2.11, Remark 4.2.14 and standard controls, we easily get the first inequality. The second inequality can be proved similarly.

We can now establish an a priori estimate for the Castro-Lannes System with a Coriolis forcing under the positivity on the Rayleigh-Taylor coefficient

$$\exists \, \mathfrak{a}_{\min} > 0 \,, \, \mathfrak{a} \ge \mathfrak{a}_{\min}. \tag{4.54}$$

Theorem 4.3.5. Let $N \geq 5$, T > 0, $b \in L^{\infty} \cap \dot{H}^{N+2}(\mathbb{R}^d)$, $P \in L_t^{\infty}(\mathbb{R}^+; \dot{H}^{N+1}(\mathbb{R}^d))$ and $(\zeta, \psi, \omega) \in E_T^N$ solution of the water waves equations (4.46) such that (ζ, b) satisfy Condition (4.22) and $\mathfrak{a}[\varepsilon\zeta, \beta b](\psi, \omega)$ satisfies (4.54) on [0, T]. We assume also that $\varepsilon, \beta, Ro, \mu$ satisfy (4.23). Then, for all $t \in [0, T]$,

$$\frac{d}{dt}\mathcal{E}^{N}(\zeta,\psi,\omega) \leq C\left(\mu_{\max},\frac{1}{h_{\min}},\varepsilon\sqrt{\mathcal{E}^{N}(\zeta,\psi,\omega)},\beta \left|\nabla b\right|_{H^{N+1}},\beta \left|b\right|_{L^{\infty}},\left|\nabla P\right|_{W^{1,\infty}_{t}H^{N}_{X}}\right) \times \left(\varepsilon\mathcal{E}^{N}(\zeta,\psi,\omega)^{\frac{3}{2}} + \max\left(\varepsilon,\beta,\frac{\varepsilon}{\operatorname{Ro}}\right)\mathcal{E}^{N}(\zeta,\psi,\omega) + \left|\nabla P\right|_{L^{\infty}_{t}H^{N}_{X}}\sqrt{\mathcal{E}^{N}(\zeta,\psi,\omega)}\right).$$
(4.55)

Proof. Compared to [34], we have here a non flat bottom, a Coriolis forcing and a non constant pressure. We focus on these terms. Inspired by [34] we can symmetrize the Castro-Lannes system. We define a modified energy

$$\mathcal{F}^{N}(\psi,\zeta,\omega) = \frac{1}{2} \Big(\left| \left| \boldsymbol{\omega} \right| \right|_{H^{N-1}}^{2} + \left| \frac{1}{\mathfrak{P}} \left(\omega_{b} \cdot N_{b}^{\mu} \right) \right|_{2}^{2} + \sum_{|\alpha| \leq 3} \left| \partial^{\alpha} \zeta \right|_{2}^{2} + \frac{1}{\mu} \int_{\mathcal{S}} \left(1 + \partial_{z} \sigma \right) \left| \partial^{\alpha} \mathbf{U}^{\mu} \right|^{2} + \sum_{k=x,y,1 \leq |\gamma| \leq N-1} \left| \mathfrak{a}_{k} \partial^{\gamma} \zeta, \partial_{k} \partial^{\gamma} \zeta \right| + \frac{1}{\mu} \int_{\mathcal{S}} \left(1 + \partial_{z} \sigma \right) \left| \partial_{k} \mathbf{U}_{(\gamma)}^{\mu} \right|^{2} \Big).$$

$$(4.56)$$

From Proposition 4.2.16 and Proposition 4.2.17 we get

$$\mathcal{E}^{N}(\psi,\zeta,\omega) \leq C\left(rac{1}{\mathfrak{a}_{\min}},M_{N}
ight)\mathcal{F}^{N}(\psi,\zeta,\omega)$$

and from Theorem 4.2.8, Theorem 4.2.11, Remark 4.2.14 and Proposition 4.3.4 we obtain that

$$\mathcal{F}^{N}(\psi,\zeta,\omega) \leq C\left(\frac{1}{h_{\min}},\beta\left|b\right|_{L^{\infty}},\beta\left|\nabla b\right|_{H^{N}},\left|\nabla P\right|_{L^{\infty}_{t}H^{N}_{X}},\varepsilon\sqrt{\mathcal{E}^{N}(\psi,\zeta,\omega)}\right)\mathcal{E}^{N}(\psi,\zeta,\omega).$$

Hence, in the following we estimate $\frac{d}{dt}\mathcal{F}^{N}(\psi,\zeta,\omega)$. We already did the work for the vorticity in Proposition 4.3.1. In the following R will be a remainder whose exact value has no importance and satisfying

$$|R|_{2} \leq C \left(\frac{1}{h_{\min}}, \beta |b|_{L^{\infty}}, \beta |\nabla b|_{H^{N+1}}, |\nabla P|_{W^{1,\infty}_{t}H^{N}_{X}}, \varepsilon \sqrt{\mathcal{E}^{N}(\psi, \zeta, \omega)}\right) \mathcal{E}^{N}(\psi, \zeta, \omega).$$
(4.57)

We start by the low order terms. Let $\alpha \in \mathbb{N}^d$, $|\alpha| \leq 3$. We apply ∂^{α} to the first equation of System (4.46) and we multiply it by ζ . Then, we apply ∂^{α} to the second equation and we multiply it by $\frac{1}{\mu} \underline{U}^{\mu} \cdot N^{\mu}$. By summing these two equations, we obtain, thanks to Theorem 4.2.8, Theorem 4.2.11, Remark 4.2.14 and the trace Lemma,

$$\frac{1}{2}\partial_{t}\left(\partial^{\alpha}\zeta,\partial^{\alpha}\zeta\right) + \left(\partial_{t}\partial^{\alpha}\psi,\frac{1}{\mu}\partial^{\alpha}\underline{\mathbf{U}}^{\mu}\cdot\mathbf{N}^{\mu}\right) + \frac{\varepsilon}{\mathrm{Ro}}\left(\frac{\nabla}{\Delta}\cdot\partial^{\alpha}\underline{\mathbf{V}}^{\perp},\frac{1}{\mu}\partial^{\alpha}\underline{\mathbf{U}}^{\mu}\cdot\mathbf{N}^{\mu}\right) \\
+ \left(\partial^{\alpha}P,\frac{1}{\mu}\partial^{\alpha}\underline{\mathbf{U}}^{\mu}\cdot\mathbf{N}^{\mu}\right) \leq \varepsilon \left|R\right|_{2}.$$
(4.58)

Furthermore, using again the same Propositions as before, we get

$$\frac{\varepsilon}{\operatorname{Ro}}\left(\frac{\nabla}{\Delta}\cdot\partial^{\alpha}\underline{V}^{\perp},\frac{1}{\mu}\partial^{\alpha}\underline{U}^{\mu}\cdot N^{\mu}\right) + \left(\partial^{\alpha}P,\frac{1}{\mu}\partial^{\alpha}\underline{U}^{\mu}\cdot N^{\mu}\right) \leq \frac{\varepsilon}{\operatorname{Ro}}\left|R\right|_{2} + M_{N}\left|\nabla P\right|_{L_{t}^{\infty}H_{X}^{N}}\sqrt{\mathcal{E}^{N}(\psi,\zeta,\omega)}.$$

Then, we have to link $(\partial_t \partial^{\alpha} \psi, \partial^{\alpha} \underline{U}^{\mu} \cdot N^{\mu})$ to $\partial_t \int_{\mathcal{S}} (1 + \partial_z \sigma) |\partial^{\alpha} U^{\mu}|^2$. Remarking that $\psi = \underline{\phi}$, where ϕ satisfies

$$\begin{cases} \nabla_{X,z}^{\mu} \cdot P(\Sigma) \nabla_{X,z}^{\mu} \phi = 0 \text{ in } \mathcal{S}, \\ \phi_{|z=0} = \psi, \, \mathbf{e}_{z} \cdot P(\Sigma) \nabla^{\mu} \phi_{|z=-1} = 0, \end{cases}$$

$$(4.59)$$

we get thanks to Green's identity

Then, notice that $\partial_k = \partial_k^{\sigma} + \partial_k \sigma \partial_z^{\sigma}$ for $k \in \{t, x, y\}$ and ∂_k^{σ} and $\nabla_{X,z}^{\sigma,\mu}$ commute. We differentiate Equation (4.59) with respect to t and we obtain thanks to Theorems 4.2.8, 4.2.11, Proposition 4.2.15 and Lemma 2.38 in [80] (irrotational theory),

$$\begin{pmatrix} \partial_t \partial^{\alpha} \psi, \frac{1}{\mu} \partial^{\alpha} \underline{\mathbf{U}}^{\mu} \cdot N^{\mu} \end{pmatrix} = \frac{1}{\mu} \int_{\mathcal{S}} (1 + \partial_z \sigma) \, \partial_t^{\sigma} \partial^{\sigma, \alpha} \nabla_{X, z}^{\sigma, \mu} \phi \cdot \partial^{\alpha} \mathbf{U}^{\mu} \\ + \left(\partial_t \partial^{\alpha} \phi_b, \frac{1}{\mu} \partial^{\alpha} \mathbf{U}_b^{\mu} \cdot N_b^{\mu} \right) + \max(\varepsilon, \beta) R.$$

Using the fact that $w_b = \mu \beta \nabla b \cdot V_b$, we get

$$\left(\partial_t \partial^\alpha \phi_b, \frac{1}{\mu} \partial^\alpha \mathbf{U}_b^\mu \cdot N_b^\mu\right) \leq \beta M_N \left|\partial_t \partial^\alpha \phi_b\right| \sqrt{\mathcal{E}^N(\psi, \zeta, \omega)}.$$

Then, by the trace Lemma, we finally obtain

$$\left(\partial_t \partial^\alpha \phi_b, \frac{1}{\mu} \partial^\alpha \mathbf{U}_b^\mu \cdot N_b^\mu\right) \leq \beta \left|R\right|_2.$$

Furthermore, remarking that $U^{\mu} = \nabla_{X,z}^{\sigma,\mu} \phi + U^{\sigma,\mu}[\varepsilon\zeta,\beta b](0,\omega)$, we obtain, thanks to Proposition 4.2.15, Theorem 4.2.8 and Theorem 4.2.11,

$$\left(\partial_t \partial^\alpha \psi, \frac{1}{\mu} \partial^\alpha \underline{\mathbf{U}}^\mu \cdot N^\mu\right) = \frac{1}{\mu} \int_{\mathcal{S}} \left(1 + \partial_z \sigma\right) \partial_t \partial^\alpha \mathbf{U}^\mu \cdot \partial^\alpha \mathbf{U}^\mu + \max\left(\varepsilon, \beta, \frac{\varepsilon}{\mathrm{Ro}}\right) R.$$

Using the following identity

$$\partial_t \int_{\mathcal{S}} (1 + \partial_z \sigma) fg = \int_{\mathcal{S}} (1 + \partial_z \sigma) \partial_t^{\sigma} fg + \int_{\mathcal{S}} (1 + \partial_z \sigma) f \partial_t^{\sigma} g + \int_{\mathbb{R}^d} \varepsilon \partial_t \zeta \underline{fg}, \qquad (4.60)$$

we obtain that

$$\frac{1}{\mu}\partial_t \int_{\mathcal{S}} (1+\partial_z \sigma) |\partial^{\alpha} \mathbf{U}^{\mu}|^2 \leq \max\left(\varepsilon, \beta, \frac{\varepsilon}{\mathrm{Ro}}\right) |R|_2 + M_N |\nabla P|_{L^{\infty}_t H^N_X} \sqrt{\mathcal{E}^N\left(\psi, \zeta, \omega\right)}.$$

To control the high order terms of $\mathcal{F}^{N}(\psi, \zeta, \omega)$ we adapt Step 2 in Proposition 4.5 in [80]. Thanks to Proposition 4.3.3, we have

$$(\partial_t + \varepsilon \underline{\mathbf{V}} \cdot \nabla) \, \partial^{\alpha} \zeta - \frac{1}{\mu} \partial_k \underline{\mathbf{U}}^{\mu}_{(\gamma)} \cdot N^{\mu} = R^1_{\alpha}, \\ (\partial_t + \varepsilon \underline{\mathbf{V}} \cdot \nabla) \left(\underline{\mathbf{U}}^{\mu}_{(\gamma)/\!\!/} \cdot e_{\mathbf{k}} \right) + \mathfrak{a} \partial^{\alpha} \zeta = -\partial^{\alpha} P + R^2_{\alpha}.$$

Then, we multiply the first equation by $\mathfrak{a}\partial^{\alpha}\zeta$ and the second by $\frac{1}{\mu}\partial_k\underline{U}^{\mu}_{(\gamma)}\cdot N^{\mu}$ and we integrate over \mathbb{R}^d . Then, using Propositions 4.2.8, 4.2.18 and 4.3.4,

$$\begin{split} \frac{1}{2}\partial_t \left(\mathfrak{a}\partial^{\alpha}\zeta,\partial\zeta\right) + \left(\left(\partial_t + \varepsilon\underline{\mathbf{V}}\cdot\nabla\right)\left(\underline{\mathbf{U}}_{(\gamma)/\!\!/}^{\mu}\cdot e_{\mathbf{k}}\right), \frac{1}{\mu}\partial_k\underline{\mathbf{U}}_{(\gamma)}^{\mu}\cdot N^{\mu}\right) &\leq \varepsilon\mathcal{E}^N(\psi,\zeta,\omega)^{\frac{3}{2}} \\ &+ \max\left(\varepsilon,\frac{\varepsilon}{\mathrm{Ro}}\right)|R|_2 + M_N|\nabla P|_{L_t^{\infty}H_X^N}\sqrt{\mathcal{E}^N(\psi,\zeta,\omega)}. \end{split}$$

We remark that

$$\left(\partial_t + \varepsilon \underline{\mathbf{V}} \cdot \nabla\right) \left(\underline{\mathbf{U}}_{(\gamma)/\!\!/}^{\mu} \cdot e_{\mathbf{k}}\right) = \underline{\left(\partial_t^{\sigma} + \frac{\varepsilon}{\mu} \mathbf{U}^{\mu} \cdot \nabla^{\sigma,\mu}\right) \left(\mathbf{U}_{(\gamma)/\!\!/}^{\mathfrak{b},\mu} \cdot e_{\mathbf{k}}\right)},$$

where $U^{\mathfrak{b},\mu}_{(\gamma)/\!\!/} = V_{(\gamma)} + w_{(\gamma)} \nabla \sigma$. Then, we have

$$\begin{split} \left(\left(\partial_t + \varepsilon \underline{\mathbf{V}} \cdot \nabla\right) \left(\underline{\mathbf{U}}_{(\gamma)/\!\!/}^{\mu} \cdot e_{\mathbf{k}} \right), \frac{1}{\mu} \partial_k \underline{\mathbf{U}}_{(\gamma)}^{\mu} \cdot N^{\mu} \right) \\ &= \frac{1}{\mu} \int_{\mathcal{S}} \left(1 + \partial_z \sigma \right) \left(\partial_t^{\sigma} + \frac{\varepsilon}{\mu} \mathbf{U}^{\mu} \cdot \nabla^{\sigma,\mu} \right) \left(\mathbf{U}_{(\gamma)/\!\!/}^{\mathfrak{b},\mu} \cdot e_{\mathbf{k}} \right) \nabla_{X,z}^{\sigma,\mu} \cdot \left(\partial_k \mathbf{U}_{(\gamma)}^{\mu} \right) \\ &+ \frac{1}{\mu} \int_{\mathcal{S}} \left(1 + \partial_z \sigma \right) \nabla_{X,z}^{\sigma,\mu} \left(\partial_t^{\sigma} + \frac{\varepsilon}{\mu} \mathbf{U}^{\mu} \cdot \nabla^{\sigma,\mu} \right) \left(\mathbf{U}_{(\gamma)/\!\!/}^{\mathfrak{b},\mu} \cdot e_{\mathbf{k}} \right) \left(\partial_k \mathbf{U}_{(\gamma)}^{\mu} \right) \\ &+ \left(\left(\partial_t + \varepsilon \mathbf{V}_b \cdot \nabla \right) \left(\mathbf{U}_{(\gamma)/\!\!/}^{\mathfrak{b},\mu} \cdot e_{\mathbf{k}} \right)_b, \frac{1}{\mu} \partial_k \left(\mathbf{U}_{(\gamma)}^{\mu} \right)_b \cdot N_b^{\mu} \right). \end{split}$$

We focus on the last term (bottom contribution). The two other terms can be controlled as in Step 2 in Proposition 4.5 in [34]. Using the same computations as in Proposition 4.2.18, we have

$$\frac{1}{\mu}\partial_k \left(\mathbf{U}^{\mu}_{(\gamma)} \right)_b \cdot N^{\mu}_b = -\mu\beta\nabla\partial^{\alpha}b \cdot \mathbf{V}_b + \text{l.o.t},$$

where l.o.t stands for lower order terms that can be controlled by the energy. Then, since $b \in \dot{H}^{N+2}(\mathbb{R}^d)$, we have by standard controls,

$$\left\|\frac{1}{\mu}\partial_k\left(\mathbf{U}^{\mu}_{(\gamma)}\right)_b\cdot N^{\mu}_b\right\|_{H^{\frac{1}{2}}} \leq \beta \left|\nabla b\right|_{H^{N+1}}\sqrt{\mathcal{E}^N(\psi,\zeta,\omega)}.$$

Furthermore, using Propositions 4.2.8, 4.2.11 and 4.2.15 and standard controls, we have

$$\left| \left(\partial_t + \varepsilon \mathbf{V}_b \cdot \nabla \right) \left(\mathbf{U}_{(\gamma)/\!\!/}^{\mathfrak{b},\mu} \cdot e_{\mathbf{k}} \right)_b \right|_{H^{-\frac{1}{2}}} \le \varepsilon \left| R \right|_2 + M_N \sqrt{\mathcal{E}^N(\psi,\zeta,\omega)},$$

and the control follows easily.

4.3.4 Existence result

We can now establish our existence theorem. Notice that thanks to Equation (4.53), we can define the Rayleigh-Taylor coefficient at time t = 0.

Theorem 4.3.6. Let A > 0, $N \ge 5$, $b \in L^{\infty} \cap \dot{H}^{N+2}(\mathbb{R}^d)$, $P \in W^{1,\infty}(\mathbb{R}^+; \dot{H}^{N+1}(\mathbb{R}^d))$, $(\zeta_0, \psi_0, \omega_0) \in \mathcal{E}_0^N$ such that $\nabla_{X,z}^{\sigma,\mu} \cdot \omega_0 = 0$. We suppose that $(\varepsilon, \beta, \mu, \operatorname{Ro})$ satisfy (4.23). We also assume that

 $\exists h_{\min}, \mathfrak{a}_{\min} > 0$, $\varepsilon \zeta_0 + 1 - \beta b \ge h_{\min}$ and $\mathfrak{a}[\varepsilon \zeta, \beta b](\psi, \omega)_{|t=0} \ge \mathfrak{a}_{\min}$

and

$$\mathcal{E}^{N}(\zeta_{0},\psi_{0},\omega_{0})+|\nabla P|_{L^{\infty}_{\star}H^{N}_{\star}}\leq A$$

Then, there exists T > 0, and a unique solution $(\zeta, \psi, \omega) \in E_T^N$ to the water waves equations (4.46) with initial data $(\zeta_0, \psi_0, \omega_0)$. Moreover,

$$T = \min\left(\frac{T_0}{\max(\varepsilon, \beta, \frac{\varepsilon}{\text{Ro}})}, \frac{T_0}{|\nabla P|_{L_t^{\infty} H_X^N}}\right), \frac{1}{T_0} = c^1 \text{ and } \sup_{t \in [0,T]} \mathcal{E}^N\left(\zeta(t), \psi(t), \omega(t)\right) = c^2,$$

with $c^j = C\left(A, \mu_{\max}, \frac{1}{h_{\min}}, \frac{1}{\mathfrak{a}_{\min}}, |b|_{L^{\infty}}, |\nabla b|_{H^{N+1}}, |\nabla P|_{W_t^{1,\infty} H_X^N}\right).$

Proof. We do not give the proof. It is very similar to Theorem 4.7 in [34]. We can regularize the system (4.46) (see Step 2 of the proof of Theorem 4.7 in [34]) and thanks to the energy estimate of Theorem 4.3.5 we get the existence. The uniqueness mainly follows from a similar proposition to Corollary 3.19 in [34] which shows that the operator $U^{\sigma,\mu}$ has a Lipschitz dependence on its coefficients.

4.4 From the Castro-Lannes system to the Euler equations, reconstruction of the pressure.

In this part, we show how we can reconstruct the pressure in the fluid domain from the Castro-Lannes formulation. This part is similar to the work of Alazard-Burq-Zuily in the irrotational case (see [3]). We work with the dimensionalized version of the Castro-Lannes formulation and we assume that the pressure at the surface is equal to zero and that we do not have a Coriolis forcing. It is not painful to generalize this part when we add them. We recall that the Castro-Lannes formulation is the following system

$$\begin{cases} \partial_t \zeta - \underline{\mathbf{U}} \cdot N = 0, \\ \partial_t \psi + g\zeta + \frac{1}{2} \left| \mathbf{U}_{\mathscr{I}} \right|^2 - \frac{1}{2} \left(1 + |\nabla \zeta|^2 \right) \underline{\mathbf{w}}^2 + \frac{\nabla}{\Delta} \cdot \left(\underline{\omega} \cdot N \underline{\mathbf{V}}^\perp \right) = 0, \\ \partial_t \boldsymbol{\omega} + \left(\mathbf{U} \cdot \nabla_{X,z} \right) \boldsymbol{\omega} = \left(\boldsymbol{\omega} \cdot \nabla_{X,z} \right) \mathbf{U}, \end{cases}$$
(4.61)

where $\mathbf{U} := \mathbf{U}[\zeta, b](\psi, \boldsymbol{\omega})$ is the unique solution in $H^1(\Omega_t)$ of

$$\begin{cases} \operatorname{curl} \mathbf{U} = \boldsymbol{\omega} \text{ in } \Omega_t, \\ \operatorname{div} \mathbf{U} = 0 \text{ in } \Omega_t, \\ U_{\mathbb{I}} = \nabla \psi + \frac{\nabla^{\perp}}{\Delta} (\underline{\omega} \cdot N), \\ U_b \cdot N_b = 0. \end{cases}$$
(4.62)

Our purpose is the reconstruction of the pressure in the fluid domain. The following proposition gives a necessary condition for the pressure \mathcal{P} .

Proposition 4.4.1. Consider a regular solution (ζ, \mathbf{U}) of the free surface Euler equations (4.1) with the boundary conditions (4.2). Then the pressure \mathcal{P} satisfies the Laplace problem

$$\begin{cases} -\Delta_{X,z}\mathcal{P} = \nabla_{X,z} \cdot \left[\left(\mathbf{U} \cdot \nabla_{X,z} \right) \mathbf{U} \right] \text{ in } \Omega_t, \\ \mathcal{P}_{|z=\zeta} = 0, \ \sqrt{1+|\nabla b|^2} \partial_n \mathcal{P}_{|z=-H+b} = -\left[\left(\mathbf{V}_b \cdot \nabla_X \right) \mathbf{U}_b \right] \cdot N_b. \end{cases}$$

Proof. We apply the divergence operator to the first equation of the free surface Euler equations (4.1) and we get the first equation. For the bottom contribution, we take the trace at the bottom of the first equation of the free surface Euler equations (4.1) and since $\mathbf{U}_b \cdot N_b = 0$, we get

$$\left[\left(\mathbf{U}\cdot\nabla_{X,z}\right)\mathbf{U}\right]_{|z=-H+b} = \left(\mathbf{V}_{b}\cdot\nabla_{X}\right)\mathbf{U}_{b}.$$

Thanks to Section A.1, we get that the previous Laplace problem has a unique solution in $\dot{H}^2(\Omega_t)$ if $\zeta, b \in H^3(\mathbb{R}^d)$ and $\mathbf{U} \in H^2(\Omega_t)$. Then we denote by $\mathcal{P} = \mathcal{P}[\zeta, b](\psi, \boldsymbol{\omega})$ the solution of the following system

$$\begin{cases} -\Delta_{X,z}\mathcal{P} = \nabla_{X,z} \cdot \left[\left(\mathbf{U} \cdot \nabla_{X,z} \right) \mathbf{U} \right] \text{ in } \Omega_t, \\ \mathcal{P}_{|z=\zeta} = 0, \sqrt{1 + |\nabla b|^2} \partial_n \mathcal{P}_{|z=-H+b} = -\left[\left(\mathbf{V}_b \cdot \nabla_X \right) \mathbf{U}_b \right] \cdot N_b, \end{cases}$$
(4.63)

where $\mathbf{U} = \mathbf{U}[\zeta, b](\psi, \boldsymbol{\omega})$. In the following, we show that if \mathcal{P} satisfies (4.63), \mathbf{U} satisfies the first equation of (4.1). We denote by \mathcal{K} the quantity

$$\mathcal{K}[\zeta, b](\psi, \boldsymbol{\omega}) = \partial_t \mathbf{U} + \nabla \mathcal{P} + (\mathbf{U} \cdot \nabla_{X, z}) \mathbf{U} + g \mathbf{e}_{\mathbf{z}}$$

where \mathcal{P} satisfies (4.63) and U verifies (4.62).

Proposition 4.4.2. Let $b \in H^3(\mathbb{R}^d)$. If (ζ, ψ, ω) is a smooth solution of the Castro-Lannes equations (4.61), the quantity $\mathcal{K} = \mathcal{K}[\zeta, b](\psi, \omega)$ satisfies the following system

$$\begin{cases} \operatorname{curl} \mathcal{K} = 0, \\ \operatorname{div} \mathcal{K} = 0, \\ \mathcal{K}_{/\!\!/} = 0, \\ \mathcal{K} \cdot N_b = 0, \end{cases}$$
(4.64)

and K is equal to zero.

Proof. Using Proposition 4.2.15, and the fact that $\partial_t \mathbf{U} \circ \Sigma = \partial_t^{\sigma} \mathbf{U}$ (see Subsection 4.2.3) we have

$$\partial_t \left(\mathbf{U}[\varepsilon\zeta,\beta b](\psi,\boldsymbol{\omega}) \right) = \mathbf{U}[\varepsilon\zeta,\beta b] \left(\partial_t \psi - \underline{\mathbf{w}} \partial_t \zeta + \frac{\nabla}{\Delta} \cdot \left(\underline{\omega_h}^{\perp} \partial_t \zeta \right), \partial_t \boldsymbol{\omega} \right).$$

It is clear that div $\mathcal{K} = 0$ by definition of \mathcal{P} and curl $\mathcal{K} = 0$ since $\boldsymbol{\omega}$ satisfies the third equation of the Castro-Lannes formulation. The boundary condition at the bottom follows by definition of the pressure \mathcal{P} . For the boundary condition at the surface, since $(\nabla \mathcal{P})_{\parallel} = \nabla \underline{\mathcal{P}} = 0$, we get

$$\mathcal{K}_{\#} = \partial_t \nabla \psi - \nabla \left(\underline{\mathbf{w}} \partial_t \zeta\right) + \nabla \frac{\nabla}{\Delta} \cdot \left(\underline{\omega_h}^{\perp} \partial_t \zeta\right) + \frac{\nabla^{\perp}}{\Delta} \left(\underline{\partial_t \boldsymbol{\omega}} \cdot N\right) + \left[\left(\mathbf{U} \cdot \nabla_{X,z}\right) \mathbf{U}\right]_{\#} + g \nabla \zeta$$

where we denote $\mathbf{U} = (\mathbf{V}, \mathbf{w})^t = \mathbf{U}[\varepsilon, b](\psi, \boldsymbol{\omega})$. Then, using the fact that

$$\nabla \frac{\nabla}{\Delta} \cdot + \nabla^{\perp} \frac{\nabla^{\perp}}{\Delta} \cdot = Id$$

and taking the trace at the surface of the equation div $\omega = 0$, we get that

$$\begin{split} \nabla \frac{\nabla}{\Delta} \cdot \left(\underline{\omega_h}^{\perp} \partial_t \zeta \right) + \frac{\nabla^{\perp}}{\Delta} \left(\underline{\partial_t \omega} \cdot N \right) &= \underline{\omega_h}^{\perp} \partial_t \zeta - \nabla^{\perp} \frac{\nabla^{\perp}}{\Delta} \cdot \left(\underline{\omega_h}^{\perp} \partial_t \zeta \right) + \frac{\nabla^{\perp}}{\Delta} \left(\underline{\partial_t \omega} \cdot N \right), \\ &= \underline{\omega_h}^{\perp} \partial_t \zeta + \frac{\nabla^{\perp}}{\Delta} \left(\underline{\omega} \cdot \partial_t N \right) + \frac{\nabla^{\perp}}{\Delta} \left(\partial_t \underline{\omega} \cdot N \right), \\ &= \underline{\omega_h}^{\perp} \partial_t \zeta + \frac{\nabla^{\perp}}{\Delta} \partial_t \left(\underline{\omega} \cdot N \right). \end{split}$$

Furthermore, using the fact that $\omega_h = \nabla w - \partial_z V$, we have

$$\underline{\omega_h}^{\perp} \partial_t \zeta - \nabla \left(\underline{\mathbf{w}} \partial_t \zeta \right) = -\underline{\partial_z \mathbf{V}} \partial_t \zeta - \underline{\partial_z \mathbf{w}} \partial_t \zeta \nabla \zeta - \underline{\mathbf{w}} \nabla \partial_t \zeta$$

and from the chain rule we get

$$\partial_t \nabla \psi + \partial_t \nabla \frac{\nabla^{\perp}}{\Delta} (\underline{\omega} \cdot N) = \partial_t \mathbf{U}_{/\!\!/} = (\partial_t \mathbf{U})_{/\!\!/} + \underline{\partial_z \mathbf{V}} \partial_t \zeta + \underline{\partial_z \mathbf{w}} \partial_t \zeta \nabla \zeta + \underline{\mathbf{w}} \nabla \partial_t \zeta.$$

Therefore, we have

$$\mathcal{K}_{/\!\!/} = (\partial_t \mathbf{U})_{/\!\!/} + [(\mathbf{U} \cdot \nabla_{X,z}) \mathbf{U}]_{/\!\!/} + g \nabla \zeta$$

and using the same computations that Subsection 4.1.1, we obtain

$$\mathcal{K}_{\#} = \partial_t \mathbf{U}_{\#} + g\nabla\zeta + \frac{1}{2}\nabla\left|\mathbf{U}_{\#}\right|^2 - \frac{1}{2}\nabla\left[\left(1 + \left|\nabla\zeta\right|^2\right)\underline{\mathbf{w}}^2\right] + \underline{\omega}\cdot N\underline{\mathbf{V}}^{\perp}.$$

Then we decompose $\mathcal{K}_{/\!\!/}$ as $\mathcal{K}_{/\!\!/} = \nabla \frac{\nabla}{\Delta} \cdot \mathcal{K}_{/\!\!/} + \nabla^{\perp} \frac{\nabla^{\perp}}{\Delta} \cdot \mathcal{K}_{/\!\!/}$ and the first term is equal to zero because ψ satisfies the second equation of the Castro-Lannes system (4.61) and the second term is equal to zero since, by taking the trace of the third equation of the Castro-Lannes system (4.61), we have the following equation for $\underline{\omega} \cdot N$

$$\partial_t \left(\underline{\omega} \cdot N \right) + \nabla \cdot \left[\left(\underline{\omega} \cdot N \right) \underline{V} \right] = 0$$

The fact that $\mathcal{K}_{/\!\!/}$ is equal to zero follows from Theorem 4.2.8.

4.5 The nonlinear shallow water equations

4.5.1 The context

In this part we justify rigorously the derivation of the nonlinear rotating shallow water equations from the water waves equations. We recall that, in this chapter, we do not consider fast Coriolis forcing, i.e Ro $\leq \varepsilon$ (see Remark 4.1.2). The nonlinear shallow water equations (or Saint Venant equations) is a model used by the mathematical and physical communities to study the water waves in shallow waters. Coupled with a Coriolis term, it is usually uses to describe shallow waters under the influence of the Coriolis force (see for instance [27], [93] or [144]). But to our knowledge, there is no mathematical justification of this fact. Without the Coriolis term, many authors mathematically justify the Saint Venant equations; for the irrotationnal case, there are, for instance the works of Iguchi [70] and Alvarez-Samaniego and Lannes ([9]). It is also done in [80]. More recently, Castro and Lannes proposed a way to justify the Saint-Venant equations without the irrotational condition and with a flat bottom ([33] and [34]). We address here the case in which the Coriolis force and a non flat bottom are present. We denote the depth

$$h(t, X) = 1 + \varepsilon \zeta(t, X) - \beta b(X), \qquad (4.65)$$

and the averaged horizontal velocity

$$\overline{\mathbf{V}} = \overline{\mathbf{V}}[\varepsilon\zeta,\beta b](\psi,\boldsymbol{\omega})(t,X) = \frac{1}{h(t,X)} \int_{z=-1+\beta b(X)}^{\varepsilon\zeta(t,X)} \mathbf{V}[\varepsilon\zeta,\beta b](\psi,\boldsymbol{\omega})(t,X,z)dz.$$
(4.66)

The Saint-Venant equations (in the nondimensionalized form) are

$$\begin{cases} \partial_t \zeta + \nabla \cdot (h\overline{\mathbf{V}}) = 0, \\ \partial_t \overline{\mathbf{V}} + \varepsilon \left(\overline{\mathbf{V}} \cdot \nabla\right) \overline{\mathbf{V}} + \nabla \zeta + \frac{\varepsilon}{\text{Ro}} \overline{\mathbf{V}}^{\perp} = -\nabla P. \end{cases}$$
(4.67)

It is well-known that the shallow water equations are wellposed (see Chapter 6 in [80] or [9] without the pressure term and the Coriolis forcing and [27]). We have the following Proposition.

Proposition 4.5.1. Let $t_0 > \frac{d}{2}$, $s \ge t_0 + 1$ and $b \in H^{s+1}(\mathbb{R}^d)$, $\zeta_0 \in H^s(\mathbb{R}^d)$, $\overline{V}_0 \in H^s(\mathbb{R}^d)^d$. We assume that Condition (4.22) is satisfied by (ζ_0, b) . Assume also that ε, β and Ro satisfy Condition (4.23). Then, there exists T > 0 and $(\zeta, \overline{V}) \in \mathcal{C}^0\left(\left[0, \frac{T}{\max(\varepsilon, \beta)}\right], H^s(\mathbb{R}^d)^{d+1}\right)$ a unique solution to the Saint-Venant equations (4.67) with initial data $(\zeta_0, \overline{V}_0)$. Furthermore, for all $t \le \frac{T}{\max(\varepsilon, \beta)}$,

$$\frac{1}{T} = c^1 \text{ and } |\zeta(t, \cdot)|_{H^s} + \left|\overline{V}(t, \cdot)\right|_{H^{s+1}} \le c^2,$$

with $c^j = C\left(\frac{1}{h_{\min}}, |\zeta_0|_{H^s}, |b|_{H^{s+1}}, |\overline{V}_0|_{H^s}\right)$.

4.5.2 WKB expansion with respect to μ

In this part, we study the dependence of U^{μ} with respect to μ . The first Proposition shows that $\overline{\mathbf{V}}$ is linked to $\underline{U}^{\mu} \cdot N^{\mu}$.

Proposition 4.5.2. Under the assumptions of Theorem 4.2.8, we have

$$\underline{\mathbf{U}}^{\mu} \cdot N^{\mu} = -\mu \nabla \cdot \left(h \overline{\mathbf{V}}\right)$$

Proof. This proof is similar to Proposition 3.35 in [80]. Consider φ smooth and compactly supported in \mathbb{R}^d . Then, a simple computation gives

$$\begin{split} \int_{\mathbb{R}^d} \varphi \mathbf{U}^{\mu} \cdot N^{\mu} dX &= \int_{\Omega} \nabla_{X,z}^{\mu} \cdot (\varphi \mathbf{U}^{\mu}) \, dX dz, \\ &= \int_{\Omega} \mu \nabla \varphi \cdot \mathbf{V} dX dz, \\ &= -\mu \int_{\mathbb{R}^d} \varphi \nabla \cdot \left(\int_{z=-1+\beta b}^{\varepsilon \zeta} \mathbf{V} \right) dX. \end{split}$$

Then we need a WKB expansion with respect to μ of \mathbf{U}^{μ} .

Proposition 4.5.3. Let $t_0 > \frac{d}{2}$, $0 \le s \le t_0$, $\zeta \in H^{t_0+2}(\mathbb{R}^d)$, $b \in L^{\infty} \cap \dot{H}^{t_0+2}(\mathbb{R}^d)$. Under the assumptions of Theorem 4.2.8, we have

$$\mathbf{U}^{\mu} = \begin{pmatrix} \sqrt{\mu} \overline{\mathbf{V}} + \mu \left(\int_{z}^{\varepsilon \zeta} \boldsymbol{\omega}_{h}^{\perp} - \mathbf{Q} \right) + \mu^{\frac{3}{2}} \widetilde{\mathbf{V}} \\ \mu \widetilde{\mathbf{w}} \end{pmatrix},$$

with

$$\mathbf{Q}(X) = \frac{1}{h(X)} \int_{z'=-1+\beta b(X)}^{\varepsilon\zeta(X)} \int_{s=z'}^{\varepsilon\zeta(X)} \boldsymbol{\omega}_h^{\perp}(X,s),$$

and

$$\left| \left| \widetilde{\mathbf{V}} \circ \Sigma \right| \right|_{H^{s,1}} + \left| \left| \widetilde{\mathbf{w}} \circ \Sigma \right| \right|_{H^{s,1}} \le C \left(\frac{1}{\mathbf{h}_{\min}}, \varepsilon \left| \zeta \right|_{H^{t_0+2}}, \beta \left| b \right|_{L^{\infty}}, \beta \left| \nabla b \right|_{H^{t_0+1}} \right) \left| \left| \mathbf{V} \circ \Sigma \right| \right|_{H^{t_0+2,1}}.$$

Proof. This proof is inspired from the computations of Part 2.2 in [33] and Part 5.7.1 [34]. First, using the previous proposition, we get that

$$\underline{\mathbf{w}} = \varepsilon \mu \nabla \zeta \cdot \underline{\mathbf{V}} - \mu \nabla \cdot \left(h \overline{\mathbf{V}} \right)$$
 .

Furthermore, using the fact that \mathbf{U}^{μ} is divergent free we have

$$\partial_z \mathbf{w} = -\mu \nabla_X \cdot \mathbf{V}.$$

Then, we obtain

$$\mathbf{w} = \varepsilon \mu \nabla \zeta \cdot \underline{\mathbf{V}} - \mu \nabla \cdot \left(h \overline{\mathbf{V}} \right) + \mu \int_{z}^{\varepsilon \zeta} \nabla_{X} \mathbf{V}$$
$$= -\mu \nabla_{X} \cdot \left(\int_{-1+\beta b}^{z} \mathbf{V} \right).$$

The control of $\widetilde{\mathbf{w}}$ follows easily. Furthermore, using the ansatz

$$\mathbf{V} = \overline{\mathbf{V}} + \sqrt{\mu} \mathbf{V}_1, \tag{4.68}$$

and plugging it into the orthogonal of the horizontal part of $\operatorname{curl}^{\mu}\mathbf{U}^{\mu}=\mu\boldsymbol{\omega}$, we get that

$$\partial_z \mathbf{V}_1 = \sqrt{\mu} \nabla_X \widetilde{\mathbf{w}} - \boldsymbol{\omega}_h^{\perp}$$

Then, integrating with respect to z the previous equation from z to $\varepsilon\zeta(X)$ we get

$$\mathbf{V}_1(X,z) = \int_{s=z}^{\varepsilon\zeta(X)} \boldsymbol{\omega}_h^{\perp}(X,s) ds + \underline{\mathbf{V}_1}(X) + \mu^{\frac{1}{2}} \mathbf{R}(X,z), \qquad (4.69)$$

where ${\bf R}$ is a remainder uniformly bounded with respect to μ and

$$\underline{\mathbf{V}}_1 = \frac{\underline{\mathbf{V}} - \mathbf{V}}{\sqrt{\mu}}.$$

Integrating Equation (4.68) with respect to z from $-1 + \beta$ to $\varepsilon \zeta$ we obtain that

$$\int_{z=-1+\beta b(X)}^{\varepsilon\zeta(X)} \mathbf{V}_1(X,z) dz = 0 , \, \forall X \in \mathbb{R}^d.$$

Then, we integrate Equation (4.69) with respect to z from $-1 + \beta b$ to $\varepsilon \zeta$ and we get

$$h\underline{\mathbf{V}_{1}} = -\int_{z'=-1+\beta b}^{\varepsilon \zeta} \int_{s=z'}^{\varepsilon \zeta} \boldsymbol{\omega}_{h}^{\perp} + \mu^{\frac{1}{2}} \widetilde{\mathbf{R}},$$

where **R** is a remainder uniformly bounded with respect to μ . Plugging the previous expression into Equation (4.69), we get the result. The control of the remainders is straightforward thanks to Lemma 4.2.9 (see also the comments about the notations of [34] in Subsection 4.2.3).

Remark 4.5.4. Under the assumptions of the previous Proposition, it is easy to check that

$$\mathbf{w} = -\mu
abla_X \cdot \left(\left[1 + z - eta b
ight] \overline{\mathbf{V}}
ight) + \mu^{rac{3}{2}} \mathbf{w}_1,$$

with

$$||\mathbf{w}_{1} \circ \Sigma||_{H^{s,1}} \le C\left(\frac{1}{\mathbf{h}_{\min}}, \varepsilon |\zeta|_{H^{t_{0}+2}}, \beta |b|_{L^{\infty}}, \beta |\nabla b|_{H^{t_{0}+1}}\right) ||\mathbf{V} \circ \Sigma||_{H^{t_{0}+2,1}}.$$
(4.70)

Then, we define the quantity

$$\mathbf{Q} = \mathbf{Q}[\varepsilon\zeta,\beta b](\psi,\boldsymbol{\omega})(t,X) = \frac{1}{h} \int_{z'=-1+\beta b}^{\varepsilon\zeta} \int_{s=z'}^{\varepsilon\zeta} \boldsymbol{\omega}_h^{\perp}.$$
(4.71)

The following Proposition shows that \mathbf{Q} satisfies the evolution equation

$$\partial_t \mathbf{Q} + \varepsilon \left(\overline{\mathbf{V}} \cdot \nabla \right) \mathbf{Q} + \varepsilon \left(\mathbf{Q} \cdot \nabla \right) \overline{\mathbf{V}} + \frac{\varepsilon}{\text{Ro}} \mathbf{Q}^{\perp} = 0, \qquad (4.72)$$

up to terms of order $\mathcal{O}(\sqrt{\mu})$.

Proposition 4.5.5. Let T > 0, $t_0 > \frac{d}{2}$, $0 \le s \le t_0$, $0 \le \mu \le 1$, $\zeta \in C^1([0,T]; H^{t_0+2}(\mathbb{R}^d))$, $b \in L^{\infty} \cap \dot{H}^{t_0+2}(\mathbb{R}^d)$. Let $\omega, \mathbf{V}, \mathbf{w} \in C^1([0,T]; H^{t_0+2}(\mathbb{R}^d))$. Suppose that we are under the assumption of Theorem 4.2.8, that ω satisfies the third equation of the Castro-Lannes system (4.20) (the vorticity equation) and that $\partial_t \zeta + \nabla \cdot (h\overline{\mathbf{V}}) = 0$, on [0,T]. Then \mathbf{Q} satisfies

$$\partial_t \mathbf{Q} + \varepsilon \left(\overline{\mathbf{V}} \cdot \nabla \right) \mathbf{Q} + \varepsilon \left(\mathbf{Q} \cdot \nabla \right) \overline{\mathbf{V}} + \frac{\varepsilon}{\text{Ro}} \mathbf{Q}^{\perp} = \sqrt{\mu} \max \left(\varepsilon, \frac{\varepsilon}{\text{Ro}} \right) \widetilde{\mathbf{R}}$$

and

$$\left\| \widetilde{\mathbf{R}} \circ \Sigma \right\|_{H^{s,1}} \le C \left(\frac{1}{\mathbf{h}_{\min}}, \varepsilon \left| \zeta \right|_{H^{t_0+2}}, \beta \left| b \right|_{L^{\infty}}, \beta \left| \nabla b \right|_{H^{t_0+1}} \right) \left\| \mathbf{V} \circ \Sigma \right\|_{H^{t_0+2,1}}.$$

Proof. This proof is inspired from Subsection 2.3 in [33]. We know that ω_h satisfies

$$\partial_t \boldsymbol{\omega}_h + \varepsilon \left(\mathbf{V} \cdot \nabla \right) \boldsymbol{\omega}_h + \frac{\varepsilon}{\mu} \mathbf{w} \partial_z \boldsymbol{\omega}_h = \varepsilon \left(\boldsymbol{\omega}_h \cdot \nabla \right) \mathbf{V} + \frac{\varepsilon}{\sqrt{\mu}} \left(\boldsymbol{\omega}_v + \frac{1}{\text{Ro}} \right) \partial_z \mathbf{V}.$$

Using Proposition 4.5.2, Remark 4.5.4 and the fact that $\boldsymbol{\omega}_v = \nabla^{\perp} \cdot \mathbf{V}$, we get

$$\partial_t \boldsymbol{\omega}_h + \varepsilon \left(\overline{\mathbf{V}} \cdot \nabla \right) \boldsymbol{\omega}_h - \varepsilon \nabla_X \cdot \left(\left[1 + z - \beta b \right] \overline{\mathbf{V}} \right) \partial_z \boldsymbol{\omega}_h = \varepsilon \left(\boldsymbol{\omega}_h \cdot \nabla \right) \overline{\mathbf{V}} - \varepsilon \left(\nabla^\perp \cdot \overline{\mathbf{V}} + \frac{1}{\mathrm{Ro}} \right) \boldsymbol{\omega}_h^\perp \\ + \sqrt{\mu} \max \left(\varepsilon, \frac{\varepsilon}{\mathrm{Ro}} \right) \mathbf{R},$$

where $\mathbf{R} \circ \Sigma$ satisfies the same estimate as $\mathbf{w}_1 \circ \Sigma$ in (4.70). If we denote $\mathbf{V}_{sh} = \int_z^{\varepsilon \zeta} \boldsymbol{\omega}_h^{\perp}$, doing the same computations as in Subsection 2.3 [33] and using the fact that $\partial_t \zeta + \nabla \cdot (h \overline{\mathbf{V}}) = 0$, we get

$$\partial_t \mathbf{V}_{sh} + \varepsilon \left(\overline{\mathbf{V}} \cdot \nabla \right) \mathbf{V}_{sh} + \varepsilon \left(\mathbf{V}_{sh} \cdot \nabla \right) \overline{\mathbf{V}} - \nabla \cdot \left(\left[1 + z - \beta b \right] \overline{\mathbf{V}} \right) + \frac{\varepsilon}{\text{Ro}} \mathbf{V}_{sh}^{\perp} = \sqrt{\mu} \max \left(\varepsilon, \frac{\varepsilon}{\text{Ro}} \right) \int_z^{\varepsilon \zeta} \mathbf{R}.$$

Then, integrating this expression with respect to z and using again the fact that $\partial_t \zeta + \nabla \cdot (h \overline{\mathbf{V}}) = 0$, we get

$$\partial_t \mathbf{Q} + \varepsilon \left(\overline{\mathbf{V}} \cdot \nabla \right) \mathbf{Q} + \varepsilon \left(\mathbf{Q} \cdot \nabla \right) \overline{\mathbf{V}} + \frac{\varepsilon}{\mathrm{Ro}} \mathbf{Q}^{\perp} = \sqrt{\mu} \max \left(\varepsilon, \frac{\varepsilon}{\mathrm{Ro}} \right) \int_{-1+\beta b}^{\varepsilon \zeta} \int_{z}^{\varepsilon \zeta} \mathbf{R},$$

and the result follows easily.

4.5.3 Rigorous derivation

The purpose of this part is to prove a rigorous derivation of the water waves equations to the shallow water equations. This part is devoted to the proof of the following theorem. We recall that Σ is defined in (4.40).

Theorem 4.5.6. Let $N \geq 6$, $0 \leq \mu \leq 1$, ε , β , Ro satisfying (4.23). We assume that we are under the assumptions of Theorem 4.3.6. Then, we can define the following quantity $\boldsymbol{\omega}_0 = \boldsymbol{\omega}_0 \circ \Sigma^{-1}$, $\boldsymbol{\omega} = \boldsymbol{\omega} \circ \Sigma^{-1}$, $\overline{\mathbf{V}}_0 = \overline{\mathbf{V}}[\varepsilon\zeta_0, \beta b](\psi_0, \boldsymbol{\omega}_0)$, $\overline{\mathbf{V}} = \overline{\mathbf{V}}[\varepsilon\zeta, \beta b](\psi, \boldsymbol{\omega})$, $\mathbf{Q}_0 = \mathbf{Q}[\varepsilon\zeta_0, \beta b](\psi_0, \boldsymbol{\omega}_0)$ and $\mathbf{Q} = \mathbf{Q}[\varepsilon\zeta, \beta b](\psi, \boldsymbol{\omega})$ and there exists a time T > 0 such that

(i) T has the form

$$T = \min\left(\frac{T_0}{\max(\varepsilon, \beta, \frac{\varepsilon}{Ro})}, \frac{T_0}{|\nabla P|_{L_t^\infty H_X^N}}\right) \text{ and } \frac{1}{T_0} = c^1.$$

(ii) There exists a unique solution $(\zeta_{SW}, \overline{\mathbf{V}}_{SW})$ of (4.67) with initial conditions $(\zeta_0, \overline{\mathbf{V}}_0)$ on [0, T].

- (iii) There exists a unique solution \mathbf{Q}_{SW} to Equation (4.72) on [0, T].
- (iv) There exists a unique solution (ζ, ψ, ω) of (4.46) with initial conditions $(\zeta_0, \psi_0, \omega_0)$ on [0, T].
- (v) The following error estimates hold, for $0 \le t \le T$,

$$\left|\left(\zeta, \overline{\mathbf{V}}, \sqrt{\mu} \mathbf{Q}\right) - \left(\zeta_{SW}, \overline{\mathbf{V}}_{SW}, \sqrt{\mu} \mathbf{Q}_{SW}\right)\right|_{L^{\infty}([0,t] \times \mathbb{R}^d)} \leq \mu \, tc^2,$$

and

$$\left|\underline{\mathbf{V}}-\overline{\mathbf{V}}+\sqrt{\mu}\mathbf{Q}\right|_{L^{\infty}\left([0,T]\times\mathbb{R}^{d}\right)}\leq\mu\,c^{3},$$
with $c^{j}=C\left(A,\mu_{\max},\frac{1}{h_{\min}},\frac{1}{\mathfrak{a}_{\min}},|b|_{L^{\infty}},|\nabla b|_{H^{N+1}},|\nabla P|_{W^{1,\infty}_{t}H^{N}_{X}}\right).$

Remark 4.5.7. Hence, in shallow waters the rotating Saint-Venant equations are a good model to approximate the water waves equations under a Coriolis forcing. Furthermore, we notice that if we start initially with a irrotational flow, at the order μ , the flow stays irrotational. It means that a Coriolis forcing (not too fast) does not generate a horizontal vorticity in shallow waters and the assumption of a columnar motion, which is the fact that the velocity is horizontal and independent of the vertical variable z, stays valid. It could be interesting to develop an asymptotic model of the water waves equations at the order μ^2 (Green-Naghdi or Boussinesq models) and study the influence a Coriolis forcing in these models. It will be done in Chapter 5.

Proof. The point *(ii)* follows from Proposition 4.5.1 and the point *(iv)* from Theorem 4.3.6. Since, Equation (4.72) is linear, the point *(iii)* is clear. We only need to show that $(\zeta, \overline{\mathbf{V}})$ satisfy the shallow water equations up to a remainder of order μ . Then, a small adaptation of Proposition 6.3 in [80] allows us to prove the point *(v)*. First, we know that

$$\partial_t \psi + \zeta + \frac{\varepsilon}{2} \left| \mathbf{U}_{\mathscr{I}}^{\mu} \right|^2 - \frac{\varepsilon}{2\mu} \left(1 + \varepsilon^2 \mu \left| \nabla \zeta \right|^2 \right) \underline{\mathbf{w}}^2 + \varepsilon \frac{\nabla}{\Delta} \cdot \left[\left(\underline{\boldsymbol{\omega}} \cdot N^{\mu} + \frac{1}{\mathrm{Ro}} \right) \underline{\mathbf{V}}^{\perp} \right] = -P,$$

and

$$\partial_t \left(\underline{\omega} \cdot N^{\mu} \right) + \varepsilon \nabla \cdot \left(\left[\underline{\omega} \cdot N^{\mu} + \frac{1}{\text{Ro}} \right] \underline{V} \right) = 0.$$

Since $U^{\mu}_{/\!\!/} = \nabla \psi + \frac{\nabla^{\perp}}{\Delta} (\underline{\omega} \cdot N^{\mu})$, we get that

$$\partial_t \mathbf{U}^{\mu}_{\mathbb{I}} + \nabla\zeta + \frac{\varepsilon}{2} \nabla \left| \mathbf{U}^{\mu}_{\mathbb{I}} \right|^2 - \frac{\varepsilon}{2\mu} \nabla \left[\left(1 + \varepsilon^2 \mu \left| \nabla \zeta \right|^2 \right) \underline{\mathbf{w}}^2 \right] + \varepsilon \left(\underline{\boldsymbol{\omega}} \cdot N^{\mu} + \frac{1}{\mathrm{Ro}} \right) \underline{\mathbf{V}}^{\perp} = -\nabla P.$$

Then, using Proposition 4.5.3 and plugging the fact that $U^{\mu}_{\parallel} = \overline{\mathbf{V}} - \sqrt{\mu}\mathbf{Q} + \mu\mathbf{R}$, we get

$$\begin{aligned} \partial_t \overline{\mathbf{V}} &+ \varepsilon \left(\overline{\mathbf{V}} \cdot \nabla \right) \overline{\mathbf{V}} + \nabla \zeta + \frac{\varepsilon}{\mathrm{Ro}} \overline{\mathbf{V}}^{\perp} + \nabla P - \sqrt{\mu} \Big(\partial_t \mathbf{Q} \\ &+ \varepsilon \left(\overline{\mathbf{V}} \cdot \nabla \right) \mathbf{Q} + \varepsilon \left(\mathbf{Q} \cdot \nabla \right) \overline{\mathbf{V}} + \frac{\varepsilon}{\mathrm{Ro}} \mathbf{Q}^{\perp} \Big) = -\mu \partial_t \mathbf{R} + \widetilde{\mathbf{R}}, \end{aligned}$$

and using the same idea as Proposition 4.5.3, it is easy to check that

$$\begin{split} \left| \left| \widetilde{\mathbf{R}} \circ \Sigma \right| \right|_{H^{2,1}} &| \partial_t \mathbf{R} \circ \Sigma ||_{H^{2,1}} \leq C \left(\frac{1}{\mathbf{h}_{\min}}, \varepsilon \left| \zeta \right|_{H^4}, \varepsilon \left| \partial_t \zeta \right|_{H^4}, \beta \left| b \right|_{L^{\infty}}, \beta \left| \nabla b \right|_{H^3} \right) \times \\ & \left(\left| \left| \mathbf{V} \circ \Sigma \right| \right|_{H^{4,1}} + \left| \left| \partial_t \mathbf{V} \circ \Sigma \right| \right|_{H^{4,1}} \right). \end{split}$$

Using Proposition 4.5.5, Theorem 4.3.6, Theorems 4.2.8 and 4.2.11 and Remark 4.2.14, we get the result . $\hfill \Box$

4.5.4 The Proudman resonance, linear properties of the shallow water equations

In Chapter 2 we studied the Proudman resonance. We recall that it is a linear amplification due to a source term (non constant pressure or moving bottom). The purpose of this part is to study it for the linear shallow water equations

$$\begin{cases} \partial_t \zeta + \nabla \cdot \overline{\mathbf{V}} = 0, \\ \partial_t \overline{\mathbf{V}} + \nabla \zeta + \frac{\varepsilon}{\text{Ro}} \overline{\mathbf{V}}^\perp = -\nabla P. \end{cases}$$
(4.73)

In the following we take d = 2 and we suppose that $\frac{\varepsilon}{\text{Ro}} = 1$ (strong rotation in the sense of [60]). We consider the asymptotic model (in the sense of Definition 4.19 in [80]), for $\delta_0 \leq 1$,

$$\mathcal{A}_{LWW} = \left\{ \left(\varepsilon, \beta, \mu, \frac{\varepsilon}{\text{Ro}}\right), \ 0 < \mu, \ \varepsilon, \ \beta \le \delta_0, \ \frac{\varepsilon}{\text{Ro}} = 1 \right\}.$$
(4.74)

Then we have the following result, which is a small adaptation of Theorem 4.5.6.

Proposition 4.5.8. Let d = 2, $N \ge 6$, $0 \le \delta_0 \le 1$, $\varepsilon, \beta, \mu, \operatorname{Ro} \in \mathcal{A}_{LWW}$. We assume that we are under the assumptions of Theorem 4.3.6. Then, we can define the following quantity $\boldsymbol{\omega}_0 = \omega_0 \circ \Sigma^{-1}$, $\boldsymbol{\omega} = \omega \circ \Sigma^{-1}$, $\overline{\mathbf{V}}_0 = \overline{\mathbf{V}}[\varepsilon\zeta_0, \beta b](\psi_0, \boldsymbol{\omega}_0)$, $\overline{\mathbf{V}} = \overline{\mathbf{V}}[\varepsilon\zeta, \beta b](\psi, \boldsymbol{\omega})$, $\mathbf{Q}_0 = \mathbf{Q}[\varepsilon\zeta_0, \beta b](\psi_0, \boldsymbol{\omega}_0)$ and $\mathbf{Q} = \mathbf{Q}[\varepsilon\zeta, \beta b](\psi, \boldsymbol{\omega})$. Then, there exists a time T > 0 such that

(i) T has the form

$$\frac{1}{T} = c^1.$$

(ii) There exists a unique solution $(\zeta_{LSW}, \overline{\mathbf{V}}_{LSW})$ of (4.73) with initial conditions $(\zeta_0, \overline{\mathbf{V}}_0)$ on [0, T].

(iii) There exists a unique solution (ζ, ψ, ω) of (4.46) with initial conditions $(\zeta_0, \psi_0, \omega_0)$ on [0, T]. (iv) The following error estimates hold, for $0 \le t \le T$,

$$\left|\left(\zeta, \overline{\mathbf{V}}\right) - \left(\zeta_{LSW}, \overline{\mathbf{V}}_{LSW}\right)\right|_{L^{\infty}([0,t] \times \mathbb{R}^2)} \leq \delta_0 \, tc^2,$$

with $c^j = C\left(A, \frac{1}{h_{\min}}, \frac{1}{\mathfrak{a}_{\min}}, |b|_{L^{\infty}}, |\nabla b|_{H^{N+1}}, |\nabla P|_{W^{1,\infty}_t H^N_X}\right).$

We denote in the following $\mathbf{V} = (u, v)^t$. We wonder now if we can catch an elevation of the sea level with the asymptotic model (4.73). We answer to this question in the following proposition.

Proposition 4.5.9. Let $P \in L^{\infty}_{t}W^{2,\infty}_{X}(\mathbb{R}^{2})$, (ζ, u, v) be a solution of (4.73). Then,

$$|(\zeta, u, v)(t, \cdot)|_{L^{\infty}} \le C \ln t,$$

where
$$C = C\left(\left|(1+|X|)^2 P\right|_{L_t^\infty H_X^2}, \left|(1+|X|)^2 \zeta_0\right|_{H^2}, \left|(1+|X|)^2 u_0\right|_{H^2}, \left|(1+|X|)^2 v_0\right|_{H^2}\right).$$

Proof. We denote by A the anti-symmetric matrix operator

$$A = \begin{pmatrix} 0 & -\partial_x & -\partial_y \\ -\partial_x & 0 & 1 \\ -\partial_y & -1 & 0 \end{pmatrix}.$$

Using the Duhamel's formula, we have

$$(\zeta, u, v)(t, \cdot) = e^{tA}(\zeta_0, u_0, v_0) + \int_0^t e^{(t-s)A} \begin{pmatrix} 0\\ \nabla P(s, \cdot) \end{pmatrix} ds.$$

Then, we notice that

$$\widehat{e^{tA}}(\xi) = e^{it\sqrt{1+|\xi|^2}}A_1(\xi) + e^{-it\sqrt{1+|\xi|^2}}A_2(\xi) + A_3(\xi),$$

where $A_1, A_2, \nabla A_1, \nabla A_2, \nabla^2 A_1, \nabla^2 A_2 \in L^{\infty}(\mathbb{R}^2)$ and, if $\xi = (\xi_1, \xi_2)^t$,

$$A_3(\xi) = \frac{1}{1+|\xi|^2} \begin{pmatrix} 1 & i\xi_2 & -i\xi_1 \\ -i\xi_2 & \xi_2^2 & -\xi_1\xi_2 \\ i\xi_1 & -\xi_1\xi_2 & \xi_1^2 \end{pmatrix}.$$

Using the fact that $A_3(\xi) \begin{pmatrix} 0\\ i\hat{P}(s,\xi)\xi \end{pmatrix} = 0$, the result follows from the following lemma (see [146] or Corollary 7.2.4 in [69]).

Lemma 4.5.10. Let $u_0 \in W^{2,1}(\mathbb{R}^2)$. Then

$$\left| \int_{\mathbb{R}} e^{i(x \cdot \xi) \pm t \sqrt{|\xi|^2 + 1}} u_0(\xi) d\xi \right|_{L^{\infty}_x} \le \frac{C}{1 + |t|} |u_0|_{W^{2,1}}.$$

Hence, we can not expect a resonance from a physical standpoint since the possible amplification is too slow. The dispersive effects due to the Coriolis forcing prevent the Proudman resonance to occur. Notice that our proof is specific to the case d = 2. In Section 5.3.2, we study the case d = 1.

Chapter 5

Long wave approximation for water waves under a Coriolis forcing

Sommaire

5.1	1 Introduction	
	5.1.1	Nondimensionalization and the Castro-Lannes formulation 132
	5.1.2	Notations
	5.1.3	Useful results
5.2	Bou	ssinesq-Coriolis equations when $\gamma = 0$
	5.2.1	Asymptotic expansion for the velocity and useful identities $\ldots \ldots 137$
	5.2.2	Equations for Q_x , Q_y and \mathbf{V}^{\sharp}
	5.2.3	The Boussinesq-Coriolis equations
	5.2.4	Full justification of the Boussinesq-Coriolis equations
5.3 Different asymptotic models in the Boussinesq regime over a flat		
bottom		
	5.3.1	Strong rotation, the Poincaré waves
	5.3.2	The Proudman resonance in presence of a Coriolis forcing 150
	5.3.3	Weak rotation, the Ostrovsky equation
	5.3.4	Very weak rotation, the KdV equation
5.4	Gree	en-Naghdi equations for $\gamma = 0$ and $\beta = \mathcal{O}(\mu) \dots \dots 159$
	5.4.1	Improvements for the equations of Q_x and Q_y
	5.4.2	Equations for E
	5.4.3	The Green-Naghdi equations
	5.4.4	A simplified model in the case of a weak rotation and medium ampli-
		tude waves

Ce chapitre est basé sur l'article [103]. Nous avons ajouté la soussection 5.3.2 où nous étudions la résonance de Proudman en présence de la force de Coriolis lorsque d = 1. Nous montrons en particulier qu'aucune résonance n'est possible lorsque la tempête se déplace à vitesse fixe.

5.1 Introduction

This paper is devoted to the study of the long wave approximation for water waves under the influence of the gravity and a Coriolis forcing. We start by deriving a generalization of the Boussinesq equations in 1D (in space) and we rigorously justify them as an asymptotic model of the water waves equations. These new Boussinesq equations are not the classical Boussinesq equations. A new term due to the vorticity and the Coriolis forcing appears that can not be neglected. Then, we study the Boussinesq regime and we derive and fully justify different asymptotic models when the bottom is flat : a linear equation linked to the Klein-Gordon equation admitting the so-called Poincaré waves; the Ostrovsky equation, which is a generalization of the KdV equation in presence of a Coriolis forcing, when the rotation is weak; and finally the KdV equation when the Ostrovsky equation. Finally, we derive a generalization of the Green-Naghdi equations in 1D in space for small topography variations and we show that this model is consistent with the water waves equations.

We study the motion of an incompressible, inviscid fluid with a constant density ρ and no surface tension under the influence of the gravity $\mathbf{g} = -g\mathbf{e_z}$ and the rotation of the Earth with a rotation vector $\mathbf{f} = \frac{f}{2}\mathbf{e_z}$. We suppose that the seabed and the surface are graphs above the still water level. The horizontal variable is $X = (x, y) \in \mathbb{R}^2$ and $z \in \mathbb{R}$ is the vertical variable. The water occupies the domain $\Omega_t := \{(X, z) \in \mathbb{R}^3, -H + b(X) < z < \zeta(t, X)\}$. The velocity in the fluid domain is denoted $\mathbf{U} = (\mathbf{V}, \mathbf{w})^t$ where \mathbf{V} is the horizontal component of \mathbf{U} and \mathbf{w} its vertical component. The equations governing such a fluid are the free surface Euler-Coriolis equations⁽¹⁾

$$\begin{cases} \partial_t \mathbf{U} + (\mathbf{U} \cdot \nabla_{X,z}) \mathbf{U} + \mathbf{f} \times \mathbf{U} = -\frac{1}{\rho} \nabla_{X,z} \mathcal{P} - g \boldsymbol{e}_{\boldsymbol{z}} \text{ in } \Omega_t, \\ \operatorname{div} \mathbf{U} = 0 \text{ in } \Omega_t, \end{cases}$$
(5.1)

with the boundary conditions

$$\begin{cases} \mathcal{P}_{|z=\zeta} = P_0, \\ \partial_t \zeta - \underline{U} \cdot N = 0, \\ U_b \cdot N_b = 0, \end{cases}$$
(5.2)

where P_0 is constant, $N = \begin{pmatrix} -\nabla\zeta\\ 1 \end{pmatrix}$, $N_b = \begin{pmatrix} -\nabla b\\ 1 \end{pmatrix}$, $\underline{\mathbf{U}} = \begin{pmatrix} \underline{\mathbf{V}}\\ \underline{\mathbf{w}} \end{pmatrix} = \mathbf{U}_{|z=\zeta}$ and $\mathbf{U}_b = \begin{pmatrix} \mathbf{V}_b\\ \mathbf{w}_b \end{pmatrix} = \mathbf{U}_{|z=-H+b}$.

Influenced by the works of Zakharov ([153]) and Craig-Sulem-Sulem ([48]), Castro and Lannes in [34] shown that we can express the free surface Euler equations thanks to the unknowns $(\zeta, U_{\mathbb{I}}, \omega)^{(2)}$ where

 $^{^{1}}$ We consider that the centrifugal potential is constant and included in the pressure term.

²In fact, Castro and Lannes used the unknowns $(\zeta, \frac{\nabla}{\Delta} \cdot U_{/\!\!/}, \omega)$. But the unknowns $(\zeta, U_{/\!\!/}, \omega)$ are better to derive shallow water asymptotic models.

$$\mathbf{U}_{\mathbb{I}} = \mathbf{\underline{V}} + \mathbf{\underline{w}} \nabla \zeta,$$

and they gave a system of three equations on these unknowns. Then, in [101] we proceeded as Castro and Lannes and, taking into account the Coriolis force, we got the following system, called the Castro-Lannes system or the water waves equations,

$$\begin{cases} \partial_t \zeta - \underline{\mathbf{U}} \cdot N = 0, \\ \partial_t \mathbf{U}_{/\!\!/} + \nabla \zeta + \frac{1}{2} \nabla \left| \mathbf{U}_{/\!\!/} \right|^2 - \frac{1}{2} \nabla \left[\left(1 + \left| \nabla \zeta \right|^2 \right) \underline{\mathbf{w}}^2 \right] + \underline{\boldsymbol{\omega}} \cdot N \, \underline{\mathbf{V}}^\perp + f \underline{\mathbf{V}}^\perp = 0, \\ \partial_t \boldsymbol{\omega} + \left(\mathbf{U} \cdot \nabla_{X,z} \right) \boldsymbol{\omega} = \left(\boldsymbol{\omega} \cdot \nabla_{X,z} \right) \mathbf{U} + f \partial_z \mathbf{U}, \end{cases}$$
(5.3)

where $\mathbf{U} = \begin{pmatrix} \mathbf{V} \\ \mathbf{w} \end{pmatrix} = \mathbf{U}[\zeta, b](\mathbf{U}_{/\!\!/}, \boldsymbol{\omega})$ is the unique solution in $H^1(\Omega_t)$ of

$$\begin{cases} \operatorname{curl} \mathbf{U} = \boldsymbol{\omega} \text{ in } \Omega_t, \\ \operatorname{div} \mathbf{U} = 0 \text{ in } \Omega_t, \\ (\underline{\mathbf{V}} + \underline{\mathbf{w}} \nabla \zeta)_{|z=\zeta} = \mathbf{U}_{/\!\!/}, \\ \mathbf{U}_b \cdot N_b = 0, \end{cases}$$
(5.4)

and with the following constraint

$$\nabla^{\perp} \cdot \mathbf{U}_{/\!\!/} = \underline{\boldsymbol{\omega}} \cdot N. \tag{5.5}$$

Our principal motivation is the study of the long waves or Boussinesq regime. Hence, we nondimensionalize the previous equations. We have six physical parameters in our problem : the typical amplitude of the surface a, the typical amplitude of the bathymetry a_{bott} , the typical longitudinal scale L_x , the typical transverse scale L_y , the characteristic water depth H and the typical Coriolis frequency f. Then we can introduce five dimensionless parameters

$$\varepsilon = \frac{a}{H}, \ \beta = \frac{a_{\text{bott}}}{H}, \ \mu = \frac{H^2}{L_x^2}, \ \gamma = \frac{L_x}{L_y} \text{ and } \text{Ro} = \frac{a\sqrt{gH}}{HfL_x}.$$

The parameter ε is called the nonlinearity parameter, β is called the bathymetric parameter, μ is called the shallowness parameter, γ is called the transversality parameter and Ro is the Rossby number. Then, we can adimensionalize the Euler equations (5.1) and the Castro-Lannes equations (5.3) (see Part 5.1.1).

We organize our paper in three parts. Section 5.2 is devoted to derive a generalization of the Boussinesq equations in 1D under a Coriolis forcing and to fully justify it. The Boussinesq equations are obtained under the assumption that μ is small, $\varepsilon, \beta = \mathcal{O}(\mu)$ (Boussinesq regime) and by neglecting all the terms of order $\mathcal{O}(\mu^2)$ in the adimensionalized Euler equations or the water waves equations (see for instance [9] in the irrotational framework). It is a system of two equations on the free surface ζ and the vertical average of the horizontal component of the velocity denoted $\overline{\mathbf{V}} = (\overline{u}, \overline{v})^t$ (defined in (5.22)). Our Boussinesq-Coriolis equations are a system of three equations on the surface ζ , the average vertical velocity $\overline{\mathbf{V}}$ and the quantity $\mathbf{V}^{\sharp} = (u^{\sharp}, v^{\sharp})^t$ (defined in (5.29)) which is introduced to catch interactions between the vorticity and the averaged velocity. These equations are the following system

$$\begin{cases} \partial_t \zeta + \partial_x \left([1 + \varepsilon \zeta - \beta b] \overline{u} \right) = 0, \\ \left(1 - \frac{\mu}{3} \partial_x^2 \right) \partial_t \overline{u} + \partial_x \zeta + \varepsilon \overline{u} \partial_x \overline{u} - \frac{\varepsilon}{\text{Ro}} \overline{v} + \frac{\varepsilon}{\text{Ro}} \mu^{\frac{3}{2}} \frac{1}{24} \partial_x^2 \frac{v^{\sharp}}{h} = 0, \\ \partial_t \overline{v} + \varepsilon \overline{u} \partial_x \overline{v} + \frac{\varepsilon}{\text{Ro}} \overline{u} = 0, \\ \partial_t \frac{\mathbf{V}^{\sharp}}{h} + \varepsilon \overline{u} \partial_x \frac{\mathbf{V}^{\sharp}}{h} + \frac{\varepsilon}{\text{Ro}} \frac{\mathbf{V}^{\sharp}}{h}^{\perp} = 0, \end{cases}$$

where $h = 1 + \epsilon \zeta - \beta b$. Then, in Section 5.3 we derive and fully justify different asymptotic models in the Boussinesq regime when the bottom is flat. We first derive in Subsection 5.3.1 a linear system (system (5.38)) linked to the Klein-Gordon equation admitting the so-called Poincaré waves. Then, in Subsection 5.3.3 we study the Ostrovsky equation

$$\partial_{\xi} \left(\partial_{\tau} f + \frac{3}{2} f \partial_{\xi} f + \frac{1}{6} \partial_{\xi}^3 f \right) = \frac{1}{2} f.$$

This equation, derived by Ostrovsky ([113]), is a generalization of the KdV equation in presence of a Coriolis forcing. We offer a rigorous justification of the Ostrovsky approximation under a weak Coriolis forcing, i.e $\frac{\varepsilon}{Ro} = \mathcal{O}(\sqrt{\mu})$. Notice that this work provides the first mathematical justification of the Ostrovsky equation. In Subsection 5.3.4 we fully justify the KdV approximation (equation (5.52)) when the rotation is very weak, i.e when $\frac{\varepsilon}{Ro} = \mathcal{O}(\mu)$. Finally, in Section 5.4 we derive a generalization of the Green-Naghdi equations (5.64) in 1D under a Coriolis forcing with small bottom variations and we show that this system is consistent with the water waves equations. The Green-Naghdi equations are originally obtained in the irrotational framework under the assumption that μ is small and by neglecting all the terms of order $\mathcal{O}(\mu^2)$ in the adimensionalized Euler equations or the water waves equations (see for instance [130] or Part 5.1.1.2 in [80] for a derivation in the irrotational framework). They were generalized in [33] in the rotational setting but without a Coriolis forcing. We add one in the paper.

5.1.1 Nondimensionalization and the Castro-Lannes formulation

We recall the five dimensionless parameter

$$\varepsilon = \frac{a}{H}, \ \beta = \frac{a_{\text{bott}}}{H}, \ \mu = \frac{H^2}{L_x^2}, \ \gamma = \frac{L_x}{L_y} \text{ and } \text{Ro} = \frac{a\sqrt{gH}}{HfL_x}.$$
 (5.6)

We nondimensionalize the variables and the unknowns. We introduce (see [80] or [101] for instance for an explanation of this nondimensionalization)

$$\begin{cases} x' = \frac{x}{L_x}, y' = \frac{y}{L_y}, z' = \frac{z}{H}, \zeta' = \frac{\zeta}{a}, b' = \frac{b}{a_{\text{bott}}}, t' = \frac{\sqrt{gH}}{L_x}t, \\ \mathbf{V}' = \sqrt{\frac{H}{g}} \frac{\mathbf{V}}{a}, w' = H\sqrt{\frac{H}{g}} \frac{w}{aL_x} \text{ and } \mathcal{P}' = \frac{\mathcal{P}}{\rho gH}. \end{cases}$$
(5.7)

In this paper, we use the following notations

$$\nabla^{\gamma} = \nabla^{\gamma}_{X'} = \begin{pmatrix} \partial_{x'} \\ \gamma \partial_{y'} \end{pmatrix}, \quad = \nabla^{\mu,\gamma}_{X',z'} = \begin{pmatrix} \sqrt{\mu} \nabla^{\gamma}_{X'} \\ \partial_{z'} \end{pmatrix}, \quad \text{curl}^{\mu,\gamma} = \nabla^{\mu,\gamma}_{X',z'} \times , \quad \text{div}^{\mu,\gamma} = \nabla^{\mu,\gamma}_{X',z'} \cdot . \tag{5.8}$$

We also define

$$\mathbf{U}^{\mu} = \begin{pmatrix} \sqrt{\mu} \mathbf{V}' \\ \mathbf{w}' \end{pmatrix}, \, \boldsymbol{\omega}' = \frac{1}{\mu} \operatorname{curl}^{\mu,\gamma} \mathbf{U}^{\mu}, \, \underline{\mathbf{U}}^{\mu} = \mathbf{U}^{\mu}_{|z'=\varepsilon\zeta'}, \, \mathbf{U}^{\mu}_{b} = \mathbf{U}^{\mu}_{|z'=-1+\beta b'}, \tag{5.9}$$

and

$$N^{\mu,\gamma} = \begin{pmatrix} -\varepsilon \sqrt{\mu} \nabla^{\gamma} \zeta' \\ 1 \end{pmatrix}, N_b^{\mu,\gamma} = \begin{pmatrix} -\beta \sqrt{\mu} \nabla^{\gamma} b' \\ 1 \end{pmatrix}.$$
 (5.10)

Notice that our nondimensionalization of the vorticity allows us to consider only weakly sheared flows (see [33], [139], [121]). The nondimensionalized fluid domain is

$$\Omega'_{t'} := \{ (X', z') \in \mathbb{R}^3 , -1 + \beta b'(X') < z' < \varepsilon \zeta'(t', X') \}.$$
(5.11)

Finally, if $\mathbf{V} = \begin{pmatrix} u \\ v \end{pmatrix} \in \mathbb{R}^2$, we define \mathbf{V} by $\mathbf{V}^{\perp} = \begin{pmatrix} -v \\ u \end{pmatrix}$. Then, the Euler-Coriolis equations (5.1) become

$$\begin{cases} \partial_{t'} \mathbf{U}^{\mu} + \frac{\varepsilon}{\mu} \left(\mathbf{U}^{\mu} \cdot \nabla_{X',z'}^{\mu,\gamma} \right) \mathbf{U}^{\mu} + \frac{\varepsilon \sqrt{\mu}}{\mathrm{Ro}} \left(\begin{array}{c} \mathbf{V}^{\prime \perp} \\ 0 \end{array} \right) = -\frac{1}{\varepsilon} \nabla_{X',z'}^{\mu,\gamma} \mathcal{P}^{\prime} - \frac{1}{\varepsilon} \boldsymbol{e}_{\boldsymbol{z}} \text{ in } \Omega_{t}^{\prime}, \\ \operatorname{div}_{X',z'}^{\mu,\gamma} \mathbf{U}^{\mu} = 0 \text{ in } \Omega_{t}^{\prime}, \end{cases}$$
(5.12)

with the boundary conditions

$$\begin{cases} \partial_{t'}\zeta' - \frac{1}{\mu}\underline{U}^{\mu} \cdot N^{\mu,\gamma} = 0, \\ U_b^{\mu} \cdot N_b^{\gamma,\mu} = 0. \end{cases}$$
(5.13)

We can also nondimensionalize the Castro-Lannes formulation. We introduce the quantity

$$\mathbf{U}^{\mu}_{\mathbb{I}} = \underline{\mathbf{V}} + \varepsilon \underline{\mathbf{w}} \nabla^{\gamma} \zeta.$$

Then, the Castro-Lannes formulation becomes (see [34] or [101] when $\gamma = 1$),

$$\begin{cases} \partial_t \zeta - \frac{1}{\mu} \underline{\mathbf{U}}^{\mu} \cdot N^{\mu,\gamma} = 0, \\ \partial_t \mathbf{U}_{l}^{\mu} + \nabla^{\gamma} \zeta + \frac{\varepsilon}{2} \nabla^{\gamma} \left| \mathbf{U}_{l}^{\mu} \right|^2 - \frac{\varepsilon}{2\mu} \nabla^{\gamma} \left[\left(1 + \varepsilon^2 \mu \left| \nabla^{\gamma} \zeta \right|^2 \right) \underline{\mathbf{w}}^2 \right] + \varepsilon \underline{\boldsymbol{\omega}} \cdot N^{\mu,\gamma} \, \underline{\mathbf{V}}^{\perp} + \frac{\varepsilon}{\mathrm{Ro}} \underline{\mathbf{V}}^{\perp} = 0, \quad (5.14) \\ \partial_t \boldsymbol{\omega} + \frac{\varepsilon}{\mu} \left(\mathbf{U}^{\mu} \cdot \nabla_{X,z}^{\mu,\gamma} \right) \boldsymbol{\omega} = \frac{\varepsilon}{\mu} \left(\boldsymbol{\omega} \cdot \nabla_{X,z}^{\mu,\gamma} \right) \mathbf{U}^{\mu} + \frac{\varepsilon}{\mu \mathrm{Ro}} \partial_z \mathbf{U}^{\mu}, \end{cases}$$

where $\mathbf{U}^{\mu} = \begin{pmatrix} \sqrt{\mu} \mathbf{V} \\ \mathbf{w} \end{pmatrix} = \mathbf{U}^{\mu} [\varepsilon \zeta, \beta b] (\mathbf{U}^{\mu}_{/\!\!/}, \boldsymbol{\omega})$ is the unique solution in $H^1(\Omega_t)$ of $\begin{cases} \operatorname{curl}^{\mu, \gamma} \mathbf{U}^{\mu} = \mu \boldsymbol{\omega} \text{ in } \Omega_t, \\ \operatorname{div}^{\mu, \gamma} \mathbf{U}^{\mu} = 0 \text{ in } \Omega_t, \\ (\underline{\mathbf{V}} + \varepsilon \underline{\mathbf{w}} \nabla^{\gamma} \zeta)_{|z=\varepsilon \zeta} = \mathbf{U}^{\mu}_{/\!\!/}, \end{cases}$

$$\begin{aligned} \operatorname{div}^{\mu,\gamma} \mathbf{U}^{\mu} &= 0 \text{ in } \Omega_t, \\ (\mathbf{\underline{V}} + \varepsilon_{\mathbf{\underline{W}}} \nabla^{\gamma} \zeta)_{|z=\varepsilon\zeta} &= \mathbf{U}_{/\!\!/}^{\mu}, \\ \mathbf{U}_b^{\mu} \cdot N_b^{\mu,\gamma} &= 0, \end{aligned}$$
(5.15)

and with the following constraint

$$\nabla^{\perp} \cdot \mathbf{U}^{\mu}_{/\!\!/} = \underline{\boldsymbol{\omega}} \cdot N^{\mu,\gamma}.$$
(5.16)

Remark 5.1.1. When, $\boldsymbol{\omega} = 0$ and $Ro = +\infty$, we get the irrotational water waves equations (see Remark 2.4 in [34]). In particular in this situation, when $\gamma = 0$ we can check that the velocity \mathbf{U}^{μ} becomes two dimensional : $\mathbf{U}^{\mu} = (\sqrt{\mu}\mathbf{V}_x, 0, \mathbf{w})^t$. This is not the case when $\boldsymbol{\omega} \neq 0$. Even if $\gamma = 0$, the vorticity transfers energy from \mathbf{V}_x to \mathbf{V}_y . The only way to get a two dimensional speed is to assume that $\boldsymbol{\omega} = (0, \omega_y, 0)^t$ (see for instance [82]).

Remark 5.1.2. Notice that if $(\zeta, U^{\mu}_{\parallel}, \boldsymbol{\omega})$ is a solution of the Castro-Lannes system (5.14), $\nabla^{\perp} \cdot U^{\mu}_{\parallel}$ satisfies the equation

$$\partial_t \nabla^{\perp} \cdot \mathbf{U}_{/\!\!/}^{\mu} + \nabla^{\gamma} \cdot \left(\varepsilon \underline{\boldsymbol{\omega}} \cdot N^{\mu,\gamma} \underline{\mathbf{V}}^{\perp} + \frac{\varepsilon}{\mathrm{Ro}} \underline{\mathbf{V}} \right) = 0$$

Furthermore, by taking the trace of the third equation of the Castro-Lannes system (5.14), we can see that $\underline{\omega} \cdot N^{\mu,\gamma}$ satisfies the equation

$$\partial_t \left(\underline{\boldsymbol{\omega}} \cdot N^{\mu, \gamma} \right) + \nabla^{\gamma} \cdot \left(\varepsilon \underline{\boldsymbol{\omega}} \cdot N^{\mu, \gamma} \underline{\mathbf{V}}^{\perp} + \frac{\varepsilon}{\operatorname{Ro}} \underline{\mathbf{V}} \right) = 0,$$

Hence, the constraint (5.16) is propagated by the equations.

We add a technical assumption. We assume that the water depth is bounded from below by a positive constant

$$\exists h_{\min} > 0 , 1 + \varepsilon \zeta - \beta b \ge h_{\min}.$$
(5.17)

We also suppose that the dimensionless parameters satisfy

$$\exists \mu_{\max}, \ 0 < \mu \le \mu_{\max}, \ 0 < \varepsilon \le 1, \ 0 \le \gamma \le 1, \ 0 \le \beta \le 1 \text{ and } \frac{\varepsilon}{\text{Ro}} \le 1.$$
(5.18)

Remark 5.1.3. We have $\frac{\varepsilon}{Ro} = \frac{fL}{\sqrt{gH}}$. As said in [101], it is quite reasonable to assume that $\frac{\varepsilon}{Ro} \leq 1$ since for water waves, the typical rotation speed due to the Coriolis forcing is less than the typical water wave celerity (see for instance [117], [61], [84]).

5.1.2 Notations

- If $\mathbf{A} \in \mathbb{R}^3$, we denote by \mathbf{A}_h its horizontal component.

- If
$$\mathbf{V} = \begin{pmatrix} u \\ v \end{pmatrix} \in \mathbb{R}^2$$
, we define the orthogonal of \mathbf{V} by $\mathbf{V}^{\perp} = \begin{pmatrix} -v \\ u \end{pmatrix}$.

- In this paper, $C\left(\cdot\right)$ is a nondecreasing and positive function whose exact value has no importance.

- Consider a vector field **A** or a function w defined on Ω . Then, we denote $\underline{\mathbf{A}} = \mathbf{A}_{|z=\varepsilon\zeta}$, $\underline{\mathbf{w}} = \mathbf{w}_{|z=\varepsilon\zeta}$ and $\mathbf{A}_b = \mathbf{A}_{|z=-1+\beta b}$, $\mathbf{w}_b = \mathbf{w}_{|z=-1+\beta b}$.

- If $s \in \mathbb{R}$ and f is a function on \mathbb{R}^2 , $|f|_{H^s}$ is its H^s -norm, $|f|_2$ is its L^2 -norm and $|f|_{L^{\infty}}$ its $L^{\infty}(\mathbb{R}^2)$ -norm.

- The operator (,)₂ is the L^2 -scalar product in \mathbb{R}^2 .

- If f is a function defined on \mathbb{R}^2 , we denote ∇f the gradient of f.

- If w is a function defined on Ω , $\nabla_{X,z}$ w is the gradient of w and ∇_X w its horizontal component.

- If u = u(X, z) is defined in Ω , we define

$$\overline{u}(X) = \frac{1}{1 + \varepsilon \zeta - \beta b} \int_{-1 + \beta b(X)}^{\varepsilon \zeta(X)} u(X, z) dz \text{ and } u^* = u - \overline{u}.$$

5.1.3 Useful results

In this paper, we fully justify different asymptotic models of the water waves equations. Then, we have to define the notion of consistence (see for instance [80]).

Definition 5.1.4. The Castro-Lannes equations (5.14) are consistent of order $\mathcal{O}(\mu^k)$ with a system of equations S for ζ and $\overline{\mathbf{V}}$ if for all sufficiently smooth solutions $\left(\zeta, \mathbf{U}^{\mu}_{/\!/}, \boldsymbol{\omega}\right)$ of the Castro-Lannes equations (5.14), the pair $\left(\zeta, \overline{\mathbf{V}}[\varepsilon\zeta, \beta b]\left(\mathbf{U}^{\mu}_{/\!/}, \boldsymbol{\omega}\right)\right)$ (defined in (5.22)) solves S up to a residual of order $\mathcal{O}(\mu^k)$.

We also need an existence result for the Castro-Lannes formulation (5.14). This is the purpose of the next theorem (Theorem 4.3.6 in Section 4.3.4). We recall that the existence of the water waves equations is always under the so-called Rayleigh-Taylor condition assuming the positivity of the Rayleigh-Taylor coefficient \mathfrak{a} (see Sections 2.2.5 and 4.3.3 for the link between \mathfrak{a} and the Rayleigh-Taylor condition) where

$$\mathfrak{a} := \mathfrak{a}[\varepsilon\zeta,\beta b](\mathbf{U}_{/\!\!/}^{\mu},\boldsymbol{\omega}) = 1 + \varepsilon \left(\partial_t + \varepsilon \underline{\mathbf{V}}[\varepsilon\zeta,\beta b](\mathbf{U}_{/\!\!/}^{\mu},\boldsymbol{\omega})\cdot\nabla\right) \underline{\mathbf{w}}[\varepsilon\zeta,\beta b](\mathbf{U}_{/\!\!/}^{\mu},\boldsymbol{\omega}).$$
(5.19)

Notice also that in Proposition 4.3.4, we show how we can define initially the Rayleigh-Taylor coefficient \mathfrak{a} (see Equation (4.53)).

Theorem 5.1.5. Let A > 0, $N \ge 5$, $b \in H^{N+2}(\mathbb{R}^2)$. We assume that $\left(\zeta_0, (U^{\mu}_{/\!/})_0, \omega_0\right) \in H^N(\mathbb{R}^2) \times H^N(\mathbb{R}^2) \times H^{N-1}(\Omega_0)$ such that $\nabla^{\mu,\gamma} \cdot \omega_0 = 0$ and Condition (5.16) is satisfied. We suppose that $(\varepsilon, \beta, \gamma, \mu, \text{Ro})$ satisfy (5.18). Finally, we assume that

 $\exists h_{\min}, \mathfrak{a}_{\min} > 0, \varepsilon \zeta_0 + 1 - \beta b \ge h_{\min} \text{ and } \mathfrak{a}[\varepsilon \zeta_0, \beta b]((\mathbf{U}^{\mu}_{\mathbb{I}})_0, \boldsymbol{\omega}_0) \ge \mathfrak{a}_{\min},$

and

$$|\zeta_0|_{H^N} + |(\mathbf{U}_{/\!\!/}^{\mu})_0|_{H^N} + ||\boldsymbol{\omega}_0||_{H^{N-1}} \le A.$$

Then, there exists T > 0 and a unique classical solution $\left(\zeta, U^{\mu}_{/\!\!/}, \boldsymbol{\omega}\right)$ to the Castro-Lannes (5.14) with initial data $\left(\zeta_0, (U^{\mu}_{/\!\!/})_0, \boldsymbol{\omega}_0\right)$. Moreover,

$$T = \frac{T_0}{\max(\varepsilon, \beta, \frac{\varepsilon}{\operatorname{Ro}})} , \frac{1}{T_0} = c^1 \text{ and } \max_{[0,T]} |\zeta(t)|_{H^N} + \left| \mathbf{U}_{\mathbb{I}}^{\mu}(t) \right|_{H^N} + ||\boldsymbol{\omega}(t)||_{H^{N-1}} = c^2,$$

with $c^j = C\left(A, \mu_{\max}, \frac{1}{h_{\min}}, \frac{1}{\mathfrak{a}_{\min}}, |b|_{H^{N+2}}\right).$

Thanks to this theorem, we know that the quantities ζ , $U^{\mu}_{/\!\!/}$, ω and then $\overline{\mathbf{V}}$ (defined in (5.22)) remain bounded uniformly with respect to the small parameters during the time evolution of the flow, which will be essential to derive rigorously asymptotic models.

5.2 Boussinesq-Coriolis equations when $\gamma = 0$

This part is devoted to the derivation and the full justification of the Boussinesq-Coriolis equations (5.31) under a Coriolis forcing and with $\gamma = 0$. These equations are an order $\mathcal{O}(\mu^2)$ approximation of the water waves equations under the assumption that $\varepsilon, \beta = \mathcal{O}(\mu)$. The corresponding regime is called *long wave regime* or Boussinesq regime. Contrary to [33], whose approach is based on the averaged Euler equations, our derivation is based on the Castro-Lannes equations (5.14). Then, the asymptotic regime is

$$\mathcal{A}_{\text{Bouss}} = \left\{ \left(\varepsilon, \beta, \gamma, \mu, \text{Ro}\right), 0 \le \mu \le \mu_0, \frac{\varepsilon}{\text{Ro}} \le 1, \varepsilon = \mathcal{O}(\mu), \beta = \mathcal{O}(\mu), \gamma = 0 \right\}.$$
(5.20)

Remark 5.2.1. In fact, we can relax the assumption $\gamma = 0$ by only assuming that $\gamma = \mathcal{O}(\mu^2)$ since we neglect all the terms of order $\mathcal{O}(\mu^2)$ in the following.

We introduce the water depth

$$h(t, X) = 1 + \varepsilon \zeta(t, X) - \beta b(X), \qquad (5.21)$$

and the averaged horizontal velocity

$$\overline{\mathbf{V}} = \overline{\mathbf{V}}[\varepsilon\zeta,\beta b](\mathbf{U}_{\mathbb{H}}^{\mu},\boldsymbol{\omega})(t,X) = \frac{1}{h(t,X)} \int_{z=-1+\beta b(X)}^{\varepsilon\zeta(t,X)} \mathbf{V}[\varepsilon\zeta,\beta b](\mathbf{U}_{\mathbb{H}}^{\mu},\boldsymbol{\omega})(t,X,z)dz.$$
(5.22)

More generally, if u is a function defined in Ω , \overline{u} is its average and $u^* = u - \overline{u}$. In the following we denote $\mathbf{V} = (u, v)^t$. As noticed in [34], we have to introduce the "shear" velocity

$$\mathbf{V}_{\rm sh} = \mathbf{V}_{\rm sh}[\varepsilon\zeta,\beta b](\mathbf{U}^{\mu}_{/\!\!/},\boldsymbol{\omega})(t,X) = (u_{\rm sh},v_{\rm sh}) = \int_{z}^{\varepsilon\zeta} \boldsymbol{\omega}_{h}^{\perp}$$
(5.23)

and its average

$$\mathbf{Q} = \left(\mathbf{Q}_x, \mathbf{Q}_y\right)^t = \overline{\mathbf{V}}_{\mathrm{sh}} = \frac{1}{h} \int_{-1+\beta b}^{\varepsilon \zeta} \int_{z'}^{\varepsilon \zeta} \boldsymbol{\omega}_h^{\perp}$$

When $\gamma = 0$, $U^{\mu}_{\mathbb{I}} = (\underline{u} + \varepsilon \underline{w} \partial_x \zeta, \underline{v})^t$. Hence in the following, we denote

$$u_{\mathbb{I}} = \underline{u} + \varepsilon \underline{\mathbf{w}} \partial_x \zeta. \tag{5.24}$$

In this section, we do the asymptotic expansion with respect to μ of different quantities. In the following, we denote by R a remainder whose exact value has no importance and which is bounded uniformly with respect to μ .

Remark 5.2.2. Notice that thanks to Theorem 5.1.5, we know that the quantities ζ , U_{\parallel}^{μ} , ω , $\overline{\mathbf{V}}$ and \mathbf{U} remain bounded uniformly with respect to the small parameters during the time evolution of the flow. Furthermore, $\partial_t \zeta$, $\partial_t U_{\parallel}^{\mu}$, $\partial_t \omega$ and $\partial_t \mathbf{U}$ also remain bounded uniformly with respect to the small parameters during this time.

5.2.1 Asymptotic expansion for the velocity and useful identities

In this part, we give an expansion of the velocity with respect to μ . First we recall the following fact (See Proposition 4.5.2).

Proposition 5.2.3. If $(\zeta, U^{\mu}_{\parallel}, \boldsymbol{\omega})$ satisfy the Castro-Lannes system (5.14), we have $U^{\mu} \cdot N^{\mu,\gamma} = -\mu \nabla^{\gamma} \cdot (h \overline{\mathbf{V}}).$

This proposition, coupled with the first equation of (5.14), gives us an equation that links ζ to $\overline{\mathbf{V}}$. In particular, when $\gamma = 0$, we get an equation that links ζ to \overline{u} . We need also an expansion of u and v with respect to μ . The following proposition is for v.

Proposition 5.2.4. If $(\zeta, U^{\mu,0}_{/\!\!/}, \omega)$ satisfy the Castro-Lannes system (5.14), we have

$$v = \overline{v} + \sqrt{\mu} v_{\rm sh}^*,$$

$$\underline{v} = \overline{v} - \sqrt{\mu} Q_y,$$

$$\omega \cdot N^{\mu,0} = \partial_x v$$

and

$$\partial_t \underline{v} + \varepsilon \underline{u} \partial_x \underline{v} + \frac{\varepsilon}{\operatorname{Ro}} \underline{u} = 0.$$

Proof. Since $\operatorname{curl}^{\mu,0} \mathbf{U}^{\mu} = \mu \boldsymbol{\omega}$, we get that

$$\sqrt{\mu}\omega_x = -\partial_z v \text{ and } \omega_z = \partial_x v.$$
 (5.25)

Then, plugging the ansatz $v = \overline{v} + \sqrt{\mu}v_1$ in the first equation and using the fact that the average of v_1 is equal to 0 we get

$$\underline{v} = \overline{v} - \sqrt{\mu} \frac{1}{h} \int_{-1+\beta b}^{\varepsilon \zeta} \int_{z'}^{\varepsilon \zeta} \boldsymbol{\omega}_x.$$

Furthermore, from the equation on the second component of $U^{\mu,0}_{/\!\!/}$, we have

$$\partial_t \underline{v} + \varepsilon \boldsymbol{\omega} \cdot N^{\mu,0} \underline{u} + \frac{\varepsilon}{\operatorname{Ro}} \underline{u} = 0.$$

Then, using the second equation of (5.25), we get that $\underline{\omega} \cdot N^{\mu,0} = \partial_x \underline{v}$ and the result follows.

The expansion of u is more complex and also involves an expansion of w. It is the purpose of the following proposition. But before, we also have to introduce the following operators

$$T\left[\varepsilon\zeta,\beta b
ight]f = \int_{z}^{\varepsilon\zeta} \partial_{x}^{2} \int_{-1+\beta b}^{z'} f \text{ and } T^{*}\left[\varepsilon\zeta,\beta b
ight]f = \left(T\left[\varepsilon\zeta,\beta b
ight]f
ight)^{*},$$

When no confusion is possible, we denote $T = T [\varepsilon \zeta, \beta b]$ and $T^* = T^* [\varepsilon \zeta, \beta b]$.

Proposition 5.2.5. If $(\zeta, U^{\mu,0}_{/\!\!/}, \omega)$ satisfy the Castro-Lannes system (5.14), we have

$$\begin{split} u &= \overline{u} + \sqrt{\mu} u_{\rm sh}^* + \mu T^* \overline{u} + \mu^{\frac{3}{2}} T^* u_{\rm sh}^* + \mu^2 R, \\ \underline{u} &= \overline{u} - \sqrt{\mu} \mathbf{Q}_x + \mu \underline{T^*} \overline{u} - \mu^{\frac{3}{2}} \overline{T} u_{\rm sh}^* + \mu^2 R, \end{split}$$

where $T^*\overline{u} = -\frac{1}{2}\left([z+1-\beta b]^2 - \frac{h^2}{3}\right)\partial_x^2\overline{u} + \beta R$. We also have $w = -\mu\partial_x\left(\int_{-1+\beta b}^z u\right),$ $\underline{w} = -\mu h\partial_x\overline{u} - \mu^{\frac{3}{2}}\partial_x hQ_x + \max(\mu^2, \beta\mu)R,$

and

$$u_{\mathbb{I}} = \overline{u} - \sqrt{\mu} \mathbf{Q}_x - \mu \frac{1}{3h} \partial_x \left(h^3 \partial_x \overline{u} \right) - \mu^{\frac{3}{2}} \left(\overline{T} u_{\mathrm{sh}}^* + \mathbf{Q}_x \left(\partial_x h \right)^2 \right) + \max(\mu^2, \beta \mu) R.$$

Proof. This proof is a small adaptation of part 2.2 in [33] and Part 4.2 in [101]. We recall the main steps. Using the fact that the velocity is divergence free and Proposition 5.2.3, we get

$$\mathbf{w} = -\mu \partial_x \left(\int_{-1+\beta b}^z u \right).$$

Furthermore, since $\operatorname{curl}^{\mu,0} \mathbf{U}^{\mu} = \mu \boldsymbol{\omega}$, we get that

$$\sqrt{\mu}\boldsymbol{\omega}_y = \partial_z u - \partial_x \mathbf{w}.$$

Then, plugging the ansatz $u = \overline{u} + \sqrt{\mu}u_1$ and using the fact that the average of u_1 is zero, we get

$$u_1 = -\left(\int_z^{\varepsilon\zeta} \boldsymbol{\omega}_y\right)^* - \frac{1}{\sqrt{\mu}} \left(\int_z^{\varepsilon\zeta} \partial_x \mathbf{w}\right)^*$$

and

$$u = \overline{u} + \sqrt{\mu}u_{\rm sh}^* + \mu T^*u. \tag{5.26}$$

Then, the expansion for u follows by applying $1 + \mu T^*$ to the previous equation. Notice that $\underline{T^*u} = -\overline{Tu}$. The computation of $T^*\overline{u}$ follows from the fact that \overline{u} does not depend on z. Finally, the expansion \underline{w} and u_{\parallel} is the direct consequence for Proposition 5.2.3 and the expansion of u.

Thanks to the previous proposition, we can also get an expansion of $\partial_t u$ and $\partial_t w$.

Proposition 5.2.6. If $(\zeta, U^{\mu,0}_{/\!\!/}, \omega)$ satisfy the Castro-Lannes system (5.14), we have

$$\partial_t \left(u - \overline{u} - \sqrt{\mu} u_{\rm sh}^* - \mu T^* \overline{u} - \mu^{\frac{3}{2}} T^* u_{\rm sh}^* \right) = \mu^2 R,$$

$$\partial_t \left(\underline{u} - \overline{u} + \sqrt{\mu} Q_x - \mu \underline{T^* \overline{u}} + \mu^{\frac{3}{2}} \overline{T} u_{\rm sh}^* \right) = \mu^2 R,$$

$$\partial_t \left(\underline{w} + \mu h \partial_x \overline{u} + \mu^{\frac{3}{2}} \partial_x h Q_x \right) = \max(\mu^2, \beta \mu) R.$$
(5.27)

Proof. From Equality (5.26) we get that

$$u = (1 - \mu T^*) \left(\overline{u} + \sqrt{\mu} u_{\rm sh}^* \right) + \mu^2 T^* T^* u.$$
(5.28)

Hence the first and the second equations follows from Remark 5.2.2. For the third equation, we get the result thanks to Proposition (5.2.3) and Remark 5.2.2.

As [33] noticed, we can not express $\overline{Tu_{\rm sh}^*}$ in terms of ζ and $\overline{\mathbf{V}}$. Then, we have to introduce

$$\mathbf{V}^{\sharp} = (u^{\sharp}, v^{\sharp})^{t} = -\frac{24}{h^{3}} \int_{-1+\beta b}^{\varepsilon \zeta} \int_{z}^{\varepsilon \zeta} \int_{-1+\beta b}^{z} (u^{*}_{\mathrm{sh}}, v^{*}_{\mathrm{sh}})^{t}, \\
= \frac{12}{h^{3}} \int_{-1+\beta b}^{\varepsilon \zeta} (1+z-\beta b)^{2} (u^{*}_{\mathrm{sh}}, v^{*}_{\mathrm{sh}})^{t}.$$
(5.29)

Notice that the previous equality follows from a double integration by parts. We have the following Lemma.

Lemma 5.2.7. We have the following equalities

$$\overline{Tu_{sh}^*} = -\left(\varepsilon\partial_x\zeta\right)^2 Q_x + \frac{1}{h}\int_{-1+\beta b}^{\varepsilon\zeta} \partial_x \int_z^{\varepsilon\zeta} \partial_x \int_{-1+\beta b}^z u_{sh}^*$$
$$= -\left(\partial_x h\right)^2 Q_x - \frac{1}{24h}\partial_x^2 \left(h^3 u^{\sharp}\right) + \beta R.$$

Proof. We have

$$\partial_x \int_z^{\varepsilon\zeta} \partial_x \int_{-1+\beta b}^z u_{\rm sh}^* = \int_z^{\varepsilon\zeta} \partial_x^2 \int_{-1+\beta b}^z u_{\rm sh}^* + \varepsilon \partial_x \zeta \partial_x \int_{-1+\beta b}^z u_{\rm sh}^*$$
(5.30)

and the first equality follows from the fact that the average of $u_{\rm sh}^*$ is zero and that $\underline{u_{\rm sh}^*} = -\mathbf{Q}_x$. The second equality follows from the same arguments.

In the following section, we give equations for Q_x , $Q_y V^{\sharp}$ since we can not express these quantities with respect to ζ and \overline{V} . These equations are essential to derive the Boussinesq-Coriolis equations.

5.2.2 Equations for $\mathbf{Q}_x, \, \mathbf{Q}_y$ and \mathbf{V}^{\sharp}

In this part we give the equations satisfied by Q_x and Q_y at order $\mathcal{O}\left(\mu^{\frac{3}{2}}\right)$. The computations are similar to Part 5.4.1 in [33]. We start by Q_x .

Proposition 5.2.8. If $(\zeta, U_{\mathbb{J}}^{\mu,0}, \omega)$ satisfy the Castro-Lannes system (5.14), then, in the Boussinesq regime (5.20), Q_x satisfies the following equation

$$\partial_t Q_x + \varepsilon \overline{u} \partial_x Q_x + \varepsilon Q_x \partial_x \overline{u} + \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} \left(\underline{v} - \overline{v} \right) = \mu^{\frac{3}{2}} R,$$

and u_{sh}^* satisfies the equation

$$\partial_t u_{\rm sh}^* + \varepsilon \overline{u} \partial_x u_{\rm sh}^* + \varepsilon u_{\rm sh}^* \partial_x \overline{u} + \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} \left(\overline{v} - v\right) = \mu^{\frac{3}{2}} R.$$

Proof. Using the second equation of the vorticity equation of the Castro-Lannes system (5.14), we have

$$\partial_t \boldsymbol{\omega}_y + \varepsilon u \partial_x \boldsymbol{\omega}_y + \frac{\varepsilon}{\mu} \mathbf{w} \partial_z \boldsymbol{\omega}_y = \varepsilon \boldsymbol{\omega}_x \partial_x v + \frac{\varepsilon}{\sqrt{\mu}} \boldsymbol{\omega}_z \partial_z v + \frac{\varepsilon}{\mathrm{Ro}\sqrt{\mu}} \partial_z v.$$

Since $\omega_x = -\frac{1}{\sqrt{\mu}}\partial_z v$ and $\omega_z = \partial_x v$ we notice that $\varepsilon \omega_x \partial_x v + \frac{\varepsilon}{\sqrt{\mu}} \omega_z \partial_z v = 0$. Using Proposition 5.2.5 we get

$$\partial_t \boldsymbol{\omega}_y + \varepsilon \overline{u} \partial_x \boldsymbol{\omega}_y - \varepsilon \partial_x \left[(1 + z - \beta b) \overline{u} \right] \partial_z \boldsymbol{\omega}_y - \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} \partial_z v = \mu^{\frac{3}{2}} R_y$$

Then, integrating with respect to z, using the fact that $\partial_t \zeta + \partial_x (h\overline{u}) = 0$ and $u_{\rm sh} = -\int_z^{\varepsilon \zeta} \omega_y$, we get

$$\partial_t u_{\rm sh} + \varepsilon \overline{u} \partial_x u_{\rm sh} + \varepsilon u_{\rm sh} \partial_x \overline{u} + \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} \left(\underline{v} - v \right) = \varepsilon \partial_x \left[\left(1 + z - \beta b \right) \overline{u} \right] \partial_z u_{\rm sh} + \mu^{\frac{3}{2}} R d_x \overline{u} + \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} \left(\underline{v} - v \right) = \varepsilon \partial_x \left[\left(1 + z - \beta b \right) \overline{u} \right] \partial_z u_{\rm sh} + \mu^{\frac{3}{2}} R d_x \overline{u} + \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} \left(\underline{v} - v \right) = \varepsilon \partial_x \left[\left(1 + z - \beta b \right) \overline{u} \right] \partial_z u_{\rm sh} + \mu^{\frac{3}{2}} R d_x \overline{u} + \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} \left(\underline{v} - v \right) = \varepsilon \partial_x \left[\left(1 + z - \beta b \right) \overline{u} \right] \partial_z u_{\rm sh} + \mu^{\frac{3}{2}} R d_x \overline{u} + \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} \left(1 + z - \beta b \right) \overline{u} \right] \partial_z u_{\rm sh} + \mu^{\frac{3}{2}} R d_x \overline{u} + \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} \left(\frac{\varepsilon}{2} + \frac{\varepsilon}{2} \right) \left(\frac{\varepsilon}{2} + \frac{$$

Integrating again with respect to z, using the fact that $\partial_t \zeta + \partial_x (h\overline{u}) = 0$ and $Q_x = \overline{u_{\rm sh}^*}$, we obtain

$$\partial_t Q_x + \varepsilon \overline{u} \partial_x Q_x + \varepsilon Q_x \partial_x \overline{u} + \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} (\underline{v} - \overline{v}) = \mu^{\frac{3}{2}} R.$$

We have a similar equation for Q_y .

Proposition 5.2.9. If $(\zeta, U^{\mu,0}_{/\!\!/}, \omega)$ satisfy the Castro-Lannes system (5.14), then, in the Boussinesq regime (5.20), Q_x satisfies the following equation

$$\partial_t \mathbf{Q}_y + \varepsilon \overline{u} \partial_x \mathbf{Q}_y + \varepsilon \mathbf{Q}_x \partial_x \overline{v} + \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} \left(\overline{u} - \underline{u}\right) = \mu^{\frac{3}{2}} R$$

and v_{sh}^* satisfies the equation

$$\partial_t v_{\rm sh}^* + \varepsilon \overline{u} \partial_x v_{\rm sh}^* + \varepsilon u_{\rm sh}^* \partial_x \overline{v} + \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} \left(u - \overline{u} \right) = \mu^{\frac{3}{2}} R.$$

Proof. Using the first equation of the vorticity equation of the Castro-Lannes system (5.14), we have

$$\partial_t \boldsymbol{\omega}_x + \varepsilon u \partial_x \boldsymbol{\omega}_x + \frac{\varepsilon}{\mu} \mathbf{w} \partial_z \boldsymbol{\omega}_x = \varepsilon \boldsymbol{\omega}_x \partial_x u + \frac{\varepsilon}{\sqrt{\mu}} \boldsymbol{\omega}_z \partial_z u + \frac{\varepsilon}{\mathrm{Ro}\sqrt{\mu}} \partial_z u$$

Then, using the fact that $\nabla^{\mu,0} \cdot \boldsymbol{\omega} = 0$ and $\nabla^{\mu,0} \cdot U^{\mu,\gamma} = 0$, we get

$$\partial_t \boldsymbol{\omega}_x - \frac{\varepsilon}{\sqrt{\mu}} \partial_z \left(u \boldsymbol{\omega}_z \right) + \frac{\varepsilon}{\mu} \partial_z \left(w \boldsymbol{\omega}_x \right) = \frac{\varepsilon}{\text{Ro}\sqrt{\mu}} \partial_z u$$

then, we integrate with respect to z and, using the fact that $\partial_t \zeta - \frac{1}{\mu} \underline{U}^{\mu} \cdot N^{\mu,0} = 0$, $\omega_x = -\frac{1}{\sqrt{\mu}} \partial_z v$ and $\omega_z = \partial_x v$, we obtain

$$\partial_t \left(\int_{-1+\beta b}^{\varepsilon \zeta} \boldsymbol{\omega}_x \right) - \frac{\varepsilon}{\sqrt{\mu}} \underline{u} \partial_x \underline{v} + \frac{\varepsilon}{\sqrt{\mu}} u \partial_x v + \frac{\varepsilon}{\mu^{\frac{3}{2}}} \mathbf{w} \partial_z v + \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} \left(u - \underline{u} \right) = 0.$$

Then, we integrate again with respect to z and, using Proposition 5.2.4 and the fact that $\partial_t \zeta - \frac{1}{\mu} \underline{U}^{\mu} \cdot N^{\mu,0} = 0$, $U_b^{\mu} \cdot N_b^{\mu,0} = 0$, and $\nabla^{\mu,0} \cdot U^{\mu} = 0$, we get

$$\partial_t \mathbf{Q}_y - \frac{\varepsilon}{\sqrt{\mu}} \underline{u} \partial_x \underline{v} + \frac{\varepsilon}{\sqrt{\mu}} \frac{1}{h} \partial_x \left(\int_{-1+\beta b}^{\varepsilon \zeta} uv \right) + \frac{1}{\sqrt{\mu} h} \partial_t h \overline{v} + \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} \left(\overline{u} - \underline{u} \right) = 0.$$

Then, thanks to Propositions 5.2.3, 5.2.4 and 5.2.5 we finally obtain that

$$\partial_t \mathbf{Q}_y + \varepsilon \overline{u} \partial_x \mathbf{Q}_y + \varepsilon \mathbf{Q}_x \partial_x \overline{v} + \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} (\overline{u} - \underline{u}) = \mu^{\frac{3}{2}} R.$$

Notice that we give in Subsection 5.4.1 a generalization of the two previous propositions to the fully nonlinear Green-Naghdi regime. Furthermore, in the following proposition we give an equation for \mathbf{V}^{\sharp} up to terms of order $\mathcal{O}(\sqrt{\mu})$.

Proposition 5.2.10. If $(\zeta, U^{\mu,0}_{\parallel}, \omega)$ satisfy the Castro-Lannes system (5.14), then \mathbf{V}^{\sharp} satisfies the following equation

$$\partial_t \mathbf{V}^{\sharp} + \varepsilon \mathbf{V}^{\sharp} \partial_x \overline{u} + \varepsilon \overline{u} \partial_x \mathbf{V}^{\sharp} + \frac{\varepsilon}{\mathrm{Ro}} \mathbf{V}^{\sharp \perp} = \max\left(\varepsilon, \frac{\varepsilon}{\mathrm{Ro}}\right) \sqrt{\mu} R.$$

Proof. The proof is similar to the computation in Part 4.4 in [33]. After multiplying by $(1 + z - \beta b)^2$ and integrating with respect to z the second equations of Propositions 5.2.8 and 5.2.9, we neglect all the term of order $\mathcal{O}(\sqrt{\mu})$. Then, using the fact that $\partial_t \zeta + \partial_x (h\overline{u}) = 0$ and $\mathbf{V} - \overline{\mathbf{V}} = \sqrt{\mu} \mathbf{V}_{sh}^* + \mu R$, we get the result.

5.2.3 The Boussinesq-Coriolis equations

We can now establish the Boussinesq-Coriolis equations when d = 1. The Boussinesq-Coriolis equations are the following system

$$\begin{cases} \partial_t \zeta + \partial_x \left(h \overline{u} \right) = 0, \\ \left(1 - \frac{\mu}{3} \partial_x^2 \right) \partial_t \overline{u} + \partial_x \zeta + \varepsilon \overline{u} \partial_x \overline{u} - \frac{\varepsilon}{\text{Ro}} \overline{v} + \frac{\varepsilon}{\text{Ro}} \mu^{\frac{3}{2}} \frac{1}{24} \partial_x^2 \frac{v^{\sharp}}{h} = 0, \\ \partial_t \overline{v} + \varepsilon \overline{u} \partial_x \overline{v} + \frac{\varepsilon}{\text{Ro}} \overline{u} = 0, \\ \partial_t \mathbf{V}^{\sharp} + \varepsilon \mathbf{V}^{\sharp} \partial_x \overline{u} + \varepsilon \overline{u} \partial_x \mathbf{V}^{\sharp} + \frac{\varepsilon}{\text{Ro}} \mathbf{V}^{\sharp \perp} = 0, \end{cases}$$
(5.31)

where \mathbf{V}^{\sharp} is defined in (5.29). We can show that the Boussinesq-Coriolis equations are an order $\mathcal{O}(\mu^2)$ approximation of the water waves equations.

Remark 5.2.11. Inspired by [82], we can renormalize V^{\sharp} by h and, using the first equation of (5.31), we get the following equation

$$\partial_t \left(\frac{\mathbf{V}^{\sharp}}{h} \right) + \varepsilon \overline{u} \partial_x \left(\frac{\mathbf{V}^{\sharp}}{h} \right) + \frac{\varepsilon}{\operatorname{Ro}} \left(\frac{\mathbf{V}^{\sharp}}{h} \right)^{\perp} = 0.$$

This remark will be useful for the local existence (Proposition 5.2.15).

Proposition 5.2.12. In the Boussinesq regime \mathcal{A}_{Bouss} (5.20), the Castro-Lannes equations (5.14) are consistent at order $\mathcal{O}(\mu^2)$ with the Boussinesq-Coriolis equations (5.31) in the sense of Definition 5.1.4.

Proof. The first equation of the Boussinesq-Coriolis equations is always satisfied for a solution of the Castro-Lannes formulation by Proposition 5.2.3. For the second equation, we use Proposition 5.2.5, Proposition 5.2.8 together with Proposition 5.2.6, Lemma 5.2.7 and Proposition 5.2.10 (we recall that $\varepsilon = \mathcal{O}(\mu)$). Notice the fact that all the terms with Q_x disappear. We also use the fact that

$$h^3 v^{\sharp} = \frac{v^{\sharp}}{h} + \mu R.$$

Then, the third equation follows from Proposition 5.2.5, 5.2.5 and 5.2.9 (all the terms with Q_y disappear also).

We notice that contrary to the classical Boussinesq equations, we have a new term due to the vorticity that we can not neglect in presence of a Coriolis forcing. In our knowledge, this term was not highlighted before in the literature.

Remark 5.2.13. In the Boussinesq-Coriolis system we could simplify the term $\partial_x^2 \frac{v^{\sharp}}{h}$ by $\partial_x^2 v^{\sharp}$ since these terms are equal up to a remainder of order $\mathcal{O}(\mu)$. However, the term $\partial_x^2 \frac{v^{\sharp}}{h}$ will be essential for the local existence (see Remark 5.2.16).

Remark 5.2.14. If we assume that $\frac{\varepsilon}{Ro} = \mathcal{O}(\sqrt{\mu})$, we can neglect the term with v^{\sharp} in the second equation of (5.31) and we obtain

$$\begin{cases} \partial_t \zeta + \partial_x \left(h\overline{u}\right) = 0, \\ \left(1 - \frac{\mu}{3} \partial_x^2\right) \partial_t \overline{u} + \partial_x \zeta + \varepsilon \overline{u} \partial_x \overline{u} - \frac{\varepsilon}{\text{Ro}} \overline{v} = 0, \\ \partial_t \overline{v} + \varepsilon \overline{u} \partial_x \overline{v} + \frac{\varepsilon}{\text{Ro}} \overline{u} = 0. \end{cases}$$
(5.32)

This system is the classical Boussinesq equations with a standard Coriolis forcing. It is consistent of order $\mathcal{O}(\mu^2)$ with the Boussinesq-Coriolis equations (5.31). We use this system in Subsections 5.3.3 and 5.3.4.

5.2.4 Full justification of the Boussinesq-Coriolis equations

In this part, we fully justify the Boussinesq-Coriolis equations (5.31). In the following we denote by u the quantity \overline{u} and by v the quantity \overline{v} . We show that the Boussinesq-Coriolis equations are wellposed. We define the energy space

$$X^{s}(\mathbb{R}) = H^{s}(\mathbb{R}) \times H^{s+1}(\mathbb{R}) \times H^{s}(\mathbb{R}) \times H^{s+1}(\mathbb{R}) \times H^{s+1}(\mathbb{R}),$$
(5.33)

endowed with the norm

$$|(\zeta, u, v, \mathbf{W})|_{X^{s}_{\mu}}^{2} = |\zeta|_{H^{s}}^{2} + |u|_{H^{s}}^{2} + \mu |\partial_{x}u|_{H^{s}}^{2} + |v|_{H^{s}}^{2} + |\mathbf{W}|_{H^{s}}^{2} + \mu |\partial_{x}\mathbf{W}|_{H^{s}}^{2}.$$
(5.34)

Proposition 5.2.15. Let A > 0, $s > \frac{1}{2} + 1$, $(\zeta_0, u_0, v_0, \mathbf{V}_0^{\sharp}) \in X^s(\mathbb{R})$ and $b \in H^{s+1}(\mathbb{R})$. We suppose that $(\varepsilon, \beta, \gamma, \mu, \operatorname{Ro}) \in \mathcal{A}_{Bouss}$. We assume that

$$\exists h_{\min} > 0$$
, $\varepsilon \zeta_0 + 1 - \beta b \ge h_{\min}$

and

$$\left| \left(\zeta_0, u_0, v_0, \frac{\mathbf{V}_0^{\sharp}}{1 + \varepsilon \zeta_0 - \beta b} \right) \right|_{X_{\mu}^s} + |b|_{H^{s+1}} \le A.$$

Then, there exists an existence time T > 0 and a unique solution $\left(\zeta, u, v, \mathbf{V}^{\sharp}\right)$ on [0, T] to the Boussinesq-Coriolis equations (5.31) with initial data $\left(\zeta_{0}, u_{0}, v_{0}, \mathbf{V}_{0}^{\sharp}\right)$ such that we have $\left(\zeta, u, v, \frac{\mathbf{V}^{\sharp}}{h}\right) \in \mathcal{C}\left([0, T]; X^{s}(\mathbb{R})\right)$ with $h = 1 + \varepsilon \zeta - \beta b$. Moreover,

$$T = \frac{T_0}{\max(\mu, \frac{\varepsilon}{\text{Ro}}\sqrt{\mu})} \ , \ \frac{1}{T_0} = c^1 \text{ and } \max_{[0,T]} \left| \left(\zeta, u, v, \frac{\mathbf{V}^{\sharp}}{h}\right)(t, \cdot) \right|_{X^s_{\mu}} = c^2$$

with $c^j = C\left(A, \mu_{\max}, \frac{1}{h_{\min}}\right)$.

Proof. We only give the energy estimates. For the existence see for instance the proof of Theorem 1 in [72]. We assume that $\left(\zeta, u, v, \mathbf{V}^{\sharp}\right)$ solves (5.31) on $\left[0, \frac{T_0}{\max(\mu, \frac{\varepsilon}{Ro}\sqrt{\mu})}\right]$ and that

$$1 + \varepsilon \zeta - \beta b \ge \frac{h_{\min}}{2}$$
 on $\left[0, \frac{T_0}{\max(\mu, \frac{\varepsilon}{\operatorname{Ro}}\sqrt{\mu})}\right]$

We denote $U = (\zeta, u, v)^t$ and we focus first on the first three equations. This part is a small adaptation of the proof of Theorem 1 in [72]. The the first three equations of the Boussinesq-Coriolis equations can be symmetrized, as an hyperbolic system, by multiplying the second and the third equations by $h = 1 + \varepsilon \zeta - \beta b$. Then, we obtain the following system

$$\mathcal{A}_0(U)\partial_t U + \mathcal{A}_1(U)\partial_x U + B_1 U + \frac{\varepsilon}{\operatorname{Ro}}B_2(U)U = \frac{\varepsilon}{\operatorname{Ro}}\mu^{\frac{3}{2}}F(h,v^{\sharp}),$$

where

$$\mathcal{A}_0(U) = \begin{pmatrix} 1 & 0 & 0\\ 0 & h - \mu \frac{h}{3} \partial_x^2 & 0\\ 0 & 0 & h \end{pmatrix}, \ \mathcal{A}_1(U) = \begin{pmatrix} \varepsilon u & h & h\\ h & \varepsilon h u & 0\\ h & 0 & \varepsilon h u \end{pmatrix}$$

and

$$B_1 = \begin{pmatrix} 0 & -\beta \partial_x b & 0 \\ 0 & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix} , B_2(U) = \begin{pmatrix} 0 & 0 & 0 \\ 0 & 0 & -h \\ 0 & h & 0 \end{pmatrix} \text{ and } F(h, v^{\sharp}) = \begin{pmatrix} 0 \\ -\frac{h}{24} \partial_x^2 \frac{v^{\sharp}}{h} \\ 0 \end{pmatrix}.$$

Then we remark that \mathcal{A}_1 is symmetric and there exists $c_1, c_2 = C\left(\frac{1}{h_{\min}}, |h|_{L^{\infty}}\right)$ such that

$$c_1 |\partial_x f|_2^2 \le \left(-\frac{1}{3}\partial_x \left(h\partial_x f\right), f\right)_2 \le c_2 |\partial_x f|_2^2.$$

Hence we introduce the symmetric matrix operator

$$S(U) = \begin{pmatrix} 1 & 0 & 0\\ 0 & h - \frac{\mu}{3}\partial_x \left(h\partial_x \cdot\right) & 0\\ 0 & 0 & h \end{pmatrix}$$

and the energy associated

$$\mathcal{E}^s(U) = \left(S(U)\Lambda^s U, \Lambda^s U\right)_2.$$

Then, we see that

$$(\Lambda^s B_2(U)U, \Lambda^s U)_2 = 0$$

and by standard product estimates we get

$$\mu^{\frac{3}{2}} \left| \left(h\Lambda^s \partial_x^2 \frac{v^{\sharp}}{h}, \Lambda^s u \right)_2 \right| \le \sqrt{\mu} C(\mathcal{E}^s(U), |b|_{H^{s+1}}) \sqrt{\mu} \left| \frac{v^{\sharp}}{h} \right|_{H^{s+1}}.$$
(5.35)

Furthermore, notice that

$$\begin{split} \mu \left| \partial_t \partial_x u \right|_{H^s} &= \mu \left| \left(1 - \frac{\mu}{3} \partial_x^2 \right)^{-1} \partial_x \left(\partial_x \zeta + \varepsilon u \partial_x u - \frac{\varepsilon}{\operatorname{Ro}} v + \frac{\varepsilon}{\operatorname{Ro}} \frac{\mu^{\frac{3}{2}}}{24} \partial_x^2 \frac{v^{\sharp}}{h} \right) \right|_{H^s}, \\ &\leq C \left(\mu_{\max}, \mathcal{E}^s(U), \sqrt{\mu} \left| \partial_x \frac{v^{\sharp}}{h} \right|_{H^s} \right). \end{split}$$

and therefore

$$\left(\frac{\mu}{3}\partial_x h\Lambda^s \partial_x \partial_t u, \Lambda^s u\right)_2 \le \mu C \left(\mathcal{E}^s(U), |b|_{H^{s+1}}, \sqrt{\mu} \left| \partial_x \frac{v^\sharp}{h} \right|_{H^s} \right).$$

Gathering all the previous estimate and proceeding as in [72] we obtain

$$\frac{d}{dt}\mathcal{E}^{s}(U) \leq \max\left(\mu, \frac{\varepsilon}{\operatorname{Ro}}\sqrt{\mu}\right)_{2} C\left(\mathcal{E}^{s}(U), |b|_{H^{s+1}}, \left|\frac{v^{\sharp}}{h}\right|_{H^{s}}, \sqrt{\mu}\left|\partial_{x}\frac{v^{\sharp}}{h}\right|_{H^{s}}\right).$$

Furthermore, using Remark 5.2.11 and the Kato-Ponce estimate, we get

$$\begin{split} \frac{d}{dt} \left| \frac{\mathbf{V}^{\sharp}}{h} \right|_{H^s}^2 &\leq \mu C \left| u \right|_{H^s} \left| \frac{\mathbf{V}^{\sharp}}{h} \right|_{H^s}^2, \\ \frac{d}{dt} \mu \left| \partial_x \frac{\mathbf{V}^{\sharp}}{h} \right|_{H^s}^2 &\leq \mu C \left(\sqrt{\mu} \left| \partial_x u \right|_{H^s} \left| \frac{\mathbf{V}^{\sharp}}{h} \right|_{H^s}^2 + \left| u \right|_{H^s} \sqrt{\mu} \left| \partial_x \frac{\mathbf{V}^{\sharp}}{h} \right|_{H^s} \right) \sqrt{\mu} \left| \partial_x \frac{\mathbf{V}^{\sharp}}{h} \right|_{H^s}. \end{split}$$

Then, the result follows.

Remark 5.2.16. Notice that the previous energy estimates do not imply that $\mathbf{V}^{\sharp} \in H^{s+1}(\mathbb{R})$. Hence, it is essential that in Inequality (5.35) we have the term $\partial_x^2 \frac{v^{\sharp}}{h}$ and not simply $\partial_x^2 v^{\sharp}$ (see Remark 5.2.13).

Then, we similarly can prove a local wellposedness result for System (5.32).

Corollary 5.2.17. Let A > 0, $s > \frac{1}{2} + 1$, $(\zeta_0, u_0, v_0) \in H^s(\mathbb{R}) \times H^{s+1}(\mathbb{R}) \times H^s(\mathbb{R})$ and $b \in H^{s+1}(\mathbb{R})$. We suppose that $(\varepsilon, \beta, \gamma, \mu, \operatorname{Ro}) \in \mathcal{A}_{Bouss}$. We assume that

$$\exists h_{\min} > 0, \varepsilon \zeta_0 + 1 - \beta b \ge h_{\min}$$

and

$$|\zeta_0|_{H^s} + |u_0|_{H^s} + \sqrt{\mu} \, |\partial_x u_0|_{H^s} + |v_0|_{H^s} + |b|_{H^{s+1}} \le A$$
Then, there exists an existence time T > 0 and a unique solution to the Boussinesq-Coriolis equations (5.31) $(\zeta, u, v) \in \mathcal{C}([0, T]; H^s(\mathbb{R}) \times H^{s+1}(\mathbb{R}) \times H^s(\mathbb{R}))$ with initial data (ζ_0, u_0, v_0) . Moreover,

$$T = \frac{T_0}{\mu} , \frac{1}{T_0} = c^1 \text{ and } \max_{[0,T]} |\zeta(t,\cdot)|_{H^s} + |u(t,\cdot)|_{H^s} + \sqrt{\mu} |\partial_x u(t,\cdot)|_{H^s} + |v(t,\cdot)|_{H^s} = c^2,$$

with $c^j = C\left(A, \mu_{\max}, \frac{1}{h_{\min}}\right).$

Furthermore, we have a stability result for the Boussinesq-Coriolis system (5.31).

$$\begin{aligned} & \text{Proposition 5.2.18. Let the assumptions of Proposition 5.2.15 satisfied. Suppose that there} \\ & exists \left(\tilde{\zeta}, \tilde{u}, \tilde{v}, \frac{\tilde{\mathbf{v}}^{\sharp}}{\tilde{h}}\right) \in \mathcal{C}\left(\left[0, \frac{T_{0}}{\max\left(\mu, \frac{\varepsilon\sqrt{\mu}}{R_{0}}\right)}\right]; X^{s}(\mathbb{R})\right) \text{ satisfying} \\ & \left\{ \begin{array}{l} \partial_{t}\tilde{\zeta} + \partial_{x}\left(\tilde{h}\tilde{u}\right) = R_{1}, \\ \left(1 - \frac{\mu}{3}\partial_{x}^{2}\right)\partial_{t}\tilde{u} + \partial_{x}\tilde{\zeta} + \varepsilon\tilde{u}\partial_{x}\tilde{u} - \frac{\varepsilon}{R_{0}}\tilde{v} + \frac{\varepsilon}{R_{0}}\mu^{\frac{3}{2}}\frac{1}{24}\partial_{x}\frac{\tilde{v}^{\sharp}}{\tilde{h}} = R_{2}, \\ \partial_{t}\tilde{v} + \varepsilon\tilde{u}\partial_{x}\tilde{v} + \frac{\varepsilon}{R_{0}}\tilde{u} = R_{3}, \\ \partial_{t}\frac{\tilde{\mathbf{V}}^{\sharp}}{\tilde{h}} + \varepsilon + \varepsilon\tilde{u}\partial_{x}\frac{\tilde{\mathbf{V}}^{\sharp}}{\tilde{h}} + \frac{\varepsilon}{R_{0}}\frac{\tilde{\mathbf{V}}^{\sharp}}{\tilde{h}} = R_{4}, \end{aligned} \right. \end{aligned}$$

where $\tilde{h} = 1 + \varepsilon \tilde{\zeta} - \beta b$ and with $R = (R_1, R_2, R_3, R_4) \in L^{\infty} \left(\left[0, \frac{T_0}{\max\left(\mu, \frac{\varepsilon \sqrt{\mu}}{R_0}\right)} \right]; X^s(\mathbb{R}) \right)$. Then, if we denote $\mathfrak{e} = \left(\zeta, u, v, \mathbf{V}^{\sharp} \right) - \left(\tilde{\zeta}, \tilde{u}, \tilde{v}, \tilde{\mathbf{V}}^{\sharp} \right)$ where $\left(\zeta, u, v, \mathbf{V}^{\sharp} \right)$ is the solution given in Proposition 5.2.15, we have

$$\left|\mathfrak{e}(t)\right|_{X_{\mu}^{s-1}} \leq C\left(A, \mu_{\max}, \frac{1}{h_{\min}}, \left|\left(\tilde{\zeta}, \tilde{u}, \tilde{v}, \frac{\tilde{\mathbf{V}}^{\sharp}}{\tilde{h}}, R\right)\right|_{L^{\infty}\left([0,t]; X_{\mu}^{s} \times X_{\mu}^{s}\right)}\right) \left(\left|\mathfrak{e}_{|t=0}\right|_{X_{\mu}^{s-1}} + t \left|R\right|_{X_{\mu}^{s}}\right).$$

Proof. This proof is a small adaptation of the one of Proposition 6.5 in [80] (see also [9]). We denote $\tilde{U} = (\tilde{\zeta}, \tilde{u}, \tilde{v})$, $\mathfrak{e}_a = U - \tilde{U}$, $R_a = (R_1, R_2, R_3)$ and we keep the notations of the proof of Proposition 5.2.15. Since the Boussinesq-Coriolis equations are symmetrizable, we have

$$\begin{cases} \mathcal{A}_{0}(U)\partial_{t}\mathfrak{e}_{a} + \mathcal{A}_{1}(U)\partial_{x}\mathfrak{e}_{a} + B_{1}\mathfrak{e}_{a} + \frac{\varepsilon}{\mathrm{Ro}}B_{2}(U)\mathfrak{e}_{a} = \frac{\varepsilon}{\mathrm{Ro}}\mu^{\frac{3}{2}}F(h,v^{\sharp} - \tilde{v}^{\sharp}) + G,\\ \partial_{t}\left(\frac{\mathbf{V}^{\sharp}}{h} - \frac{\tilde{\mathbf{V}}^{\sharp}}{\tilde{h}}\right) + \varepsilon u\partial_{x}\left(\frac{\mathbf{V}^{\sharp}}{h} - \frac{\tilde{\mathbf{V}}^{\sharp}}{\tilde{h}}\right) + \frac{\varepsilon}{\mathrm{Ro}}\left(\frac{\mathbf{V}^{\sharp}}{h} - \frac{\tilde{\mathbf{V}}^{\sharp}}{\tilde{h}}\right)^{\perp} = H, \end{cases}$$

where

$$G = F(h, \tilde{v}^{\sharp}) - F(\tilde{h}, \tilde{v}^{\sharp}) - R_a - (\mathcal{A}_0(U) - \mathcal{A}_0(\tilde{U}))\partial_t \tilde{U} - (\mathcal{A}_1(U) - \mathcal{A}_1(\tilde{U}))\partial_x \tilde{U} - \frac{\varepsilon}{\mathrm{Ro}} (B_2(U) - B_2(\tilde{U}))U, H = \varepsilon(\tilde{u} - u)\partial_x \frac{\tilde{\mathbf{V}}^{\sharp}}{\tilde{h}} + R_4.$$

Then, using standard products estimates, we get (notice that $s > \frac{1}{2} + 1$)

$$\left(\Lambda^{s-1}G,\Lambda^{s-1}\mathfrak{e}_{a}\right)_{2} \leq \left(\left|R\right|_{X_{\mu}^{s}} + \mu C\left(\mathcal{E}^{s}(\tilde{U}),\mathcal{E}^{s-1}(\partial_{t}\tilde{U}),\left|\frac{\tilde{v}^{\sharp}}{\tilde{h}}\right|_{H^{s}},\sqrt{\mu}\left|\partial_{x}\frac{\tilde{v}^{\sharp}}{\tilde{h}}\right|_{H^{s}}\right)|\mathfrak{e}|_{X^{s-1}}\right)|\mathfrak{e}|_{X^{s-1}}$$

and

$$\left(\Lambda^{s-1}H,\Lambda^{s-1}\left(\frac{\mathbf{V}^{\sharp}}{h}-\frac{\tilde{\mathbf{V}}^{\sharp}}{\tilde{h}}\right)\right)_{2} \leq \left(|R|_{X^{s}_{\mu}}+\mu C\left(\mathcal{E}^{s}(\tilde{U}),\left|\frac{\tilde{\mathbf{V}}^{\sharp}}{\tilde{h}}\right|_{H^{s}},\sqrt{\mu}\left|\partial_{x}\frac{\tilde{v}^{\sharp}}{\tilde{h}}\right|_{H^{s}}\right)|\mathfrak{e}|_{X^{s-1}}\right)|\mathfrak{e}|_{X^{s-1}}.$$

Then, the result follows from energy estimates and the Gronwall's lemma.

The two previous results and Theorem 5.1.5 allow us to fully justify the Boussinesq-Coriolis equations. We recall that the operators $\overline{\mathbf{V}}[\varepsilon\zeta_0,\beta b](\mathbf{U}^{\mu,0}_{/\!\!/},\boldsymbol{\omega})$ and $\mathbf{V}_{\mathrm{sh}}[\varepsilon\zeta,\beta b](\mathbf{U}^{\mu,0}_{/\!\!/},\boldsymbol{\omega})(t,X)$ are defined in (5.22) and (5.23) respectively.

Theorem 5.2.19. Let $N \ge 7$ and $(\varepsilon, \beta, \gamma, \mu, \operatorname{Ro}) \in \mathcal{A}_{Bouss}$. We assume that we are under the assumptions of Theorem 5.1.5. Then, we can define the following quantity $(u_0, v_0)^t = \overline{\mathbf{V}}[\varepsilon\zeta_0, \beta b]((\mathbf{U}_{\mathbb{A}}^{\mu,0})_0, \boldsymbol{\omega}_0), (u, v)^t = \overline{\mathbf{V}}[\varepsilon\zeta, \beta b](\mathbf{U}_{\mathbb{A}}^{\mu,0}, \boldsymbol{\omega}), \mathbf{V}_0^{\sharp} = \mathbf{V}^{\sharp}[\varepsilon\zeta_0, \beta b](((\mathbf{U}_{\mathbb{A}}^{\mu,0})_0, \boldsymbol{\omega}_0), \boldsymbol{\omega}_0), \boldsymbol{\omega}_0), \mathbf{V}_0^{\sharp} = \mathbf{V}^{\sharp}[\varepsilon\zeta, \beta b](\mathbf{U}_{\mathbb{A}}^{\mu,0}, \boldsymbol{\omega}_0), and there exists a time <math>T > 0$ such that

(i) T has the form

$$T = rac{T_0}{\max(\mu, rac{\varepsilon}{Ro})}, \ and \ rac{1}{T_0} = c^1$$

(ii) There exists a unique classical solution $(\zeta_B, u_B, v_B, \mathbf{V}_B^{\sharp})$ of (5.31) on [0, T] with the initial data $(\zeta_0, u_0, v_0, \mathbf{V}_0^{\sharp})$.

(iii) There exists a unique classical solution $\left(\zeta, \mathbf{U}^{\mu,0}_{\mathbb{A}}, \boldsymbol{\omega}\right)$ of System (5.14) with initial data $\left(\zeta_{0}, (\mathbf{U}^{\mu,0}_{\mathbb{A}})_{0}, \boldsymbol{\omega}_{0}\right)$ on [0,T].

(iv) The following error estimate holds, for $0 \le t \le T$,

$$\left| \left(\zeta, u, v, \mathbf{V}^{\sharp} \right) - \left(\zeta_B, u_B, v_B, \mathbf{V}_B^{\sharp} \right) \right|_{L^{\infty}([0,t] \times \mathbb{R})} \leq \mu^2 t \, c^2,$$

with $c^j = C\left(A, \mu_{\max}, \frac{1}{h_{\min}}, \frac{1}{\mathfrak{a}_{\min}}, |b|_{H^{N+2}}\right).$

This theorem shows that the solutions of the water waves system (5.14) remain close to the solutions of the Boussinesq-Coriolis equations (5.31) over times $\mathcal{O}\left(\frac{1}{\max(\mu, \frac{\varepsilon}{R_0})}\right)$ with an accuracy of order $\mathcal{O}(\mu)$. Hence, if one considers a system and wants to show that the solutions of this system remain close to the solutions of the waves equations over times $\mathcal{O}\left(\frac{1}{\max(\mu, \frac{\varepsilon}{R_0})}\right)$ with an accuracy of order $\mathcal{O}(\mu)$, it is sufficient to compare the solutions of this system with the solutions of the Boussinesq-Coriolis equations (5.31). It is our approach in the following.

5.3 Different asymptotic models in the Boussinesq regime over a flat bottom

The Boussinesq-Coriolis equations (5.31) are particularly interesting for the evolution of offshore water waves. Without vorticity, we get the so-called Boussinesq equations. When we add a rotation, and in particular Coriolis effects, a standard assumption made by physicists is to also assume that the Rossby radius, or Obukhov radius, $\frac{\sqrt{gH}}{f}$ is greater than the typical length of the waves L (see for instance [117], [61], [84]). Then, different regimes for the Coriolis parameter were considered depending on whether the rotation is weak or not ([113], [60], [64]). In this paper, we consider three different regimes (noticed in [60]), a strong rotation ($\frac{\varepsilon}{\text{Ro}} \leq \mathcal{O}(\sqrt{\mu})$) and very weak rotation ($\frac{\varepsilon}{\text{Ro}} = \mathcal{O}(\mu)$). We derive and fully justify different asymptotic models when the bottom is flat : a linear equation admitting the so-called Poincaré waves (5.39) ; the Ostrovsky equation (5.43), which is a generalization of the KdV equation (5.52) in presence of a Coriolis forcing, when the rotation is weak; and the KdV equation when the rotation is very weak.

5.3.1 Strong rotation, the Poincaré waves

In this part we are interested in the behaviour of long water waves under a strong Coriolis forcing (in the sense of [60]). We suppose that $\frac{\varepsilon}{B_0}$ is of order 1. The asymptotic regime is

$$\mathcal{A}_{\text{Poin}} = \left\{ \left(\varepsilon, \beta, \gamma, \mu, \text{Ro}\right), 0 \le \mu \le \mu_0, \varepsilon = \mu, \beta = \gamma = 0, \frac{\varepsilon}{\text{Ro}} = 1 \right\}.$$
 (5.36)

Then, the Boussinesq-Coriolis equations (5.31) become

$$\begin{cases} \partial_t \zeta + \partial_x \left(\left(1 + \mu \zeta \right) u \right) = 0, \\ \left(1 - \frac{\mu}{3} \partial_x^2 \right) \partial_t u + \partial_x \zeta + \mu u \partial_x u - v + \frac{\mu^3}{24} \partial_x^2 \frac{v^\sharp}{h} = 0, \\ \partial_t v + \mu u \partial_x v + u = 0, \\ \partial_t \mathbf{V}^\sharp + \mu \mathbf{V}^\sharp \partial_x u + \mu u \partial_x \mathbf{V}^\sharp + \mathbf{V}^{\sharp \perp} = 0. \end{cases}$$
(5.37)

Our purpose is to justify the so-called Poincaré waves or Sverdrup waves ([134]), which are inertia-gravity waves. These are linear waves induced by the Coriolis forcing. If we drop all the terms of order $\mathcal{O}(\mu)$ in the Boussinesq-Coriolis equation we get the linear system

$$\begin{cases} \partial_t \zeta + \partial_x u = 0, \\ \partial_t u + \partial_x \zeta - v = 0, \\ \partial_t v + u = 0. \end{cases}$$
(5.38)

If we denote $U = (\zeta, u, v)^t$, by taking the Fourier transform, we get

$$\partial_t \widehat{U} = \mathcal{A} \widehat{U} \text{ with } \mathcal{A} = \begin{pmatrix} 0 & -i\xi & 0 \\ -i\xi & 0 & 1 \\ 0 & -1 & 0 \end{pmatrix}$$

and then,

$$\widehat{U} = \mathcal{S}(t,\xi)\widehat{U}_{0} = \begin{pmatrix} \frac{\xi^{2}\cos(\sqrt{\xi^{2}+1}t)+1}{\xi^{2}+1} & -i\xi\frac{\sin(\sqrt{\xi^{2}+1}t)}{\sqrt{\xi^{2}+1}} & i\xi\frac{\cos(\sqrt{\xi^{2}+1}t)-1}{\xi^{2}+1} \\ -i\xi\frac{\sin(\sqrt{\xi^{2}+1}t)}{\sqrt{\xi^{2}+1}} & \cos(\sqrt{\xi^{2}+1}t) & \frac{\sin(\sqrt{\xi^{2}+1}t)}{\sqrt{\xi^{2}+1}} \\ -i\xi\frac{\cos(\sqrt{\xi^{2}+1}t)-1}{\xi^{2}+1} & -\frac{\sin(\sqrt{\xi^{2}+1}t)}{\sqrt{\xi^{2}+1}} & \frac{\xi^{2}+\cos(\sqrt{\xi^{2}+1}t)}{\xi^{2}+1} \end{pmatrix} \widehat{U}_{0}.$$
(5.39)

Commonly, Poincaré waves are waves of the form

$$U(t,x) = e^{i(xk \pm t\sqrt{k^2+1})}U_0$$

They are solutions of the Klein-Gordon equation. In this setting, Poincaré waves correspond to solutions of System (5.38) of the form

$$\widehat{U}(t,\xi) = e^{i\pm t\sqrt{\xi^2+1}}\widehat{U}_0(\xi)$$

Therefore, a solution of System (5.38) is a sum of two Poincaré waves if and only if

$$\begin{pmatrix} \frac{1}{\xi^2+1} & 0 & -\frac{i\xi}{\xi^2+1} \\ 0 & 0 & 0 \\ \frac{i\xi}{\xi^2+1} & 0 & \frac{\xi^2}{\xi^2+1} \end{pmatrix} \widehat{U}_0 = 0,$$

which is equivalent to

$$\zeta_0 = \partial_x v_0. \tag{5.40}$$

In the following, we denote by S(t) the semi-group of the linear Boussinesq-Coriolis equation. The end of this part is devoted to the full justification of Poincaré waves. The following lemma shows that Condition (5.40) is propagated by the flow of System (5.38).

Lemma 5.3.1. Let (ζ, u, v) be a solution of (5.38) such that $(\zeta, u, v)_{|t=0} = 0$ satisfies Condition (5.40). Then, for all $t \in \mathbb{R}$,

$$\zeta(t,\cdot) = \partial_x v(t,\cdot).$$

We also have the following dispersion result (see for instance [146] and [95] or Corollary 7.2.4 in [69]).

Lemma 5.3.2. Let $u_0 \in W^{2,1}(\mathbb{R})$. Then

$$\left| \int_{\mathbb{R}} e^{ix\xi \pm t\sqrt{\xi^2 + 1}} u_0(\xi) d\xi \right|_{L^{\infty}_x} \leq \frac{C}{\sqrt{1 + |t|}} \, |u_0|_{W^{2,1}} \, d\xi \leq \frac{C}{\sqrt{1 + |t|}} \, |u_0|_{W^{2,1}} \, d\xi \leq \frac{C}{\sqrt{1 + |t|}} \, d\xi \leq \frac{C}{$$

We can give the main result of this part.

Theorem 5.3.3. Let $\mu_0 > 0$, $\zeta_0, u_0, v_0, \mathbf{V}_0^{\sharp} \in H^6(\mathbb{R})$, $x\zeta_0, xu_0, xv_0 \in H^4(\mathbb{R})$, such that ζ_0, v_0 satisfy Condition (5.40), $1 + \varepsilon \zeta \ge h_{\min} > 0$ and $0 < \mu < \mu_0$. Then, there exists a time T > 0, such that there exists

(i) a unique classical solution $\left(\zeta_B, u_B, v_B, \mathbf{V}_B^{\sharp}\right)$ of (5.37) with initial data $\left(\zeta_0, u_0, v_0, \mathbf{V}_0^{\sharp}\right)$ on $\left[0, \frac{T}{\sqrt{\mu}}\right]$.

(ii) a unique solution (ζ, u, v) of (5.38) with initial data (ζ_0, u_0, v_0) on $\left[0, \frac{T}{\sqrt{\mu}}\right]$. Moreover, we have the following error estimate for all $0 \le t \le \frac{T}{\sqrt{\mu}}$,

$$|(\zeta_B, u_B, v_B) - (\zeta, u, v)|_{L^{\infty}([0,t] \times \mathbb{R})} \le C\left(\frac{\mu t}{1 + \sqrt{t}} + \mu^2 t^2 + \mu^{\frac{3}{2}} t\right).$$

where $C = C\left(T, \frac{1}{h_{\min}}, \mu_0, |\zeta_0|_{H^6}, |u_0|_{H^6}, |v_0|_{H^6}, |\mathbf{V}_0^{\sharp}|_{H^6}, |x\zeta_0|_{H^4}, |xu_0|_{H^4}, |xv_0|_{H^4}\right)$.

Proof. The first point follows from Proposition 5.2.15. For the error estimate, if we denote by $U = (\zeta_B, u_B, v_B)^t$, U satisfies the linear Boussinesq-Coriolis equation up to a remainder of order μ and a remainder of order $\mu^{\frac{3}{2}}$. Then, using the Duhamel's formula we get

$$U(t) = \mathcal{S}(t)U_0 + \mu \int_0^t \mathcal{S}(t-\tau) \begin{pmatrix} -\partial_x \left(\zeta_B u_B\right)(\tau) \\ -u_B(\tau)\partial_x u_B(\tau) + \frac{1}{3}\partial_x^2 \partial_\tau u_B(\tau) \\ -u_B\partial_x v_B \end{pmatrix} + \mu^{\frac{3}{2}} \int_0^t \mathcal{S}(t-\tau)R$$

where R is a remainder bounded uniformly with respect to μ . Then, using again the Duhamel's formula on the first integral we get

$$\begin{split} U(t) &= \mathcal{S}(t)U_0 - \mu \int_0^t \mathcal{S}(t-\tau) \begin{pmatrix} \partial_x \left((\mathcal{S}_1(\tau)U_0)(\mathcal{S}_2(\tau)U_0) \right) \\ (\mathcal{S}_2(\tau)U_0)\partial_x (\mathcal{S}_2(\tau)U_0) \\ (\mathcal{S}_2(\tau)U_0)\partial_x (\mathcal{S}_3(\tau)U_0) \end{pmatrix} \\ &+ \mu \int_0^t \mathcal{S}(t-\tau) \begin{pmatrix} 0 \\ \frac{1}{3}\partial_x^2 \partial_\tau \mathcal{S}_2(\tau)U_0 \\ 0 \end{pmatrix} + \mu^2 \int_0^t \int_0^\tau \tilde{R} + \mu^{\frac{3}{2}} \int_0^t \mathcal{S}(t-\tau)\tilde{R} \\ &= \mathcal{S}(t)U_0 - \mu I_1(t) + \mu I_2(t) + \mu^2 I_3(t) + \mu^{\frac{3}{2}} I_4(t), \end{split}$$

where $S_i(t)$ is the *i*th row of S(t). We start by estimating I_1 . We have

$$I_1(t) = \int_0^t \mathcal{S}(t-\tau) \begin{pmatrix} \partial_x \left(\zeta(\tau)u(\tau)\right) \\ u(\tau)\partial_x u(\tau) \\ u(\tau)\partial_x v(\tau) \end{pmatrix}.$$

Then, we notice that $\partial_x (\zeta(\tau)u(\tau)) = \partial_x (u(\tau)\partial_x v(\tau))$ since $\zeta(\tau) = \partial_x v(\tau)$ by Lemma 5.3.1. Therefore, using Lemma 5.3.2 and products estimates, we get

$$\begin{aligned} |I_1(t)|_{L^{\infty}} &\leq \int_0^t \frac{1}{\sqrt{1+t-\tau}} \left| \begin{pmatrix} \partial_x \left((\mathcal{S}_1(\tau)U_0)(\mathcal{S}_2(\tau)U_0) \right) \\ (\mathcal{S}_2(\tau)U_0)\partial_x(\mathcal{S}_2(\tau)U_0) \\ (\mathcal{S}_2(\tau)U_0)\partial_x(\mathcal{S}_3(\tau)U_0) \end{pmatrix} \right|_{W^{2,1}} \\ &\leq C \left(|\zeta_0|_{H^3}, |u_0|_{H^3}, |v_0|_{H^3}, \left| \mathbf{V}_0^{\sharp} \right|_{H^3} \right) \frac{t}{\sqrt{1+t}}. \end{aligned}$$

For I_2 , using Lemma 5.3.2 we get

$$|I_2| \le C\left(|\zeta_0|_{H^4}, |u_0|_{H^4}, |v_0|_{H^4}, |x\zeta_0|_{H^4}, |xu_0|_{H^4}, |xv_0|_{H^4}\right) \frac{t}{\sqrt{1+t}}$$

Finally, using Proposition 5.2.15, we have

$$\begin{aligned} |I_3(t)|_{H^1} &\leq C \left(|\zeta_0|_{H^6}, |u_0|_{H^6}, |v_0|_{H^6}, \left| \mathbf{V}_0^{\sharp} \right|_{H^6} \right) t^2 \\ |I_4(t)|_{H^1} &\leq C \left(|\zeta_0|_{H^4}, |u_0|_{H^4}, |v_0|_{H^4}, \left| \mathbf{V}_0^{\sharp} \right|_{H^4} \right) t. \end{aligned}$$

Gathering these four estimates, we get the result.

Hence, using Theorem 5.2.19, we justify that poincaré waves remain close to the solutions of the water waves equations (5.14) over times $\mathcal{O}_{\mu}(1)$ with an accuracy of order $\mathcal{O}(\mu)$. Furthermore, if one can show that a solution of the water waves equations (5.14), with initial data satisfying Condition (5.40), exists over a time $\mathcal{O}\left(\frac{1}{\sqrt{\mu}}\right)$, we show that this solution remains close, with an accuracy of order $\mathcal{O}\left(\mu^{\frac{3}{4}}\right)$, to the solution of the linear Boussinesq-Coriolis equations with the same initial data. The reader interested in more linear properties of the water waves equations in shallow water can refer to Chapter 4 in [93].

5.3.2 The Proudman resonance in presence of a Coriolis forcing

In Section 2.3, we studied the Proudman resonance when the Coriolis forcing is negligible. We recall that it is a linear amplification due to a source term (non constant pressure or moving bottom). In subsection 4.5.4 we showed that the Coriolis forcing prevents the Proudman resonance when d = 2. In this subsection, we study the case d = 1. By proceeding as in Proposition 4.5.8, we can show that the system

$$\begin{cases} \partial_t \zeta + \partial_x u = 0, \\ \partial_t u + \partial_x \zeta - v = -\partial_x P, \\ \partial_t v + u = 0, \end{cases}$$
(5.41)

is an asymptotic model of the water waves equations in the regime

$$\mathcal{A}_{LWW} = \left\{ \left(\varepsilon, \beta, \gamma, \mu, \frac{\varepsilon}{\text{Ro}}\right), \ 0 < \mu, \ \varepsilon, \ \beta \le \delta_0, \ \frac{\varepsilon}{\text{Ro}} = 1, \gamma \le \delta_0^2 \right\},\tag{5.42}$$

with an accuracy $\mathcal{O}(\delta_0 t)$ on times T independent of δ_0 . In Section 2.3, we show that a pressure of the form $P(t, x) = P_0(x - t)$ creates a resonance when the Coriolis forcing is negligible. In the next Proposition, we shows that no physical resonance is possible for pressure of the form $P(t, x) = P_0(x - \alpha t)$.

Proposition 5.3.4. Let $P_0 \in H^3(\mathbb{R}) \cap W^{3,1}(\mathbb{R})$, $\alpha \ge 0$, $P(t,x) = P_0(x - \alpha t)$ and $U = (\zeta, u, v)$ be the solution of (5.41) with initial data equal to 0. Then, if $\alpha \le 1$,

$$|(\zeta, u, v)(t, \cdot)|_{L^{\infty}} \le C |P_0|_{H^3}$$

Furthermore, there exists a constant C > 0, such that for all $\alpha > 1$,

$$|(\zeta, u, v)(t, \cdot)|_{L^{\infty}} \le \frac{C}{\alpha + 1} \left(|P_0|_{H^2} + |P_0|_{W^{3,1}} \right) \left(1 + \ln(t) \right)$$

Proof. We keep the notations of Subsection 5.3.1. We begin with the case $\alpha < 1$. Using the Duhamel's formula, we get

$$\begin{split} U(t,x) = & \int_{\mathbb{R}} \int_{0}^{t} S(t-s,\xi) \begin{pmatrix} 0\\ -i\xi \widehat{P}(s,\xi) \\ 0 \end{pmatrix} e^{ix\xi} ds d\xi \\ = & \int_{\mathbb{R}} \int_{-t}^{0} e^{i(x-\alpha t)\xi} e^{i\tau \left(-\alpha\xi + \sqrt{1+\xi^{2}}\right)} i\xi \widehat{P}_{0}(\xi) m_{1}(\xi) + e^{i(x-\alpha t)\xi} e^{-i\tau \left(\alpha\xi + \sqrt{1+\xi^{2}}\right)} i\xi \widehat{P}_{0}(\xi) m_{2}(\xi) d\tau d\xi \\ = & \int_{\mathbb{R}} e^{i(x-\alpha t)\xi} \xi \widehat{P}_{0}(\xi) \frac{1 - e^{it \left(\alpha\xi - \sqrt{1+\xi^{2}}\right)}}{\sqrt{1+\xi^{2}} - \alpha\xi} + e^{i(x-t)\xi} \xi \widehat{P}_{0}(\xi) \frac{e^{it \left(\alpha\xi + \sqrt{1+\xi^{2}}\right)} - 1}{\sqrt{1+\xi^{2}} + \alpha\xi} m_{2}(\xi) d\xi, \end{split}$$

where m_1 and m_2 are bounded vectors in $L^{\infty}(\mathbb{R})$. Then, the first inequality follows easily from the fact that

$$\left|\frac{1}{\sqrt{1+\xi^2}-\alpha\xi}\right| \le C\left(1+|\xi|\right).$$

For the second inequality, we also have

$$U(t,x) = \int e^{i(x-\alpha t)\xi} \xi \widehat{P}_0(\xi) \frac{1 - e^{it\left(\alpha\xi - \sqrt{1+\xi^2}\right)}}{\sqrt{1+\xi^2} - \alpha\xi} + e^{i(x-t)\xi} \xi \widehat{P}_0(\xi) \frac{e^{it\left(\alpha\xi + \sqrt{1+\xi^2}\right)} - 1}{\sqrt{1+\xi^2} + \alpha\xi} m_2(\xi) d\xi$$

We focus on the first part of the previous equation since the other part is similar. We begin with the case $1 < \alpha \leq 2$. We denote $g(\xi) = \sqrt{1 + \xi^2} - \alpha \xi$ and $\xi_{\alpha} = \frac{1}{\sqrt{\alpha^2 - 1}}$ its unique root. The function g is decreasing and we denote by ϕ its inverse. We can compute ϕ explicitly, for $y \in \mathbb{R}$,

$$\phi(y) = \frac{-\alpha y + \sqrt{\alpha^2 - 1 + y^2}}{\alpha^2 - 1}.$$

Let $\chi : \mathbb{R} \to \mathbb{R}$ be a positive, compactly supported in [-1, 1], smooth, even function equal to one in $\left[-\frac{1}{2}, \frac{1}{2}\right]$. Then, we decompose our integral in two parts

$$\int_{\mathbb{R}} e^{i(x-\alpha t)\xi} \xi \widehat{P}_0(\xi) \frac{1-e^{it\left(\alpha\xi-\sqrt{1+\xi^2}\right)}}{\sqrt{1+\xi^2}-\alpha\xi} d\xi = I+J,$$
$$I = \int_{\mathbb{R}} e^{i(x-\alpha t)\xi} \chi(\xi-\xi_\alpha) \xi \widehat{P}_0(\xi) \frac{1-e^{it\left(\alpha\xi-\sqrt{1+\xi^2}\right)}}{\sqrt{1+\xi^2}-\alpha\xi} d\xi.$$

Using the change of variable $\xi = \phi(z)$, we get

$$|I| = \left| \int_{\xi_{\alpha}-1}^{\xi_{\alpha}+1} e^{i(x-\alpha t)\xi} \chi(\xi-\xi_{\alpha})\xi \widehat{P}_{0}(\xi) \frac{1-e^{it\left(\alpha\xi-\sqrt{1+\xi^{2}}\right)}}{\sqrt{1+\xi^{2}}-\alpha\xi} d\xi \right|$$
$$\leq C \int_{g(\xi_{\alpha}+1)}^{g(\xi_{\alpha}-1)} \left| \chi(\phi(z)-\xi_{\alpha})\phi'(z)\phi(z)\widehat{P}_{0}(\phi(z)) \right| \left| \frac{\sin(\frac{t}{2}z)}{z} \right| dz$$

Then, we remark that there exists a constant $C(\alpha) > 0$, such that for all $y \in \mathbb{R}$, $|\phi'(y)| \le C(\alpha) \left(1 + |\phi(y)|^2\right)$ and $|g(\xi_{\alpha} + 1)| + |g(\xi_{\alpha} - 1)| \le C(\alpha)$. We get

$$\begin{aligned} |I| &\leq C(\alpha) \left| \xi \left(1 + |\xi|^2 \right) \widehat{P}_0 \right|_{L^{\infty}} \int_{-tC(\alpha)}^{tC(\alpha)} \frac{1}{1 + |z|} dz \\ &\leq C(\alpha) \left| P_0 \right|_{W^{3,1}} \left(1 + \ln(t) \right). \end{aligned}$$

Furthermore, we notice that there exists a constant C > 0, $\left| \frac{(1-\chi(\xi-\xi_{\alpha}))}{g(\xi)} \right| \leq C$. Then

$$\begin{aligned} |J| &= \left| \int_{\mathbb{R}} e^{i(x-\alpha t)\xi} (1-\chi(\xi-\xi_{\alpha}))\xi \widehat{P}_{0}(\xi) \frac{1-e^{it\left(\alpha\xi-\sqrt{1+\xi^{2}}\right)}}{\sqrt{1+\xi^{2}}-\alpha\xi} d\xi \right| \\ &\leq C \int_{\mathbb{R}} \left| \xi \widehat{P}_{0}(\xi) \right| d\xi, \\ &\leq C \left| P_{0} \right|_{H^{2}}, \end{aligned}$$

and the result follows. Finally, when $\alpha \geq 2$, we remark that, $\forall y \in \mathbb{R}$, $|\phi'(y)| \leq \frac{1}{\alpha-1}$ and $|\xi_{\alpha} \pm 1| \leq 2$. We get

$$\begin{split} |I| &= \left| \int_{\mathbb{R}} e^{i(x-\alpha t)\xi} \chi(\xi-\xi_{\alpha})\xi \widehat{P}_{0}(\xi) \frac{1-e^{it\left(\alpha\xi-\sqrt{1+\xi^{2}}\right)}}{\sqrt{1+\xi^{2}}-\alpha\xi} d\xi \right| \\ &\leq C \int_{-2}^{2} \left| \chi(\phi(z)-\xi_{\alpha})\phi'(z)\phi(z)\widehat{P}_{0}(\phi(z)) \right| \left| \frac{\sin(\frac{t}{2}z)}{z} \right| dz \\ &\leq C \frac{1}{\alpha-1} \left| \xi \widehat{P}_{0} \right|_{L^{\infty}} \int_{-2t}^{2t} \frac{1}{1+|z|} dz \\ &\leq \frac{C}{\alpha-1} \left| P_{0}' \right|_{L^{1}} (1+\ln(t)) \,. \end{split}$$

The control of J is similar to the previous case and the result follows easily.

Hence, we see that we can not expect a resonance from a physical standpoint when we consider traveling pressure since the possible amplification is too slow. However, a resonance is possible for particular profiles. Inspired by Subsection 2.3.3, we consider a pressure of the form

$$P(t,\cdot) = e^{-it\sqrt{1+D^2}}P_0$$

The next proposition shows that with such a profile we get a resonance with a factor of amplification of \sqrt{t} .

Proposition 5.3.5. Let $P_0 \in H^1(\mathbb{R}) \cap W^{3,1}(\mathbb{R})$ such that $xP_0 \in H^1(\mathbb{R})$, $P(t, \cdot) = e^{-it\sqrt{1+D^2}}P_0$ and $U = (\zeta, u, v)$ be the solution of (5.41) with initial data equal to 0. Then,

$$|U(t,\cdot)|_{L^{\infty}} \le \frac{t}{\sqrt{1+t}} \left(|P|_{H^1} + |xP_0|_{H^1} \right).$$

Furthermore,

$$\lim_{t \to \infty} \frac{1}{\sqrt{t}} |U(t, \cdot)|_{L^{\infty}} \ge C(P_0) > 0.$$

Proof. We keep the notations of Subsection 5.3.1. Using the Duhamel's formula, we get

$$\begin{split} U(t,x) = & \int_{\mathbb{R}} \int_{0}^{t} S(t-s,\xi) \begin{pmatrix} 0\\ -i\xi \widehat{P}(s,\xi)\\ 0 \end{pmatrix} e^{ix\xi} ds d\xi \\ &= \int_{\mathbb{R}} \int_{0}^{t} e^{ix\xi} i\xi \widehat{P}_{0}(\xi) \left(e^{-is\sqrt{1+\xi^{2}}} e^{i(t-s)\sqrt{1+\xi^{2}}} m_{1}(\xi) + e^{i(s-t)\sqrt{1+\xi^{2}}} e^{-is\sqrt{1+\xi^{2}}} m_{2}(\xi) \right) ds d\xi \\ &= I_{1}(t,x) + I_{2}(t,x). \end{split}$$

The first inequality follows from Lemma 5.3.2. For the second inequality, we first notice that

$$|I_1(t,x)| = \left| \int_{\mathbb{R}} \int_0^t e^{ix\xi} i\xi \widehat{P}_0(\xi) e^{i(t-2s)\sqrt{1+\xi^2}} m_1(\xi) ds d\xi \right|$$
$$\leq C \int_{\mathbb{R}} \left| \xi \widehat{P}_0(\xi) \right| d\xi.$$

Furthermore, we have

$$I_{2}(t,x) = t \int_{\mathbb{R}} \int_{0}^{t} e^{i\left(x\xi - t\sqrt{1+\xi^{2}}\right)} i\xi \widehat{P}_{0}(\xi) m_{2}(\xi) d\xi.$$

Let $\xi_0 > 0$ such that $\xi_0 \hat{P}_0(\xi_0) m_2(\xi_0) \neq 0$ and $x = \frac{t\xi_0}{\sqrt{1+\xi_0^2}}$. Then, using a stationary phase argument, we easily get

$$\left| I_2\left(t, \frac{t\xi_0}{\sqrt{1+\xi_0^2}}\right) \right| = \left| \xi_0 \hat{P}_0(\xi_0) m_2(\xi_0) \right| \sqrt{2\pi t (1+\xi_0)^{\frac{3}{2}}} + o\left(\sqrt{t}\right),$$

follows.

and the result follows.

5.3.3 Weak rotation, the Ostrovsky equation

Without Coriolis forcing and vorticity, it is well-known, that the KdV equation is a good approximation of the water waves equation under the assumption that ε and μ have the same order ([43], [75], [129], [19], Part 7.1 in [80]). When the Coriolis forcing is taken into account, Ostrovsky ([113]) derived an equation for long waves, which is an adaptation of the KdV equation

$$\partial_{\xi} \left(\partial_{\tau} f + \frac{3}{2} f \partial_{\xi} f + \frac{1}{6} \partial_{\xi}^3 f \right) = \frac{1}{2} f.$$
(5.43)

This equation is called the Ostrovsky equation or rKdV-equation in the physical literature. Initially developed for internal water waves, several authors also studied it for surface water waves ([114], [60], [85], [64]). The purpose of this part is to fully justify it. Inspired by [60] we consider the asymptotic regime

$$\mathcal{A}_{\text{Ost}} = \left\{ \left(\varepsilon, \beta, \gamma, \mu, \text{Ro}\right), 0 \le \mu \le \mu_0, \varepsilon = \mu, \beta = \gamma = 0, \frac{\varepsilon}{\text{Ro}} = \sqrt{\mu} \right\}.$$
 (5.44)

Then, the Boussinesq-Coriolis equations become (see Remark 5.2.14)

$$\begin{cases} \partial_t \zeta + \partial_x \left([1 + \mu \zeta] u \right) = 0, \\ \left(1 - \frac{\mu}{3} \partial_x^2 \right) \partial_t u + \partial_x \zeta + \mu u \partial_x u - \sqrt{\mu} v = 0, \\ \partial_t v + \mu u \partial_x v + \sqrt{\mu} u = 0. \end{cases}$$
(5.45)

In order to motivate our approach, let us recall that we are interested in the one-dimensional propagation of water waves in the long wave regime. If we drop all the terms of order $\mathcal{O}(\sqrt{\mu})$ in the Boussinesq-Coriolis, we see that

$$\begin{cases} \partial_t \zeta + \partial_x u = 0, \\ \partial_t u + \partial_x \zeta = 0, \\ \partial_t v = 0. \end{cases}$$

Hence, if we assume that v is initially zero, we get a wave equation and the propagation of traveling water waves with speed ± 1 . Then it is natural to study how these traveling water waves are perturbed when we add weakly nonlinear effects, i.e when we consider the System (5.45). In this paper, we consider only water waves with speed 1. We consider a WKB expansion for (ζ, u, v) . We seek an approximate solution $(\zeta_{app}, u_{app}, v_{app})$ of (5.45) under the form

$$\begin{aligned} \zeta_{app}(t,x) &= f(x-t,\mu t) + \mu \zeta_{(1)}(t,x,\mu t), \\ u_{app}(t,x) &= f(x-t,\mu t) + \mu u_{(1)}(t,x,\mu t), \\ v_{app}(t,x) &= \sqrt{\mu} v_{(1/2)}(t,x,\mu t). \end{aligned}$$
(5.46)

where $f = f(\xi, \tau)$ is our modulated traveling water waves, and the others terms are correctors. Then, we plug the ansatz in System (5.45) and we get

$$\partial_{t}\zeta_{app} + \partial_{x}\left(\left[1 + \mu\zeta_{app}\right]u_{app}\right) = \mu R_{(1)}^{1} + \mu^{2}R_{1}, \\ \left(1 - \frac{\mu}{3}\partial_{x}^{2}\right)\partial_{t}u_{app} + \partial_{x}\zeta_{app} + \mu u_{app}\partial_{x}u_{app} - \sqrt{\mu}v_{app} = \mu R_{(1)}^{2} + \mu^{2}R_{2}, \quad (5.47) \\ \partial_{t}v_{app} + \mu u_{app}\partial_{x}v_{app} + \sqrt{\mu}u_{app} = \sqrt{\mu}R_{(1/2)}^{3} + \mu^{\frac{3}{2}}R_{3},$$

where

$$\begin{aligned} R^{1}_{(1)} &= \partial_{t}\zeta_{(1)} + \partial_{x}u_{(1)} + \partial_{\tau}f + 2f\partial_{\xi}f, \\ R^{2}_{(1)} &= \partial_{t}u_{(1)} + \partial_{x}\zeta_{(1)} + \partial_{\tau}f + \frac{1}{3}\partial_{\xi}^{3}f + f\partial_{\xi}f - v_{(1/2)}, \\ R^{3}_{(1/2)} &= \partial_{t}v_{(1/2)} + f, \end{aligned}$$

and

$$R_{1} = \partial_{\tau}\zeta_{(1)} + \partial_{x}\left(fu_{(1)} + f\zeta_{(1)} + \mu\zeta_{(1)}u_{(1)}\right),$$

$$R_{2} = \partial_{\tau}u_{(1)} - \frac{1}{3}\partial_{\xi}^{3}\partial_{\tau}f - \frac{1}{3}\partial_{x}^{3}\partial_{t}u_{(1)} - \mu\frac{1}{3}\partial_{x}^{3}\partial_{\tau}u_{(1)} + \partial_{x}\left(fu_{(1)}\right) + \mu u_{(1)}\partial_{x}u_{(1)},$$

$$R_{3} = \partial_{\tau}v_{(1/2)} + \left(f + \sqrt{\mu}u_{(1)}\right)\partial_{x}v_{(1/2)} + u_{(1)}.$$
(5.48)

Then, the idea is to choose the correctors with $R^1_{(1)}(t, x, \tau) = R^2_{(1)}(t, x, \tau) = 0$ and $R^3_{(1/2)}(t, x, \tau) = 0$ for all $x \in \mathbb{R}, t \in \left[0, \frac{T}{\mu}\right]$ and $\tau \in [0, T]$.

Remark 5.3.6. In fact, we should add $\sqrt{\mu}\zeta_{(1/2)}(t, x, \mu t)$, $\sqrt{\mu}u_{(1/2)}(t, x, \mu t)$, $v_{(0)}(t, x, \mu t)$, and $\mu v_{(1)}(t, x, \mu t)$ to the ansatz (5.46) for ζ_{app} , u_{app} , v_{app} and v_{app} respectively. However, if we plug them in System (5.45) and we want to cancel all the terms of order $\sqrt{\mu}$ and μ , we get

$$\begin{split} \partial_t \zeta_{(1/2)} &+ \partial_x u_{(1/2)} = 0, \\ \partial_t u_{(1/2)} &+ \partial_x \zeta_{(1/2)} + v_{(0)} = 0, \\ \partial_t v_{(0)} &= 0, \\ \partial_t v_{(1)} &+ \partial_\tau v_{(0)} + f \partial_x v_{(0)} + u_{(1/2)} = 0 \end{split}$$

which leads to $\zeta_{(1/2)} = u_{(1/2)} = v_{(0)} = v_{(1)} = 0$ if these quantities are initially zero.

Then, if we assume that $v_{(1/2)}$ and f vanish at $x = \infty$, the condition $R^3_{(1/2)} = 0$ is equivalent to the equation

$$\partial_t \partial_x v_{(1/2)}(t, x, \tau) + \partial_\xi f(x - t, \tau) = 0.$$

Since, $\partial_t (f(x-t,\tau)) = -\partial_\xi f(x-t,\tau)$, we can take

$$\partial_x v_{(1/2)}(t, x, \tau) = \partial_x v_{(1/2)}^0(x) - f^0(x) + f(x - t, \tau),$$
(5.49)

where $v_{(1/2)}^0$ and f^0 are the initial data of $v_{(1/2)}$ and f respectively. Then, we have to introduce the following spaces.

Definition 5.3.7. For $s \in \mathbb{R}$, we define the Hilbert spaces $\partial_x H^s(\mathbb{R})$ as

$$\partial_x H^s(\mathbb{R}) = \left\{ f \in H^{s-1}(\mathbb{R}), \ f = \partial_x \tilde{f} \ with \ \tilde{f} \in H^s(\mathbb{R}) \right\},$$

and \tilde{f} is denoted $\partial_x^{-1} f$ in the following. In the same way, we define $\partial_x^2 H^s(\mathbb{R})$.

Then, if we assume that $f(\cdot, \tau) \in \partial_x H^s(\mathbb{R})$ for all $\tau \in [0, T]$, we have

$$v_{(1/2)}(t,x,\tau) = v_{(1/2)}^0(x) - \partial_x^{-1} f^0(x) + \partial_x^{-1} f(x-t,\tau),$$

Furthermore, from $R_{(1)}^1 = R_{(1)}^2 = 0$, if we denote $w_{\pm} = \zeta_{(1)} \pm u_{(1)}$ we get

$$(\partial_t + \partial_x) w_+ + \left(2\partial_\tau f + 3f\partial_\xi f + \frac{1}{3}\partial_\xi^3 f - \partial_\xi^{-1} f \right) (x - t, \tau) - \left(v_{(1/2)}^0 - \partial_\xi^{-1} f^0 \right) (x) = 0,$$

$$(\partial_t - \partial_x) w_- + \left(\frac{1}{2}\partial_\xi f^2 - \frac{1}{3}\partial_\xi^3 f + \partial_\xi^{-1} f \right) (x - t, \tau) + \left(v_{(1/2)}^0 - \partial_\xi^{-1} f^0 \right) (x) = 0.$$

$$(5.50)$$

The following lemma (Lemma 7.6 in [80]) is the key point to control u and v.

Lemma 5.3.8. Let $c_1 \neq c_2$. Let $h_1, h_2, h_3 \in L^2(\mathbb{R})$ with $h_2 = H'_2$ and $H_2 \in L^2(\mathbb{R})$. We consider the unique solution g of

$$\begin{cases} (\partial_t + c_1 \partial_x)g = h_1(x - c_1 t) + h_2(x - c_2 t) + h_3(x - c_2 t), \\ g_{|t=0} = 0. \end{cases}$$

Then, $\lim_{t\to\infty} \left|\frac{1}{t}g(t,\cdot)\right|_2 = 0$ if and only if $h_1 \equiv 0$ and in that case

$$|g(t,\cdot)|_2 \le \frac{C}{|c_1 - c_2|} \left(|H_2|_2 \frac{t}{1+t} + |h_3|_{H^2} \frac{t}{1+\sqrt{t}} \right).$$

Then, in order to avoid a linear growth for the solution of (5.50), we also have to impose that

$$\partial_{\tau}f + \frac{3}{2}f\partial_{\xi}f + \frac{1}{6}\partial_{\xi}^{3}f = \frac{1}{2}\partial_{\xi}^{-1}f, \qquad (5.51)$$

which is the Ostrovsky equation. Before giving a full justification of the Ostrovsky equation, we need a local wellposedness result of this equation. The following proposition is a generalization of Theorem 2.1 in [89] and Theorem 2.6 in [143] (see also [87] for weak solutions).

Proposition 5.3.9. Let $s \ge 3$ and $f_0 \in \partial_x H^s(\mathbb{R})$. Then, there exists a time T > 0 and a unique solution $f \in \mathcal{C}([0,T]; \partial_x H^s(\mathbb{R})))$ to the Ostrovsky equation (5.51) and one has

$$\left| \partial_{\xi}^{-1} f(t, \cdot) \right|_{H^s} \le C \left(T, \left| \partial_{\xi}^{-1} f_0 \right|_{H^s} \right).$$

Moreover, if $f_0 \in \partial_x^2 H^{s+1}(\mathbb{R}), f \in \mathcal{C}([0,T]; \partial_x^2 H^{s+1}(\mathbb{R})))$ and one has

$$\left|\partial_{\xi}^{-2}f(t,\cdot)\right|_{H^{s+1}} \leq C\left(T, \left|\partial_{\xi}^{-2}f_{0}\right|_{H^{s+1}}\right)$$

Proof. We only prove the second point of the Proposition. We denote by S(t) the semi-group of the linearized Ostrovsky equation

$$\partial_{\tau}f + \frac{1}{6}\partial_{\xi}^{3}f - \frac{1}{2}\partial_{\xi}^{-1}f = 0,$$

and it is easy to check that this semi-group acts unitary on $H^s(\mathbb{R})$. We denote $\tilde{f} = \partial_{\tau} f$. Then, \tilde{f} satisfies the equation

$$\partial_{\tau}\tilde{f} + \frac{3}{2}\partial_{\xi}\left(\tilde{f}f\right) + \frac{1}{6}\partial_{\xi}^{3}\tilde{f} - \frac{1}{2}\partial_{\xi}^{-1}\tilde{f} = 0.$$

Using the Duhamel's formula we obtain

wh

$$\partial_{\xi}^{-1}\tilde{f}(t,\cdot) = S(t)\partial_{\xi}^{-1}\tilde{f}_0 + \frac{3}{2}\int_0^t S(t-s)\left(f\tilde{f}\right)(s,\cdot)ds.$$

Since $\partial_{\xi}^{-1}\tilde{f}_0 = -\frac{3}{4}f_0^2 - \partial_{\xi}^2 f_0 + \partial_{\xi}^{-2}f_0 \in L^2(\mathbb{R})$, we get the result since we have

$$\frac{1}{2}\partial_{\xi}^{-2}f = \partial_{\xi}^{-1}\partial_{\tau}f + \frac{3}{4}f^2 + \frac{1}{6}\partial_{\xi}^2f.$$

Notice that contrary to the KdV equation, we do not have a global existence. We can now give the main result of this part.

Theorem 5.3.10. Let $f^0 \in \partial_x^2 H^{10}(\mathbb{R})$, such that $1 + \varepsilon f \ge h_{\min} > 0$, $v^0 \in \partial_x H^6(\mathbb{R})$ and $\mu_0 > 0$. Then, there exists a time T > 0, such that for all $0 < \mu < \mu_0$, we have

(i) a unique classical solution (ζ_B, u_B, v_B) of (5.45) with initial data $(f^0, f^0, \sqrt{\mu}v^0)$ on $\left[0, \frac{T}{\mu}\right]$.

(ii) a unique classical solution f of (5.51) with initial data f^0 on [0, T].

(iii) If we define $(\zeta_{Ost}, u_{Ostr})(t, x) = (f(x - t, \mu t), f(x - t, \mu t))$ we have the following error estimate for all $0 \le t \le \frac{T}{\mu}$,

$$|(\zeta_B, u_B) - (\zeta_{Ost}, u_{Ost})|_{L^{\infty}([0,t] \times \mathbb{R})} \le C\left((1 + \sqrt{\mu}t)\frac{\mu t}{1+t} + \mu^{\frac{3}{2}}t\right)$$

ere $C = C\left(T, \frac{1}{h_{\min}}, \mu_0, \left|\partial_x^{-2}f^0\right|_{H^{10}}, \left|\partial_x^{-1}v^0\right|_{H^6}\right).$

Proof. In all the proof, C will be a constant as in the theorem. The first and second point follow from Corollary 5.2.17 and 5.3.9. In order to get the error estimate, we have to control the remainders R_1, R_2, R_3 , defined in (5.48). First, using Lemma 5.3.8, the fact that we can express the quantities $\frac{1}{2}\partial_{\xi}f^2 - \frac{1}{3}\partial_{\xi}^3f$, $\partial_{\xi}^{-1}f$ and v_0 as derivatives with respect to x and the fact that f satisfies the Ostrovsky equation (5.51), we have

$$|\zeta_{(1)}|_2 + |u_{(1)}|_2 \le C \frac{t}{1+t}.$$

But we can also control all the derivatives with respect to τ or x of u and v be differentiating (5.50). Hence, we get a control for the remainders R_1, R_2 . For R_3 , we use the fact that $v = \partial_x^{-1} f$. We finally, obtain

$$|R_1|_{H^2} + |R_2|_{H^2} + |R_3|_{H^2} \le C\left(\frac{t}{1+t} + \mu t + 1\right),$$

Then, thanks to Proposition 5.2.18 and remark 5.2.14, we get

$$|(\zeta_B, u_B, v_B) - (\zeta_{app}, u_{app}, v_{app})|_{L^{\infty}([0,t] \times \mathbb{R})} \le C\mu^{\frac{3}{2}} t\left(\frac{t}{1+t} + \mu t + 1\right).$$

Moreover, we have

$$|(\zeta_{app}, u_{app}) - (\zeta_{Ost}, v_{Ost})|_{L^{\infty}([0,t] \times \mathbb{R})} \le \mu \frac{t}{1+t}.$$

Then, the result follows easily.

This theorem, combined with Theorem 5.2.19, shows that the solutions of the water waves equations (5.14) is well approximated over times $\mathcal{O}\left(\frac{1}{\sqrt{\mu}}\right)$ with an accuracy of order $\mathcal{O}(\mu)$ by the Ostrovsky approximation if we have a small Coriolis forcing. Notice that contrary to the KdV equation, the Ostrovsky equation does not admit solitons ([154], [58]). Notice also that this approach is similar to the one of the KP equations (see for instance [83], [9] or Part 7.2.1 in [80]). The fact that $f^0 \in \partial_x H^8$ is essential and physical since a solution of the Ostrovsky equation has to be mean free. However, we suppose here that $f^0 \in \partial_x^2 H^9(\mathbb{R})$ and $v^0 \in \partial_x H^5(\mathbb{R})$ which is more restrictive. In fact, using the strategy developed in [15] for the KP approximation we can hope to release this assumption.

5.3.4 Very weak rotation, the KdV equation

As we said before, without Coriolis forcing, it is well-known, that the KdV equation is a good approximation of the water waves equations. In this part we show that if $\frac{\varepsilon}{\text{Ro}}$ is small enough, we get the KdV equation as an asymptotic model. We recall that the KdV equation

$$\partial_{\tau}f + \frac{3}{2}f\partial_{\xi}f + \frac{1}{6}\partial_{\xi}^{3}f = 0.$$
 (5.52)

Inspired by [60], we show that $\frac{\varepsilon}{\text{Ro}} = \mathcal{O}(\mu)$ is sufficient. we consider the asymptotic regime

$$\mathcal{A}_{KdV} = \left\{ \left(\varepsilon, \beta, \gamma, \mu, \operatorname{Ro}\right), 0 \le \mu \le \mu_0, \varepsilon = \mu, \beta = \gamma = 0, \frac{\varepsilon}{\operatorname{Ro}} = \mu \right\}.$$
(5.53)

Then, the Boussinesq-Coriolis equations become (see Remark 5.2.14)

$$\begin{cases} \partial_t \zeta + \partial_x \left([1 + \mu \zeta] u \right) = 0, \\ \left(1 - \frac{\mu}{3} \partial_x^2 \right) \partial_t u + \partial_x \zeta + \mu u \partial_x u - \mu v = 0, \\ \partial_t v + \mu u \partial_x v + \mu u = 0. \end{cases}$$
(5.54)

Proceeding as in the previous part, we seek an approximate solution $(\zeta_{app}, u_{app}, v_{app})$ of (5.54) under the form

$$\begin{aligned} \zeta_{app}(t,x) &= f(x-t,\mu t) + \mu \zeta_{(1)}(t,x,\mu t), \\ u_{app}(t,x) &= f(x-t,\mu t) + \mu u_{(1)}(t,x,\mu t), \\ v_{app}(t,x) &= \mu v_{(1)}(t,x,\mu t). \end{aligned}$$
(5.55)

Then, we plug the ansatz in Sytem (5.54) and we get

$$\partial_t \zeta_{app} + \partial_x \left([1 + \mu \zeta_{app}] u_{app} \right) = \mu R_{(1)}^1 + \mu^2 R_1, \left(1 - \frac{\mu}{3} \partial_x^2 \right) \partial_t u_{app} + \partial_x \zeta_{app} + \mu u_{app} \partial_x u_{app} - \mu v_{app} = \mu R_{(1)}^2 + \mu^2 R_2,$$
(5.56)
$$\partial_t v_{app} + \mu u_{app} \partial_x v_{app} + \mu u_{app} = \mu R_{(1)}^3 + \mu^2 R_3,$$

where

$$\begin{aligned} R^{1}_{(1)} &= \partial_t \zeta_{(1)} + \partial_x u_{(1)} + \partial_\tau f + 2f \partial_\xi f, \\ R^{2}_{(1)} &= \partial_t u_{(1)} + \partial_x \zeta_{(1)} + \partial_\tau f + \frac{1}{3} \partial_\xi^3 f + f \partial_\xi f, \\ R^{3}_{(1)} &= \partial_t v_{(1)} + f, \end{aligned}$$

and

$$\begin{split} R_{1} &= \partial_{\tau} \zeta_{(1)} + \partial_{x} \left(f u_{(1)} + f \zeta_{(1)} + \mu \zeta_{(1)} u_{(1)} \right), \\ R_{2} &= \partial_{\tau} u_{(1)} - \frac{1}{3} \partial_{\xi}^{3} \partial_{\tau} f - \frac{1}{3} \partial_{x}^{3} \partial_{t} u_{(1)} - \mu \frac{1}{3} \partial_{x}^{3} \partial_{\tau} u_{(1)} + \partial_{x} \left(f u_{(1)} \right) + \mu u_{(1)} \partial_{x} u_{(1)} - v_{(1)}, \\ R_{3} &= \partial_{\tau} v_{(1)} + \mu \left(f + \mu u_{(1)} \right) \partial_{x} v_{(1)} + u_{(1)}. \end{split}$$

Remark 5.3.11. We should add $v_{(0)}(t, x, \mu t)$ to the ansatz (5.55) for v_{app} . However, if we plug it in System (5.54) we get $\partial_t v_{(0)} = 0$ which leads to $v_{(0)} = 0$ if the quantity is initially zero. Hence, we make this assumption in the following.

As before, we assume that $R_{(1)}^1(t, x, \tau) = R_{(1)}^2(t, x, \tau) = R_{(1)}^3(t, x, \tau) = 0$ for all $x \in \mathbb{R}$, $t \in \left[0, \frac{T}{\mu}\right]$ and $\tau \in [0, T]$ which leads to $v_{(1)} = v_{(1)}^0 - \partial_x^{-1} f^0 + \partial_x^{-1} f$ and, if we denote $w_{\pm} = \zeta_{(1)} \pm u_{(1)}$ we get

$$(\partial_t + \partial_x) w_+ + \left(2\partial_\tau f + 3f\partial_\xi f + \frac{1}{3}\partial_\xi^3 f\right) (x - t, \tau) = 0,$$

$$(\partial_t - \partial_x) w_- + \left(f\partial_\xi f - \frac{1}{3}\partial_\xi^3 f\right) (x - t, \tau) = 0$$

and to avoid a linear growth of u or v we need that f satisfies (5.52). We also have a existence result for the KdV equation (see for instance [76]).

Proposition 5.3.12. Let $s \ge 1$, $f_0 \in H^s(\mathbb{R})$ and T > 0. Then, there exists a unique solution to the KdV equation (5.52) $f \in \mathcal{C}([0,T]; H^s(\mathbb{R})))$ and one have

$$|f|_{H^s} \le C(T, |f^0|_{H^s}).$$

Moreover, if $s \geq 2$ and $f_0 \in \partial_x H^{s+1}(\mathbb{R}), f \in \mathcal{C}([0,T];\partial_x H^{s+1}(\mathbb{R})))$ and we have

$$\left|\partial_{x}^{-1}f\right|_{H^{s+1}} \leq C\left(T, \left|\partial_{x}^{-1}f^{0}\right|_{H^{s+1}}\right)$$

Then, proceeding as in the previous part, we obtain the following theorem.

Theorem 5.3.13. Let $f^0 \in \partial_x H^9(\mathbb{R})$, such that such that $1 + \varepsilon f^0 \ge h_{\min} > 0$, $v^0 \in H^5(\mathbb{R})$ and $\mu_0 > 0$. Then, there exists a time T > 0, such that for all $0 < \mu < \mu_0$, we have (i) a unique classical solution (ζ_B, u_B, v_B) of (5.54) with initial data $(f^0, f^0, \mu v^0)$ on $[0, \frac{T}{\mu}]$.

(ii) a unique classical solution f of (5.52) with initial data f^0 on [0, T].

(iii) If we define $(\zeta_{KdV}, u_{KdV})(t, x) = (f(x - t, \mu t), f(x - t, \mu t))$ we have the following error estimate for all $0 \le t \le \frac{T}{\mu}$,

$$|(\zeta_B, u_B) - (\zeta_{KdV}, u_{KdV})|_{L^{\infty}([0,t] \times \mathbb{R})} \le C\left(\frac{\mu t}{1+t} + \mu^2 t\right)$$

where $C = C\left(T, \frac{1}{h_{\min}}, \mu_0, \left|\partial_x^{-1} f^0\right|_{H^9}, \left|v^0\right|_{H^5}\right).$

This theorem, combined with Theorem 5.2.19, shows that the solutions of the water waves equations (5.14) is well approximated over times $\mathcal{O}\left(\frac{1}{\mu}\right)$ with an accuracy of order $\mathcal{O}(\mu)$ by the KdV approximation if we have a very small Coriolis forcing. Notice that contrary to the irrotational case, the transverse velocity v is not zero (noticed also in [60]). Furthermore, in our situation, the initial data for the KdV equation has to be of zero mean which means that we can not expect the propagation of solitons on a large time (they have a constant sign) if $\frac{\varepsilon}{R_0}$ and μ have the same order.

5.4 Green-Naghdi equations for $\gamma = 0$ and $\beta = \mathcal{O}(\mu)$

This part is devoted to the derivation and justification of the Green-Naghdi equations (5.64) under a Coriolis forcing, with $\gamma = 0$ and for small amplitude topography variations ($\beta = \mathcal{O}(\mu)$). The Green-Naghdi equations are originally obtained in the irrotational framework under the assumption that μ is small (no assumption on ε) and by neglecting all the terms of order $\mathcal{O}(\mu^2)$ in the water waves equations (see for instance [130] or Part 5.1.1.2 in [80]). It is a system of two equations on the surface ζ and the averaged horizontal velocity $\overline{\mathbf{V}}$. These equations were generalized in [33] in presence of vorticity but without a Coriolis forcing. This new system is a cascade of equations that involves a second order tensor and a third order tensor. After deriving these equations, we show that they are an order $\mathcal{O}(\mu^2)$ approximation of the water waves equations. We consider the asymptotic regime for the 1D Green-Naghdi equations

$$\mathcal{A}_{\rm GN} = \left\{ \left(\varepsilon, \beta, \gamma, \mu, \operatorname{Ro}\right), 0 \le \mu \le \mu_0, 0 \le \varepsilon, \frac{\varepsilon}{\operatorname{Ro}} \le 1, \beta = \mathcal{O}\left(\mu\right), \gamma = 0 \right\}.$$
(5.57)

The next subsection is devoted to extending Proposition 5.2.8 and 5.2.9.

5.4.1 Improvements for the equations of \mathbf{Q}_x and \mathbf{Q}_y

We start by extending Proposition 5.2.8.

Proposition 5.4.1. If $(\zeta, U^{\mu,0}_{/\!\!/}, \omega)$ satisfy the Castro-Lannes system (5.14), then Q_x satisfies the following equation

$$\begin{split} \partial_t Q_x + \varepsilon \overline{u} \partial_x Q_x + \varepsilon Q_x \partial_x \overline{u} + \frac{\varepsilon}{\text{Ro}\sqrt{\mu}} \left(\underline{v} - \overline{v}\right) &= -\varepsilon \sqrt{\mu} \frac{1}{h} \partial_x \int_{-1+\beta b}^{\varepsilon \zeta} \left(u_{\text{sh}}^*\right)^2 \\ &+ \varepsilon \sqrt{\mu} Q_x \partial_x Q_x + \varepsilon \mu \frac{1}{3} \partial_x \left(h^2 Q_x \partial_x^2 \overline{u}\right) \\ &+ \varepsilon \mu \frac{1}{6} h^2 u^{\sharp} \partial_x^3 \overline{u} + \varepsilon \mu \frac{1}{8h} \partial_x \left(h^3 u^{\sharp}\right) \partial_x^2 \overline{u} \\ &+ \varepsilon \max \left(\beta \sqrt{\mu}, \mu^{\frac{3}{2}}\right) R, \end{split}$$

and u_{sh}^* satisfies the equation

$$\begin{aligned} \partial_t u_{\rm sh}^* + \varepsilon \overline{u} \partial_x u_{\rm sh}^* + \varepsilon u_{\rm sh}^* \partial_x \overline{u} + \frac{\varepsilon}{\text{Ro}\sqrt{\mu}} \left(\overline{v} - v\right) = & \varepsilon \sqrt{\mu} \frac{1}{h} \partial_x \int_{-1+\beta b}^{\varepsilon \zeta} \left(u_{\rm sh}^*\right)^2 - \varepsilon \sqrt{\mu} u_{\rm sh}^* \partial_x u_{\rm sh}^* \\ & + \varepsilon \partial_x \left(\int_{-1+\beta b}^z \left[\overline{u} + \sqrt{\mu} u_{\rm sh}^*\right] \right) \partial_z u_{\rm sh}^* \\ & + \varepsilon \mu R. \end{aligned}$$

Proof. Using the second equation of the vorticity equation of the Castro-Lannes system (5.14), we have

$$\partial_t \boldsymbol{\omega}_y + \varepsilon u \partial_x \boldsymbol{\omega}_y + \frac{\varepsilon}{\mu} \mathbf{w} \partial_z \boldsymbol{\omega}_y = \varepsilon \boldsymbol{\omega}_x \partial_x v + \frac{\varepsilon}{\sqrt{\mu}} \boldsymbol{\omega}_z \partial_z v + \frac{\varepsilon}{\mathrm{Ro}\sqrt{\mu}} \partial_z v$$

Since $\boldsymbol{\omega}_x = -\frac{1}{\sqrt{\mu}}\partial_z v$ and $\boldsymbol{\omega}_z = \partial_x v$ we notice that $\varepsilon \boldsymbol{\omega}_x \partial_x v + \frac{\varepsilon}{\sqrt{\mu}} \boldsymbol{\omega}_z \partial_z v = 0$. Using Proposition 5.2.5 we get

$$\partial_t \boldsymbol{\omega}_y + \varepsilon \overline{u} \partial_x \boldsymbol{\omega}_y - \varepsilon \partial_x \left[(1 + z - \beta b) \overline{u} \right] \partial_z \boldsymbol{\omega}_y - \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} \partial_z v + \varepsilon \sqrt{\mu} A_1 + \varepsilon \mu A_2 = \varepsilon \max\left(\mu^{\frac{3}{2}}, \beta \sqrt{\mu} \right) R,$$

where

$$A_{1} = u_{\rm sh}^{*} \partial_{x} \omega_{y} - \partial_{x} \left(\int_{-1+\beta b}^{z} u_{\rm sh}^{*} \right) \partial_{z} \omega_{y},$$

$$A_{2} = -\frac{1}{2} \left([1+z-\beta b]^{2} - \frac{h^{2}}{3} \right) \partial_{x}^{2} \overline{u} \partial_{x} \omega_{y} + \frac{1}{2} \partial_{x} \left(\int_{-1+\beta b}^{z} \left([1+z-\beta b]^{2} - \frac{h^{2}}{3} \right) \partial_{x}^{2} \overline{u} \right) \partial_{z} \omega_{y}.$$

Then, integrating with respect to z, using the fact that $\partial_t \zeta + \partial_x (h\overline{u}) = 0$ and $u_{\rm sh} = -\int_z^{\varepsilon \zeta} \omega_y$, we get

$$\partial_t u_{\rm sh} + \varepsilon \overline{u} \partial_x u_{\rm sh} + \varepsilon u_{\rm sh} \partial_x \overline{u} + \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} \left(\underline{v} - v\right) = \varepsilon \partial_x \left[\left(1 + z - \beta b\right) \overline{u} \right] \partial_z u_{\rm sh} + \varepsilon \sqrt{\mu} \int_z^{\varepsilon \zeta} A_1 + \varepsilon \mu \int_z^{\varepsilon \zeta} A_2 + \varepsilon \max\left(\mu^{\frac{3}{2}}, \beta \sqrt{\mu}\right) R.$$

Integrating again with respect to z, using the fact that $\partial_t \zeta + \partial_x (h\overline{u}) = 0$ and $Q_x = \overline{u_{\rm sh}^*}$, we obtain

$$\begin{split} \partial_t Q_x + \varepsilon \overline{u} \partial_x Q_x + \varepsilon Q_x \partial_x \overline{u} + \frac{\varepsilon}{\text{Ro}\sqrt{\mu}} \left(\underline{v} - \overline{v}\right) = \varepsilon \sqrt{\mu} \frac{1}{h} \int_{-1+\beta b}^{\varepsilon \zeta} \int_z^{\varepsilon \zeta} A_1 \\ + \varepsilon \mu \frac{1}{h} \int_{-1+\beta b}^{\varepsilon \zeta} \int_z^{\varepsilon \zeta} A_2 + \varepsilon \max\left(\mu^{\frac{3}{2}}, \beta \sqrt{\mu}\right) R. \end{split}$$

The end of the proof is devoted to the computation of the others terms. We have

$$\int_{z}^{\varepsilon\zeta} A_{1} = \int_{z}^{\varepsilon\zeta} u_{\rm sh}^{*} \partial_{x} \omega_{y} - \partial_{x} \left(\int_{-1+\beta b}^{z} u_{\rm sh}^{*} \right) \partial_{z} \omega_{y}$$
$$= \int_{z}^{\varepsilon\zeta} \partial_{x} \left(u_{\rm sh}^{*} \omega_{y} \right) - \varepsilon\zeta \mathbf{Q}_{x} \underline{\omega_{y}} + \partial_{x} \left(\int_{-1+\beta b}^{z} u_{\rm sh}^{*} \right) \omega_{y}.$$

Since $\boldsymbol{\omega}_y = \partial_z u_{\mathrm{sh}}^*$, we obtain

$$\int_{z}^{\varepsilon\zeta} A_{1} = \mathbf{Q}_{x}\partial_{x}\mathbf{Q}_{x} - u_{\mathrm{sh}}^{*}\partial_{x}u_{\mathrm{sh}}^{*} + \partial_{x}\left(\int_{-1+\beta b}^{z} u_{\mathrm{sh}}^{*}\right)\partial_{z}u_{\mathrm{sh}}^{*}.$$

then, integrating again with respect to z, we obtain

$$\frac{1}{h} \int_{-1+\beta b}^{\varepsilon \zeta} \int_{z}^{\varepsilon \zeta} A_{1} = \mathcal{Q}_{x} \partial_{x} \mathcal{Q}_{x} - \frac{1}{h} \partial_{x} \int_{-1+\beta b}^{\varepsilon \zeta} \left(u_{\mathrm{sh}}^{*}\right)^{2}.$$

Furthermore, we have

$$\begin{split} \int_{z}^{\varepsilon\zeta} A_{2} &= -\frac{1}{2} \int_{z}^{\varepsilon\zeta} \left(\left[1 + z' - \beta b \right]^{2} - \frac{h^{2}}{3} \right) \partial_{x}^{2} \overline{u} \partial_{x} \omega_{y} \\ &+ \frac{1}{2} \int_{z}^{\varepsilon\zeta} \partial_{x} \left(\int_{-1+\beta b}^{z} \left(\left[1 + z' - \beta b \right]^{2} - \frac{h^{2}}{3} \right) \partial_{x}^{2} \overline{u} \right) \partial_{z} \omega_{y} \\ &= -\frac{1}{2} \int_{z}^{\varepsilon\zeta} \partial_{x} \left[\left(\left[1 + z' - \beta b \right]^{2} - \frac{h^{2}}{3} \right) \partial_{x}^{2} \overline{u} \omega_{y} \right] - \varepsilon \partial_{x} \zeta \frac{h^{2}}{3} \partial_{x}^{2} \overline{u} \underline{\omega}_{y} \\ &- \frac{1}{2} \partial_{x} \left(\int_{-1+\beta b}^{z} \left(\left[1 + z' - \beta b \right]^{2} - \frac{h^{2}}{3} \right) \partial_{x}^{2} \overline{u} \right) \omega_{y}. \end{split}$$

Since $\boldsymbol{\omega}_y = \partial_z u_{\mathrm{sh}}^*$, we obtain

$$\begin{split} \int_{z}^{\varepsilon\zeta} A_{2} &= \int_{z}^{\varepsilon\zeta} \partial_{x} \left(\left[1 + z' - \beta b \right] \partial_{x}^{2} \overline{u} u_{\mathrm{sh}}^{*} \right) + \frac{1}{2} \partial_{x} \left(\left(\left[1 + z' - \beta b \right]^{2} - \frac{h^{2}}{3} \right) \partial_{x}^{2} \overline{u} u_{\mathrm{sh}}^{*} \right) \\ &- \frac{1}{2} \partial_{x} \left(\int_{-1 + \beta b}^{z} \left(\left[1 + z' - \beta b \right]^{2} - \frac{h^{2}}{3} \right) \partial_{x}^{2} \overline{u} \right) \partial_{z} u_{\mathrm{sh}}^{*} \\ &+ \frac{1}{3} \partial_{x} \left(h^{2} \partial_{x}^{2} \overline{u} \mathbf{Q}_{x} \right) - \varepsilon \partial_{x} \zeta h \partial_{x}^{2} \overline{u} \mathbf{Q}_{x}. \end{split}$$

Then we integrate again with respect to z and we divide h. We obtain

$$\begin{split} \frac{1}{h} \int_{-1+\beta b}^{\varepsilon \zeta} \int_{z}^{\varepsilon \zeta} A_{2} &= \frac{1}{h} \int_{-1+\beta b}^{\varepsilon \zeta} \int_{z}^{\varepsilon \zeta} \partial_{x} \left(\left[1+z'-\beta b \right] \partial_{x}^{2} \overline{u} u_{\mathrm{sh}}^{*} \right) \\ &+ \frac{1}{2h} \int_{-1+\beta b}^{\varepsilon \zeta} \partial_{x} \left(\left(\left[1+z'-\beta b \right]^{2} - \frac{h^{2}}{3} \right) \partial_{x}^{2} \overline{u} u_{\mathrm{sh}}^{*} \right) \\ &+ \frac{1}{2h} \int_{-1+\beta b}^{\varepsilon \zeta} \partial_{x} \left(\left(\left[1+z'-\beta b \right]^{2} - \frac{h^{2}}{3} \right) \partial_{x}^{2} \overline{u} \right) u_{\mathrm{sh}}^{*} \\ &+ \frac{1}{3} \partial_{x} \left(h^{2} \partial_{x}^{2} \overline{u} Q_{x} \right) - \frac{4}{3} h \partial_{x} h \partial_{x}^{2} \overline{u} Q_{x} + \beta R. \end{split}$$

Then, using the fact that

$$\int_{-1+\beta b}^{\varepsilon\zeta} \int_{z}^{\varepsilon\zeta} \int_{-1+\beta b}^{z'} \partial_x u_{\rm sh}^* = \partial_x \int_{-1+\beta b}^{\varepsilon\zeta} \int_{z}^{\varepsilon\zeta} \int_{-1+\beta b}^{z'} u_{\rm sh}^* + \beta R,$$

we finally get

$$\frac{1}{h} \int_{-1+\beta b}^{\varepsilon \zeta} \int_{z}^{\varepsilon \zeta} A_{2} = \frac{1}{3} \partial_{x} \left(h^{2} \mathbf{Q}_{x} \partial_{x}^{2} \overline{u} \right) + \frac{1}{6} h^{2} u^{\sharp} \partial_{x}^{3} \overline{u} + \frac{1}{8h} \partial_{x} \left(h^{3} u^{\sharp} \right) \partial_{x}^{2} \overline{u} + \beta R,$$

and the first equation follows. The second equation follows similarly using the fact that $u_{\rm sh}^* = u_{\rm sh} - Q_x$.

We can also extend Proposition 5.2.9.

Proposition 5.4.2. If $(\zeta, U^{\mu,0}_{\mathbb{A}}, \omega)$ satisfy the Castro-Lannes system (5.14), then Q_x satisfies the following equation

$$\begin{split} \partial_{t}\mathbf{Q}_{y} + \varepsilon \overline{u}\partial_{x}\mathbf{Q}_{y} + \varepsilon \mathbf{Q}_{x}\partial_{x}\overline{v} + \frac{\varepsilon}{\mathrm{Ro}\sqrt{\mu}}\left(\overline{u} - \underline{u}\right) &= \varepsilon\sqrt{\mu}\mathbf{Q}_{x}\partial_{x}\mathbf{Q}_{y} - \varepsilon\sqrt{\mu}\frac{1}{3}h^{2}\partial_{x}^{2}\overline{u}\partial_{x}\overline{v} \\ &- \varepsilon\sqrt{\mu}\frac{1}{h}\partial_{x}\left(\int_{-1+\beta b}^{\varepsilon\zeta} u_{\mathrm{sh}}^{*}v_{\mathrm{sh}}^{*}\right) \\ &- \varepsilon\mu\left(\partial_{x}h\right)^{2}\mathbf{Q}_{x}\partial_{x}\overline{v} + \varepsilon\mu\frac{h^{2}}{3}\partial_{x}^{2}\overline{u}\partial_{x}\mathbf{Q}_{y} \\ &- \varepsilon\mu\frac{1}{24h}\partial_{x}^{2}(h^{3}u^{\sharp})\partial_{x}\overline{v} + \varepsilon\mu\frac{1}{24h}\partial_{x}(h^{3}v^{\sharp}\partial_{x}^{2}\overline{u}) \\ &+ \varepsilon\max\left(\mu^{\frac{3}{2}},\beta\sqrt{\mu}\right)R, \end{split}$$

and v_{sh}^* satisfies the equation

$$\begin{split} \partial_t v_{\rm sh}^* + \varepsilon \overline{u} \partial_x v_{\rm sh}^* + \varepsilon u_{\rm sh}^* \partial_x \overline{v} + \frac{\varepsilon}{{\rm Ro}\sqrt{\mu}} \left(u - \overline{u}\right) = & \varepsilon \sqrt{\mu} \frac{1}{h} \partial_x \left(\int_{-1+\beta b}^{\varepsilon \zeta} u_{\rm sh}^* v_{\rm sh}^* \right) - \varepsilon \sqrt{\mu} u_{\rm sh}^* \partial_x v_{\rm sh}^* \\ & + \varepsilon \partial_x \left(\int_{-1+\beta b}^z \left[\overline{u} + \sqrt{\mu} u_{\rm sh}^* \right] \right) \partial_z v_{\rm sh}^* \\ & + \varepsilon \sqrt{\mu} \frac{1}{2} \left(\left[1 + z - \beta b \right]^2 - \frac{h^2}{3} \right) \partial_x^2 \overline{u} \partial_x \overline{v} \\ & + \varepsilon \left(\mu, \beta \sqrt{\mu} \right) R. \end{split}$$

Proof. Using the first equation of the vorticity equation of the Castro-Lannes system (5.14), we have

$$\partial_t \boldsymbol{\omega}_x + \varepsilon u \partial_x \boldsymbol{\omega}_x + \frac{\varepsilon}{\mu} \mathbf{w} \partial_z \boldsymbol{\omega}_x = \varepsilon \boldsymbol{\omega}_x \partial_x u + \frac{\varepsilon}{\sqrt{\mu}} \boldsymbol{\omega}_z \partial_z u + \frac{\varepsilon}{\mathrm{Ro}\sqrt{\mu}} \partial_z u.$$

Then, using the fact that $\nabla^{\mu,0} \cdot \boldsymbol{\omega} = 0$ and $\nabla^{\mu,0} \cdot U^{\mu,\gamma} = 0$, we get

$$\partial_t \boldsymbol{\omega}_x - \frac{\varepsilon}{\sqrt{\mu}} \partial_z \left(u \boldsymbol{\omega}_z \right) + \frac{\varepsilon}{\mu} \partial_z \left(w \boldsymbol{\omega}_x \right) = \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} \partial_z u.$$

then, we integrate with respect to z and, using the fact that $\partial_t \zeta - \frac{1}{\mu} \underline{U}^{\mu} \cdot N^{\mu,0} = 0$, $\boldsymbol{\omega}_x = -\frac{1}{\sqrt{\mu}} \partial_z v$ and $\boldsymbol{\omega}_z = \partial_x v$, we obtain

$$\partial_t \left(\int_{-1+\beta b}^{\varepsilon \zeta} \boldsymbol{\omega}_x \right) - \frac{\varepsilon}{\sqrt{\mu}} \underline{u} \partial_x \underline{v} + \frac{\varepsilon}{\sqrt{\mu}} u \partial_x v + \frac{\varepsilon}{\mu^{\frac{3}{2}}} \mathbf{w} \partial_z v + \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} \left(u - \underline{u} \right) = 0.$$

Then, we integrate again with respect to z and, using Proposition 5.2.4 and the fact that $\partial_t \zeta - \frac{1}{\mu} \underline{U}^{\mu} \cdot N^{\mu,0} = 0$, $U_b^{\mu} \cdot N_b^{\mu,0} = 0$, and $\nabla^{\mu,0} \cdot U^{\mu} = 0$, we get

$$\partial_t \mathbf{Q}_y - \frac{\varepsilon}{\sqrt{\mu}} \underline{u} \partial_x \underline{v} + \frac{\varepsilon}{\sqrt{\mu}} \frac{1}{h} \partial_x \left(\int_{-1+\beta b}^{\varepsilon \zeta} uv \right) + \frac{1}{\sqrt{\mu} h} \partial_t h \overline{v} + \frac{\varepsilon}{\operatorname{Ro}\sqrt{\mu}} \left(\overline{u} - \underline{u} \right) = 0.$$

Then, thanks to Propositions 5.2.3, 5.2.4 and 5.2.5 we finally obtain that

$$\begin{split} \partial_t \mathbf{Q}_y + \varepsilon \overline{u} \partial_x \mathbf{Q}_y + \varepsilon \mathbf{Q}_x \partial_x \overline{v} + \frac{\varepsilon}{\mathrm{Ro}\sqrt{\mu}} (\overline{u} - \underline{u}) &= \varepsilon \sqrt{\mu} \mathbf{Q}_x \partial_x \mathbf{Q}_y - \varepsilon \sqrt{\mu} \frac{1}{3} h^2 \partial_x^2 \overline{u} \partial_x \overline{v} \\ &\quad - \varepsilon \sqrt{\mu} \frac{1}{h} \partial_x \left(\int_{-1 + \beta b}^{\varepsilon \zeta} u_{\mathrm{sh}}^* v_{\mathrm{sh}}^* \right) \\ &\quad + \varepsilon \mu \overline{T} u_{\mathrm{sh}}^* \partial_x \overline{v} + \varepsilon \mu \frac{h^2}{3} \partial_x^2 \overline{u} \partial_x \mathbf{Q}_y \\ &\quad + \varepsilon \mu \frac{1}{2h} \partial_x \left(\int_{-1 + \beta b}^{\varepsilon \zeta} v_{\mathrm{sh}}^* \left([1 + z - \beta b]^2 - \frac{h^2}{3} \right) \partial_x^2 \overline{u} \right) \\ &\quad + \varepsilon \max \left(\mu^{\frac{3}{2}}, \beta \sqrt{\mu} \right) R. \end{split}$$

Finally, we can compute that

$$\frac{1}{2} \int_{-1+\beta b}^{\varepsilon \zeta} v_{\rm sh}^* \left(\left[1+z-\beta b \right]^2 - \frac{h^2}{3} \right) = \frac{1}{24} h^3 v^{\sharp},$$

and the first equation follows from Lemma 5.2.7. The second equation follows similarly using the fact that $v_{\rm sh}^* = v_{\rm sh} - Q_y$.

As noticed in [33], the quantity E defined by

$$E = \begin{pmatrix} E_{xx} & E_{xy} \\ E_{xy} & E_{yy} \end{pmatrix} = \int_{-1+\beta b}^{\varepsilon \zeta} \mathbf{V}_{\mathrm{sh}}^* \otimes \mathbf{V}_{\mathrm{sh}}^*$$
(5.58)

appears in the equations of Q_x and Q_y and can not be express with respect to ζ , $\overline{\mathbf{V}}$ and \mathbf{V}^{\sharp} . The following subsection is devoting to giving an equation for E.

5.4.2 Equations for E

In this part, we derive an equation for E up to terms of order $\mathcal{O}(\mu)$. We have to introduce the quantity F

$$F = (F_{ijk})_{i,j,k} = \int_{-1+\beta b}^{\varepsilon \zeta} \mathbf{V}_{\mathrm{sh}}^* \otimes \mathbf{V}_{\mathrm{sh}}^* \otimes \mathbf{V}_{\mathrm{sh}}^*.$$
(5.59)

The following proposition gives an equation for E.

Proposition 5.4.3. If $(\zeta, U_{\parallel}^{\mu,0}, \omega)$ satisfy the Castro-Lannes system (5.14), then E satisfies the following equation

$$\partial_t E + \varepsilon \overline{u} \partial_x E + \varepsilon l\left(E, \partial_x \overline{\mathbf{V}}\right) + \varepsilon \sqrt{\mu} \partial_x F_{\cdot,\cdot,1} + \frac{\varepsilon}{\mathrm{Ro}} E^S = \left(\varepsilon \sqrt{\mu} \partial_x \overline{v} + \frac{\varepsilon \sqrt{\mu}}{\mathrm{Ro}}\right) \mathcal{D}(\mathbf{V}^\sharp, \overline{u}) \\ + \max\left(\varepsilon \mu, \varepsilon \beta \sqrt{\mu}, \frac{\varepsilon}{\mathrm{Ro}} \mu\right) R,$$

where

$$E^{S} = \int_{-1+\beta b}^{\varepsilon \zeta} \mathbf{V}_{\mathrm{sh}}^{\perp} \otimes \mathbf{V}_{\mathrm{sh}} + \mathbf{V}_{\mathrm{sh}} \otimes \mathbf{V}_{\mathrm{sh}}^{\perp} = \begin{pmatrix} -2E_{xy} & E_{xx} - E_{yy} \\ E_{xx} - E_{yy} & 2E_{xy} \end{pmatrix}$$
(5.60)

$$l\left(E,\partial_{x}\overline{\mathbf{V}}\right) = \begin{pmatrix} 3\partial_{x}\overline{u}E_{xx} + 2\partial_{x}\overline{v}E_{xy} & 2\partial_{x}\overline{u}E_{xy} + \partial_{x}\overline{v}E_{yy} \\ 2\partial_{x}\overline{u}E_{xy} + \partial_{x}\overline{v}E_{yy} & \partial_{x}\overline{u}E_{yy} \end{pmatrix}$$
(5.61)

and

$$\mathcal{D}(\mathbf{V}^{\sharp}, \overline{u}) = \partial_x^2 \overline{u} \begin{pmatrix} 0 & u^{\sharp} \\ u^{\sharp} & 2v^{\sharp} \end{pmatrix}.$$
(5.62)

Proof. The proof is similar to the computation in Part 4.5.2 and Part 5.4.1 in [33]. We compute $\partial_t E$ and we use the second equations of Propositions 5.4.1 and 5.4.2 up to terms of order $\mathcal{O}(\mu)$. For the Coriolis contribution, we use the expansion of u and v given in Proposition 5.2.5 and 5.2.4.

The quantity F appears in the equation of E and can not be expressed with respect to ζ , $\overline{\mathbf{V}}$, \mathbf{V}^{\sharp} and E. The next proposition gives an equation for F up to terms of order $\mathcal{O}(\sqrt{\mu})$.

Proposition 5.4.4. If $(\zeta, U_{\parallel}^{\mu,0}, \omega)$ satisfy the Castro-Lannes system (5.14), then F_{ijk} satisfies the following equation

$$\partial_t F_{ijk} + \varepsilon (\overline{u} \partial_x F_{ijk} + \partial_x \overline{u} F_{ijk} + F_{1kj} \partial_x \mathbf{V}_i + F_{i1k} \partial_x \mathbf{V}_j + F_{ij1} \partial_x \mathbf{V}_k) + \frac{\varepsilon}{\text{Ro}} F^S = \max\left(\varepsilon, \frac{\varepsilon}{\text{Ro}}\right) \sqrt{\mu} R,$$

where

$$F^{S} = \int_{-1+\beta b}^{\varepsilon \zeta} \mathbf{V}_{\rm sh}^{\perp} \otimes \mathbf{V}_{\rm sh} \otimes \mathbf{V}_{\rm sh} + \mathbf{V}_{\rm sh} \otimes \mathbf{V}_{\rm sh}^{\perp} \otimes \mathbf{V}_{\rm sh} + \mathbf{V}_{\rm sh} \otimes \mathbf{V}_{\rm sh} \otimes \mathbf{V}_{\rm sh}^{\perp}.$$
 (5.63)

Proof. The proof is similar to the computation in Part 4.5.3 and Part 5.4.2 in [33]. We compute $\partial_t F$ and we use the second equations of Propositions 5.4.1 and 5.4.2 up to terms of order $\mathcal{O}(\sqrt{\mu})$. For the Coriolis contribution, we use the expansion of u and v in Proposition 5.2.5 and 5.2.4.

5.4.3 The Green-Naghdi equations

We can now establish the Green-Naghdi equations when d = 1. The Green-Naghdi equations are the following system

$$\begin{cases} \partial_{t}\zeta + \partial_{x}\left(h\overline{u}\right) = 0, \\ (1+\mu\mathcal{T})(\partial_{t}\overline{u} + \varepsilon\overline{u}\partial_{x}\overline{u}) + \partial_{x}\zeta - \frac{\varepsilon}{\mathrm{Ro}}\overline{v} + \varepsilon\mu\mathcal{Q}(\overline{u}) + \varepsilon\mu\partial_{x}E_{xx} + \varepsilon\mu^{\frac{3}{2}}\mathcal{C}_{1}\left(u^{\sharp},\overline{u}\right) + \frac{\varepsilon}{\mathrm{Ro}}\frac{\mu^{\frac{3}{2}}}{24h}\partial_{x}^{2}(h^{3}v^{\sharp}) = 0, \\ \partial_{t}\overline{v} + \varepsilon\overline{u}\partial_{x}\overline{v} + \frac{\varepsilon}{\mathrm{Ro}}\overline{u} + \varepsilon\mu\partial_{x}E_{xy} + \varepsilon\mu^{\frac{3}{2}}\mathcal{C}_{2}\left(v^{\sharp},\partial_{x}^{2}\overline{u}\right) = 0, \\ \partial_{t}\mathbf{V}^{\sharp} + \varepsilon\mathbf{V}^{\sharp}\partial_{x}\overline{u} + \varepsilon\overline{u}\partial_{x}\mathbf{V}^{\sharp} + \frac{\varepsilon}{\mathrm{Ro}}\mathbf{V}^{\sharp\perp} = 0, \\ \partial_{t}E + \varepsilon\overline{u}\partial_{x}E + \varepsilon\,l\left(E,\partial_{x}\overline{\mathbf{V}}\right) + \varepsilon\sqrt{\mu}\partial_{x}F_{\cdot,\cdot,1} + \frac{\varepsilon}{\mathrm{Ro}}E^{S} = \left(\varepsilon\sqrt{\mu}\partial_{x}\overline{v} + \frac{\varepsilon}{\mathrm{Ro}}\sqrt{\mu}\right)\mathcal{D}(\mathbf{V}^{\sharp},\overline{u}), \\ \partial_{t}F_{ijk} + \varepsilon\overline{u}\partial_{x}F_{ijk} + \varepsilon\partial_{x}\overline{u}F_{ijk} + \varepsilon F_{1kj}\partial_{x}\mathbf{V}_{i} + \varepsilon F_{i1k}\partial_{x}\mathbf{V}_{j} + \varepsilon F_{ij1}\partial_{x}\mathbf{V}_{k} + \frac{\varepsilon}{\mathrm{Ro}}F^{S} = 0. \end{cases}$$

$$(5.64)$$

where

$$\begin{aligned}
\mathcal{T} &= -\frac{1}{3h} \partial_x \left(h^3 \partial_x \cdot \right), \\
\mathcal{Q}(\overline{u}) &= \frac{2}{3h} \partial_x \left(h^3 \left[\partial_x \overline{u} \right]^2 \right), \\
\mathcal{C}_1 \left(u^{\sharp}, \overline{u} \right) &= -\frac{1}{6h} \partial_x \left(2h^3 u^{\sharp} \partial_x^2 \overline{u} + \partial_x (h^3 u^{\sharp}) \partial_x \overline{u} \right), \\
\mathcal{C}_2 \left(v^{\sharp}, w \right) &= -\frac{1}{24h} \partial_x \left(h^3 v^{\sharp} w \right), \\
l \left(E, \partial_x \overline{\mathbf{V}} \right) &= \begin{pmatrix} 3\partial_x \overline{u} E_{xx} + 2\partial_x \overline{v} E_{xy} & 2\partial_x \overline{u} E_{xy} + \partial_x \overline{v} E_{yy} \\ 2\partial_x \overline{u} E_{xy} + \partial_x \overline{v} E_{yy} & \partial_x \overline{u} E_{yy} \end{pmatrix}, \\
\mathcal{D}(\mathbf{V}^{\sharp}, \overline{u}) &= \partial_x^2 \overline{u} \begin{pmatrix} 0 & u^{\sharp} \\ u^{\sharp} & 2v^{\sharp} \end{pmatrix}
\end{aligned}$$
(5.65)

and

$$E^{S} = \int_{-1+\beta b}^{\varepsilon \zeta} \mathbf{V}_{sh}^{\perp} \otimes \mathbf{V}_{sh} + \mathbf{V}_{sh} \otimes \mathbf{V}_{sh}^{\perp} = \begin{pmatrix} -2E_{xy} & E_{xx} - E_{yy} \\ E_{xx} - E_{yy} & 2E_{xy} \end{pmatrix},$$

$$F^{S} = \int_{-1+\beta b}^{\varepsilon \zeta} \mathbf{V}_{sh}^{\perp} \otimes \mathbf{V}_{sh} \otimes \mathbf{V}_{sh} + \mathbf{V}_{sh} \otimes \mathbf{V}_{sh}^{\perp} \otimes \mathbf{V}_{sh} + \mathbf{V}_{sh} \otimes \mathbf{V}_{sh} \otimes \mathbf{V}_{sh},$$

(5.66)

and \mathbf{V}^{\sharp} is defined in (5.29), E in (5.58) and F in (5.59). Notice that the first, the second and the third equations of System (5.64) are the classical Green-Naghdi equations with new terms due to the vorticity (terms with \mathbf{V}^{\sharp} and E). The last equations are important to get a close system. We can now state that the Green-Naghdi equations are an order $\mathcal{O}(\mu^2)$ approximation of the water waves equations.

Proposition 5.4.5. In the Green-Naghdi regime with small topography variations \mathcal{A}_{GN} , the Castro-Lannes equations (5.14) are consistent at order $\mathcal{O}(\mu^2)$ with the Green-Naghdi equations (5.64) in the sense of Definition 5.1.4.

Proof. The proof is similar to the one in Proposition 5.2.12. The first equation of the Green-Naghdi equations is always satisfied for a solution of the Castro-Lannes formulation by Proposition 5.2.3. For the second equation, we use Proposition 5.2.5, Proposition 5.4.1 together with Proposition 5.2.6, Lemma 5.2.7 and Proposition 5.2.10. Notice the fact that all the terms with Q_x disappear. The third equation follows from Proposition 5.2.4, 5.2.5 and 5.4.2 (all the terms with Q_y disappear also). The last equations follows from Propositions 5.2.10, 5.4.3 and 5.4.4.

Remark 5.4.6. Notice that even without a Coriolis forcing, we can not decrease the number of equations in the previous Green-Naghdi equations. However, if one suppose also that the vorticity is initially of the form $(0, \omega_y, 0)^t$, which corresponds to the propagation of 2D water waves, we can significantly simplify the Green-Naghdi equations (See Section 4 in [33] and [82]).

5.4.4 A simplified model in the case of a weak rotation and medium amplitude waves

As noticed in [33], if we assume that $\varepsilon = \mathcal{O}(\sqrt{\mu})$ we can simplify the Green-Naghdi equations. This regime corresponds to medium amplitude waves (in the terminology of [80]). We also assume that $\frac{\varepsilon}{\text{Ro}} = \mathcal{O}(\sqrt{\mu})$. Then, we can simplify the Green-Naghdi system (5.64) by dropping all the terms of $\mathcal{O}(\mu^2)$ and we get

$$\begin{cases} \partial_t \zeta + \partial_x \left(h\overline{u}\right) = 0, \\ (1 + \mu \mathcal{T}) \left(\partial_t \overline{u} + \varepsilon \overline{u} \partial_x \overline{u}\right) + \partial_x \zeta - \frac{\varepsilon}{\text{Ro}} \overline{v} + \varepsilon \mu \mathcal{Q}(\overline{u}) + \varepsilon \mu \partial_x E_{xx} = 0, \\ \partial_t \overline{v} + \varepsilon \overline{u} \partial_x \overline{v} + \frac{\varepsilon}{\text{Ro}} \overline{u} + \varepsilon \mu \partial_x E_{xy} = 0, \\ \partial_t E + \varepsilon \overline{u} \partial_x E + \varepsilon l \left(E, \partial_x \overline{\mathbf{V}}\right) + \frac{\varepsilon}{\text{Ro}} E^S = 0. \end{cases}$$

$$(5.67)$$

Notice that in this regime, we catch effects of the vorticity on $\overline{\mathbf{V}}$ thanks to the second order tensor E. Without vorticity, this regime is particularly interesting since it is related to the Camassa-Holm equation and the Degasperis-Procesi equation (see for instance [41]). It could be interesting to understand how we can adapt these two scalar equations in presence of a Coriolis forcing.

Appendix A Laplace problems for water waves

Sommaire

Laplace problems
Formulation of the problems
Regularity estimates of the solutions
Shape derivatives of the $B^{\mathfrak{d}}$
e Dirichlet-Neumann and the Neumann-Neumann operators . 180
Main properties
Regularity Estimates
Shape derivatives

This appendix is devoted to the study of Laplace problems that appear in the water waves problems. After obtaining the existence and uniqueness of solutions, we give precise regularity estimates. The second section is devoted to the study of non local operators that appear in the water waves problems. We give regularity estimate and shape derivatives of these operators. Notice that it is very important to obtain precise regularity estimates if one wants to get a local wellposedness for the water waves equations.

We recall that the water occupies the domain $\Omega_t := \{(X, z) \in \mathbb{R}^{d+1}, -1 + \beta b(t, X) < z < \varepsilon \zeta(t, X)\}$. We assume also that the water depth is bounded from below by a positive constant

$$\exists h_{\min} > 0 , \varepsilon \zeta + 1 - \beta b \ge h_{\min}.$$
(A.1)

Finally, we suppose that there $\mu_{\max} > 0$, such that

$$0 < \varepsilon \le 1$$
, $0 \le \beta \le 1$ and $\mu \le \mu_{\max}$. (A.2)

In this chapter, C is a constant and for a function f in a normed space $(X, |\cdot|)$ or a parameter γ , $C(|f|, \gamma)$ is a constant depending on |f| and γ whose exact value has non importance. The norm $|\cdot|_2$ is the L^2 -norm and $|\cdot|_{\infty}$ is the L^{∞} -norm in \mathbb{R}^d . Let $f \in \mathcal{C}^0(\mathbb{R}^d)$ and $m \in \mathbb{N}$ such that $\frac{f}{1+|x|^m} \in L^{\infty}(\mathbb{R}^d)$. We define the Fourier multiplier $f(D): H^m(\mathbb{R}^d) \to L^2(\mathbb{R}^d)$ as

$$\forall u \in H^m(\mathbb{R}^d), \ \widehat{f(D)u}(\xi) = f(\xi)\widehat{u}(\xi).$$

In \mathbb{R}^d we denote the gradient operator by ∇ and in Ω or $S = \mathbb{R}^d \times (-1, 0)$ the gradient operator is denoted $\nabla_{X,z}$. Finally, we denote by $\Lambda := \sqrt{1 + |D|^2}$ with $D = -i\nabla$.

In this chapter, (,) is the standard $L^2(\mathbb{R}^d)$ scalar product.

A.1 The Laplace problems

A.1.1 Formulation of the problems

In this part, we extend the results of Chapter 2 in [80] and Section 4 of [71] (see also [78] and [70]). We suppose that the parameters ε , μ and β satisfy Condition (A.2). In this chapter, we have to study two Laplace problems. The first one is the problem

$$\begin{cases} \Delta^{\mu} \Phi^{S} = 0 \text{ in } \Omega_{t} ,\\ \Phi^{S}_{|z=\varepsilon\zeta} = \psi , \sqrt{1+\varepsilon^{2} |\nabla\zeta|^{2}} \partial_{n} \Phi^{S}_{|z=-1+\beta b} = 0. \end{cases}$$
(A.3)

The second one is the problem

$$\begin{cases} \Delta^{\mu} \Phi^{B} = 0 \text{ in } \Omega_{t} ,\\ \Phi^{B}_{|z=\varepsilon\zeta} = 0 , \sqrt{1+\beta^{2}|\nabla b|^{2}} \partial_{n} \Phi^{B}_{|z=-1+\beta b} = B, \end{cases}$$
(A.4)

and here $B := \partial_t b$. Notice that n is here the upward normal. We work with Beppo Levi spaces. We define the Beppo Levi spaces as, for $s \ge 0$,

$$\dot{H}^{s}(\mathbb{R}^{d}) := \left\{ \psi \in L^{2}_{\text{loc}}(\mathbb{R}^{d}), \, \nabla \psi \in H^{s-1}(\mathbb{R}^{d}) \right\}.$$

We refer to [53] and Proposition 2.3 in [80] for general results about these spaces but we recall the following properties of these spaces.

Proposition A.1.1. Let $s \geq 1$. Then, $H^s(\mathbb{R}^d)$ is dense in $\dot{H}^s(\mathbb{R}^d)$. Furthermore, $\dot{H}^1(\mathbb{R}^d \times (-1,0))/\mathbb{R}$ is a Hilbert space for the norm $|\nabla_{X,z} \cdot |_{L^2}$.

In order to give precise regularity estimates, we fix the domain. We transform these problems into variable coefficients elliptic problems on $S := \mathbb{R}^d \times (-1, 0)$ (the flat strip). We introduce a regularizing diffeomorphism. Let $\theta : \mathbb{R} \to \mathbb{R}$ be a positive, compactly supported, smooth, even function equal to one near 0. For $\delta > 0$ we define

$$\Sigma := \begin{array}{cc} S & \longrightarrow & \Omega \\ (X,z) & \mapsto (X,z + [\theta(\delta z | D |) \varepsilon \zeta(X) - \theta(\delta(z+1) | D |) \beta b(X)] \, z + \varepsilon \theta(\delta z | D |) \zeta(X)) \, . \end{array}$$

We omit the dependence on t here. We denote by M_0 a constant of the form

$$M = C\left(\frac{1}{h_{\min}}, \mu_{\max}, |\zeta|_{H^{t_0+1}(\mathbb{R}^d)}, |b|_{H^{t_0+1}(\mathbb{R}^d)}\right).$$

In order to study the Laplace problems in S, we have to treat the regularity in the direction X and in the direction z one at a time. We introduce the following spaces.

Definition A.1.2. Let $s \in \mathbb{R}$. We define the Banach spaces $(H^{s,k}(S), |\cdot|_{s,k}), k = 0, 1, by$

$$H^{s,1}(S) := L^2_z H^s_X(S) \cap H^1_z H^{s-1}_X(S), \text{ and } |u|^2_{s,1} = |\Lambda^s u|^2_2 + |\Lambda^{s-1} \partial_z u|^2_2,$$

and

$$H^{s,0}(S) := L_z^2 H_X^s(S), and |u|_{s,0}^2 = |\Lambda^s u|_2^2.$$

Remark A.1.3. We have the following inclusion (see Proposition 2.10 in [80]) for $s \in \mathbb{R}$

$$H^{s+\frac{1}{2},1}(S) \subset L^{\infty}_{z}H^{s}_{X}(S).$$

 Σ satisfies the following properties (see Propositions 2.16 and 2.18 in [80] for the proof).

Proposition A.1.4. Let $t_0 > \frac{d}{2}$, ζ , $b \in H^{t_0+1}(\mathbb{R}^d)$ such that Condition (A.1) is satisfied and $s \leq t_0 + \frac{1}{2}$. Then, we can extend continuously Σ as a mapping $\overline{S} \to \overline{\Omega}$. Furthermore, for $\delta > 0$ small enough, Σ is a regularizing diffeomorphism and we have

$$|J_{\Sigma}|_{L^{\infty}(S)} \leq M$$
, $|\det(J_{\Sigma})|_{L^{\infty}(S)} \geq \frac{h_{\min}}{2}$ and $\partial_{z}\Sigma \geq \frac{1}{2}$

Finally, if we denote by

$$\sigma(X,z) := \left[\theta(\delta z |D|) \varepsilon \zeta(X) - \theta(\delta(z+1)|D|)\beta b(X)\right] z + \varepsilon \theta(\delta z |D|) \zeta(X),$$

we have

$$\begin{split} \left| \nabla_{X,z}^{\mu} \sigma \right|_{H^{s,1}(S)} &\leq C \left(\mu_{\max}, \frac{1}{\delta} \right) \left(\left| \varepsilon \zeta \right|_{H^{s+\frac{1}{2}}} + \left| \beta b \right|_{H^{s+\frac{1}{2}}} \right), \\ \left| \Lambda^{s-\frac{1}{2}} \nabla_{X,z}^{\mu} \sigma \right|_{L_{z}^{\infty} L_{X}^{2}(S)} &\leq C \left(\mu_{\max}, \frac{1}{\delta} \right) \left(\left| \varepsilon \zeta \right|_{H^{s+\frac{1}{2}}} + \left| \beta b \right|_{H^{s+\frac{1}{2}}} \right), \\ \left| \Lambda^{s-\frac{3}{2}} \partial_{z} \nabla_{X,z}^{\mu} \sigma \right|_{L_{z}^{\infty} L_{X}^{2}(S)} &\leq C \left(\mu_{\max}, \frac{1}{\delta} \right) \left(\left| \varepsilon \zeta \right|_{H^{s+\frac{1}{2}}} + \left| \beta b \right|_{H^{s+\frac{1}{2}}} \right). \end{split}$$

Remark A.1.5. Notice that if $h \in H^{s}(\mathbb{R}^{d})$, then

$$\theta(\delta z|D|)h \in L^{\infty}_{z}H^{s}_{X}(S) \cap L^{2}_{z}H^{s+\frac{1}{2}}_{X}(S)$$

See Lemma 2.20 in [80].

Remark A.1.6. The smallness of δ depends on the inverse of $(|\varepsilon\zeta|_{H^{t_0+1}} + |\beta b|_{H^{t_0+1}})$.

In the following, we fix $\delta > 0$ such that Proposition A.1.4 is valid. Then, we can transform our equations. We denote by $\phi^S := \Phi^S \circ \Sigma$ and $\phi^B := \Phi^B \circ \Sigma$. We obtain that ϕ^S and ϕ^B satisfy

$$\begin{cases} \nabla^{\mu}_{X,z} \cdot P(\Sigma) \nabla^{\mu}_{X,z} \phi^{S} = 0 \text{ in } S, \\ \phi^{S}_{|z=0} = \psi , \partial_{n} \phi^{S}_{|z=-1} = 0, \end{cases}$$
(A.5)

and

$$\begin{cases} \nabla^{\mu}_{X,z} \cdot P(\Sigma) \nabla^{\mu}_{X,z} \phi^B = 0 \text{ in } S, \\ \phi^B_{|z=0} = 0, \ \partial_n \phi^B_{|z=-1} = B, \end{cases}$$
(A.6)

with $P(\Sigma) = I_{d+1 \times d+1} + Q(\Sigma) \in S_{d+1}^{++}(\mathbb{R})$ and

$$Q(\Sigma) := \begin{pmatrix} \partial_z \sigma I_{d \times d} & -\sqrt{\mu} \nabla_X \sigma \\ -\sqrt{\mu} \nabla_X \sigma^t & \frac{-\partial_z \sigma + \mu |\nabla_X \sigma|^2}{1 + \partial_z \sigma} \end{pmatrix}.$$
 (A.7)

Notice that $P(\Sigma)$ is well defined thanks to Proposition A.1.4 and that $\partial_n := \mathbf{e}_z \cdot (P(\Sigma)\nabla^{\mu} \cdot)$. We have to know the regularity of $P(\Sigma)$. It is the subject of the next proposition.

Proposition A.1.7. Let $t_0 > \frac{d}{2}$, $\zeta, b \in H^{t_0+1}(\mathbb{R}^d)$ such that Condition (A.1) is satisfied. Then,

$$|Q(\Sigma)|_{H^{t_0+\frac{1}{2},1}}, \left|\Lambda^{t_0}Q(\Sigma)\right|_{L^{\infty}_z L^2_X(S)}, \left|\Lambda^{t_0-1}\partial_z Q(\Sigma)\right|_{L^{\infty}_z L^2_X(S)} \le M.$$

Furthermore, $P(\Sigma)$ is coercive. There exist a constant $k(\Sigma) > 0$ such that $\frac{1}{k(\Sigma)} \leq M$ and

$$\forall \Theta \in \mathbb{R}^{d+1} \text{ , } \forall (X,z) \in \mathcal{S} \text{ , } P(\Sigma)(X,z) \Theta \cdot \Theta \geq k(\Sigma) |\Theta|^2.$$

Proof. By the product estimate B.2.3, $|\nabla_X \sigma|^2 \in H^{t_0+\frac{1}{2},1}(S)$. Then, using Products estimates B.2.2 and B.2.5 we obtain

$$\begin{split} \left| \Lambda^{t_0 + \frac{1}{2}} Q(\Sigma) \right|_{L^2(S)} &\leq \max(\sqrt{\mu_{\max}}, 1) \ C\left(\left| \Lambda^{t_0 + \frac{1}{2}} \nabla^{\mu}_{X, z} \sigma \right|_{L^2(S)} \right), \\ \left| \Lambda^{t_0 - \frac{1}{2}} \partial_z Q(\Sigma) \right|_{L^2(S)} &\leq \max(\sqrt{\mu_{\max}}, 1) \ C\left(\left| \Lambda^{t_0 - \frac{1}{2}} \partial_z \nabla^{\mu}_{X, z} \sigma \right|_{L^2(S)} \right), \\ \left| \Lambda^{t_0} Q(\Sigma) \right|_{L^{\infty}_z L^2_X(S)} &\leq \max(\sqrt{\mu_{\max}}, 1) \ C\left(\left| \Lambda^{t_0} \nabla^{\mu}_{X, z} \sigma \right|_{L^{\infty}_z L^2_X(S)} \right), \\ \left| \Lambda^{t_0 - 1} \partial_z Q(\Sigma) \right|_{L^{\infty}_z L^2_X(S)} &\leq \max(\sqrt{\mu_{\max}}, 1) \ C\left(\left| \Lambda^{t_0 - 1} \partial_z \nabla^{\mu}_{X, z} \sigma \right|_{L^{\infty}_z L^2_X(S)} \right). \end{split}$$

We get the first estimates thanks to Proposition A.1.4. A straightforward computing gives the second point (see Lemma 2.26 in [80]). $\hfill \Box$

In order to define variational formulations of the Laplace problems, it is useful to introduce an extension result.

Definition A.1.8. Let χ be a smooth compactly supported real function that is equal to 1 near 0 and $\psi \in \dot{H}^{\frac{1}{2}}(\mathbb{R}^d)$. We denote by

$$\psi^{\dagger}(.,z) := \psi + \int_0^1 \chi'(s\sqrt{\mu}z|D|) ds\sqrt{\mu}z|D|\psi, \qquad (A.8)$$

where -1 < z < 0. We introduce also $\mathfrak{P} := \frac{|D|}{\sqrt{1+\sqrt{\mu}|D|}}$. We will see that \mathfrak{P} acts as the square root of the Dirichlet-Neumann operator in $\dot{H}^s(\mathbb{R}^d)$. Lemma 2.34 in [80] gives the following regularity result for ψ^{\dagger} .

Proposition A.1.9. Let $s \in \mathbb{R}$ and $\psi \in \dot{H}^{s+\frac{1}{2}}(\mathbb{R}^d)$. Then, $\nabla^{\mu}_{X,z}\psi^{\dagger} \in H^{s,1}(S)$ and

$$|\Lambda^s \nabla^{\mu}_{X,z} \psi^{\dagger}|_{L^2(S)} \le C \sqrt{\mu} |\mathfrak{P}\psi|_{H^s(\mathbb{R}^d)} \text{ and } |\Lambda^{s-1} \partial_z \nabla^{\mu}_{X,z} \psi^{\dagger}|_{L^2(S)} \le C \mu |\mathfrak{P}\psi|_{H^s(\mathbb{R}^d)}.$$

We can now define the variational formulations of Problem (A.5) and (A.6). We introduce

$$H^{1}_{0,surf}(S) := \overline{\mathcal{D}(S \cup \{z = -1\})}^{||_{H^{1}(S)}} = \overline{\mathcal{D}(S \cup \{z = -1\})}^{||_{\dot{H}^{1}(S)}}$$

See Proposition 2.3 (3) in [80] for a proof of the second equality.

Definition A.1.10. Let $\psi \in \dot{H}^{\frac{1}{2}}(\mathbb{R}^d)$. We say that ϕ is a variational solution of (A.5) if $\phi = \tilde{\phi} + \psi^{\dagger}$ with $\tilde{\phi} \in H^1_{0,surf}(S)$ and for all $\varphi \in H^1_{0,surf}(S)$,

$$\int_{S} \nabla^{\mu} \widetilde{\phi} \cdot P(\Sigma) \nabla^{\mu} \varphi = -\int_{S} \nabla^{\mu} \psi^{\dagger} \cdot P(\Sigma) \nabla^{\mu} \varphi.$$

Definition A.1.11. Let $B \in H^{-\frac{1}{2}}(\mathbb{R}^d)$. We say that $\phi \in H^1_{0,surf}(S)$ is a variational solution of (A.6) if for all $\varphi \in H^1_{0,surf}(S)$,

$$\int_{S} \nabla^{\mu} \phi \cdot P(\Sigma) \nabla^{\mu} \varphi = -\langle B, \varphi|_{z=-1} \rangle_{H^{-\frac{1}{2}} - H^{\frac{1}{2}}}.$$

We have the following trace result.

Lemma A.1.12. For all $\varphi \in H^1_{0,surf}(S)$ we have

$$\left| \sqrt{1 + \sqrt{\mu} |D|} \varphi_{|z=-1} \right|_{L^2(\mathbb{R}^d)} \le 2 \left| \nabla^{\mu}_{X,z} \varphi \right|_{L^2(S)}$$

Proof. Let $u \in \mathcal{D}(S \cup \{z = -1\})$. We have

$$\begin{split} \int_{\xi} (1+\sqrt{\mu}|\xi|) |\widehat{u}(\xi,-1)|^2 d\xi &\leq 2 \int_{\xi} (1+\sqrt{\mu}|\xi|) \int_{z=-1}^{0} |\partial_z \widehat{u}(\xi,z)| |\widehat{u}(\xi,z)| dz d\xi \\ &\leq \int_{\xi} \int_{z=-1}^{0} |\partial_z \widehat{u}(\xi,z)|^2 dz + \int_{\xi} \int_{z=-1}^{0} (1+\sqrt{\mu}|\xi|)^2 |\widehat{u}(\xi,z)|^2 dz d\xi \\ &\leq \int_{\xi} \int_{z=-1}^{0} |\partial_z \widehat{u}(\xi,z)|^2 dz d\xi + 2 \int_{\xi} \int_{z=-1}^{0} (1+\mu|\xi|^2) |\widehat{u}(\xi,z)|^2 dz d\xi \end{split}$$

Furthermore, one can check that

$$\int_{\xi} \int_{z=-1}^{0} |\widehat{u}(\xi, z)|^2 dz d\xi \le \frac{1}{2} \int_{\xi} \int_{z=-1}^{0} |\partial_z \widehat{u}(\xi, z)|^2 dz d\xi.$$

Finally,

$$\int_{\xi} (1+\sqrt{\mu}|\xi|) |\widehat{u}(\xi,-1)|^2 d\xi \le 2 \int_{\xi} \int_{z=-1}^{0} |\partial_z \widehat{u}(\xi,z)|^2 dz d\xi + 2 \int_{\xi} \int_{z=-1}^{0} \mu |\xi|^2 |\widehat{u}(\xi,z)|^2 dz d\xi ,$$

and the result follows by density.

We can now establish existence and uniqueness results.

Proposition A.1.13. Let $\psi \in \dot{H}^{\frac{1}{2}}(\mathbb{R}^d)$, $B \in H^{-\frac{1}{2}}(\mathbb{R}^d)$ and ζ , $b \in H^{t_0+1}(\mathbb{R}^d)$ satisfying (A.1). Then, problems (A.5) and (A.6) have a unique variational solution named respectively $\psi^{\mathfrak{h}} \in \dot{H}^1(\mathbb{R}^d)$ and $B^{\mathfrak{d}} \in H^1_{0,surf}(S)$.

Proof. Because S is bounded in the direction z and that $P(\Sigma)$ is uniformly coercive, the results follow from the Lax-Milgram's theorem, Proposition (A.1.1) and Poincaré's inequality in $H^1_{0,surf}(S)$.

It is tempting to extend the variational formulations A.1.10 and A.1.11 for $\phi \in \dot{H}^1(S)/\mathbb{R}$. It is worked for $\psi^{\mathfrak{h}}$ but it requires that $B \in \left(\dot{H}^{\frac{1}{2}}(\mathbb{R}^d)/\mathbb{R}\right)'$ for $B^{\mathfrak{d}}$. In fact, we can identify $\left(\dot{H}^{\frac{1}{2}}(\mathbb{R}^d)/\mathbb{R}\right)'$ as the Banach space $H_*^{-\frac{1}{2}}(\mathbb{R}^d)$, where

$$H_*^{-\frac{1}{2}}(\mathbb{R}^d) := \left\{ u \in H^{-\frac{1}{2}}(\mathbb{R}^d), \ \exists \underline{u} \in H^{\frac{1}{2}}(\mathbb{R}^d), \ u = |D|\underline{u} \right\},\tag{A.9}$$

endowed with norm $|\cdot|_{H^{-\frac{1}{2}}_*} := \left|\frac{\cdot}{|D|}\right|_{H^{\frac{1}{2}}}$. See for instance [29] for a proof of this result. Notice that $\underline{u} \in H^{\frac{1}{2}}(\mathbb{R}^d)$ is unique since

$$\ker(|D|) = \ker(-\Delta) = \{0\} \text{ in } L^2(\mathbb{R}^d).$$

Then, for $u \in H^{-\frac{1}{2}}_*(\mathbb{R}^d)$, $\frac{u}{|D|}$ makes sense and we can define, for $v \in \dot{H}^{\frac{1}{2}}(\mathbb{R}^d)$

$$(u,v)_{H^{-\frac{1}{2}}_{*}(\mathbb{R}^{d})-\dot{H}^{\frac{1}{2}}(\mathbb{R}^{d})} = \left(|D|v,\frac{u}{|D|}\right)_{H^{-\frac{1}{2}}(\mathbb{R}^{d})-H^{\frac{1}{2}}(\mathbb{R}^{d})}.$$

Notice that if $\varphi \in H^{\frac{1}{2}}(\mathbb{R}^d)$, we have

$$(u,\varphi)_{H_*^{-\frac{1}{2}}-\dot{H}^{1/2}} = (u,\varphi)_{H^{-1/2}-H^{1/2}}.$$

We have also a density result for $H_*^{-\frac{1}{2}}(\mathbb{R}^d)$. For $s \ge 0$, we denote by $H_*^{s-\frac{1}{2}}(\mathbb{R}^d)$ the space $H_*^{-\frac{1}{2}}(\mathbb{R}^d) \cap H^{s-\frac{1}{2}}(\mathbb{R}^d)$. Notice that

$$H^{s-\frac{1}{2}}_{*}(\mathbb{R}^{d}) := \left\{ u \in H^{s-\frac{1}{2}}(\mathbb{R}^{d}), \ \exists \underline{u} \in H^{s+\frac{1}{2}}(\mathbb{R}^{d}), \ u = |D|\underline{u} \right\}.$$
 (A.10)

Lemma A.1.14. Let s > 0. Then, $\mathcal{S}_0(\mathbb{R}^d) := \{ u \in \mathcal{S}(\mathbb{R}^d), \ \widehat{u}(0) = 0 \}$ is dense in $H^{-\frac{1}{2}}_*(\mathbb{R}^d)$ and $H^{s-\frac{1}{2}}(\mathbb{R}^d)$ is dense in $H^{-\frac{1}{2}}_*(\mathbb{R}^d)$.

Proof. If $u \in H_*^{-\frac{1}{2}}(\mathbb{R}^d)$, $u = |D|\underline{u}$ with $\underline{u} \in H^{\frac{1}{2}}(\mathbb{R}^d)$ and we take

$$u_n := \left(\sqrt{\frac{1}{n^2} + |D|^2} - \frac{1}{n}\right) \underline{v}_n ,$$

with $\underline{v}_n \in \mathcal{S}(\mathbb{R}^d)$ converging to \underline{u} in $H^{\frac{1}{2}}(\mathbb{R}^d)$. The result follows from the dominated convergence theorem. Furthermore, if $u \in H_*^{-\frac{1}{2}}(\mathbb{R}^d)$, we take

$$\underline{u}_n := \frac{\underline{u}}{\left(1 + \frac{|D|^2}{n^2}\right)^{\frac{s}{2}}} ,$$

and $u_n = |D|\underline{u}_n \in H^{s-\frac{1}{2}}_*(\mathbb{R}^d)$ converges to u in $H^{-\frac{1}{2}}_*(\mathbb{R}^d)$ thanks to the dominated convergence theorem.

Remark A.1.15. The density of $S_0(\mathbb{R}^d)$ implies that $\int_{\mathbb{R}^d} u$ makes sense and is equal to 0 for all $u \in H_*^{-\frac{1}{2}}(\mathbb{R}^d)$.

In the case where $B \in H^{-\frac{1}{2}}_{*}(\mathbb{R}^{d}), P(\Sigma)\nabla^{\mu}B^{\mathfrak{d}}$ has the same property.

Proposition A.1.16. Let $B \in H^{-\frac{1}{2}}_{*}(\mathbb{R}^{d})$. Then $\frac{P(\Sigma)\nabla^{\mu}B^{\mathfrak{d}}}{|D|} \in H^{1,0}(S)$.

Proof. Let $\varphi \in H^1_{0,surf}(S)$. Then, $\widetilde{\varphi} := \frac{\sqrt{1+|D|^2}}{\sqrt{\frac{1}{n}+|D|^2}}\varphi \in H^1_{0,surf}(S)$, and applying the variational formulation (A.1.11) to $\widetilde{\varphi}$ we have, thanks to the trace result A.1.12,

$$\begin{split} \left| \int_{S} \frac{\sqrt{1+|D|^{2}}}{\sqrt{\frac{1}{n}+|D|^{2}}} P(\Sigma) \nabla^{\mu} B^{\mathfrak{d}} \cdot \nabla^{\mu} \varphi \right| &= \left| \int_{\mathbb{R}^{d}} \frac{\sqrt{1+|D|^{2}}}{\sqrt{\frac{1}{n}+|D|^{2}}} B \cdot \varphi_{|z=-1} \right|, \\ &\leq \left| \frac{\sqrt{1+|D|^{2}}}{\sqrt{1+\sqrt{\mu}|D|}} \frac{B}{\sqrt{\frac{1}{n}+|D|^{2}}} \right|_{L^{2}} \left| \sqrt{1+\sqrt{\mu}|D|} \varphi_{|z=-1} \right|_{L^{2}}, \\ &\leq C \left| \frac{1}{\sqrt{1+\sqrt{\mu}|D|}} \frac{B}{|D|} \right|_{H^{1}} \left| \nabla^{\mu} \varphi \right|_{L^{2}(S)}, \\ &\leq C \left| \frac{1}{\sqrt{1+\sqrt{\mu}|D|}} B \right|_{H^{0}_{*}} \left| \nabla^{\mu} \varphi \right|_{L^{2}(S)}. \end{split}$$

Then, by duality, we obtain that

$$\left|\frac{P(\Sigma)\nabla^{\mu}B^{\mathfrak{d}}}{\sqrt{\frac{1}{n}+|D|^2}}\right|_{H^{1,0}(S)} \leq C \left|\frac{1}{\sqrt{1+\sqrt{\mu}|D|}}B\right|_{H^0_*(\mathbb{R}^d)},$$

and the result follows.

A.1.2 Regularity estimates of the solutions

We need some regularity results for the solutions of the Laplace problems. The first result is Corollary 2.40 in [80]. We recall that $M = C\left(\frac{1}{h_{\min}}, \mu_{\max}, |\zeta|_{H^{t_0+1}(\mathbb{R}^d)}, |b|_{H^{t_0+1}(\mathbb{R}^d)}\right)$.

Theorem A.1.17. Let $t_0 > \frac{d}{2}$ and $0 \le s \le t_0 + \frac{1}{2}$. Let $\zeta, b \in H^{t_0+1}(\mathbb{R}^d)$ be such that Condition (A.1) is satisfied. Then, for all $\psi \in \dot{H}^{s+\frac{1}{2}}(\mathbb{R}^d)$, we have

$$\left|\Lambda^{s} \nabla^{\mu}_{X,z} \psi^{\mathfrak{h}}\right|_{L^{2}(S)} \leq \sqrt{\mu} M \left|\mathfrak{P}\psi\right|_{H^{s}(\mathbb{R}^{d})}.$$

Futhermore, if $s \geq \max(0, 1 - t_0)$, we have

$$\left|\Lambda^{s-1}\partial_z \nabla^{\mu}_{X,z}\psi^{\mathfrak{h}}\right|_{L^2(S)} \leq \sqrt{\mu}M \left|\mathfrak{P}\psi\right|_{H^s(\mathbb{R}^d)}$$

Remark A.1.18. If we summarize the previous theorem in a Sobolev framework, we see that for a domain with a regularity $H^{t_0+1}(\mathbb{R}^d)$, we can expect at most a $H^{t_0+\frac{3}{2}}$ regularity for the Laplace solution. This gain of one half derivative will be crucial for the local wellposedness of the water waves equations.

We can prove the same estimates for $B^{\mathfrak{d}}$. Notice that the next result is an extension of Proposition 4.15 in [71].

Theorem A.1.19. Let $t_0 > \frac{d}{2}$ and $0 \le s \le t_0 + \frac{1}{2}$. Let $\zeta, b \in H^{t_0+1}(\mathbb{R}^d)$ be such that Condition (A.1) is satisfied. Then, for all $B \in H^{s-\frac{1}{2}}(\mathbb{R}^d)$, we have

$$\left|\Lambda^s \nabla^{\mu}_{X,z} B^{\mathfrak{d}}\right|_{L^2(S)} \leq M \left| \frac{1}{\sqrt{1 + \sqrt{\mu} |D|}} B \right|_{H^s(\mathbb{R}^d)}.$$

Furthermore, if $s \ge \max(0, 1 - t_0)$, we have

$$\left|\Lambda^{s-1}\partial_z \nabla^{\mu}_{X,z} B^{\mathfrak{d}}\right|_{L^2(S)} \le M \left|\frac{1}{\sqrt{1+\sqrt{\mu}|D|}} B\right|_{H^s(\mathbb{R}^d)}$$

 $\label{eq:Finally} \textit{ fi } B \in H^{s-\frac{1}{2}}_{*}(\mathbb{R}^{d}), \; \frac{P(\Sigma)\nabla^{\mu}B^{\mathfrak{d}}}{|D|} \in H^{s+1,0}(S) \; \textit{ (and in } H^{s+1,1}(S) \; \textit{ if } s \geq 1-t_{0} \; \textit{)}.$

Remark A.1.20. Notice that, if $\mu > 0$, $\nabla^{\mu}_{X,z}B^{\mathfrak{d}}$ is a half more regular than B, whereas when μ goes to 0, $\partial_z B^{\mathfrak{d}}$ has the same regularity than B. This means that in the shallow water limit (μ goes to zero), we lose the regularity we gained. However, we will see in the next part that we can regain this loss of regularity for the Neumann-Neumann operator and that it will not a problem after all.

Proof. Let $\delta > 0$ and χ be a smooth compactly supported real function that is equal to 1 near 0. We introduce the smoothing operator $\Lambda^s_{\delta} := \chi(\delta\Lambda)\Lambda^s$. We know that $B^{\mathfrak{d}} \in H^1_{0,surf}(S)$. Therefore, using $\Lambda^{2s}_{\delta}B^{\mathfrak{d}}$ a test function, we have

$$\int_{S} \nabla^{\mu} B^{\mathfrak{d}} \cdot P(\Sigma) \nabla^{\mu} \Lambda_{\delta}^{2s} B^{\mathfrak{d}} = - \int_{\mathbb{R}^{d}} B(\Lambda_{\delta}^{2s} B^{\mathfrak{d}})_{|z=-1}$$

Since $P(\Sigma)$ is symmetric, Λ^s_{δ} commutes with ∇^{μ} and is independent of z we obtain that

$$\begin{split} \int_{S} P(\Sigma) \Lambda_{\delta}^{s} \nabla^{\mu} B^{\mathfrak{d}} \cdot \nabla^{\mu} \Lambda_{\delta}^{s} B^{\mathfrak{d}} &= -\int_{S} [\Lambda_{\delta}^{s}, Q(\Sigma)] \nabla^{\mu} B^{\mathfrak{d}} \cdot \nabla^{\mu} \Lambda_{\delta}^{s} B^{\mathfrak{d}} \\ &- \int_{\mathbb{R}^{d}} \frac{\Lambda_{\delta}^{s}}{\sqrt{1 + \sqrt{\mu} |D|}} B\left(\sqrt{1 + \sqrt{\mu} |D|} \Lambda_{\delta}^{s} B^{\mathfrak{d}} \right)_{|z=-1} \end{split}$$

Then by coercivity of $P(\Sigma)$ and trace inequality A.1.12

$$\begin{split} k(\Sigma) \left| \Lambda_{\delta}^{s} \nabla^{\mu} B^{\mathfrak{d}} \right|_{2}^{2} &\leq \left| [\Lambda_{\delta}^{s}, Q(\Sigma)] \nabla^{\mu} B^{\mathfrak{d}} \right|_{2} \left| \Lambda_{\delta}^{s} \nabla^{\mu} B^{\mathfrak{d}} \right|_{2} \\ &+ 2 \left| \Lambda_{\delta}^{s} \nabla^{\mu} B^{\mathfrak{d}} \right|_{2} \left| \frac{\Lambda_{\delta}^{s}}{\sqrt{1 + \sqrt{\mu} |D|}} B \right|_{L^{2}(\mathbb{R}^{d})} \end{split}$$

and

$$k(\Sigma) \left| \Lambda^s_{\delta} \nabla^{\mu} B^{\mathfrak{d}} \right|_2 \leq \left| [\Lambda^s_{\delta}, Q(\Sigma)] \nabla^{\mu} B^{\mathfrak{d}} \right|_2 + 2 \left| \frac{\Lambda^s_{\delta}}{\sqrt{1 + \sqrt{\mu} |D|}} B \right|_{L^2(\mathbb{R}^d)}$$

We have to distinguish two cases.

a) $0 \le s \le t_0$:

The commutator estimate B.3.2 (with $T_0 = t_0$) and Proposition A.1.7 give

$$\begin{split} k(\Sigma) \left| \Lambda_{\delta}^{s} \nabla^{\mu} B^{\mathfrak{d}} \right|_{2} &\leq C \left| Q(\Sigma) \right|_{L^{\infty} H^{t_{0}}} \left| \Lambda_{\delta}^{s-\varepsilon} \nabla^{\mu} B^{\mathfrak{d}} \right|_{2} + 2 \left| \frac{\Lambda_{\delta}^{s}}{\sqrt{1 + \sqrt{\mu} |D|}} B \right|_{L^{2}(\mathbb{R}^{d})} \\ &\leq M \left| \Lambda_{\delta}^{s-\varepsilon} \nabla^{\mu} B^{\mathfrak{d}} \right|_{2} + 2 \left| \frac{\Lambda_{\delta}^{s}}{\sqrt{1 + \sqrt{\mu} |D|}} B \right|_{L^{2}(\mathbb{R}^{d})} \end{split}$$

for some $\varepsilon > 0$ small enough $(\varepsilon < t_0 - \frac{d}{2})$. Using a finite induction on s and taking the limit when δ goes to 0, the first inequality follows. For the second estimate, we only need to give a control of $\partial_z^2 B^{\mathfrak{d}}$. We use Equation (A.6) satisfied by $B^{\mathfrak{d}}$. We express $P(\Sigma)$ as

$$P(\Sigma) := \begin{pmatrix} (1+a(X,z))I_{d\times d} & \mathbf{q}(X,z) \\ \mathbf{q}^t(X,z) & 1+q_{d+1}(X,z) \end{pmatrix}.$$

A simple computing gives

$$(1+q_{d+1})\partial_z^2 B^{\mathfrak{d}} = -\sqrt{\mu}\nabla_X \cdot \left((1+a)\sqrt{\mu}\nabla_X B^{\mathfrak{d}}\right) - \sqrt{\mu}\nabla_X \cdot \left(\partial_z B^{\mathfrak{d}} \mathbf{q}\right) -\sqrt{\mu}\partial_z \mathbf{q} \cdot \nabla_X B^{\mathfrak{d}} - \sqrt{\mu}\partial_z \nabla_X B^{\mathfrak{d}} \cdot \mathbf{q} - \partial_z q_{d+1}\partial_z B^{\mathfrak{d}}.$$

We have $a, \mathbf{q}, q_{d+1} \in L_z^{\infty} H_X^{t_0}(S), \partial_z \mathbf{q}, \partial_z q_{d+1} \in L_z^{\infty} H_X^{t_0-1}(S)$ and $1 + q_{d+1} \ge k(\Sigma)$. Then, since $s \ge 1 - t_0$ and $\nabla_X B^{\mathfrak{d}} \in H^{s,1}(S)$, by the product estimates B.2.2 and B.2.4 (with $T_0 = t_0$), we obtain the result.

b) $t_0 \le s \le t_0 + \frac{1}{2}$:

The commutator estimate B.3.3 (with $T_0 = t_0 + \frac{1}{2}$ and $t_1 > \frac{1}{2}$) and Proposition A.1.7 give

$$k(\Sigma) \left| \Lambda_{\delta}^{s} \nabla^{\mu} B^{\mathfrak{d}} \right|_{L^{2}} \leq M \left[\left| \Lambda_{\delta}^{s+\frac{1}{2}-t_{1}} \nabla^{\mu} B^{\mathfrak{d}} \right|_{L^{2}} + \left| \Lambda_{\delta}^{s-1+\frac{1}{2}-t_{1}} \partial_{z} \nabla^{\mu} B^{\mathfrak{d}} \right|_{L^{2}} + 2 \left| \frac{\Lambda_{\delta}^{s}}{\sqrt{1+\sqrt{\mu}|D|}} B \right|_{L^{2}} \right].$$

We denote by $\varepsilon := \frac{1}{2} - t_1$. We obtain the first inequality for $t_0 \leq s \leq t_0 + \varepsilon$ thanks to the previous case. Furthermore, we saw that

$$(1+q_{d+1})\partial_z^2 B^{\mathfrak{d}} = -\sqrt{\mu}\nabla_X^{\mu} \cdot \left((1+a)\sqrt{\mu}\nabla_X B^{\mathfrak{d}}\right) - \sqrt{\mu}\nabla_X \cdot \left(\partial_z B^{\mathfrak{d}} \mathbf{q}\right) -\sqrt{\mu}\partial_z \mathbf{q} \cdot \nabla_X B^{\mathfrak{d}} - \sqrt{\mu}\partial_z \nabla_X B^{\mathfrak{d}} \cdot \mathbf{q} - \partial_z q_{d+1}\partial_z B^{\mathfrak{d}}.$$

We have $a, \mathbf{q}, q_{d+1} \in L_z^2 H_X^{t_0 + \frac{1}{2}}(S), \partial_z \mathbf{q}, \partial_z q_{d+1} \in L_z^2 H_X^{t_0 - \frac{1}{2}}(S)$ and $1 + q_{d+1} \ge k(\Sigma)$. Then, since $s \ge 1 - t_0$ and $\nabla_X B^{\mathfrak{d}} \in L_z^{\infty} H_X^{s - \frac{1}{2}}(S)$, by the product estimates B.2.2 and B.2.5 (with $T_0 = t_0$), and we obtain the second inequality for $t_0 \le s \le t_0 + \varepsilon$. Using a finite induction, we obtain the first and the second inequality.

Finally, if $B \in H^{s-\frac{1}{2}}_{*}(\mathbb{R}^{d})$, we proceed as in the proof of Proposition A.1.16.

Remark A.1.21. Theorems A.1.17 and A.1.19 show that if $s \ge 1$, which corresponds to $\psi \in \dot{H}^{\frac{3}{2}}(\mathbb{R}^d)$ and $B \in H^{\frac{1}{2}}(\mathbb{R}^d)$, we have $B^{\mathfrak{d}} \in H^2(S)$ and $\psi^{\mathfrak{h}} \in \dot{H}^2(S)$. Hence, we can check that Φ^S solution (A.3) and Φ^B solution (A.4) are in $H^2(\Omega)$.

A.1.3 Shape derivatives of the $B^{\mathfrak{d}}$

In order to get shape derivatives of the Dirichlet-Neumann and the Neumann operator, we need some for the solutions of the Laplace problems. For the Laplace problem (A.5), we refer to Appendix A in [80] and we only study the shape derivative of $B^{\mathfrak{d}}$. Let $t_0 > \frac{d}{2}$, $s \ge 0$ and $B \in H^{s-\frac{1}{2}}(\mathbb{R}^d)$. We denote by Γ the set of functions (ζ, b) in $H^{t_0+1}(\mathbb{R}^d)$ satisfying (A.1). We introduce the map :

$$\mathfrak{D}(B) := \begin{cases} \Gamma \to L_z^2 H_X^{s+1}(S) \\ [\zeta, b] \mapsto B^{\mathfrak{d}}. \end{cases}$$
(A.11)

We begin with a result for $P(\Sigma)$.

Proposition A.1.22. Let $t_0 > \frac{d}{2}$, $\zeta, b \in H^{t_0+1}(\mathbb{R}^d)$ and $0 \le s \le t_0 + \frac{1}{2}$. Then, the map

$$Q(\cdot) := \begin{cases} \Gamma \to L_z^{\infty} H_X^{t_0}(S) \cap L_z^2 H_X^{t_0 + \frac{1}{2}}(S) \\ [\zeta, b] \mapsto Q(\Sigma) \end{cases}$$

is smooth. Furthermore, for all $j \in \mathbb{N}^*$, and $(\mathbf{h}, \mathbf{k}) := (h_1, .., h_j, k_1, .., k_j) \in H^{t_0+1}(\mathbb{R}^d)^{2j}$, we have

$$\begin{split} \left| d^{j} Q_{(\zeta,b)} \cdot (\boldsymbol{h}, \boldsymbol{k}) \right|_{H^{t_{0}+\frac{1}{2},1}} &\leq M \prod_{i \geq 1} |(\varepsilon h_{i}, \beta k_{i})|_{H^{t_{0}+1}} \,, \\ \left| \Lambda^{t_{0}} d^{j} Q_{(\zeta,b)} \cdot (\boldsymbol{h}, \boldsymbol{k}) \right|_{L_{z}^{\infty} L_{X}^{2}(S)} &\leq M \prod_{i \geq 1} |(\varepsilon h_{i}, \beta k_{i})|_{H^{t_{0}+1}} \,, \\ \left| \Lambda^{s} d^{j} Q_{(\zeta,b)} \cdot (\boldsymbol{h}, \boldsymbol{k}) \right|_{L^{2}(S)} &\leq M \left| (\varepsilon h_{1}, \beta k_{1}) \right|_{H^{s+\frac{1}{2}}} \prod_{i \geq 2} |(\varepsilon h_{i}, \beta k_{i})|_{H^{t_{0}+1}} \,, \\ \left| \Lambda^{s-\frac{1}{2}} d^{j} Q_{(\zeta,b)} \cdot (\boldsymbol{h}, \boldsymbol{k}) \right|_{L_{z}^{\infty} L_{X}^{2}(S)} &\leq M \left| (\varepsilon h_{1}, \beta k_{1}) \right|_{H^{s+\frac{1}{2}}} \prod_{i \geq 2} |(\varepsilon h_{i}, \beta k_{i})|_{H^{t_{0}+1}} \,. \end{split}$$

Proof. The fact that Q is smooth is clear. Since,

$$Q(\Sigma) = \begin{pmatrix} \partial_z \sigma I_{d \times d} & -\sqrt{\mu} \nabla_X \sigma \\ -\sqrt{\mu} \nabla_X \sigma^t & \frac{-\partial_z \sigma + \mu |\nabla_X \sigma|^2}{1 + \partial_z \sigma} \end{pmatrix},$$

with $\sigma(X, z) := [\theta(\delta z | D |) \varepsilon \zeta(X) - \theta(\delta(z+1) | D |) \beta b(X)] z + \varepsilon \theta(\delta z | D |) \zeta(X)$. All the terms excepted $Q_{d+1,d+1}$ are linear and then, if we differentiate more than twice $(j \ge 2)$, this terms disappear. We only have to consider

$$Q_{d+1,d+1} := \frac{-\partial_z \sigma + \mu \left| \nabla_X \sigma \right|^2}{1 + \partial_z \sigma}.$$

We notice that if we differentiate j times, we obtain a sum of products of terms of the form $|D|\theta(\delta z|D|)h_i$ and $|D|\theta(\delta(z+1)|D|)k_i$ which are $L_z^{\infty}H_X^{t_0} \cap L_z^2H_X^{t_0+\frac{1}{2}}$ -functions. Then, using Remark A.1.5, we have, for $i \geq 2$, $|D|\theta(\delta z|D|)h_i$ and $|D|\theta(\delta(z+1)|D|)k_i$ in $L_z^{\infty}H_z^{t_0} \cap L_z^2H_z^{t_0+\frac{1}{2}}(\mathbb{R}^d)$ and, by the product estimate B.2.3, product of terms of this form stays in $L_z^{\infty}H_z^{t_0} \cap L_z^2H_z^{t_0+\frac{1}{2}}$. Furthermore, if we take (h_1, k_1) in $H^{s+\frac{1}{2}}(\mathbb{R}^d)$, $|D|\theta(\delta z|D|)h_1$ and $|D|\theta(\delta(z+1)|D|)k_1$ in $L_z^{\infty}H_X^{s-\frac{1}{2}} \cap L_z^2H_x^{s}(\mathbb{R}^d)$, by Products estimates B.2.3 and B.2.2 we obtain the third and the fourth inequality. However, if we take (h_1, k_1) in $H^{t_0+1}(\mathbb{R}^d)$, we obtain the first and the second inequality.

We can now prove shape derivatives estimates for $B^{\mathfrak{d}}$.

Theorem A.1.23. Let $t_0 > \frac{d}{2}$, $\zeta, b \in H^{t_0+1}(\mathbb{R}^d)$, $0 \le s \le t_0 + \frac{1}{2}$ and $B \in H^{s-\frac{1}{2}}(\mathbb{R}^d)$. Then, $\mathfrak{D}(B)$ is smooth. Furthermore, for all $j \in \mathbb{N}^*$, and $(\mathbf{h}, \mathbf{k}) := (h_1, ..., h_j, k_1, ..., k_j) \in H^{t_0+1}(\mathbb{R}^d)^{2j}$, we have

$$\left|\Lambda^{s} \nabla^{\mu}_{X,z} d^{j} \mathfrak{D}(B)_{(\zeta,b)} \cdot (\boldsymbol{h}, \boldsymbol{k})\right|_{L^{2}(S)} \leq M \prod_{i} \left| (\varepsilon h_{i}, \beta k_{i}) \right|_{H^{t_{0}+1}} \left| \frac{1}{\sqrt{1 + \sqrt{\mu}|D|}} B \right|_{H^{s}(\mathbb{R}^{d})}$$

Futhermore, if $s \geq \max(0, 1 - t_0)$, we have

$$\left|\Lambda^{s-1}\partial_z \nabla^{\mu}_{X,z} d^j \mathfrak{D}(B)_{(\zeta,b)} \cdot (\boldsymbol{h}, \boldsymbol{k})\right|_{L^2(S)} \leq M \prod_i |(\varepsilon h_i, \beta k_i)|_{H^{t_0+1}} \left| \frac{1}{\sqrt{1+\sqrt{\mu}|D|}} B \right|_{H^s(\mathbb{R}^d)}.$$

Proof. We recall that $B^{\mathfrak{d}}$ satisfies the variational formulation of Definition A.1.11. We differentiate formally with respect to (\mathbf{h}, \mathbf{k}) this formulation and we obtain, for all $\varphi \in H^1_{0,surf}(S)$,

$$\int_{S} \sum_{\substack{j_1+j_2=j\\I_1\sqcup I_2=[1,j]}} C_{j_1,j_2,I_1,I_2} \ d^{j_1} P_{(\zeta,b)}.(\mathbf{h},\mathbf{k})_{I_1} \nabla^{\mu} d^{j_2} \mathfrak{D}(B)_{(\zeta,b)}.(\mathbf{h},\mathbf{k})_{I_2} \cdot \nabla^{\mu} \varphi \ dx = 0,$$

where C_{j_1,j_2,I_1,I_2} are constants. Notice that the coordinates of $(\mathbf{h}, \mathbf{k})_{I_1}$ and $(\mathbf{h}, \mathbf{k})_{I_2}$ form a permutation of the coordinates of (\mathbf{h}, \mathbf{k}) . We denote by \mathbf{g} and v

$$\mathbf{g} := \sum_{\substack{j_1+j_2=j, \ j_2 < j \\ I_1 \sqcup I_2 = [1,j]}} C_{j_1,j_2,I_1,I_2} \ d^{j_1} Q_{(\zeta,b)} \cdot (\mathbf{h}, \mathbf{k})_{I_1} \nabla^{\mu} d^{j_2} \mathfrak{D}(B)_{(\zeta,b)} \cdot (\mathbf{h}, \mathbf{k})_{I_2},$$

and

$$v := d^j \mathfrak{D}(B)_{(\zeta,b)} \cdot (\mathbf{h}, \mathbf{k}).$$

Notice also that $C_{0,j,\emptyset,[1,j]} = 1$. Then we have

$$\int_{S} P(\Sigma) \nabla^{\mu} v \cdot \nabla^{\mu} \varphi \, dx = -\int_{S} \mathbf{g} \cdot \nabla^{\mu} \varphi \, dx.$$

Therefore, we can use the same method that Theorem A.1.19. We take $\varphi = \Lambda_{\delta}^{2s} B^{\mathfrak{d}}$ and a simple computing gives

$$k(\Sigma) \left| \Lambda^s_{\delta} \nabla^{\mu} v \right|_{L^2} \le \left| \left[\Lambda^s_{\delta}, Q(\Sigma) \right] v \right|_2 + \left| \Lambda^s_{\delta} \mathbf{g} \right|_{L^2(S)},$$

We have to distinguish two cases.

a) $0 \le s \le t_0$:

The commutator estimate B.3.2 (with $T_0 = t_0$), Proposition A.1.7 and a finite induction on s gives

$$|\Lambda^s \nabla^\mu v|_{L^2(S)} \le M |\Lambda^s \mathbf{g}|_{L^2(S)}.$$

Furthermore, using the product estimate B.2.2 we obtain

$$|\Lambda^{s} \mathbf{g}|_{L^{2}(S)} \leq C \sum_{\substack{j_{1}+j_{2}=j, \ j_{2} < j \\ I_{1} \sqcup I_{2}=[1,j]}} |d^{j_{1}}Q_{(\zeta,b)}.(\mathbf{h},\mathbf{k})_{I_{1}}|_{L^{\infty}_{z}H^{t_{0}}_{X}} |\Lambda^{s} \nabla^{\mu} d^{j_{2}}\mathfrak{D}(B)_{(\zeta,b)}.(\mathbf{h},\mathbf{k})_{I_{2}}|_{L^{2}(S)}.$$

Using the first inequality of Proposition A.1.22, Theorem A.1.19 and an induction on j, we obtain the first inequality. Furthermore, in order to obtain the second inequality, we only need to estimate $\partial_z^2 v$. We know that v satisfies,

$$\nabla^{\mu}_{X,z} \cdot P(\Sigma) \nabla^{\mu}_{X,z} v = \nabla^{\mu}_{X,z} \cdot \mathbf{g}.$$

Using the same notation than Theorem A.1.19, we have

$$(1+q_{d+1})\partial_z^2 v = -\sqrt{\mu}\nabla_X^{\mu} \cdot (a\sqrt{\mu}\nabla_X v) - \sqrt{\mu}\nabla_X \cdot (\partial_z v \mathbf{q}) -\sqrt{\mu}\partial_z \mathbf{q} \cdot \nabla_X v - \sqrt{\mu}\partial_z \nabla_X v \cdot \mathbf{q} - \partial_z q_{d+1}\partial_z v - \nabla_{X,z} \cdot \mathbf{g},$$

and $\nabla_{X,z} \cdot \mathbf{g} \in L_z^2 H_X^{s-1}(\mathbb{R}^d)$ thanks to the first inequality. Then, the second inequality follows (same estimate as Theorem A.1.19).

b)
$$s = t_0 + \frac{1}{2}$$
:

The commutator estimate B.3.3 (with $T_0 = t_0 + \frac{1}{2}$ and $t_1 > \frac{1}{2}$) and Proposition A.1.7 give

$$k(\Sigma) \left| \Lambda_{\delta}^{t_0 + \frac{1}{2}} \nabla^{\mu} v \right|_{L^2} \leq M \left[\left| \Lambda_{\delta}^{t_0 + 1 - t_1} \nabla^{\mu} v \right|_{L^2} + \left| \Lambda_{\delta}^{t_0 - t_1} \partial_z \nabla^{\mu} v \right|_{L^{\frac{1}{2}}} \left| \Lambda^{t_0 + \frac{1}{2}} \mathbf{g} \right|_{L^2} \right].$$

Furthermore, by the product estimates B.2.3,

$$\begin{split} \left| \Lambda^{t_0 + \frac{1}{2}} \mathbf{g} \right|_{L^2(S)} &\leq C \sum_{\substack{j_1 + j_2 = j, \ j_2 < j \\ I_1 \sqcup I_2 = [1,j]}} \left| d^{j_1} Q_{(\zeta,b)} \cdot (\mathbf{h}, \mathbf{k})_{I_1} \right|_{L_z^\infty H_X^{t_0}} \left| \Lambda^{t_0 + \frac{1}{2}} \nabla^{\mu} d^{j_2} \mathfrak{D}(B)_{(\zeta,b)} \cdot (\mathbf{h}, \mathbf{k})_{I_2} \right|_{L^2(S)} \\ &+ \left| \Lambda^{t_0 + \frac{1}{2}} d^{j_1} Q_{(\zeta,b)} \cdot (\mathbf{h}, \mathbf{k})_{I_1} \right|_{L^2} \left| \nabla^{\mu} d^{j_2} \mathfrak{D}(B)_{(\zeta,b)} \cdot (\mathbf{h}, \mathbf{k})_{I_2} \right|_{L_z^\infty H_X^{t_0}} \end{split}$$

and hence

$$\begin{split} \left| \Lambda^{t_0 + \frac{1}{2}} \mathbf{g} \right|_{L^2(S)} &\leq C \sum_{\substack{j_1 + j_2 = j, \ j_2 < j \\ I_1 \sqcup I_2 = [1,j]}} \left| d^{j_1} Q_{(\zeta,b)} \cdot (\mathbf{h}, \mathbf{k})_{I_1} \right|_{L_x^{\infty} H_X^{t_0}} \left| \Lambda^{t_0 + \frac{1}{2}} \nabla^{\mu} d^{j_2} \mathfrak{D}(B)_{(\zeta,b)} \cdot (\mathbf{h}, \mathbf{k})_{I_2} \right|_{L^2(S)} \\ &+ \left| \Lambda^{t_0 + \frac{1}{2}} d^{j_1} Q_{(\zeta,b)} \cdot (\mathbf{h}, \mathbf{k})_{I_1} \right|_{L^2} \left| \nabla^{\mu} d^{j_2} \mathfrak{D}(B)_{(\zeta,b)} \cdot (\mathbf{h}, \mathbf{k})_{I_2} \right|_{H_X^{t_0 + \frac{1}{2}, 1}}. \end{split}$$

Then, the result follows using small adaptations of the proof of Theorem A.1.19.

c) $0 < s < t_0 + \frac{1}{2}$:

The result follows from the previous case by interpolations.

The following theorem gives also shape derivatives estimates for $B^{\mathfrak{d}}$.

Theorem A.1.24. Let $t_0 > \frac{d}{2}$, $\zeta, b \in H^{t_0+1}(\mathbb{R}^d)$, $0 \le s \le t_0$ and $B \in H^{t_0}(\mathbb{R}^d)$. Then, $\mathfrak{D}(B)$ is smooth. Furthermore, for all $j \in \mathbb{N}^*$, and $(\mathbf{h}, \mathbf{k}) := (h_1, ..., h_j, k_1, ..., k_j) \in H^{t_0+1}(\mathbb{R}^d)^{2j}$, we have

$$\left|\Lambda^{s} \nabla^{\mu}_{X,z} d^{j} \mathfrak{D}(B)_{(\zeta,b)} \cdot (\boldsymbol{h}, \boldsymbol{k})\right|_{L^{2}} \leq M |(\varepsilon h_{1}, \beta k_{1})|_{H^{s+\frac{1}{2}}} \prod_{i \geq 2} |(\varepsilon h_{i}, \beta k_{i})|_{H^{t_{0}+1}} \left| \frac{1}{\sqrt{1+\sqrt{\mu}|D|}} B \right|_{H^{t_{0}+\frac{1}{2}}}.$$

Futhermore, if $s \ge \max(0, 1 - t_0)$, we have

$$\left|\Lambda^{s-1}\partial_z \nabla^{\mu}_{X,z} d^{j} \mathfrak{D}(B)_{(\zeta,b)} \cdot (\mathbf{h}, \mathbf{k})\right|_{L^2} \leq \sqrt{\mu} M |(\varepsilon h_1, \beta k_1)|_{H^{s+\frac{1}{2}}} \prod_{i \geq 2} |(\varepsilon h_i, \beta k_i)|_{H^{t_0+1}} \left| \frac{1}{\sqrt{1+\sqrt{\mu}|D|}} B \right|_{H^{t_0+\frac{1}{2}}} \cdot \frac{1}{|\mathbf{h}|^{t_0+\frac{1}{2}}} \| \mathbf{h}\|_{H^{t_0+\frac{1}{2}}} \cdot \frac{1}{|\mathbf{h}|^{t_0+\frac{1}{2}}} \| \mathbf{h}\|_{H^{t_0+\frac{1}{2}}} \cdot \frac{1}{|\mathbf{h}|^{t_0+\frac{1}{2}}} \| \mathbf{h}\|_{H^{t_0+\frac{1}{2}}} \cdot \frac{1}{|\mathbf{h}|^{t_0+\frac{1}{2}}} \| \mathbf{h}\|_{H^{t_0+\frac{1}{2}}} \| \mathbf{h}\|_{H^{t_0+\frac{1}{2}}} \cdot \frac{1}{|\mathbf{h}|^{t_0+\frac{1}{2}}} \| \mathbf{h}\|_{H^{t_0+\frac{1}{2}}} \| \mathbf{h}\|_{H^{t$$

Proof. We use the same notation that the previous theorem. By the product estimate B.2.2, we have

$$\begin{split} |\Lambda^{s} \mathbf{g}|_{L^{2}(S)} &\leq C \sum_{\substack{j_{1}+j_{2}=j, \ j_{2} < j \\ I_{1} \sqcup I_{2}=[1,j] \\ (1,1) \in I_{1} \\ + C \sum_{\substack{j_{1}+j_{2}=j, \ j_{2} < j \\ I_{1} \sqcup I_{2}=[1,j] \\ (1,1) \in I_{2} \\ (1,1) \in I_{2} \\ (1,1) \in I_{2} \\ \end{split}} \left| d^{j_{1}}Q_{(\zeta,b)}.(\mathbf{h},\mathbf{k})_{I_{1}} \right|_{L^{\infty}_{z}H^{t_{0}}_{X}} \left| \Lambda^{s} \nabla^{\mu} d^{j_{2}}\mathfrak{D}(B)_{(\zeta,b)}.(\mathbf{h},\mathbf{k})_{I_{2}} \right|_{L^{2}(S)}. \end{split}$$

Since $|\cdot|_{L_z^{\infty}H_X^{t_0}} \leq C|\cdot|_{H^{t_0+\frac{1}{2},1}}$, we can control the first sum with the estimate of the theorem A.1.23 and the result follow by the second inequality of Proposition A.1.22 and a finite induction. \Box

A.2 The Dirichlet-Neumann and the Neumann-Neumann operators

We refer to Chapter 3 of [80] for more details about the Dirichlet-Neumann operator. We extend the result of Section 3 in [71]. We define the Dirichlet-Neumann operator $G_{\mu}[\varepsilon\zeta,\beta b]$ by

$$G_{\mu}[\varepsilon\zeta,\beta b](\psi) := \sqrt{1+\varepsilon^2 |\nabla\zeta|^2} \partial_{\mathbf{n}} \Phi^S_{|z=\varepsilon\zeta} = -\mu\varepsilon\nabla\zeta\cdot\nabla_X \Phi^S_{|z=\varepsilon\zeta} + \partial_z \Phi^S_{|z=\varepsilon\zeta} , \qquad (A.12)$$

where Φ^S satisfies (A.3). We define also the Neumann-Neumann operator $G^{NN}_{\mu}[\varepsilon\zeta,\beta b]$ by

$$G^{NN}_{\mu}[\varepsilon\zeta,\beta b](B) := \sqrt{1+\varepsilon^2 |\nabla\zeta|^2} \partial_{\mathbf{n}} \Phi^B_{|z=\varepsilon\zeta} = -\mu \nabla(\varepsilon\zeta) \cdot \nabla_X \Phi^B_{|z=\varepsilon\zeta} + \partial_z \Phi^B_{|z=\varepsilon\zeta} , \quad (A.13)$$

where Φ^B satisfies (A.4).

Remark A.2.1. . Notice that

$$\frac{1}{\mu}G_{\mu}[0,0](\psi) = |D|^2 \frac{\tanh(\sqrt{\mu}|D|)}{\sqrt{\mu}|D|} \psi \text{ and } G_{\mu}^{NN}[0,0](B) = \frac{1}{\cosh(\sqrt{\mu}|D|)}B$$

A.2.1 Main properties

We can express the Dirichlet-Neumann and the Neumann-Neumann operator with the formalism of the previous section. For $\psi \in \dot{H}^{\frac{3}{2}}(\mathbb{R}^d)$ and $B \in H^{\frac{1}{2}}(\mathbb{R}^d)$ we have

$$G_{\mu}[\varepsilon\zeta,\beta b](\psi) = \left(\mathbf{e}_{z} \cdot P(\Sigma)\nabla^{\mu}\psi^{\mathfrak{h}}\right)_{|z=0}, \qquad (A.14)$$

and

$$G^{NN}_{\mu}[\varepsilon\zeta,\beta b](B) = \left(\mathbf{e}_{z} \cdot P(\Sigma)\nabla^{\mu}B^{\mathfrak{d}}\right)_{|z=0}.$$
(A.15)

The following two results is a summarize of Paragraph **3.1** in [80]. It gives some basic properties of the Dirichlet-Neumann operator. The first result is a symmetry property and a dual formulation of the Dirichlet-Neumann operator.
Proposition A.2.2. Let $\psi \in \dot{H}^{\frac{1}{2}}(\mathbb{R}^d)$ and $\zeta, b \in H^{t_0+1}(\mathbb{R}^d)$ such that (A.1) is satisfied. $G_{\mu}[\varepsilon\zeta,\beta b](\cdot)$ can be extended to $\dot{H}^{\frac{1}{2}}(\mathbb{R}^d)$ with the dual formulation

$$G_{\mu}[\varepsilon\zeta,\beta b](\psi) = \begin{cases} \dot{H}^{\frac{1}{2}}(\mathbb{R}^{d})/\mathbb{R} \longrightarrow \mathbb{R} \\ \varphi \longmapsto \int_{S} \nabla^{\mu}_{X,z} \psi^{\mathfrak{h}} \cdot P(\Sigma) \nabla^{\mu}_{X,z} \varphi^{\dagger} dx, \end{cases}$$
(A.16)

where φ^{\dagger} is defined in (A.8). Furthermore, $G_{\mu}[\varepsilon\zeta,\beta b](\cdot)$ is a symmetric positive operator and, for all ψ_1, ψ_2, ψ in $\dot{H}^{\frac{1}{2}}(\mathbb{R}^d)$,

$$\begin{aligned} (G_{\mu}[\varepsilon\zeta,\beta b](\psi_{1}),\psi_{2})_{(\dot{H}^{1/2})'-\dot{H}^{1/2}} &= (G_{\mu}[\varepsilon\zeta,\beta b](\psi_{2}),\psi_{1})_{(\dot{H}^{1/2})'-\dot{H}^{1/2}},\\ |(G_{\mu}[\varepsilon\zeta,\beta b](\psi_{1}),\psi_{2})| &\leq \sqrt{(G_{\mu}[\varepsilon\zeta,\beta b](\psi_{1}),\psi_{1})} \sqrt{(G_{\mu}[\varepsilon\zeta,\beta b](\psi_{2}),\psi_{2})},\\ (G_{\mu}[\varepsilon\zeta,\beta b](\psi),\psi) &\geq 0,\\ G_{\mu}[\varepsilon\zeta,\beta b](\psi) \in H^{-\frac{1}{2}}(\mathbb{R}^{d}). \end{aligned}$$

Finally, if $G_{\mu}[\varepsilon\zeta,\beta b](\psi_1), \ G_{\mu}[\varepsilon\zeta,\beta b](\psi_2) \in L^2(\mathbb{R}^d)$, then

$$(G_{\mu}[\varepsilon\zeta,\beta b](\psi_1),\psi_2)_{(\dot{H}^{1/2})'-\dot{H}^{1/2}} = (G_{\mu}[\varepsilon\zeta,\beta b](\psi_1),\psi_2)_2.$$

Remark A.2.3. Notice that, instead of φ^{\dagger} , we can take $\varphi^{\mathfrak{h}}$ in the dual formulation.

We see that $G_{\mu}[\varepsilon\zeta,\beta b]$ maps continuously $\dot{H}^{\frac{1}{2}}(\mathbb{R}^d)$ into $H^{-\frac{1}{2}}_{*}(\mathbb{R}^d)$. Furthermore, we can also compute explicitly an operator whose norm is equivalent to the square root of the Dirichlet-Neumann operator. We recall that $\mathfrak{P} := \frac{|D|}{\sqrt{1+\sqrt{\mu}|D|}}$ and that

$$M = C\left(\frac{1}{h_{\min}}, \mu_{\max}, |\zeta|_{H^{t_0+1}}, |b|_{H^{t_0+1}}\right)$$

Proposition A.2.4. Let $t_0 > \frac{d}{2}$ and $\zeta, b \in H^{t_0+1}(\mathbb{R}^d)$ such that (A.1) is satisfied. Then, for all $\psi \in \dot{H}^{\frac{1}{2}}(\mathbb{R}^d)$, we have

$$\left(\frac{1}{\mu}G_{\mu}[\varepsilon\zeta,\beta b](\psi),\psi\right) \leq M \left|\mathfrak{P}\psi\right|_{2}^{2} \ and \ \left|\mathfrak{P}\psi\right|_{2}^{2} \leq M\left(\frac{1}{\mu}G_{\mu}[\varepsilon\zeta,\beta b](\psi),\psi\right).$$

Conversely the Neumann-Neumann operator is not symmetric but it does not have a loss of regularity. We need an extension result in $\dot{H}^{\frac{1}{2}}(\mathbb{R}^d)$ in order to give a dual formulation of the Neumann-Neumann operator.

Definition A.2.5. Let $\varphi \in H^{-\frac{1}{2}}(\mathbb{R}^d)$. We define $\varphi^{\#}$ as

$$\varphi^{\#} = \frac{\sinh([z+1]\sqrt{\mu}|D|)}{\sinh(\sqrt{\mu}|D|)}\varphi$$

Remark A.2.6. $\varphi^{\#}$ satisfies weakly

$$\begin{cases} \Delta^{\mu} \varphi^{\#} = 0 \ in \ S, \\ \varphi^{\#}_{|z=-1} = \varphi \ ; \ \varphi^{\#}_{|z=0} = 0 \end{cases}$$

We can prove easily regularity results for $\varphi^{\#}$ similar to $\varphi^{\mathfrak{h}}$.

Proposition A.2.7. Let $s \ge 0$ and $\varphi \in H^{s-\frac{1}{2}}(\mathbb{R}^d)$. Then,

$$\begin{split} \left| \Lambda^{s-1} \nabla^{\mu}_{X,z} \varphi^{\#} \right|_{L^{2}(S)} &\leq C(\mu_{\max}) \left| \sqrt{1 + \sqrt{\mu} |D|} \varphi \right|_{H^{s-1}(\mathbb{R}^{d})}, \\ \left| \Lambda^{s-2} \partial_{z} \nabla^{\mu}_{X,z} \varphi^{\#} \right|_{L^{2}(S)} &\leq C(\mu_{\max}) \mu^{\frac{1}{4}} \left| \sqrt{1 + \sqrt{\mu} |D|} \varphi \right|_{H^{s-1}(\mathbb{R}^{d})} \end{split}$$

We can now give a dual formulation of the Neumann-Neumann operator. We introduce the Dirichlet-Dirichlet operator, for $\psi \in \dot{H}^{\frac{1}{2}}(\mathbb{R}^d)$,

$$G^{DD}_{\mu}[\varepsilon\zeta,\beta b](\psi) := (\psi^{\mathfrak{h}})_{|z=-1}.$$
(A.17)

The following result extends Proposition 3.3 in [71].

Proposition A.2.8. Let $t_0 > \frac{d}{2}$, $B \in H_*^{-\frac{1}{2}}(\mathbb{R}^d)$ and $\zeta, b \in H^{t_0+1}(\mathbb{R}^d)$ such that (A.1) is satisfied. $G_{\mu}^{NN}[\varepsilon\zeta,\beta b](\cdot)$ can be extended to $H_*^{-\frac{1}{2}}(\mathbb{R}^d)$ with the dual formulation

$$G^{NN}_{\mu}[\varepsilon\zeta,\beta b](B) = \begin{cases} \dot{H}^{\frac{1}{2}}(\mathbb{R}^{d})/\mathbb{R} \longrightarrow \mathbb{R} \\ \psi \longmapsto \left(\frac{P(\Sigma)\nabla^{\mu}_{X,z}B^{\mathfrak{d}}}{|D|},\nabla^{\mu}_{X,z}(|D|\psi)^{\#}\right)_{H^{1,0}-H^{-1,0}}. \end{cases}$$
(A.18)

Furthermore, the adjoint of $G^{NN}_{\mu}[\varepsilon\zeta,\beta b]$ is $G^{DD}_{\mu}[\varepsilon\zeta,\beta b]$. For all $B \in H^{-\frac{1}{2}}_{*}(\mathbb{R}^{d})$ and $\psi \in \dot{H}^{\frac{1}{2}}(\mathbb{R}^{d})$,

$$\left(G_{\mu}^{NN}[\varepsilon\zeta,\beta b](B),\psi\right)_{H_{*}^{-\frac{1}{2}}-\dot{H}^{\frac{1}{2}}(\mathbb{R}^{d})} = \left(B,G_{\mu}^{DD}[\varepsilon\zeta,\beta b](\psi)\right)_{H_{*}^{-1/2}-\dot{H}^{1/2}}.$$

Remark A.2.9. We see that $G^{NN}_{\mu}[\varepsilon\zeta,\beta b](\cdot)$ maps continuously $H^{-\frac{1}{2}}_{*}(\mathbb{R}^{d})$ into itself. Furthermore, if $B \in H^{-\frac{1}{2}}(\mathbb{R}^{d})$ and $\varphi \in H^{\frac{1}{2}}(\mathbb{R}^{d})$, we have

$$\left(G^{NN}_{\mu}[\varepsilon\zeta,\beta b](B),\varphi\right)_{H^{-\frac{1}{2}}-H^{\frac{1}{2}}} = \int_{S} P(\Sigma)\nabla^{\mu}_{X,z}B^{\mathfrak{d}} \cdot \nabla^{\mu}_{X,z}\varphi^{\#}dx$$

Hence, $G^{NN}_{\mu}[\varepsilon\zeta,\beta b](\cdot)$ maps continuously $H^{-\frac{1}{2}}(\mathbb{R}^d)$ into itself.

Remark A.2.10. Notice that the formulation (A.18) makes sense thanks to Proposition A.1.16. Proof. We take $B \in H^{\frac{1}{2}}_{*}(\mathbb{R}^{d})$ and $\psi \in H^{\frac{3}{2}}(\mathbb{R}^{d})$. We compute

$$\begin{split} \int_{\mathbb{R}^d} G^{NN}_{\mu} [\varepsilon\zeta, \beta b](B) \ \psi \ dx &= \int_{\mathbb{R}^d} \left(\mathbf{e}_z \cdot P(\Sigma) \nabla^{\mu} B^{\mathfrak{d}} \right)_{|z=0} \left(\psi^{\#} \right)_{|z=0} dx \\ &- \int_{\mathbb{R}^d} \left(\mathbf{e}_z \cdot P(\Sigma) \nabla^{\mu} B^{\mathfrak{d}} \right)_{|z=-1} \left(\psi^{\#} \right)_{|z=-1} dx, \\ &= \int_S P(\Sigma) \nabla^{\mu}_{X,z} B^{\mathfrak{d}} \cdot \nabla^{\mu}_{X,z} \psi^{\#} dx + \int_S \left(\nabla^{\mu}_{X,z} \cdot P(\Sigma) \nabla^{\mu}_{X,z} B^{\mathfrak{d}} \right) \varphi^{\#} dx, \\ &= \int_S P(\Sigma) \nabla^{\mu}_{X,z} B^{\mathfrak{d}} \cdot \nabla^{\mu}_{X,z} \psi^{\#} dx, \\ &= \int_S \frac{P(\Sigma) \nabla^{\mu}_{X,z} B^{\mathfrak{d}}}{|D|} \cdot \nabla^{\mu}_{X,z} (|D|\psi)^{\#} dx, \end{split}$$

since $(\cdot)^{\#}$ and D commute. Furthermore,

$$\begin{split} \int_{\mathbb{R}^d} G^{NN}_{\mu} [\varepsilon\zeta, \beta b](B) \ \psi \ -B \ G^{DD}_{\mu} [\varepsilon\zeta, \beta b](\psi) &= \int_{\mathbb{R}^d} \left(\mathbf{e}_z \cdot P(\Sigma) \nabla^{\mu} B^{\mathfrak{d}} \right)_{|z=0} \left(\psi^{\mathfrak{h}} \right)_{|z=0} dx \\ &- \int_{\mathbb{R}^d} \left(\mathbf{e}_z \cdot P(\Sigma) \nabla^{\mu} B^{\mathfrak{d}} \right)_{|z=-1} \left(\psi^{\mathfrak{h}} \right)_{|z=-1} dx, \\ &= \int_{S} \left(\nabla^{\mu}_{X,z} \cdot P(\Sigma) \nabla^{\mu}_{X,z} B^{\mathfrak{d}} \right) \psi^{\mathfrak{h}} dx \\ &+ \int_{S} P(\Sigma) \nabla^{\mu}_{X,z} B^{\mathfrak{d}} \cdot \nabla^{\mu}_{X,z} \psi^{\mathfrak{h}} dx, \\ &= \int_{S} P(\Sigma) \nabla^{\mu}_{X,z} B^{\mathfrak{d}} \cdot \nabla^{\mu}_{X,z} \psi^{\mathfrak{h}} dx, \end{split}$$

and

$$\begin{split} \int_{S} P(\Sigma) \nabla^{\mu}_{X,z} B^{\mathfrak{d}} \cdot \nabla^{\mu}_{X,z} \psi^{\mathfrak{h}} dx &= \int_{S} B^{\mathfrak{d}} \left(\nabla^{\mu}_{X,z} \cdot P(\Sigma) \nabla^{\mu}_{X,z} \psi^{\mathfrak{h}} \right) dx \\ &+ \int_{\mathbb{R}^{d}} \left(\mathbf{e}_{z} \cdot P(\Sigma) \nabla^{\mu} \psi^{\mathfrak{h}} \right)_{|z=0} \left(B^{\mathfrak{d}} \right)_{|z=0} dx \\ &- \int_{\mathbb{R}^{d}} \left(\mathbf{e}_{z} \cdot P(\Sigma) \nabla^{\mu} \psi^{\mathfrak{h}} \right)_{|z=-1} \left(B^{\mathfrak{d}} \right)_{|z=-1} dx, \\ &= 0. \end{split}$$

Finally, thanks to Proposition A.1.16 and Proposition A.2.7, we have

$$\begin{split} \left| \left(\frac{P(\Sigma) \nabla_{X,z}^{\mu} B^{\mathfrak{d}}}{|D|}, \nabla_{X,z}^{\mu} (|D|\psi)^{\#} \right)_{H^{1,0} - H^{-1,0}} \right| &\leq \left| \frac{P(\Sigma) \nabla_{X,z}^{\mu} B^{\mathfrak{d}}}{|D|} \right|_{H^{1,0}} \left| \nabla_{X,z}^{\mu} (|D|\psi)^{\#} \right|_{H^{-1,0}}, \\ &\leq C \left| \frac{1}{\sqrt{1 + \sqrt{\mu}|D|}} B \right|_{H^{\mathfrak{d}}_{\mathfrak{s}}} \left| \sqrt{1 + \sqrt{\mu}|D|} |D|\psi \right|_{H^{-1}}, \\ &\leq C(\mu_{\max}) \left| \frac{1}{\sqrt{1 + \sqrt{\mu}|D|}} B \right|_{H^{\mathfrak{d}}_{\mathfrak{s}}} \left| |\mathfrak{P}\psi|_{L^{2}}. \end{split}$$

Since $H^{\frac{3}{2}}(\mathbb{R}^d)$ is dense in $\dot{H}^{\frac{1}{2}}(\mathbb{R}^d)/\mathbb{R}$ (Proposition A.1.1) and $H^{\frac{1}{2}}_*(\mathbb{R}^d)$ is dense in $H^{-\frac{1}{2}}_*(\mathbb{R}^d)$ (Lemma A.1.14), we get the result by density.

In order to study shape derivatives of Dirichlet-Neumann and Neumann-Neumann operators, we have to introduce the Neumann-Dirichlet operator. For $B \in H^{-\frac{1}{2}}(\mathbb{R}^d)$, we define

$$G^{ND}_{\mu}[\varepsilon\zeta,\beta b](B) := \left(B^{\mathfrak{d}}\right)_{|z=-1}.$$
(A.19)

The following result is a symmetry property and a dual formulation of the Neumann-Dirichlet operator which extends Proposition 3.3 in [71].

Proposition A.2.11. Let $B \in H^{-\frac{1}{2}}(\mathbb{R}^d)$ and $\zeta, b \in H^{t_0+1}(\mathbb{R}^d)$ such that (A.1) is satisfied. $G^{ND}_{\mu}[\varepsilon\zeta,\beta b](B)$ can be view as

$$G^{ND}_{\mu}[\varepsilon\zeta,\beta b](B) = \begin{cases} H^{-\frac{1}{2}}(\mathbb{R}^d) \longrightarrow \mathbb{R} \\ C \longmapsto -\int_{S} P(\Sigma)\nabla^{\mu}_{X,z}B^{\mathfrak{d}} \cdot \nabla^{\mu}_{X,z}C^{\mathfrak{d}}dx. \end{cases}$$
(A.20)

Furthermore, $G^{ND}_{\mu}[\varepsilon\zeta,\beta b]$ is a negative operator and, for all B_1 , B_2 , B in $H^{-\frac{1}{2}}(\mathbb{R}^d)$,

$$\begin{split} \left(G^{ND}_{\mu}[\varepsilon\zeta,\beta b](B_1),B_2\right)_{(H^{-1/2})'-H^{1/2}} &= \left(G^{ND}_{\mu}[\varepsilon\zeta,\beta b](B_2),B_1\right)_{(H^{-1/2})'-H^{1/2}},\\ \left|\left(G^{ND}_{\mu}[\varepsilon\zeta,\beta b](B_1),B_2\right)\right| &\leq \sqrt{\left(G^{ND}_{\mu}[\varepsilon\zeta,\beta b](B_1),B_1\right)} \sqrt{\left(G^{ND}_{\mu}[\varepsilon\zeta,\beta b](B_2),B_2\right)} \\ \left(G^{ND}_{\mu}[\varepsilon\zeta,\beta b](B),B\right) &\geq 0. \end{split}$$

Proof. We take $B \in H^{\frac{1}{2}}(\mathbb{R}^d)$. Since $B_1^{\mathfrak{d}}$ satisfies (A.6), we have

$$\begin{split} \left(G^{ND}_{\mu}[\varepsilon\zeta,\beta b](B_{1}),B_{2}\right) &= \int_{\mathbb{R}^{d}} \left(B^{\mathfrak{d}}_{1}\right)_{|z=-1} \left(\partial_{n}B^{\mathfrak{d}}_{2}\right)_{|z=-1} - \int_{\mathbb{R}^{d}} \left(B^{\mathfrak{d}}_{1}\right)_{|z=0} \left(\partial_{n}B^{\mathfrak{d}}_{2}\right)_{|z=0},\\ &= -\int_{S} P(\Sigma)\nabla^{\mu}B^{\mathfrak{d}}_{1} \cdot \nabla^{\mu}B^{\mathfrak{d}}_{2} dx. \end{split}$$

This expression is symmetric and $P(\Sigma)$ is coercive so it is clear that $G^{ND}_{\mu}[\varepsilon\zeta,\beta b](\cdot)$ is a symmetric positive operator. Furthermore, we have

$$\begin{aligned} \left(G^{ND}_{\mu}[\varepsilon\zeta,\beta b](B_{1}),B_{2}\right) &= -\int_{S}P(\Sigma)\nabla^{\mu}B^{\mathfrak{d}}_{1}\cdot B^{\mathfrak{d}}_{2}dx, \\ &= -\int_{S}P(\Sigma)^{1/2}\nabla^{\mu}B^{\mathfrak{d}}_{1}\cdot P(\Sigma)^{1/2}\nabla^{\mu}B^{\mathfrak{d}}_{2}dx, \\ &\leq \left|P(\Sigma)^{1/2}\nabla^{\mu}B^{\mathfrak{d}}_{1}\right|_{L^{2}}\left|P(\Sigma)^{1/2}\nabla^{\mu}B^{\mathfrak{d}}_{2}\right|_{L^{2}}. \end{aligned}$$

The fact that

$$|P(\Sigma)^{1/2}\nabla^{\mu}B_1^{\mathfrak{d}}|_{L^2}^2 = -\left(G_{\mu}^{ND}[\varepsilon\zeta,\beta b](B_1),B_1\right)$$

implies the second inequality.

A.2.2 Regularity Estimates

In this part we give some controls of the Dirichlet-Neumann and the Neumann-Neumann operators. The first Proposition is Theorem 3.15 in [80].

Proposition A.2.12. Let $t_0 > \frac{d}{2}$, $0 \le s \le t_0 + \frac{1}{2}$ and $\zeta, b \in H^{t_0+1}(\mathbb{R}^d)$. Then, $G_{\mu}[\varepsilon\zeta,\beta b]$ maps continuously $\dot{H}^{s+\frac{1}{2}}(\mathbb{R}^d)$ into $H^{s-\frac{1}{2}}_*(\mathbb{R}^d)$

$$|G_{\mu}[\varepsilon\zeta,\beta b](\psi)|_{H^{s-\frac{1}{2}}} \le \mu^{\frac{3}{4}} M|\mathfrak{P}\psi|_{H^{s}}$$

Furthermore, if $0 \leq s \leq t_0$, $G_{\mu}[\varepsilon\zeta,\beta b]$ maps continuously $\dot{H}^{s+1}(\mathbb{R}^d)$ into $H^{s-\frac{1}{2}}_*(\mathbb{R}^d)$

$$\left|G_{\mu}[\varepsilon\zeta,\beta b](\psi)\right|_{H^{s-\frac{1}{2}}} \leq \mu M \left|\mathfrak{P}\psi\right|_{H^{s+\frac{1}{2}}}$$

We have a similar estimation for the Neumann-Neumann operator.

Proposition A.2.13. Let $t_0 > \frac{d}{2}$, $0 \le s \le t_0 + \frac{1}{2}$ and $\zeta, b \in H^{t_0+1}(\mathbb{R}^d)$ such that Condition (A.1) is satisfied. Then, $G^{NN}_{\mu}[\varepsilon\zeta,\beta b]$ maps continuously $H^{s-\frac{1}{2}}_{*}(\mathbb{R}^d)$ into itself

$$|G^{NN}_{\mu}[\varepsilon\zeta,\beta b](B)|_{H^{s-\frac{1}{2}}_{*}} \le M |B|_{H^{s-\frac{1}{2}}_{*}}.$$

Remark A.2.14. Notice for all μ (even μ goes to 0), $G^{NN}_{\mu}[\varepsilon\zeta,\beta b](B)$ has the same regularity than B.

Proof. The dual formulation of $G^{NN}_{\mu}[\varepsilon\zeta,\beta b]$ shows that it maps continuously $H^{-\frac{1}{2}}_{*}(\mathbb{R}^{d})$ into itself. We have to prove that the regularity of B is kept by G^{NN}_{μ} . We argue by duality. Let take $B \in H^{s-\frac{1}{2}}(\mathbb{R}^{d})$ and φ in $H^{s-\frac{1}{2}}(\mathbb{R}^{d})$. We have

$$\left(\Lambda^{s-1/2}G^{NN}_{\mu}[\varepsilon\zeta,\beta b](B),\varphi\right) = \int_{S} P(\Sigma)\nabla^{\mu}_{X,z}B^{\mathfrak{d}} \cdot \nabla^{\mu}_{X,z}\left(\Lambda^{s-1/2}\varphi\right)^{\#} dx.$$

Since $(\cdot)^{\#}$, Λ and $\nabla^{\mu}_{X,z}$ commute, we obtain, using Proposition A.2.7,

$$\begin{split} \left| \left(\Lambda^{s-1/2} G^{NN}_{\mu} [\varepsilon\zeta, \beta b](B), \varphi \right) \right| &= \left| \int_{S} \frac{\sqrt{1 + \sqrt{\mu} |D|}}{(1 + |D|^2)^{\frac{1}{4}}} \Lambda^s P(\Sigma) \nabla^{\mu}_{X,z} B^{\mathfrak{d}} \cdot \nabla^{\mu}_{X,z} \left(\frac{1}{\sqrt{1 + \sqrt{\mu} |D|}} \varphi \right)^{\#} dx \right|, \\ &\leq C(\mu_{\max}) \left| \varphi \right|_{L^2} \left| \sqrt{1 + \sqrt{\mu} |D|} \Lambda^{s-\frac{1}{2}} P(\Sigma) \nabla^{\mu}_{X,z} B^{\mathfrak{d}} \right|_{L^2}. \end{split}$$

Then, we have

$$\begin{split} \left| \Lambda^{s-1/2} G^{NN}_{\mu} [\varepsilon \zeta, \beta b](B) \right|_{L^2(S)} \leq & C(\mu_{\max}) \left| \left[\sqrt{1 + \sqrt{\mu} |D|} \Lambda^{s-\frac{1}{2}}, Q(\Sigma) \right] \nabla^{\mu}_{X,z} B^{\mathfrak{d}} \right|_{L^2} \\ & + C(\mu_{\max}) \left| Q(\Sigma) \sqrt{1 + \sqrt{\mu} |D|} \Lambda^{s-\frac{1}{2}} \nabla^{\mu}_{X,z} B^{\mathfrak{d}} \right|_{L^2}. \end{split}$$

If $s \leq t_0$, we use Proposition B.3.2 (with a slight modification) and the product estimate B.2.2. We obtain

$$\left|\Lambda^{s-1/2} G^{NN}_{\mu}[\varepsilon\zeta,\beta b](B)\right|_{L^{2}(S)} \leq C \left(\left|Q(\Sigma)\right|_{L^{\infty}H^{t_{0}}} \left|\sqrt{1+\sqrt{\mu}|D|}\Lambda^{s-\frac{1}{2}} \nabla^{\mu}_{X,z} B^{\mathfrak{d}}\right|_{2}\right).$$

However, if $t_0 \le s \le t_0 + \frac{1}{2}$, we use Proposition B.3.3 (with a slight modification) and the product estimate B.2.2. We obtain,

$$\begin{split} \left| \Lambda^{s-1/2} G^{NN}_{\mu} [\varepsilon \zeta, \beta b](B) \right|_{L^2(S)} &\leq \left. C \left| Q(\Sigma) \right|_{L^2 H^{t_0 + \frac{1}{2}}} \left| \sqrt{1 + \sqrt{\mu} |D|} \Lambda^{s-1} \nabla^{\mu}_{X,z} B^{\mathfrak{d}} \right|_{L^{\infty}_{x} L^2_{X}} \right. \\ &+ \left. C \left| Q(\Sigma) \right|_{L^{\infty}_{z} H^{t_0}_{X}} \left| \sqrt{1 + \sqrt{\mu} |D|} \Lambda^{s-\frac{1}{2}} \nabla^{\mu}_{X,z} B^{\mathfrak{d}} \right|_{L^2}. \end{split}$$

In any case, using Theorem A.1.19, we get the result.

We can also give some regularity estimates for $G_{\mu}^{DD}[\varepsilon\zeta,\beta b]$ since it is the adjoint of $G_{\mu}^{NN}[\varepsilon\zeta,\beta b]$.

Proposition A.2.15. Let $t_0 > \frac{d}{2}$, $0 \le s \le t_0 + \frac{1}{2}$ and $\zeta, b \in H^{t_0+1}(\mathbb{R}^d)$ such that Condition (A.1) is satisfied. Then, $G^{DD}_{\mu}[\varepsilon\zeta,\beta b]$ maps continuously $\dot{H}^{s+\frac{1}{2}}(\mathbb{R}^d)$ into itself

$$\left|\nabla G^{DD}_{\mu}[\varepsilon\zeta,\beta b](\psi)\right|_{H^{s-\frac{1}{2}}} \le M \left|\nabla\psi\right|_{H^{s-\frac{1}{2}}}.$$

Finally, we can give some regularity estimates for $G^{ND}_{\mu}[\varepsilon\zeta,\beta b]$.

Proposition A.2.16. Let $t_0 > \frac{d}{2}$, $0 \le s \le t_0 + \frac{1}{2}$ and $\zeta, b \in H^{t_0+1}(\mathbb{R}^d)$ such that Condition (A.1) is satisfied. Then, $G^{ND}_{\mu}[\varepsilon\zeta,\beta b]$ maps continuously $H^{s-\frac{1}{2}}(\mathbb{R}^d)$ into $H^{s+\frac{1}{2}}(\mathbb{R}^d)$

$$\left|G^{ND}_{\mu}[\varepsilon\zeta,\beta b](B)\right|_{H^{s+\frac{1}{2}}} \le M \left|B\right|_{H^{s-\frac{1}{2}}}.$$

We can extend these estimates to $\underline{w}[\varepsilon\zeta,\beta b]$, the vertical velocity at the surface and to $\underline{V}[\varepsilon\zeta,\beta b]$ the horizontal velocity at the surface. These operators appear naturally when we differentiate the Dirichlet-Neumann and the Neumann-Neumann operator with respect to the surface ζ . We define

$$\underline{w}[\varepsilon\zeta,\beta b] := \begin{cases} \dot{H}^{s+\frac{1}{2}}(\mathbb{R}^d) \times H^{s-\frac{1}{2}}(\mathbb{R}^d) \to & H^{s-\frac{1}{2}}(\mathbb{R}^d) \\ (\psi,B) & \mapsto \frac{G_{\mu}[\varepsilon\zeta,\beta b](\psi) + \mu G^{NN}_{\mu}[\varepsilon\zeta,\beta b](B) + \varepsilon\mu\nabla\zeta\cdot\nabla\psi}{1 + \varepsilon^2\mu|\nabla\zeta|^2}, \end{cases}$$
(A.21)

and

$$\underline{V}[\varepsilon\zeta,\beta b] := \begin{cases} \dot{H}^{s+\frac{1}{2}}(\mathbb{R}^d) \times H^{s-\frac{1}{2}}_*(\mathbb{R}^d) \to H^{s-\frac{1}{2}}(\mathbb{R}^d) \\ (\psi,B) &\mapsto \nabla \psi - \varepsilon \underline{w}[\varepsilon\zeta,\beta b](\psi,B) \nabla \zeta. \end{cases}$$
(A.22)

Proposition A.2.17. Let $t_0 > \frac{d}{2}$, $0 \le s \le t_0 + \frac{1}{2}$ and $\zeta, b \in H^{t_0+1}(\mathbb{R}^d)$ such that Condition (A.1) is satisfied. Then, $\underline{w}[\varepsilon\zeta,\beta b]$ maps continuously $\dot{H}^{s+\frac{1}{2}}(\mathbb{R}^d) \times H^{s-\frac{1}{2}}(\mathbb{R}^d)$ into $H^{s-\frac{1}{2}}(\mathbb{R}^d)$

$$\left|\underline{w}[\varepsilon\zeta,\beta b](\psi,B)\right|_{H^{s-\frac{1}{2}}} \le M\left(\mu^{\frac{3}{4}}|\mathfrak{P}\psi|_{H^{s}}+\mu|B|_{H^{s-\frac{1}{2}}}\right)$$

Furthermore, if $1 \leq s \leq t_0$, $\underline{w}[\varepsilon\zeta, \beta b]$ maps continuously $\dot{H}^{s+1}(\mathbb{R}^d) \times H^{s-\frac{1}{2}}_*(\mathbb{R}^d)$ into $H^{s-\frac{1}{2}}(\mathbb{R}^d)$

$$\left|\underline{w}[\varepsilon\zeta,\beta b](\psi,B)\right|_{H^{s-\frac{1}{2}}} \le M\mu\left(\left|\mathfrak{P}\psi\right|_{H^{s+\frac{1}{2}}} + \left|B\right|_{H^{s-\frac{1}{2}}}\right)$$

Finally, we have the same estimate holds for $\underline{V}[\varepsilon\zeta,\beta b]$.

In the same way, we can extend also these estimates to $\underline{\tilde{w}}[\varepsilon\zeta,\beta b]$, the vertical velocity at the bottom and to $\underline{\tilde{V}}[\varepsilon\zeta,\beta b]$ the horizontal velocity at the bottom. These operators appear naturally when we differentiate the Dirichlet-Neumann and the Neumann-Neumann operator with respect to the bottom b. We define

$$\underbrace{\widetilde{w}}[\varepsilon\zeta,\beta b] := \begin{cases} \dot{H}^{s+\frac{1}{2}}(\mathbb{R}^d) \times H^{s-\frac{1}{2}}(\mathbb{R}^d) \to H^{s-\frac{1}{2}}(\mathbb{R}^d) \\ (\psi,B) &\mapsto \frac{\mu B + \beta\mu\nabla b \cdot \nabla \left(G^{DD}_{\mu}[\varepsilon\zeta,\beta b](\psi) + \mu G^{ND}_{\mu}[\varepsilon\zeta,\beta b](B)\right)}{1 + \beta^2\mu|\nabla b|^2}, \end{aligned}$$
(A.23)

and

$$\underbrace{\widetilde{V}}[\varepsilon\zeta,\beta b] := \begin{cases}
\dot{H}^{s+\frac{1}{2}}(\mathbb{R}^d) \times H^{s-\frac{1}{2}}(\mathbb{R}^d) \to H^{s-\frac{1}{2}}(\mathbb{R}^d) \\
(\psi,B) \mapsto \nabla \left(G^{DD}_{\mu}[\varepsilon\zeta,\beta b](\psi) + \mu G^{ND}_{\mu}[\varepsilon\zeta,\beta b](B) \right) - \beta \widetilde{\underline{w}}[\varepsilon\zeta,\beta b](\psi,B) \nabla b.$$
(A.24)

Proposition A.2.18. Let $t_0 > \frac{d}{2}$, $0 \le s \le t_0 + \frac{1}{2}$ and $\zeta, b \in H^{t_0+1}(\mathbb{R}^d)$ such that Condition (A.1) is satisfied. Then, $\underline{\widetilde{w}}[\varepsilon\zeta,\beta b]$ maps continuously $\dot{H}^{s+\frac{1}{2}}(\mathbb{R}^d) \times H^{s-\frac{1}{2}}(\mathbb{R}^d)$ into $H^{s-\frac{1}{2}}(\mathbb{R}^d)$

$$\left|\underline{\widetilde{w}}[\varepsilon\zeta,\beta b](\psi,B)\right|_{H^{s-\frac{1}{2}}} \le M\left(\left|\nabla\psi\right|_{H^{s-\frac{1}{2}}} + \mu|B|_{H^{s-\frac{1}{2}}}\right).$$

Finally, we have the same continuity result for $\underline{V}[\varepsilon\zeta,\beta b]$.

We end this section with a commutator estimate for the Dirichlet-Neumann operator.

Proposition A.2.19. Let $t_0 > \frac{d}{2}$ and $\zeta, b \in H^{t_0+2}(\mathbb{R}^d)$ such that Condition (A.1) is satisfied. Then, for all $V \in H^{t_0+1}(\mathbb{R}^d)^2$ and $u \in H^{\frac{1}{2}}(\mathbb{R}^d)$, we have

$$\left(V \cdot \nabla u, \frac{1}{\mu} G_{\mu}[\varepsilon\zeta, \beta b](u)\right) \leq C\left(M, \varepsilon \left|\zeta\right|_{H^{t_0+2}}, \beta \left|b\right|_{H^{t_0+2}}\right) \left|V\right|_{W^{1,\infty}} \left|\mathfrak{P}u\right|_{2}^{2}.$$

A.2.3 Shape derivatives

Let $t_0 > \frac{d}{2}$. Given $\psi \in \dot{H}^{\frac{3}{2}}(\mathbb{R}^d)$ and $B \in H^{\frac{1}{2}}_*(\mathbb{R}^d)$. We denote by Γ the set of functions (ζ, b) in $H^{t_0+1}(\mathbb{R}^d)$ satisfying (A.1). We introduce two maps :

$$G_{\mu}(\psi) := \begin{cases} \Gamma \to H^{\frac{1}{2}}_{*}(\mathbb{R}^{d}) \\ (\zeta, b) \mapsto G_{\mu}[\varepsilon\zeta, \beta b](\psi), \end{cases}$$
(A.25)

which is the Dirichlet-Neumann operator and

$$G^{NN}_{\mu}(B) := \begin{cases} \Gamma \to H^{\frac{1}{2}}_{*}(\mathbb{R}^{d}) \\ (\zeta, b) \mapsto G^{NN}_{\mu}[\varepsilon\zeta, \beta b](B), \end{cases}$$
(A.26)

which is the Neumann-Neumann operator. We can also define $\underline{w}(\psi, B)$ and $\underline{V}(\psi, B)$.

In order to linearize the water waves equations, we need a shape derivative formula for the Dirichlet-Neumann and the Neumann-Neumann operators. The following proposition is a summarize of Theorems 3.5 and 3.6 in [71] and Theorem 3.21 in [80].

Proposition A.2.20. Let $t_0 > \frac{d}{2}$, $\zeta, b \in H^{t_0+1}(\mathbb{R}^d)$, $\psi \in \dot{H}^{\frac{3}{2}}(\mathbb{R}^d)$ and $B \in H^{\frac{1}{2}}(\mathbb{R}^d)$. Then, $G_{\mu}(\psi)$ and $G_{\mu}^{NN}(B)$ are Fréchet differentiable. For $(h,k) \in H^{t_0+1}(\mathbb{R}^d)$, we have

$$dG_{\mu}(\psi).(h,0) + \mu dG_{\mu}^{NN}(B).(h,0) = -\varepsilon G_{\mu}[\varepsilon\zeta,\beta b](h\underline{w}[\varepsilon\zeta,\beta b](\psi,B))$$
(A.27)

$$-\varepsilon\mu\nabla\cdot(h\,\underline{V}[\varepsilon\zeta,\beta b](\psi,B)),\tag{A.28}$$

and

$$dG_{\mu}(\psi).(0,k) + \mu dG_{\mu}^{NN}(B).(0,k) = -\beta \mu G_{\mu}^{NN}[\varepsilon\zeta,\beta b] \left(\nabla \cdot \left(k \, \underline{\widetilde{V}}[\varepsilon\zeta,\beta b](\psi,B)\right)\right),$$

Furthermore,

$$dG^{DD}_{\mu}(\psi).(h,0) + \mu dG^{ND}_{\mu}(B).(h,0) = -\varepsilon G^{DD}_{\mu}[\varepsilon\zeta,\beta b](h\,\underline{w}[\varepsilon\zeta,\beta b](\psi,B))$$

Remark A.2.21. Notice that each term in the r.h.s of the formula (A.27) lies on $H^{-\frac{1}{2}}(\mathbb{R}^d)$ where as $dG_{\mu}(\psi)$ and $dG_{\mu}^{NN}(B)$ are in $H^{\frac{1}{2}}(\mathbb{R}^d)$. In fact, there are a cancellation of the singular terms.

Thanks to these formulae we can give some controls to the first shape derivatives of the operators. For instance, we give an estimate for $d\underline{\tilde{w}}$ and $d\underline{\tilde{V}}$.

Proposition A.2.22. Let $t_0 > \frac{d}{2}$ and $(\zeta, b) \in H^{t_0+1}(\mathbb{R}^d)$ such that Condition (A.1) is satisfied. Then, for $0 \le s \le t_0 + \frac{1}{2}$, for $\psi \in \dot{H}^{s+\frac{1}{2}}(\mathbb{R}^d)$ and $B \in H^{s-\frac{1}{2}}(\mathbb{R}^d)$, we have

$$\left| d \underline{\widetilde{V}}(\psi, B).(h, k) \right|_{H^{s-\frac{1}{2}}}, \ \left| d \underline{\widetilde{w}}(\psi, B).(h, k) \right|_{H^{s-\frac{1}{2}}} \leq M \left| (h, k) \right|_{H^{t_0+1}} \left(\left| \nabla \psi \right|_{H^{s-\frac{1}{2}}} + \left| B \right|_{H^{s-\frac{1}{2}}} \right).$$

Proof. This result follows from Proposition A.2.20 and Proposition A.2.13.

We end this part by giving some controls of the shape derivatives of G_{μ} and G_{μ}^{NN} . We do not use the previous method, we differentiate j times directly the dual formulation of both operators. The following proposition is Proposition 3.28 in [80].

Proposition A.2.23. Let $t_0 > \frac{d}{2}$ and $(\zeta, b) \in H^{t_0+1}(\mathbb{R}^d)$ such that Condition (A.1) is satisfied. Then

• (1) For all $0 \le s \le t_0 + \frac{1}{2}$ and $\psi \in \dot{H}^{s+\frac{1}{2}}(\mathbb{R}^d)$

$$|d^{j}G_{\mu}.(\boldsymbol{h},\boldsymbol{k})(\psi)|_{H^{s-\frac{1}{2}}} \leq M\mu^{\frac{3}{4}} \prod_{i\geq 1} |(\varepsilon h_{i},\beta k_{i})|_{H^{t_{0}+1}} |\mathfrak{P}\psi|_{H^{s}},$$

• (2) For all $0 \leq s \leq t_0$ and $\psi \in \dot{H}^{s+1}(\mathbb{R}^d)$

$$|d^{j}G_{\mu}\cdot(\boldsymbol{h},\boldsymbol{k})(\psi)|_{H^{s-\frac{1}{2}}} \leq M\mu \prod_{i\geq 1} |(\varepsilon h_{i},\beta k_{i})|_{H^{t_{0}+1}} |\mathfrak{P}\psi|_{H^{s+\frac{1}{2}}},$$

• (3) For all $0 \leq s \leq t_0$ and $\psi \in \dot{H}^{t_0+1}(\mathbb{R}^d)$

$$|d^{j}G_{\mu}.(\boldsymbol{h},\boldsymbol{k})(\psi)|_{H^{s-\frac{1}{2}}} \leq M\mu^{\frac{3}{4}} |(\varepsilon h_{1},\beta k_{1})|_{H^{s+\frac{1}{2}}} \prod_{i\geq 2} |(\varepsilon h_{i},\beta k_{i})|_{H^{t_{0}+1}} |\mathfrak{P}\psi|_{H^{t_{0}+\frac{1}{2}}}.$$

• (4) For all $0 \le s \le t_0 - \frac{1}{2}$ and $\psi \in \dot{H}^{t_0+1}(\mathbb{R}^d)$

$$|d^{j}G_{\mu}.(\boldsymbol{h},\boldsymbol{k})(\psi)|_{H^{s-\frac{1}{2}}} \leq M\mu|(\varepsilon h_{1},\beta k_{1})|_{H^{s+1}}\prod_{i\geq 2}|(\varepsilon h_{i},\beta k_{i})|_{H^{t_{0}+1}}|\mathfrak{P}\psi|_{H^{t_{0}+\frac{1}{2}}}.$$

Proposition A.2.24. Let $t_0 > \frac{d}{2}$ and $(\zeta, b) \in H^{t_0+1}(\mathbb{R}^d)$ such that Condition (A.1) is satisfied. Then for all $0 \le s \le t_0 + \frac{1}{2}$ and $B \in H^{s-\frac{1}{2}}(\mathbb{R}^d)$, we have

$$\left| d^{j} G^{NN}_{\mu} . (\boldsymbol{h}, \boldsymbol{k})(B) \right|_{H^{s-\frac{1}{2}}} \leq M \prod_{i \geq 1} \left| (\varepsilon h_{i}, \beta k_{i}) \right|_{H^{t_{0}+1}} \left| B \right|_{H^{s-\frac{1}{2}}}.$$

Furthermore, if $0 \leq s \leq t_0$ and $B \in H^{t_0}(\mathbb{R}^d)$,

$$\left| d^{j} G^{NN}_{\mu} . (\boldsymbol{h}, \boldsymbol{k})(B) \right|_{H^{s-\frac{1}{2}}} \leq M \left| (\varepsilon h_{1}, \beta k_{1}) \right|_{H^{s+\frac{1}{2}}} \prod_{i \geq 2} \left| (\varepsilon h_{i}, \beta k_{i}) \right|_{H^{t_{0}+1}} |B|_{H^{t_{0}}} .$$

Proof. Differentiating the dual formulation of $G^{NN}_{\mu}[\varepsilon\zeta,\beta b]$ for $B \in H^{s-\frac{1}{2}}(S)$, $\Lambda^{s-\frac{1}{2}}\varphi \in L^{2}(S)$ and arguing by duality, we obtain that

$$\left| d^{j} G^{N\!N}_{\mu} (\mathbf{h}, \mathbf{k})(B) \right|_{H^{s-\frac{1}{2}}} \leq C(\mu_{\max}) \left| \sqrt{1 + \sqrt{\mu} |D|} \Lambda^{s-\frac{1}{2}} \sum_{\substack{j_{1}+j_{2}=j\\I_{1} \sqcup I_{2}=[1,j]}} d^{j_{1}} P_{(\zeta,b)} \cdot (\mathbf{h}, \mathbf{k})_{I_{1}} \nabla^{\mu} d^{j_{2}} \mathfrak{D}(B)_{(\zeta,b)} \cdot (\mathbf{h}, \mathbf{k})_{I_{2}} \right|_{L^{2}}$$

Notice that the coordinates of $(\mathbf{h}, \mathbf{k})_{I_1}$ and $(\mathbf{h}, \mathbf{k})_{I_2}$ form a permutation of the coordinates of (\mathbf{h}, \mathbf{k}) . If $0 \le s \le t_0$, using Commutator estimate B.3.2 and the product estimate B.2.2 we get

$$\left| d^{j} G^{NN}_{\mu} (\mathbf{h}, \mathbf{k})(B) \right|_{H^{s-\frac{1}{2}} \leq C} \sum_{\substack{j_{1}+j_{2}=j\\I_{1} \sqcup I_{2}=[1,j]}} \left| d^{j_{1}} P_{(\zeta,b)} . (\mathbf{h}, \mathbf{k})_{I_{1}} \right|_{L^{\infty}_{z} H^{t_{0}}_{X}} \left| \sqrt{1 + \sqrt{\mu} |D|} \Lambda^{s-\frac{1}{2}} \nabla^{\mu} d^{j_{2}} \mathfrak{D}(B)_{(\zeta,b)} . (\mathbf{h}, \mathbf{k})_{I_{2}} \right|_{L^{2}}$$

Then, the first estimate of Proposition A.1.22 and Theorem A.1.23 give the first inequality. For the second inequality, we have to distinguish two cases : if $(1,1) \in I_1$ we use the third estimate of Proposition A.1.22 and Theorem A.1.23 whereas if $(1,1) \in I_2$ we use the first estimate of Proposition A.1.22 and Theorem A.1.24. If now $s = t_0 + \frac{1}{2}$, using Commutator estimate B.3.3 (with $t_1 = \frac{1}{2}$) and the product estimate B.2.2 (with Remark A.1.3) we get

$$\left| d^{j} G^{NN}_{\mu} (\mathbf{h}, \mathbf{k})(B) \right|_{H^{s-\frac{1}{2}}} \leq C \sum_{\substack{j_{1}+j_{2}=j\\I_{1}\sqcup I_{2}=[1,j]}} \left| d^{j_{1}} P_{(\zeta,b)} \cdot (\mathbf{h}, \mathbf{k})_{I_{1}} \right|_{H^{t_{0}+\frac{1}{2},1}} \left| \sqrt{1 + \sqrt{\mu} |D|} \nabla^{\mu} d^{j_{2}} \mathfrak{D}(B)_{(\zeta,b)} \cdot (\mathbf{h}, \mathbf{k})_{I_{2}} \right|_{H^{t_{0},1}}.$$

Using, the first and the second estimate of Proposition A.1.22 and Theorem A.1.23, we obtain the first inequality for $s = t_0 + \frac{1}{2}$. The case $t_0 \le s \le t_0 + \frac{1}{2}$ follows by interpolation.

A straightforward corollary is a control of $d\underline{w}$ and $d\underline{V}$.

Corollary A.2.25. Let $t_0 > \frac{d}{2}$ and $(\zeta, b) \in H^{t_0+1}(\mathbb{R}^d)$ such that Condition (A.1) is satisfied. Then

• (1) For all
$$0 \le s \le t_0 + \frac{1}{2}$$
, $\psi \in \dot{H}^{s+\frac{1}{2}}(\mathbb{R}^d)$ and $B \in H^{s-\frac{1}{2}}(\mathbb{R}^d)$,

$$|d^{j}\underline{w}.(\boldsymbol{h},\boldsymbol{k})(\psi)|_{H^{s-\frac{1}{2}}} \leq M \prod_{i\geq 1} |(\varepsilon h_{i},\beta k_{i})|_{H^{t_{0}+1}} \left(\mu^{\frac{3}{4}} |\mathfrak{P}\psi|_{H^{s}} + \mu|B|_{H^{s-\frac{1}{2}}} \right),$$

• (2) For all $0 \le s \le t_0, \ \psi \in \dot{H}^{s+1}(\mathbb{R}^d)$ and $B \in H^{s-\frac{1}{2}}(\mathbb{R}^d)$,

$$|d^{j}\underline{w}.(\boldsymbol{h},\boldsymbol{k})(\psi)|_{H^{s-\frac{1}{2}}} \leq M \prod_{i\geq 1} |(\varepsilon h_{i},\beta k_{i})|_{H^{t_{0}+1}} \left(\mu |\mathfrak{P}\psi|_{H^{s+\frac{1}{2}}} + \mu |B|_{H^{s-\frac{1}{2}}} \right),$$

• (3) For all $0 \le s \le t_0$, $\psi \in \dot{H}^{t_0+1}(\mathbb{R}^d)$ and $B \in H^{t_0}(\mathbb{R}^d)$,

$$|d^{j}\underline{w}.(\boldsymbol{h},\boldsymbol{k})(\psi)|_{H^{s-\frac{1}{2}}} \leq M|(\varepsilon h_{1},\beta k_{1})|_{H^{s+\frac{1}{2}}} \prod_{i\geq 2} |(\varepsilon h_{i},\beta k_{i})|_{H^{t_{0}+1}} \left(\mu^{\frac{3}{4}}|\mathfrak{P}\psi|_{H^{t_{0}+\frac{1}{2}}} + \mu|B|_{H^{t_{0}}}\right).$$

• (4) For all $0 \le s \le t_0 - \frac{1}{2}$, $\psi \in \dot{H}^{t_0+1}(\mathbb{R}^d)$ and $B \in H^{t_0}(\mathbb{R}^d)$,

$$|d^{j}\underline{w}.(\boldsymbol{h},\boldsymbol{k})(\psi)|_{H^{s-\frac{1}{2}}} \leq M|(\varepsilon h_{1},\beta k_{1})|_{H^{s+1}} \prod_{i\geq 2} |(\varepsilon h_{i},\beta k_{i})|_{H^{t_{0}+1}} \left(\mu|\mathfrak{P}\psi|_{H^{t_{0}+\frac{1}{2}}} + \mu|B|_{H^{t_{0}}}\right).$$

Finally, the same estimates holds for $d^j \underline{V}.(\mathbf{h}, \mathbf{k})(\psi)$.

Appendix B

Useful estimates

Sommaire

B.1	Estimate for \mathfrak{P}	192
B.2	Product estimates	192
B.3	Commutator estimates	193
$\mathbf{B.4}$	Estimates for \mathcal{H}_{μ}	194

In this part, we give some useful estimates, product and commutator estimates. We refer to Appendix B in [80], [79] and Chapter II in [8] for the proofs.

B.1 Estimate for \mathfrak{P}

The first estimates are useful to control $\mathfrak{P}f$. We recall that $\mathfrak{P} = \frac{|D|}{\sqrt{1+\sqrt{\mu}|D|}}$.

Proposition B.1.1. Let $f \in H^1(\mathbb{R}^d)$ and $g \in H^{\frac{1}{2}}(\mathbb{R}^d)$. Then,

$$|\mathfrak{P}g|_{2} \leq \mu^{-\frac{1}{4}} |g|_{H^{\frac{1}{2}}}, \ |\mathfrak{P}f|_{H^{\frac{1}{2}}} \leq \max(1, \mu^{-\frac{1}{4}}) |\nabla f|_{2} \ and \ |\nabla \psi|_{H^{s}} \leq \max(1, \mu^{\frac{1}{4}}) \left|\mathfrak{P}\psi\right|_{H^{s+\frac{1}{2}}}$$

We aslo need a product estimate for \mathfrak{P} .

Lemma B.1.2. Let $u \in W^{1,\infty}(\mathbb{R}^d)$ and $v \in H^{\frac{1}{2}}(\mathbb{R}^d)$. Then,

$$\left|\sqrt{\mu}\mathfrak{P}\left(uv\right)\right|_{2} \leq C\left(\mu_{\max}\right)\left|u\right|_{W^{1,\infty}(\mathbb{R}^{d})}\left|\sqrt{1+\sqrt{\mu}|D|}v\right|_{2}.$$

The following proposition is commutator estimate for \mathfrak{P} .

Lemma B.1.3. Let $t_0 > \frac{d}{2}$, $u \in H^{t_0+1}(\mathbb{R}^d)$ and $v \in H^{\frac{1}{2}}(\mathbb{R}^d)$. Then,

$$|[\mathfrak{P}, u] v|_2 \le C |u|_{H^{t_0+1}} \left(|v|_2 + |\mathfrak{P} v|_2 \right).$$

We also give a regularity estimate for functions in $H^{-\frac{1}{2}}_{*}(\mathbb{R}^d)$. We recall that

$$H_*^{-\frac{1}{2}}(\mathbb{R}^d) = \left\{ u \in H^{-\frac{1}{2}}(\mathbb{R}^d) \ , \ \exists v \in H^{\frac{1}{2}}(\mathbb{R}^d) \ , \ u = |D|v \right\},$$

and we denote v by $v = \frac{1}{|D|}u$.

Lemma B.1.4. Let
$$s \ge 0$$
 and $u \in H_*^{-\frac{1}{2}}(\mathbb{R}^d) \cap H^{s-\frac{1}{2}}(\mathbb{R}^d)$. Then $u \in H_*^{s-\frac{1}{2}}(\mathbb{R}^d)$ and

$$\left|\frac{1}{\mathfrak{P}}u\right|_{H^s} \leq \left|\frac{1}{\mathfrak{P}}u\right|_2 + \left|\sqrt{1+\sqrt{\mu}|D|}u\right|_{H^{s-1}}$$

B.2 Product estimates

We need some product estimates in \mathbb{R}^d . The following Proposition is Proposition B.2 in [80].

Proposition B.2.1. Let $s, s_1, s_2 \in \mathbb{R}$ such that $s \leq s_1, s \leq s_2, s_1 + s_2 \geq 0$ and $s < s_1 + s_2 - \frac{d}{2}$. Then, there exists a constant C > 0 such that for all $f \in H^{s_1}(\mathbb{R}^d)$ and for all $g \in H^{s_2}(\mathbb{R}^d)$, we have $fg \in H^s(\mathbb{R}^d)$ and

$$|fg|_{H^s} \le C|f|_{H^{s_1}}|g|_{H^{s_2}}$$

We also need some product estimates in $S = \mathbb{R}^d \times (-1, 0)$. The following Propositions are the Corollary B.5 in [80]. We recall that Λ^s is the Fourier multiplier

$$\Lambda^s = \left(1 + |D|^2\right)^{\frac{s}{2}}.$$

Proposition B.2.2. Let $s, s_1, s_2 \in \mathbb{R}$ such that $s \leq s_1, s \leq s_2, s_1 + s_2 \geq 0, s < s_1 + s_2 - \frac{d}{2}$ and $p \in \{2, +\infty\}$. Then, there exists a constant C > 0 such that for all $f \in L_z^{\infty} H_X^{s_1}(S)$ and for all $g \in L_z^p H_X^{s_2}(S)$, we have $fg \in L_z^p H_X^{s_3}(S)$ and

$$|\Lambda^{s}(fg)|_{L^{p}_{z}L^{2}_{x}(S)} \leq C|\Lambda^{s_{1}}f|_{L^{\infty}_{z}L^{2}_{x}(S)}|\Lambda^{s_{2}}g|_{L^{p}_{z}L^{2}_{x}(S)}.$$

Proposition B.2.3. Let $T_0 > \frac{d}{2}$ and $s \ge -T_0$. Then, there exists a constant C > 0 such that for all $f,g \in L_z^{\infty} H_X^{T_0}(S) \cap H^{s,0}(S)$, we have $fg \in H^{s,0}(S)$ and

$$|\Lambda^{s}(fg)|_{L^{2}(S)} \leq C.\left(|f|_{L^{\infty}_{z}H^{T_{0}}_{X}(S)}|\Lambda^{s}g|_{L^{2}(S)} + \mathbf{1}_{\{s>T_{0}\}}|g|_{L^{\infty}_{z}H^{T_{0}}_{X}(S)}|\Lambda^{s}f|_{L^{2}(S)}\right)$$

Notice that if $s \leq T_0$, $g \in H^{s,0}(S)$ is enough.

The following propositions gives estimates for 1/(1+g) in the flat strip S. We refers to Corollary B.6 in [80].

Proposition B.2.4. Let $T_0 > \frac{d}{2}$, $-T_0 \le s \le T_0$, $k_0 > 0$ and $p \in \{2, +\infty\}$. Then, for all $f \in L_z^p H_X^s(S)$ and $g \in L_z^\infty H_X^{T_0}(S)$ with $1 + g \ge k_0$, we have

$$\left|\Lambda^{s} \frac{f}{1+g}\right|_{L^{p}_{z}L^{2}(S)} \leq C\left(\frac{1}{k_{0}}, |g|_{L^{\infty}_{z}H^{T_{0}}_{X}}\right) |\Lambda^{s} f|_{L^{p}_{z}L^{2}(S)}.$$

Proposition B.2.5. Let $T_0 > \frac{d}{2}$, $s \ge -T_0$ and $k_0 > 0$. Then, for all $f,g \in L_z^{\infty}H_X^{T_0}(S) \cap H^{s,0}(S)$ with $1 + g \ge k_0$, we have

$$\left|\frac{f}{1+g}\right|_{H^{s,0}} \le C\left(\frac{1}{k_0}, |g|_{L^{\infty}_{z}H^{T_0}_{X}}\right) \left(|f|_{H^{s,0}} + \mathbf{1}_{\{s>T_0\}}|f|_{L^{\infty}_{z}H^{T_0}_{X}}|g|_{H^{s,0}}\right).$$

Notice that if $s \leq T_0$, $f \in H^{s,0}(S)$ is enough.

B.3 Commutator estimates

We also need some commutator estimates. We recall the classical Coifman-Meyer estimate. Lemma B.3.1. Let $s > \frac{d}{2} + 1$. Then, for $f, g \in L^2(\mathbb{R}^d)$,

$$\left| \left[\Lambda^{s}, f \right] g \right|_{2} \le C \left| f \right|_{H^{s}} \left| g \right|_{H^{s-1}}.$$

We need some commutator estimates in S. The following Propositions are Corollary B.17 in [80]. We denote by Λ_{δ}^{s} , the Fourier multiplier

$$\Lambda^s_\delta = \chi(\delta D)\Lambda^s,$$

where $\delta > 0$ and χ a smooth positive compactly supported function.

Proposition B.3.2. Let $T_0 > \frac{d}{2}$, $\delta \ge 0$, $0 < t_1 \le 1$ with $t_1 < T_0 - \frac{d}{2}$ and $-\frac{d}{2} < s \le T_0 + t_1$. Then for all $u \in L_z^{\infty} H_X^{T_0}$ and $v \in H^{s-t_1,0}(S)$ we have

$$\left\| \left[\Lambda_{\delta}^{s}, u \right] v \right\|_{L^{2}(S)} \leq C \left\| \Lambda_{\delta}^{T_{0}} u \right\|_{L^{\infty}_{z} L^{2}_{X}(S)} \left\| \Lambda_{\delta}^{s-t_{1}} v \right\|_{L^{2}(S)}.$$

Proposition B.3.3. Let $T_0 > \frac{d}{2}$, $\delta \ge 0$, $0 < t_1 \le 1$ with $t_1 < T_0 - \frac{d}{2}$ and $-\frac{d}{2} < s \le T_0 + t_1$. Then for all $u \in H^{T_0,0}$ and $v \in L_z^{\infty} H_X^{s-t_1}$ we have

$$\left|\left[\Lambda_{\delta}^{s}, u\right] v\right|_{L^{2}(S)} \leq C \left|\Lambda_{\delta}^{T_{0}} u\right|_{L^{2}(S)} \left|\Lambda_{\delta}^{s-t_{1}} v\right|_{L_{z}^{\infty} L_{X}^{2}(S)}$$

B.4 Estimates for \mathcal{H}_{μ}

In this part, we give some estimate for the operator \mathcal{H}_{μ} and some standard product and commutator estimates. For the estimate for \mathcal{H}_{μ} , we refer to part III in [125]. For the others estimates we refer to [8] and [79]. We recall that \mathcal{H}_{μ} is defined by

$$\mathcal{H}_{\mu} = -\frac{\tanh(\sqrt{\mu}D)}{D}\partial_x.$$

First, we show that \mathcal{H}_{μ} is a zero-order operator.

Proposition B.4.1. Let $s \ge 0$ and μ such that $0 < \mu_{\min} \le \mu \le \mu_{\max}$. Then,

$$\left|\mathcal{H}_{\mu}u\right|_{H^{s}} \leq C\left(\mu_{\max}\right)\left|u\right|_{H^{s}}.$$

Furthermore, for all $s \ge r \ge 0$,

$$\left| \left(\mathcal{H}_{\mu}^2 + 1 \right) u \right|_{H^s} \le C_r \left(\frac{1}{\mu_{\min}} \right) |u|_{H^r}.$$

Then, we give a commutator estimate for \mathcal{H}_{μ} .

Proposition B.4.2. Let $s \ge 0$, $t_0 > \frac{1}{2}$, $r \ge 0$, and $\mu \ge \mu_{\min} > 0$. Then,

$$|[\mathcal{H}_{\mu}, a] u|_{2} \leq C |a|_{H^{t_{0}}} |f|_{2},$$

$$\left| |\xi|^s [\widehat{\mathcal{H}_{\mu}, a]} u \right|_2 \le C \left(\frac{1}{\mu_{\min}} \right) |a|_{H^{r+s}} \left| \frac{(1+|\xi|)^{t_0}}{|\xi|^r} \widehat{u} \right|_2,$$

and

$$\left| |\xi|^s [\widehat{\mathcal{H}_{\mu}, a]} u \right|_2 \le C \left(\frac{1}{\mu_{\min}} \right) |a|_{H^{r+s+t_0}} \left| \frac{1}{|\xi|^r} \widehat{u} \right|_2.$$

Bibliography

- [1] T. Alazard. Stabilization of gravity water waves. Preprint arXiv:1603.08541, 2016.
- [2] T. Alazard, P. Baldi, and D. Han-Kwan. Control of the water waves. Preprint arXiv:1506.08520, 2015.
- [3] T. Alazard, N. Burq, and C. Zuily. The water-wave equations: from Zakharov to Euler. In Studies in phase space analysis with applications to PDEs, volume 84 of Progr. Nonlinear Differential Equations Appl., pages 1–20. Birkhäuser/Springer, New York, 2013.
- [4] T. Alazard, N. Burq, and C. Zuily. On the Cauchy problem for gravity water waves. Invent. Math., 198(1):71–163, 2014.
- [5] T. Alazard, N. Burq, and C. Zuily. Cauchy theory for the gravity water waves system with nonlocalized initial data. Annales de l'Institut Henri Poincare (C) Non Linear Analysis, 33(2):337 – 395, 2016.
- [6] T. Alazard and G. Métivier. Paralinearization of the Dirichlet to Neumann operator, and regularity of three-dimensional water waves. *Comm. Partial Differential Equations*, 34(10-12):1632–1704, 2009.
- [7] S. Alinhac. Existence d'ondes de raréfaction pour des systèmes quasi-linéaires hyperboliques multidimensionnels. Comm. Partial Differential Equations, 14(2):173–230, 1989.
- [8] S. Alinhac and P. Gérard. Opérateurs pseudo-différentiels et théorème de Nash-Moser. Savoirs Actuels. [Current Scholarship]. InterEditions, Paris; Éditions du Centre National de la Recherche Scientifique (CNRS), Meudon, 1991.
- B. Alvarez-Samaniego and D. Lannes. Large time existence for 3D water-waves and asymptotics. Invent. Math., 171(3):485–541, 2008.
- [10] D. M. Ambrose, J. L. Bona, and D. P. Nicholls. On ill-posedness of truncated series models for water waves. Proc. R. Soc. Lond. Ser. A Math. Phys. Eng. Sci., 470(2166):20130849, 16, 2014.
- [11] E. Audusse, F. Bouchut, M. O. Bristeau, R. Klein, and B. Perthame. A fast and stable wellbalanced scheme with hydrostatic reconstruction for shallow water flows. SIAM J. Sci. Comput., 25(6):2050–2065, June 2004.
- [12] A. J. C. Barré de Saint-Venant. Sur la houle et le clapotis. C. R. Acad. Sc. Paris, 73:521–528, 1871.
- [13] A. J. C. Barré de Saint-Venant. Théorie du mouvement non permanent des eaux, avec applications aux crues des rivières et à l'introduction des marées dans leur lit. C. R. Acad. Sc. Paris, 73:147–151, 1871.
- [14] A. J. Bechle and C. H. Wu. The lake michigan meteotsunamis of 1954 revisited. Natural Hazards, 74(1):155–177, 2014.
- [15] W. Ben Youssef and D. Lannes. The long wave limit for a general class of 2d quasilinear hyperbolic problems. *Communications in Partial Differential Equations*, 27(5-6):979–1020, 2002.
- [16] S. Benzoni-Gavage and D. Serre. Multidimensional hyperbolic partial differential equations. Oxford Mathematical Monographs. The Clarendon Press, Oxford University Press, Oxford, 2007. Firstorder systems and applications.

- [17] J. L. Bona, M. Chen, and J.-C. Saut. Boussinesq equations and other systems for small-amplitude long waves in nonlinear dispersive media. I. Derivation and linear theory. J. Nonlinear Sci., 12(4):283–318, 2002.
- [18] J. L. Bona, M. Chen, and J.-C. Saut. Boussinesq equations and other systems for small-amplitude long waves in nonlinear dispersive media. II. The nonlinear theory. *Nonlinearity*, 17(3):925–952, 2004.
- [19] J. L. Bona, T. Colin, and D. Lannes. Long wave approximations for water waves. Arch. Ration. Mech. Anal., 178(3):373–410, 2005.
- [20] P. Bonneton, E. Barthelemy, F. Chazel, R. Cienfuegos, D. Lannes, F. Marche, and M. Tissier. Recent advances in serre-green naghdi modelling for wave transformation, breaking and runup processes. *European Journal of Mechanics - B Fluids*, 30(6):589 – 597, 2011. Special Issue: Nearshore Hydrodynamics.
- [21] P. Bonneton, F. Chazel, D. Lannes, F. Marche, and M. Tissier. A splitting approach for the fully nonlinear and weakly dispersive green-naghdi model. *Journal of Computational Physics*, 230(4):1479 – 1498, 2011.
- [22] P. Bonneton and D. Lannes. Derivation of asymptotic two-dimensional time-dependent equations for surface water wave propagation. *Phys. Fluids*, 21, 2009.
- [23] F. Bouchut. Nonlinear stability of finite volume methods for hyperbolic conservation laws and well-balanced schemes for sources. Frontiers in mathematics. Birkauser, 1 edition, 2004.
- [24] A. Bouharguane and B. Mélinand. A splitting method for deep waters with bathymetry. In preparation, 2016.
- [25] J. Boussinesq. Théorie générale des mouvements qui sont propagés dans un canal rectangulaire horizontal. C. R. Acad. Sci., Paris, 73:256–260, 1871.
- [26] J. Boussinesq. Théorie des ondes et des remous qui se propagent le long d'un canal rectangulaire horizontal, en communiquant au liquide contenu dans ce canal des vitesses sensiblement pareilles de la surface au fond. *Liouville J. (2)*, 17:55–109, 1872.
- [27] D. Bresch. Shallow-water equations and related topics. In Handbook of differential equations: evolutionary equations. Vol. V, Handb. Differ. Equ., pages 1–104. Elsevier/North-Holland, Amsterdam, 2009.
- [28] D. Bresch and G. Métivier. Anelastic limits for euler-type systems. Applied Mathematics Research eXpress, 2010(2):119–141, 2010.
- [29] B. Buffoni, M. D. Groves, S. M. Sun, and E. Wahlén. Existence and conditional energetic stability of three-dimensional fully localised solitary gravity-capillary water waves. J. Differential Equations, 254(3):1006–1096, 2013.
- [30] T. Bugge, R. H. Belderson, and N. H. Kenyon. The storegga slide. *Philosophical Transactions of the Royal Society of London*, 325(1586):357–388, 1988.
- [31] A. Bulut. An optimal decay estimate for the linearized water wave equation in 2d. to appear in Proceedings of the American Mathematical Society, 2016.
- [32] R. Carles. On Fourier time-splitting methods for nonlinear Schrödinger equations in the semiclassical limit. SIAM J. Numer. Anal., 51(6):3232–3258, 2013.
- [33] A. Castro and D. Lannes. Fully nonlinear long-wave models in the presence of vorticity. Journal of Fluid Mechanics, 759:642–675, 11 2014.
- [34] A. Castro and D. Lannes. Well-posedness and shallow-water stability for a new Hamiltonian formulation of the water waves equations with vorticity. *Indiana Univ. Math. J.*, 64(4):1169–1270, 2015.
- [35] M. Cathala. Asymptotic shallow water models with non smooth topographies. Monatshefte für Mathematik, pages 1–29, 2014.

- [36] P. Chartier, L. Le treust, and F. Méhats. Uniformly accurate time-splitting methods for the semiclassical schrödinger equation part 2 : numerical analysis of the linear case. arXiv : 1601.04825v1, 2016.
- [37] J.-Y. Chemin, B. Desjardins, I. Gallagher, and E. Grenier. Mathematical Geophysics: An Introduction to Rotating Fluids and the Navier-Stokes Equations, volume 32 of Oxford Lecture Series in Mathematics and Its Applications. Oxford University Press, 2006.
- [38] W. Choi. Nonlinear evolution equations for two-dimensional surface waves in a fluid of finite depth. J. Fluid Mech., 295:381–394, 1995.
- [39] D. Christodoulou and H. Lindblad. On the motion of the free surface of a liquid. Comm. Pure Appl. Math., 53(12):1536–1602, 2000.
- [40] L. Chupin. Roughness effect on Neumann boundary condition. Asymptot. Anal., 78(1-2):85–121, 2012.
- [41] A. Constantin and D. Lannes. The hydrodynamical relevance of the camassa-holm and degasperisprocess equations. Archive for Rational Mechanics and Analysis, 192(1):165–186, 2008.
- [42] D. Coutand and S. Shkoller. Well-posedness of the free-surface incompressible Euler equations with or without surface tension. J. Amer. Math. Soc., 20(3):829–930, 2007.
- [43] W. Craig. An existence theory for water waves and the Boussinesq and Korteweg-de Vries scaling limits. Comm. Partial Differential Equations, 10(8):787–1003, 1985.
- [44] W. Craig, P. Guyenne, D. P. Nicholls, and C. Sulem. Hamiltonian long-wave expansions for water waves over a rough bottom. Proc. R. Soc. Lond. Ser. A Math. Phys. Eng. Sci., 461(2055):839–873, 2005.
- [45] W. Craig, P. Guyenne, D. P. Nicholls, and C. Sulem. Hamiltonian long-wave expansions for water waves over a rough bottom. Proc. R. Soc. Lond. Ser. A Math. Phys. Eng. Sci., 461:839–873, 2005.
- [46] W. Craig, D. Lannes, and C. Sulem. Water waves over a rough bottom in the shallow water regime. Ann. Inst. H. Poincaré Anal. Non Linéaire, 29(2):233–259, 2012.
- [47] W. Craig and C. Sulem. Numerical simulation of gravity waves. J. Comput. Phys., 108(1):73–83, 1993.
- [48] W. Craig, C. Sulem, and P.-L. Sulem. Nonlinear modulation of gravity waves: a rigorous approach. Nonlinearity, 5(2):497–522, 1992.
- [49] S. Cuccagna. Dispersion for schrödinger equation with periodic potential in 1d. Communications in Partial Differential Equations, 33(11):2064–2095, 2008.
- [50] S. Cuccagna. On dispersion for klein gordon equation with periodic potential in 1d. Hokkaido Mathematical Journal, 37:627–645, 2008.
- [51] R. Dautray and J. L. Lions. Analyse mathématique et calcul numérique pour les sciences et les techniques. Vol. 5. INSTN: Collection Enseignement. [INSTN: Teaching Collection]. Masson, Paris, 1988. Spectre des opérateurs. [The operator spectrum], With the collaboration of Michel Artola, Michel Cessenat, Jean Michel Combes and Bruno Scheurer, Reprinted from the 1984 edition.
- [52] T. de Poyferré. Blow-up conditions for gravity water-waves. arXiv:1407.6881, 2014.
- [53] J. Deny and J. L. Lions. Les espaces du type de Beppo Levi. Ann. Inst. Fourier, Grenoble, 5:305–370 (1955), 1954.
- [54] I. Didenkulova, I. Nikolkina, E. Pelinovsky, and N. Zahibo. Tsunami waves generated by submarine landslides of variable volume: analytical solutions for a basin of variable depth. *Natural Hazards* and Earth System Sciences, 10(11):2407–2419, 2010.
- [55] I. I. Didenkulova, I. F. Nikolkina, and E. N. Pelinovsky. Resonant amplification of tsunami waves generated by an underwater landslide. *Doklady Earth Sciences*, 436(1):66–69, 2011.
- [56] D. G. Ebin. The equations of motion of a perfect fluid with free boundary are not well posed. Comm. Partial Differential Equations, 12(10):1175–1201, 1987.

- [57] I. V. Fine, A. B. Rabinovich, R. E. Thomson, and E. A. Kulikov. Submarine Landslides and Tsunamis, chapter Numerical Modeling of Tsunami Generation by Submarine and Subaerial Landslides, pages 69–88. Springer Netherlands, Dordrecht, 2003.
- [58] V.N. Galkin and Yu.A. Stepanyants. On the existence of stationary solitary waves in a rotating fluid. Journal of Applied Mathematics and Mechanics, 55(6):939 – 943, 1991.
- [59] I. Gallagher and L. Saint-Raymond. Chapter 5 on the influence of the earth's rotation on geophysical flows. In S. Friedlander and D. Serre, editors, *Handbook of Mathematical Fluid Dynamics*, volume 4, pages 201 – 329. North-Holland, 2007.
- [60] J. P. Germain and D. P. Renouard. On permanent nonlinear waves in a rotating fluid. Fluid Dynamics Research, 7(5-6):263, 1991.
- [61] A.E. Gill. Atmosphere-Ocean Dynamics. International Geophysics. Elsevier Science, 1982.
- [62] A. E. Green and P. M. Naghdi. A derivation of equations for wave propagation in water of variable depth. Journal of Fluid Mechanics, 78:237–246, 11 1976.
- [63] R. Grimshaw, E. Pelinovsky, and X. Tian. Interaction of a solitary wave with an external force. *Physica D: Nonlinear Phenomena*, 77(4):405 – 433, 1994.
- [64] R.H.J. Grimshaw, L.A. Ostrovsky, V.I. Shrira, and Yu. A. Stepanyants. Long nonlinear surface and internal gravity waves in a rotating ocean. *Surveys in Geophysics*, 19(4):289–338, 1998.
- [65] P. Guyenne and D. P. Nicholls. Numerical simulation of solitary waves on plane slopes. Math. Comput. Simul., 69:269–281, 2005.
- [66] P. Guyenne and D. P. Nicholls. A high-order spectral method for nonlinear water waves over moving bottom topography. SIAM J. Sci. Computer, no. 1:81–101, 2007-08.
- [67] T. Hibiya and K. Kajiura. Origin of theabiki phenomenon (a kind of seiche) in nagasaki bay. Journal of the Oceanographical Society of Japan, 38(3):172–182, 1982.
- [68] H. Holden, C. Lubich, and N. H. Risebro. Operator splitting for partial differential equations with burgers nonlinearity. *Math. Comput.*, 82(281), 2013.
- [69] L. Hörmander. Lectures on nonlinear hyperbolic differential equations, volume 26 of Mathématiques & Applications (Berlin) [Mathematics & Applications]. Springer-Verlag, Berlin, 1997.
- [70] T. Iguchi. A shallow water approximation for water waves. J. Math. Kyoto Univ., 49(1):13–55, 2009.
- [71] T. Iguchi. A mathematical analysis of tsunami generation in shallow water due to seabed deformation. Proc. Roy. Soc. Edinburgh Sect. A, 141(3):551–608, 2011.
- [72] S. Israwi. Large time existence for 1D Green-Naghdi equations. Nonlinear Anal., 74(1):81–93, 2011.
- [73] A. Jansa, S. Monserrat, and D. Gomis. The rissaga of 15 june 2006 in ciutadella (menorca), a meteorological tsunami. Advances in Geosciences, 12:1–4, Jun 2007.
- [74] T. Kano and T. Nishida. Sur les ondes de surface de l'eau avec une justification mathématique des équations des ondes en eau peu profonde. J. Math. Kyoto Univ., 19(2):335–370, 1979.
- [75] T. Kano and T. Nishida. A mathematical justification for Korteweg-de Vries equation and Boussinesq equation of water surface waves. Osaka J. Math., 23(2):389–413, 1986.
- [76] C. E. Kenig, G. Ponce, and L. Vega. Well-posedness and scattering results for the generalized korteweg-de vries equation via the contraction principle. *Communications on Pure and Applied Mathematics*, 46(4):527–620, 1993.
- [77] J. L. Lagrange. Mémoire sur la théorie du mouvement des fluides. Oeuvres de Lagrange, Gauthier-Villars, Paris, France. J.A. SERRET Editeur, 1781.
- [78] D. Lannes. Well-posedness of the water-waves equations. J. Amer. Math. Soc., 18(3):605–654 (electronic), 2005.

- [79] D. Lannes. Sharp estimates for pseudo-differential operators with symbols of limited smoothness and commutators. J. Funct. Anal., 232(2):495–539, 2006.
- [80] D. Lannes. The water waves problem, volume 188 of Mathematical Surveys and Monographs. American Mathematical Society, Providence, RI, 2013. Mathematical analysis and asymptotics.
- [81] D. Lannes and B. Alvarez-Samaniego. A nash-moser theorem for singular evolution equations. application to the serre and green-naghdi equations. *Indiana Univ. Math. J.*, 57:97–132, 2008.
- [82] D. Lannes and F. Marche. Nonlinear wave-current interactions in shallow water. Studies in Applied Mathematics, 2016.
- [83] D. Lannes and J. C. Saut. Weakly transverse Boussinesq systems and the Kadomtsev-Petviashvili approximation. *Nonlinearity*, 19(12):2853–2875, 2006.
- [84] P.H. LeBlond and L.A. Mysak. Waves in the Ocean. Elsevier Oceanography Series. Elsevier Science, 1981.
- [85] A. I. Leonov. The effect of the earth's rotation on the propagation of weak nonlinear surface and internal long oceanic waves. Annals of the New York Academy of Sciences, 373(1):150–159, 1981.
- [86] B. Levin and M. Nosov. *Physics of Tsunamis*, volume XI of *Earth Sciences and Geography*. Springer Science and Business Media, 2009.
- [87] Y. Li, J. Huang, and W. Yan. The cauchy problem for the ostrovsky equation with negative dispersion at the critical regularity. *Journal of Differential Equations*, 259(4):1379 – 1408, 2015.
- [88] Yi A. Li. A shallow-water approximation to the full water wave problem. Communications on Pure and Applied Mathematics, 59(9):1225–1285, 2006.
- [89] F. Linares and A. Milanés. Local and global well-posedness for the ostrovsky equation. Journal of Differential Equations, 222(2):325 – 340, 2006.
- [90] H. Lindblad. Well-posedness for the motion of an incompressible liquid with free surface boundary. Ann. of Math. (2), 162(1):109–194, 2005.
- [91] Y. Liu and D.K.P. Yue. On generalized Bragg scattering of surface waves by bottom ripples. Journal of Fluid Mechanics, 356:297–326, 1998.
- [92] C. Lubich. On splitting methods for Schrödinger-Poisson and cubic nonlinear Schrödinger equations. Math. Comp., 77(264):2141–2153, 2008.
- [93] A. Majda. Introduction to PDEs and waves for the atmosphere and ocean, volume 9 of Courant Lecture Notes in Mathematics. New York University, Courant Institute of Mathematical Sciences, New York; American Mathematical Society, Providence, RI, 2003.
- [94] N. Makarenko. A second long-wave approximation in the cauchy poisson problem (in russian). Dyn. Contin. Media, 1986:56–72, 1986.
- [95] A. I. Markovskii. Remarks on the lp-lq estimates of solutions of the klein-gordon equation. Ukrainian Mathematical Journal, 44(2):205–214, 1992.
- [96] N. Masmoudi and F. Rousset. Uniform regularity and vanishing viscosity limit for the free surface navier-stokes equations. *Submitted, arXiv: 1202.0657*, 2013.
- [97] Y. Matsuno. Nonlinear evolutions of surface gravity waves on fluid of finite depth. Phys. Rev. Lett., 69(4):609–611, 1992.
- [98] Y. Matsuno. Nonlinear evolution of surface gravity waves over an uneven bottom. J. Fluid Mech., 249:121–133, 1993.
- [99] Y. Matsuno. Two-dimensional evolution of surface gravity waves on a fluid of arbitrary depth. Phys. Rev. E, 47:4593–4596, 1993.
- [100] C. C. Mei. Resonant reflection of surface water waves by periodic sandbars. Journal of Fluid Mechanics, 152:315–335, 3 1985.
- [101] B. Mélinand. Coriolis effect on water waves. arXiv:1511.07407, 2015.

- [102] B. Mélinand. A mathematical study of meteo and landslide tsunamis : The proudman resonance. Nonlinearity, 28(11):4037–4080, 2015.
- [103] B. Mélinand. Long wave approximation for water waves under a coriolis forcing and the ostrovsy equation. arXiv:1603.08782, 2016.
- [104] B. Mésognon-Gireau. The cauchy problem on large time for a boussinesq-peregrine equation with large topography variations. arXiv:1512.02429v1, 2015.
- [105] B. Mésognon-Gireau. The cauchy problem on large time for the water waves equations with large topography variations. Annales de l'Institut Henri Poincare (C) Non Linear Analysis, 2015.
- [106] B. Mésognon-Gireau. A dispersive estimate for the linearized water-waves equations in finite depth. arXiv:1512.02424, 2015.
- [107] B. Mésognon-Gireau. The singular limit of the water-waves equations in the rigid lid regime. arXiv:1512.02424, 2015.
- [108] J. W. Miles and Y. K. Lee. Helmholtz resonance of harbours. Journal of Fluid Mechanics, 67:445– 464, 2 1975.
- [109] S. Monserrat, I. Vilibić, and A. B. Rabinovich. Meteotsunamis: atmospherically induced destructive ocean waves in the tsunami frequency band. *Natural Hazards and Earth System Science*, 6(6):1035–1051, 2006.
- [110] V. I. Nalimov. The cauchy-poisson problem (in russian). Dyn Splosh. Sredy, 18:104–210, 1974.
- [111] D. P. Nicholls and F. Reitich. Stability of high-order perturbative methods for the computation of dirichlet-neumann operators. *Journal of Computational Physics*, 170(1):276–298, 2001.
- [112] O. Nwogu. Alternative form of boussinesq equations for nearshore wave propagation. Journal of Waterway, Port, Coastal, and Ocean Engineering, 119(6):618–638, 1993.
- [113] L. Ostrovsky. Nonlinear internal waves in a rotating ocean. Oceanology, 18(2):119–125, 1978.
- [114] L. A. Ostrovsky and Yu. A. Stepanyants. Nonlinear Waves 3: Physics and Astrophysics Proceedings of the Gorky School 1989, chapter Nonlinear Surface and Internal Waves in Rotating Fluids, pages 106–128. Springer Berlin Heidelberg, Berlin, Heidelberg, 1990.
- [115] L. V. Ovsjannikov. To the shallow water theory foundation. Arch. Mech., 26:407–422, 1974.
- [116] L. V. Ovsjannikov. Cauchy problem in a scale of banach spaces and its application to the shallow water theory justification. Applications of Methods of Functional Analysis to Problems in Mechanics, Lecture Notes in Math., 503:426–437, 1976.
- [117] J. Pedlosky. Geophysical Fluid Dynamics. Springer study edition. Springer New York, 1992.
- [118] E. Pelinovsky, T. Talipova, A. Kurkin, and C. Kharif. Nonlinear mechanism of tsunami wave generation by atmospheric disturbances. *Natural Hazards and Earth System Sciences*, 1(4):243– 250, 2001.
- [119] D. H. Peregrine. Long waves on a beach. Journal of Fluid Mechanics, 27:815–827, 3 1967.
- [120] J Proudman. The effects on the sea of changes in atmospheric pressure. Geophysical Journal International, 3:197–209, 1929.
- [121] G. L. Richard and S. L. Gavrilyuk. A new model of roll waves: comparison with Brock's experiments. J. Fluid Mech., 698:374–405, 2012.
- [122] M. Rodriguez, N. Chamot-Rooke, H. Hébert, M. Fournier, and P. Huchon. Owen ridge deep-water submarine landslides: implications for tsunami hazard along the oman coast. *Natural Hazards and Earth System Sciences*, 13(2):417–424, 2013.
- [123] D. L. Ropp and J. N. Shadid. Stability of operator splitting methods for systems with indefinite operators: advection-diffusion-reaction systems. J. Comput. Phys., 228(9):3508–3516, 2009.
- [124] A. Sacchetti. Spectral splitting method for nonlinear Schrödinger equations with singular potential. J. Comput. Phys., 227(2):1483–1499, 2007.

- [125] J. C. Saut and L. Xu. Well-posedness on large time for a modified full dispersion system of surface waves. J. Math. Phys., 53(11):115606, 23, 2012.
- [126] J.C. Saut, C. Wang, and L. Xu. The Cauchy problem on large time for surface waves Boussinesq systems ii. arXiv:1511.08824, 2016.
- [127] J.C. Saut and L. Xu. The Cauchy problem on large time for surface waves Boussinesq systems. J. Math. Pures Appl. (9), 97(6):635–662, 2012.
- [128] G. Schneider and C. E. Wayne. Justification of the NLS approximation for a quasilinear water wave model. J. Differential Equations, 251(2):238–269, 2011.
- [129] G. Schneider and C. E. Wayne. Corrigendum: The long-wave limit for the water wave problem I. The case of zero surface tension [mr1780702]. Comm. Pure Appl. Math., 65(5):587–591, 2012.
- [130] F. J. Seabra-Santos, D. P. Renouard, and A. M. Temperville. Numerical and experimental study of the transformation of a solitary wave over a shelf or isolated obstacle. *Journal of Fluid Mechanics*, 176:117–134, 3 1987.
- [131] F. Serre. Contribution à l'étude des écoulements permanents et variables dans les canaux. La Houille Blanche, 3:374–388, 1953.
- [132] R. A. Smith. An operator expansion formalism for nonlinear surface waves over variable depth. J. Fluid Mech., 363:333–347, 1998.
- [133] E. M. Stein. Harmonic analysis: real-variable methods, orthogonality, and oscillatory integrals, volume 43 of Princeton Mathematical Series. Princeton University Press, Princeton, NJ, 1993. With the assistance of Timothy S. Murphy, Monographs in Harmonic Analysis, III.
- [134] H. Sverdrup. Dynamics of tides on the north siberian shelf. Geophys. Publ., 4:1 75, 1927.
- [135] T. R. Taha and M. J. Ablowitz. Analytical and numerical aspects of certain nonlinear evolution equations. II. Numerical, nonlinear Schrödinger equation. J. Comput. Phys., 55(2):203–230, 1984.
- [136] D. R. Tappin, S. T. Grilli, J. C. Harris, R. J. Geller, Masterlark T., J.T. Kirby, F. Shi, G. Ma, K.K.S. Thingbaijam, and P. M. Mai. Did a submarine landslide contribute to the 2011 tohoku tsunami ? *Marine Geology*, 357:344 – 361, 2014.
- [137] D. R. Tappin, P. Watts, and S. T. Grilli. The papua new guinea tsunami of 17 july 1998: anatomy of a catastrophic event. *Natural Hazards and Earth System Sciences*, 8(2):243–266, 2008.
- [138] G. Taylor. The instability of liquid surfaces when accelerated in a direction perpendicular to their planes. I. Proc. Roy. Soc. London. Ser. A., 201:192–196, 1950.
- [139] V. M. Teshukov. Gas-dynamic analogy for vortex free-boundary flows. Journal of Applied Mechanics and Technical Physics, 48(3):303–309, 2007.
- [140] R. Thai. Autour des équations de navier-stokes-coriolis avec surface libre. thèse de doctorat, 2013.
- [141] S. Tinti and E. Bortolucci. Energy of water waves induced by submarine landslides. Pure and Applied Geophysics, 157:281–318, 2000.
- [142] S. Tinti, E. Bortolucci, and A. Armigliato. Numerical simulation of the landslide-induced tsunami of 1988 on vulcano island, italy. *Bull Volcanol*, 61:121–137, 1999.
- [143] V. Varlamov and Y. Liu. Cauchy problem for the ostrovsky equation. Discrete and Continuous Dynamical Systems, 10(3):731–753, 2004.
- [144] I. Vilibic. Numerical simulations of the proudman resonance. Continental Shelf Research, 28:574– 581, 2008.
- [145] T. Vučetić, I. Vilibic, S. Tinti, and A. Maramai. The great adriatic flood of 21 june 1978 revisited: An overview of the reports. *Physics and Chemistry of the Earth, Parts A/B/C*, 34(17–18):894 – 903, 2009.
- [146] W. Wahl. Lp-decacy rates for homogeneous wave-equations. Mathematische Zeitschrift, 120:93– 106, 1971.

- [147] X. Wang. Global solution for the 3d gravity waves system above a flat bottom. arXiv:1508.06227, 2015.
- [148] X. Wang. On 3d water waves system above a flat bottom. arXiv:1508.06223, 2015.
- [149] G. B. Whitham. Linear and Nonlinear Waves. John Wiley and Sons, Inc., 1999.
- [150] S. Wu. Well-posedness in Sobolev spaces of the full water wave problem in 2-D. Invent. Math., 130(1):39–72, 1997.
- [151] S. Wu. Well-posedness in Sobolev spaces of the full water wave problem in 3-D. J. Amer. Math. Soc., 12(2):445–495, 1999.
- [152] H. Yosihara. Gravity waves on the free surface of an incompressible perfect fluid of finite depth. *RIMS Kyoto*, 18:49–96, 1982.
- [153] V.E Zakharov. Stability of periodic waves of finite amplitude on the surface of a deep fluid. J. Applied Mech. Tech. Phys., 9:190–194, 1968.
- [154] P. Zhang and Y. Liu. Symmetry and uniqueness of the solitary-wave solution for the ostrovsky equation. Archive for Rational Mechanics and Analysis, 196(3):811–837, 6 2010.
- [155] P. Zhang and Z. Zhang. On the local wellposedness of 3-d water wave problem with vorticity. Science in China Series A: Mathematics, 50(8):1065–1077, 2007.
- [156] P. Zhang and Z. Zhang. On the free boundary problem of three-dimensional incompressible euler equations. Communications on Pure and Applied Mathematics, 61(7):877–940, 2008.

The author has been partially funded by the ANR project Dyficolti ANR-13-BS01-0003.